

Technische Universität München

Department of Mathematics



Master's Thesis

EM algorithm and its extensions for Gaussian and vine copula mixture models

Ka Wing Ho

Supervisor: Prof. Claudia Czado Advisor: Prof. Claudia Czado, Özge Sahin Submission Date: December 15, 2021

I assure the single handed composition of this master's thesis only supported by declared resources.

Garching,

Acknowledgement

I would like to express my gratitude to my thesis supervisor Prof. Claudia Czado, that I have this valuable opportunity to write my master's thesis at the Chair of Mathematical Statistics. In the last six months, we had a lot of meetings and she can always give me useful advice and feedback, so that I can keep working on and improve my thesis. Furthermore, I would like to thank my advisor Özge Sahin very much for all the guidance, her support and help in the programming for simulation. Every time I ask her questions about the thesis through email, she always replies me very soon with useful answers, even in the evening and on non-working days. Last but not least, I thank my family and my friends for their constant support on my thesis and my study in TUM.

Abstract

In the 21th century, the importance of finite mixture models in the statistical analysis of data keeps increasing, so that the number of articles on mixture model applications appearing in the statistical and general scientific literature increases steadily, for example, cluster analysis, unsupervised pattern recognition, speech recognition, medical imaging and other applications.

The expectation and maximization (EM) algorithm is a well-known and convenient way for parameter estimation in mixture models. However, the EM algorithm is an iterative algorithm requiring starting values. Different starting values for the EM algorithm can significantly impact the resulting solution. In addition to initialization strategies, the extensions of EM algorithm with different iteration processes from classic EM algorithm also affect the resulting solution. Furthermore, the majority of finite mixture models cannot capture clusters, which are non-elliptical and asymmetric tail dependencies. Due to the higher flexibility of vine copulas, the vine copula mixture model (VCMM) algorithm proposed by Sahin and Czado [2021] is more suitable for modelling Non-Gaussian multivariate data with clusters.

This thesis aims to study the performance of the classic Gaussian mixture model (GMM) algorithm and the vine copulas mixture model (VCMM) algorithm with different EM algorithms and initialization strategies for clustering data with different characteristics and real data sets. According to our results, clustering Gaussian data with GMM algorithm using different EM algorithms and initialization strategies both have a significant effect on classification rate. For the best fit, we recommend the expectation/conditional maximisation either (ECME) algorithm together with the optimization method Nelder-Mead and initialization by k-means clustering. However, the is not the case in VCMM algorithm. For clustering data with various characteristics and real data sets by VCMM algorithm, different EM algorithms don't affect the performance regarding to classification rate for clustering significantly, but computation time. We found that heuristic based optimization method (Nelder-Maad) is taking more time than with gradient based optimization method (BFGS) in many situations. Moreover, we found that some initialization strategies for VCMM algorithm outperform other strategies for clustering data with different characteristics. The result is summarised as a flow chart in the Figure 3.3.19 and the recommendation of the initialization strategy for data clustering in the Figure 3.3.20 for clustering fit over GMM in two data sets and the performance of the selected models from the Figure 3.3.20 for clustering the two real data sets.

1	Intr	roduction	1
2	The	eoretical Background	1
2.1 Parameter Estimation Techniques			
		2.1.1 Maximum Likelihood Estimation	1
		2.1.2 The Classical Expectation-Maximization (EM) Algorithm	3
		2.1.3 Extensions of the EM Algorithm	7
	2.2	Optimization	12
	2.3	Copulas	13
		2.3.1 Concept of Copulas	14
		2.3.2 Vine Copulas	25
	2.4	Mixture Models	27
		2.4.1 Mixture Model Formulation and Parameter Estimation	27
		2.4.2 Multivariate Gaussian Mixture Models (GMM)	32
		2.4.3 Vine Copula Mixture Models (VCMM)	35
	2.5	Clustering using Gaussian Mixture Models	39
		2.5.1 Characteristics of clusters in model based Gaussian mixture modelling	39
		2.5.2 Issues of singularity and clusters collapse in Gaussian mixture modelling for ECME $$	44
	2.6	Performance measures	48
3	Sim	ulation study	50
Ŭ	3.1	EM Algorithm for GMM algorithm	50
	0.1	3.1.1 Data simulation and the experiment setup	50
		3.1.2 Result visualisation and the performance	61
	3.2	EM Algorithm and initialization strategies for GMM algorithm	70
	3.3	EM Algorithm and initialization strategies for VCMM algorithm	78
	0.0	3.3.1 Data simulation and the experiment setup	78
		3.3.2 Result visualisation and the performance	90
	ъ		110
4	Rea	al data sets	119
	4.1		119
	4.2	Introduction to the modelling methods for clustering	123
	4.3	Result and performance of clustering	124
5	Con	nclusion	131
A	App	pendices	132
	А	Appendix for Section 3.2 GMM Algorithm	132
		A.1 Box plots of the classification rate	132
		A.2 Box plots of the normalised BIC	133
		A.3 Box plots of the computation time	134
	В	Appendix for Section 3.3 VCMM Algorithm	135
		B.1 Abbreviation for marginal distributions	135
		B.2 Abbreviation for copula families	135
		B.3 Box plots of the classification rate	136
		B.4 Box plots of the normalised BIC	141
		B.5 Box plots of the computation time	146
		B.6 Ranking of the performance of EM algorithms and initialization strategies	151
C Appendix for Section 4 Real data sets		Appendix for Section 4 Real data sets	158
		C.1 Estimated vine copula model for AIS data	158

References		172
C.4	Pair plots of BCW data with initialization and final clustering	167
C.3	Pair plots of AIS data with initialization and final clustering	162
C.2	Estimated vine copula model for BCW data	160

1 Introduction

This thesis consists of five parts. In Chapter 2, we give the necessary theoretical background, Chapter 3 contains the simulation studies, Chapter 4 does the clustering for real data sets, Chapter 5 concludes.

More specifically, Chapter 2 mainly introduces the mathematical definitions, theory and methods used throughout the thesis. Chapter 2 consists of six subsections. In Section 2.1, we introduce maximum likelihood estimation and discuss in detail the expectation maximization (EM) algorithm and its extensions. In Section 2.2, the method of Lagrange multipliers for solving optimization problems is introduced. Section 2.3 gives an overview of copulas and vine copulas. The formulation for the Gaussian mixture model (GMM) and the vine copulas mixture model (VCMM) and their corresponding steps for EM algorithms are presented in Section 2.4. Section 2.5 describes the characteristics for clusters in model based Gaussian mixtures. Further we discuss modelling issues we encounter some times when we implement the extension of EM algorithm - ECME. Section 2.6 discusses some performance measures used in the simulation study for performance assessment.

Chapter 3 shows the required steps of how the simulation experiments are implemented. There are three subsections in Section 3. In Section 3.1, four data settings following a Gaussian mixture model are generated for testing the performance of GMM algorithm by using different EM algorithms. In Section 3.2, we will use the same data settings as in the Section 3.1 for performance assessment with different initialization strategies. In the last Section 3.3, data sets for nine different settings are generated to assess the performance of the VCMM algorithm, with using different EM algorithms and initialization strategies and the best initialization strategy is suggested for the data with different characteristics.

In Chapter 4, two real data sets with two clusters are classified by GMM and VCMM algorithm, different EM algorithms and initialization strategies for performance assessment.

We will conclude the thesis in Chapter 5.

2 Theoretical Background

In this chapter, we will introduce mathematical concepts, definitions and theories used in the simulation studies.

2.1 Parameter Estimation Techniques

In this section, we will introduce some methods for parameter estimation in statistical modelling. Maximum likelihood estimation is a well-known approach to do so, but it is quite challenging on data in the presence of missing variables, for example, mixture models. Because our paper mainly focus on mixture models, EM algorithm and its extension as the iterative algorithms will be introduced and used for the parameter estimation in this thesis.

2.1.1 Maximum Likelihood Estimation

We begin by assuming that $X \in \mathbb{R}^p$ is the random vector and the corresponding random sample matrix $\mathcal{X} = [x_1, x_2, ..., x_n] \in \mathbb{R}^{p \times n}$ of n independent and identically distributed (i.i.d.) observations being realizations of X is given. Furthermore, it is assumed that X follows cumulative distribution function (cdf) $F(x; \theta)$ with probability density function (pdf) $f(x; \theta)$ with corresponding parameters θ . The parameter $\theta = [\theta_1, \theta_2, ..., \theta_k]^T \in \Theta$ is a vector of unknown parameters needed to be estimated and Θ denotes the parameter space.

In statistics, maximum likelihood estimation (MLE) is one of the common technique to estimate the

parameter θ of the (pdf), by maximizing the likelihood of the realization of the sample. Intuitively, the estimated value of the parameter by the MLE is the one most likely to have produced the sample or observations.

The likelihood function is defined as

$$L(\boldsymbol{\theta} \mid \boldsymbol{x}) = L(\boldsymbol{\theta} \mid \boldsymbol{x_1}, \boldsymbol{x_2}, ..., \boldsymbol{x_n}) := \prod_{i=1}^n f(\boldsymbol{x_i} \mid \boldsymbol{\theta}),$$

or equivalently the corresponding log-likelihood function given by

$$l(\boldsymbol{\theta} \mid \boldsymbol{x}) = l(\boldsymbol{\theta} \mid \boldsymbol{x_1}, \boldsymbol{x_2}, ..., \boldsymbol{x_n}) := \sum_{i=1}^{n} \ln(f(\boldsymbol{x_i} \mid \boldsymbol{\theta})).$$
(2.1.1)

Based on the definition of the MLE mentioned above, the estimated parameter θ_{MLE} can be denoted as

$$\hat{\boldsymbol{\theta}}_{MLE} = \operatorname{arg\,max}_{\boldsymbol{\theta} \in \Theta} l(\boldsymbol{\theta} \mid \boldsymbol{x_1}, \boldsymbol{x_2}, ..., \boldsymbol{x_n}).$$

Since the logarithm is a monotonic function, the maximum of log-likelihood function occurs at the same value of θ as the maximum of likelihood.

We can obtain the estimated parameter $\hat{\theta}_{MLE}$ by taking the first derivative of the log-likelihood function with respect to θ , and finding out the solution of the following equation by setting the values of the partial derivatives vector to the zero vector.

$$\frac{\partial l(\boldsymbol{\theta} \mid \boldsymbol{x})}{\partial \boldsymbol{\theta}} = \boldsymbol{0}$$

Because the log-likelihood function leads to a summation structure for which it is easier to take the derivatives, we prefer to maximize the log-likelihood function instead of likelihood function.

Example 2.1.1 (MLE of univariate normal distribution)

The probability density function of univariate normal distribution with mean μ and variance σ^2 is:

$$f(x) = \frac{1}{\sigma\sqrt{2\pi}} \exp\left(-\frac{1}{2}\left(\frac{x-\mu}{\sigma}\right)^2\right)$$

Suppose that $x_1, x_2, ..., x_n$ represents n random samples (i.i.d) from an univariate normal distribution with mean μ and variance σ^2 . The likelihood for the samples with $\boldsymbol{\theta} = (\mu, \sigma^2)$ is given by

$$L(\boldsymbol{\theta} \mid \boldsymbol{x}) = L(\mu, \sigma^2 \mid x_1, x_2, ..., x_n)$$

$$=\prod_{i=1}^{n} \frac{1}{\sigma\sqrt{2\pi}} \exp\left(-\frac{1}{2}\left(\frac{x_i-\mu}{\sigma}\right)^2\right) = \left(\frac{1}{\sigma\sqrt{2\pi}}\right)^n \exp\left(-\frac{1}{2\sigma^2}\sum_{i=1}^{n} (x_i-\mu)^2\right)$$

and the corresponding log-likelihood is

$$l(\boldsymbol{\theta} \mid \boldsymbol{x}) = \ln L(\boldsymbol{\theta} \mid \boldsymbol{x}) = -\frac{n}{2} \ln (2\pi) - n \ln \sigma - \frac{1}{2\sigma^2} \sum_{i=1}^{n} (x_i - \mu)^2.$$
(2.1.2)

In order to obtain the estimated mean $\hat{\mu}$, we can maximize the log-likelihood by taking the first derivative with respect to μ and then set to 0.

$$\frac{\partial l}{\partial \mu} = -\frac{1}{2\sigma^2} \sum_{i=1}^n \left(x_i - \mu \right)^2 = 0$$

Solving this equation, we get

$$\hat{\mu} = \frac{\sum_{i=1}^{n} x_i}{n}$$

Similarly, to obtain the estimated $\hat{\sigma}$, we take the first derivatives of the log-likelihood with respect to σ and then set to 0. We have

$$\frac{\partial l}{\partial \sigma} = -\frac{n}{\sigma} + \sum_{i=1}^{n} (x_i - \mu)^2 \sigma^{-3} = 0.$$

Solving this equation, we get

$$\hat{\sigma}^2 = \frac{\sum_{i=1}^n (x_i - \hat{\mu})^2}{n}$$

Maximum likelihood estimation has many advantages and works in many situations. However, the limitation of maximum likelihood estimation is that we assume the data used for parameter estimation is fully observable. In other words, if we have missing data and/or latent variables, which are unobserved or hidden variables influencing other random variables, then computing a maximum likelihood estimate becomes hard. [Murphy, 2012] In Section 2.1.2, another approach for parameter estimation in the presence of latent variables called **expectation-maximization (EM) algorithm** is introduced.

2.1.2 The Classical Expectation-Maximization (EM) Algorithm

The expectation-maximization is a method for deriving an iterative procedure, called EM algorithm, to maximize the likelihood function in the situation of missing data and/or latent variables.

We let the complete data consists of $y_{comp} = (x, z)$, where x denotes the observed but incomplete data and z denotes the unobserved or missing data. The complete-data log likelihood is then denoted by $l(\theta; x, z)$, where θ is the unknown parameter vector for which we wish to find the ML estimate. In the presence of missing data, estimating the parameters by MLE is difficult. Since the complete data y_{comp} is not observed, evaluation and maximization of complete-data log likelihood l are not possible. In this situation, the EM algorithm is an useful method to solve this problem.

In general, the EM algorithm attempts to maximize $l(\boldsymbol{\theta} \mid \boldsymbol{y}_{comp})$, which is the log-likelihood function of the complete data, iteratively, by replacing the complete data log likelihood by its conditional expectation given the observed data \boldsymbol{x} . Because the iteration repeats an expectation step (E step) followed by a maximization step (M step), that is the reason why it is named expectation-maximization (EM) algorithm and it is defined as follows:

E-step: The E-step of the EM algorithm computes the expected value of $l(\boldsymbol{\theta} \mid \boldsymbol{Y}_{comp} = (\boldsymbol{X}, \boldsymbol{Z}))$ given the observed data \boldsymbol{x} , and the current parameter estimate $\boldsymbol{\theta}_{old}$. In particular, we define

$$\begin{aligned} Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old}) &\coloneqq E[l(\boldsymbol{\theta} \mid \boldsymbol{Y}_{comp}) \mid \boldsymbol{\theta}_{old}, \boldsymbol{X} = \boldsymbol{x}] \\ &= \int l(\boldsymbol{\theta} \mid \boldsymbol{y}_{comp} = (\boldsymbol{x}, \boldsymbol{z})) \; p(\boldsymbol{z} \mid \boldsymbol{\theta}_{old}, \boldsymbol{x}) \; d\boldsymbol{z}, \end{aligned}$$

where $p(\boldsymbol{z} \mid \boldsymbol{\theta}_{old}, \boldsymbol{x})$ is the conditional density for the missing data \boldsymbol{Z} given the observed data \boldsymbol{x} , and assuming $\boldsymbol{\theta} = \boldsymbol{\theta}_{old}$.

M-step: The M-step of the EM algorithm chooses $\boldsymbol{\theta}_{new}$ to be any value of $\boldsymbol{\theta}$ that maximizes $Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old})$. That is

$$Q(\boldsymbol{\theta}_{new} \mid \boldsymbol{\theta}_{old}) \geq Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old}),$$

or equivalently, the θ_{new} can be denoted as

$$\boldsymbol{\theta}_{new} := \arg \max_{\boldsymbol{\theta}} Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old}).$$

More specifically, on the first iteration, we first let θ_{old} to be some initial value of θ . After one E and M-step is carried out, we set $\theta_{old} = \theta_{new}$. The two steps will be repeated until the stopping conditions fulfills. Two stopping conditions will be introduced in the following:

Stopping condition 1 : Convergence of parameter estimation θ

The EM algorithm stops if two successive estimate differ smaller by a prespecified ϵ , such that

$$\|\boldsymbol{\theta}_{new} - \boldsymbol{\theta}_{old}\| < \epsilon \tag{2.1.3}$$

Stopping condition 2 : Convergence of log-likelihood $l(\theta \mid x)$

The EM algorithm stops when the relative change of the log-likelihood between two successive iterations is smaller than a prespecified ϵ , such that

$$\frac{l(\boldsymbol{\theta}_{new} \mid \boldsymbol{x}) - l(\boldsymbol{\theta}_{old} \mid \boldsymbol{x})}{l(\boldsymbol{\theta}_{old} \mid \boldsymbol{x})} < \epsilon$$
(2.1.4)

At that time, we can conclude that θ_{new} is the optimal estimate of the EM process.

Example 2.1.2 (Missing Data in an Univariate Normal Sample)

Suppose $\boldsymbol{x} := (y_1, ..., y_n)^T$ is *n* i.i.d. samples from univariate nominal distribution with mean μ and variance σ^2 . In Example 2.1.1, the log-likelihood was derived in Equation (2.1.2). However, in this example, we want to perform the EM algorithm instead. To do so, we assume the complete data is $\boldsymbol{y}_{comp} = (\boldsymbol{x}, \boldsymbol{z}) := (y_1, ..., y_n, y_{n+1}, ..., y_m)^T$, so that $\boldsymbol{z} := (y_{n+1}, ..., y_m)^T$ are missing, where \boldsymbol{y}_{comp} are assumed to be i.i.d.

E-step: The E step of the EM algorithm computes the expected value of $l(\boldsymbol{\theta} \mid \boldsymbol{y}_{comp} = (\boldsymbol{x}, \boldsymbol{Z}))$ given the observed data \boldsymbol{x} and the current parameter estimate $\boldsymbol{\theta}_{old} = (\mu_{old}, \sigma_{old}^2)$. We first compute $l(\boldsymbol{\theta} \mid \boldsymbol{y}_{comp})$:

$$\begin{split} l(\boldsymbol{\theta} \mid \boldsymbol{y}_{comp} = (\boldsymbol{x}, \boldsymbol{z})) &= -\frac{m}{2} \ln (2\pi) - \frac{m}{2} \ln \sigma^2 - \frac{1}{2\sigma^2} \sum_{i=1}^m (y_i - \mu)^2 \\ &= -\frac{m}{2} \ln (2\pi) - \frac{m}{2} \ln \sigma^2 - \frac{1}{2\sigma^2} \sum_{i=1}^m y_i^2 + \frac{\mu}{\sigma^2} \sum_{i=1}^m y_i - \frac{m\mu^2}{2\sigma^2} \\ &= -\frac{m}{2} \ln (2\pi) - \frac{m}{2} \ln \sigma^2 - \frac{m\mu^2}{2\sigma^2} - \frac{1}{2\sigma^2} \left(\sum_{i=1}^n y_i^2 + \sum_{i=n+1}^m y_i^2 \right) \\ &+ \frac{\mu}{\sigma^2} \left(\sum_{i=1}^n y_i + \sum_{i=n+1}^m y_i \right) \end{split}$$

Since

$$E[Y_i \mid \boldsymbol{x}, \boldsymbol{\theta}_{obs}] = \begin{cases} y_i & i = 1, ..., n \\ \mu_{old} & i = n + 1, ..., m \end{cases}$$
$$E[Y_i^2 \mid \boldsymbol{x}, \boldsymbol{\theta}_{obs}] = \begin{cases} y_i^2 & i = 1, ..., n \\ \mu_{old}^2 + \sigma_{old}^2 & i = n + 1, ..., m, \end{cases}$$

we have

 $Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old}) \coloneqq E[l(\boldsymbol{\theta} \mid \boldsymbol{x}, \boldsymbol{Z}) \mid \boldsymbol{x}, \boldsymbol{\theta}_{old}]$

$$= -\frac{m}{2}\ln(2\pi) - \frac{m}{2}\ln\sigma^2 - \frac{m\mu^2}{2\sigma^2} - \frac{1}{2\sigma^2}\sum_{i=1}^n y_i^2 - \frac{(m-n)}{2\sigma^2}(\mu_{old}^2 + \sigma_{old}^2) + \frac{\mu}{\sigma^2}\sum_{i=1}^n y_i - \frac{(m-n)\mu_{old}\mu}{\sigma^2}$$

We define

$$S_1(\mu_{old}) = (m-n)\mu_{old} + \sum_{i=1}^n y_i$$

and

$$S_2(\mu_{old}, \sigma_{old}^2) := (m-n)(\sigma_{old}^2 + \mu_{old}^2) + \sum_{i=1}^n y_i^2.$$

$$\Rightarrow Q(\theta \mid \theta_{old}) = -\frac{m}{2}\ln(2\pi) - \frac{m}{2}\ln\sigma^2 - \frac{m\mu^2}{2\sigma^2} - \frac{1}{2\sigma^2}S_2(\mu_{old}, \sigma_{old}^2) + \frac{\mu}{\sigma^2}S_1(\mu_{old})$$
(2.1.5)

M-step: The M-step of the EM algorithm chooses θ_{new} to be any value of θ that maximizes $Q(\theta \mid \theta_{old})$. The θ_{new} can be denoted as

$$\boldsymbol{\theta}_{new} := \arg \max_{\boldsymbol{\theta}} Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old}). \tag{2.1.6}$$

The M step here, taking the derivatives with respect to μ and σ^2 separately, does not maximize $Q(\theta \mid \theta_{old})$ over the parameter space Θ jointly, which is the definition of the M step in the classical EM algorithm. Therefore, this method is actually ECM algorithm which will be introduced later. But, at this moment, we regard this as a standard EM algorithm because this is the way how people estimate the parameters by EM algorithm.

To get μ_{new} needed in Equation (2.1.6), take the derivatives of $Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old})$ with respect to μ and set to 0:

$$\frac{S_1(\mu_{old})}{\sigma^2} - \frac{m\mu}{\sigma^2} = 0$$

$$\mu_{new} = \frac{S_1(\mu_{old})}{m}$$

$$= \frac{\sum_{i=1}^n y_i + (m-n)\mu_{old}}{m}$$
(2.1.7)

To get σ_{new}^2 needed in Equation (2.1.6), take the derivatives of $Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old})$ with respect to σ^2 and set to 0:

$$-\frac{m}{2\sigma^{2}} + \frac{1}{2(\sigma^{2})^{2}} S_{2}(\mu_{old}, \sigma_{old}^{2}) - \frac{\mu}{(\sigma^{2})^{2}} S_{1}(\mu_{old}) + \frac{m\mu^{2}}{2(\sigma^{2})^{2}} = 0$$

$$-m\sigma^{2} + S_{2}(\mu_{old}, \sigma_{old}^{2}) - 2\mu S_{1}(\mu_{old}) + m\mu^{2} = 0$$

$$-m\sigma^{2} + S_{2}(\mu_{old}, \sigma_{old}^{2}) - 2m\mu^{2} + m\mu^{2} = 0$$

$$\implies \sigma_{new}^{2} = \frac{S_{2}(\mu_{old}, \sigma_{old}^{2})}{m} - \mu^{2}$$

$$= \frac{\sum_{i=1}^{n} y_{i}^{2} + (m-n)(\sigma_{old}^{2} + \mu_{new}^{2})}{m} - \mu_{new}^{2}$$
(2.1.8)

To illustrate the application of the EM algorithm to this type of problem, we now apply it to the following random data set. Here, there are m = 15 observations, but 5 of them are missing which are indicated by NA, thus observed number n = 10.

$$\boldsymbol{y} = \{5, 3, 10, 0, -4, 5, 6, 3, -5, 7, \text{NA}, \text{NA}, \text{NA}, \text{NA}, \text{NA}\}$$

We assume that the observations follow normal distribution, so we can apply the expression in Equation (2.1.7) and Equation (2.1.8) to estimate its mean μ and variance σ^2 . Also, we choose $\mu^{(0)} = 10$ and $\sigma^{2(0)} = 10$ as the initial values for the iteration and the stopping condition 1 from Equation (2.1.3) is used. Table 2.1.1 shows the convergence results of applying EM to this problem and we can know that, as iteration t increases, the $|\mu^{(t)} - \mu^{(t-1)}|$ and $|\sigma^{2(t)} - \sigma^{2(t-1)}|$ in each iteration tend to 0, so that $\mu^{(t)}$ and $\sigma^{2(t)}$ converge. Therefore, we can conclude that, by the EM algorithm, the estimated $\hat{\mu} \approx 3$ and estimated $\hat{\sigma}^2 \approx 20.4$ are the optimal estimate of EM process.

Actually, each EM step monotonically increases the likelihood function. This is the reason why repeating EM steps until convergence can maximize the likelihood function. To prove this, the definition of the E-step and M-step can be used. In the M-step, θ_{new} is chosen to maximizes $Q(\theta \mid \theta_{old})$, that is

$$Q(\boldsymbol{\theta}_{new} \mid \boldsymbol{\theta}_{old}) \geq Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old}) \quad \forall \boldsymbol{\theta}$$

And θ can be chosen as θ_{old} , such that

$$Q(\boldsymbol{\theta}_{new} \mid \boldsymbol{\theta}_{old}) \geq Q(\boldsymbol{\theta}_{old} \mid \boldsymbol{\theta}_{old}).$$

Iteration t	$\mu^{(t)}$	$\mid \mu^{(t)} - \mu^{(t-1)} \mid$	$\sigma^{2^{(t)}}$	$\mid \sigma^{2^{(t)}} - \sigma^{2^{(t-1)}} \mid$
0	10	-	10	-
1	5.333333	4.666667	27.82222	17.82222
2	3.777778	1.555555	24.08395	3.73827
3	3.259259	0.518519	21.76241	2.32154
4	3.08642	0.172839	20.86907	0.89334
5	3.028807	0.057613	20.55802	0.31105
6	3.009602	0.019205	20.45286	0.10516
7	3.003201	0.006401	20.41764	0.03522
8	3.001067	0.002134	20.40588	0.01176
9	3.000356	0.000711	20.40196	0.00392
10	3.000119	0.000237	20.40065	0.00131

Table 2.1.1: The EM iteration values for Example 2.1.2

In the E-step, we defined $Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old}) := \int l(\boldsymbol{\theta} \mid \boldsymbol{y}_{comp}) \ p(\boldsymbol{z} \mid \boldsymbol{\theta}_{old}, \boldsymbol{x}) \ d\boldsymbol{z}$. So, we have

$$\int l(\boldsymbol{\theta}_{new} \mid \boldsymbol{y}_{comp}) \ p(\boldsymbol{z} \mid \boldsymbol{\theta}_{old}, \boldsymbol{x}) \ d\boldsymbol{z} \geq \int l(\boldsymbol{\theta}_{old} \mid \boldsymbol{y}_{comp}) \ p(\boldsymbol{z} \mid \boldsymbol{\theta}_{old}, \boldsymbol{x}) \ d\boldsymbol{z},$$

which implies

$$\int \left[\ l(\boldsymbol{\theta}_{new} \mid \boldsymbol{y}_{comp}) - l(\boldsymbol{\theta}_{old} \mid \boldsymbol{y}_{comp}) \ \right] \ p(\boldsymbol{z} \mid \boldsymbol{\theta}_{old}, \boldsymbol{x}) \ d\boldsymbol{z} \ge 0$$

In the Section 2.1.1, log-likelihood function $l(\boldsymbol{\theta} \mid \boldsymbol{x})$ is defined as $\ln[\prod_{i=1}^{n} f(\boldsymbol{x}_i \mid \boldsymbol{\theta})]$. For simplicity, we define $l(\boldsymbol{\theta} \mid \boldsymbol{y})$ as $\ln p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta})$, where $p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta})$ is the probability density function of complete data \boldsymbol{y}_{comp} .

$$\int \ln \left[\frac{p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta}_{new})}{p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta}_{old})} \right] \ p(\boldsymbol{z} \mid \boldsymbol{\theta}_{old}, \boldsymbol{x}) \ d\boldsymbol{z} \ge 0$$

By $\ln x \le x - 1$ for all $x \ge 0$ with equality if and only if x = 1, we have

$$\int \left[\frac{p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta}_{new})}{p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta}_{old})} - 1 \right] p(\boldsymbol{z} \mid \boldsymbol{\theta}_{old}, \boldsymbol{x}) d\boldsymbol{z} \ge 0$$

$$\iff \int \left[\frac{p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta}_{old})}{p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta}_{old})} - 1 \right] \frac{p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta}_{old})}{p(\boldsymbol{x} \mid \boldsymbol{\theta}_{old})} d\boldsymbol{z} \ge 0$$

$$\iff \frac{1}{p(\boldsymbol{x} \mid \boldsymbol{\theta}_{old})} \int \left[p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta}_{new}) - p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta}_{old}) \right] d\boldsymbol{z} \ge 0$$

Since $\int p(\boldsymbol{y}_{comp} \mid \boldsymbol{\theta}) \, d\boldsymbol{z} = \int p(\boldsymbol{x}, \boldsymbol{z} \mid \boldsymbol{\theta}) \, d\boldsymbol{z} = p(\boldsymbol{x} \mid \boldsymbol{\theta}),$

$$p(\boldsymbol{x} \mid \boldsymbol{\theta}_{new}) - p(\boldsymbol{x} \mid \boldsymbol{\theta}_{old}) \ge 0$$
$$\implies L(\boldsymbol{\theta}_{new} \mid \boldsymbol{x}) \ge L(\boldsymbol{\theta}_{old} \mid \boldsymbol{x})$$

We can see that, in the M-step, finding a $\boldsymbol{\theta}$ by maximizing $Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old})$ over $\boldsymbol{\theta}$ is equivalent to maximize the likelihood function $L(\boldsymbol{\theta} \mid \boldsymbol{x})$ over $\boldsymbol{\theta}$, such that

$$Q(\boldsymbol{\theta}_{new} \mid \boldsymbol{\theta}_{old}) \ge Q(\boldsymbol{\theta}_{old} \mid \boldsymbol{\theta}_{old}) \quad \forall \boldsymbol{\theta} \implies L(\boldsymbol{\theta}_{new} \mid \boldsymbol{x}) \ge L(\boldsymbol{\theta}_{old} \mid \boldsymbol{x})$$
(2.1.9)

This is the reason why the EM algorithm works, such that each iteration increasing the likelihood.

In this section, we have shown that the advantages of the EM algorithm: 1) In the presence of missing data and/or latent variables, the complete-data maximum likelihood estimation is often computationally simple. 2) Its convergence is stable, because of each iteration increasing the likelihood. However, the classic EM algorithm also has its limitations, for example, when the associated complete-data maximum likelihood estimation itself is complicated, EM is less attractive because the M-step is computationally unattractive. [Meng and Rubin, 1993] In the next Section 2.1.3, some extensions of the EM algorithm improving its limitations will be introduced.

2.1.3 Extensions of the EM Algorithm

In this section, further modifications and extensions to the EM algorithm are considered. In particular, we focus on 1. Expectation Conditional Maximization (ECM) algorithm, 2. multi-cycle ECM (MCECM) algorithm and 3. Expectation/Conditional Maximization Either (ECME) algorithm.

1. The Expectation Conditional Maximization (ECM) Algorithm

In the M step of the classical EM algorithm, it is required to choose θ_{new} which maximizes $Q(\theta \mid \theta_{old})$. In the Example 2.1.2, we want to choose μ , σ^2 together to maximize $Q(\theta \mid \theta_{old})$. Actually, we just can do it numerically because a analytic solution to maximize a function for 2 parameters or more does not exist in general. The M step in the Example 2.1.2, taking the derivatives with respect to μ and σ^2 separately (or fixing the μ to estimate the σ^2), cannot really maximize $Q(\theta \mid \theta_{old})$ over the parameter space Θ , which is the definition of the M step in the classical EM algorithm. Strictly speaking, the method shown in the example is not the classical EM algorithm. Because of the convenience of finding a analytic solution, people do the so-called "EM algorithm" in that way and we still call it "EM algorithm".

Due to the limitation of the classical EM algorithm in the M step, Meng and Rubin [1993] proposed a modification of the EM algorithm called expectation conditional maximization (ECM) algorithm. In the ECM algorithm, the M step is replaced by several computationally simpler conditional maximization (CM) steps. Each CM-step maximizes $Q(\theta \mid \theta_{old})$ found on the preceding E-step subject to the constraints on θ . compared to the classic EM algorithm, ECM algorithm is often faster, simpler and more stable for maximization in the M step, because the CM maximization are over smaller dimensional spaces.

More precisely, the expectation conditional maximization (ECM) algorithm at iteration t + 1 is defined as follows:

E-step: The E step in the classic EM algorithm and the ECM algorithm are exactly the same. Compute the expected value of $l(\boldsymbol{\theta} \mid \boldsymbol{y}_{comp} = (\boldsymbol{x}, \boldsymbol{Z}))$ given the observed data \boldsymbol{x} , and the current parameter estimate $\boldsymbol{\theta}^{(t)}$:

$$Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t)}) \coloneqq E[l(\boldsymbol{\theta} \mid \boldsymbol{x}, \boldsymbol{Z}) \mid \boldsymbol{\theta}^{(t)}, \boldsymbol{x}]$$

CM-steps: The M step in the ECM algorithm is replaced by S > 1 steps and each step is called CM step, so that there are S CM steps in each iteration. At the sth CM-step of the (t + 1)th iteration, $\theta^{(t+\frac{s}{S})}$ is chosen to maximize $Q(\theta \mid \theta^{(t)})$ which is subject to the constraint $g_s(\theta) = g_s(\theta^{(t+\frac{s-1}{S})})$, or equivalently, the $\theta^{(t+\frac{s}{S})}$ is defined as

$$\begin{split} \boldsymbol{\theta}^{(t+\frac{s}{S})} &:= \arg \max_{\boldsymbol{\theta}} Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t)}) \\ &\text{subject to } g_s(\boldsymbol{\theta}) = g_s(\boldsymbol{\theta}^{(t+\frac{s-1}{S})}) \end{split}$$

where $G = \{g_s(\boldsymbol{\theta}), s = 1, ..., S\}$ is a set of S preselected functions. Thus, $\boldsymbol{\theta}^{(t+\frac{s}{S})}$ satisfies

$$Q(\boldsymbol{\theta}^{(t+\frac{s}{S})} \mid \boldsymbol{\theta}^{(t)}) \ge Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t)}) \text{ for all } \boldsymbol{\theta} \in \{\boldsymbol{\theta} \in \boldsymbol{\Theta} : g_s(\boldsymbol{\theta}) = g_s(\boldsymbol{\theta}^{(t+\frac{s-1}{S})})\}.$$
(2.1.10)

where $\boldsymbol{\Theta}$ is parameter space.

In the final CM-step of the (t + 1)th iteration, the value $\theta^{(t+\frac{S}{S})} = \theta^{(t+1)}$, is taken to be the input on the (t+2)th iteration.

From Equation (2.1.10), we know that

$$Q(\boldsymbol{\theta}^{(t+1)} \mid \boldsymbol{\theta}^{(t)}) \ge Q(\boldsymbol{\theta}^{(t+\frac{S-1}{S})} \mid \boldsymbol{\theta}^{(t)})$$
$$\ge Q(\boldsymbol{\theta}^{(t+\frac{S-2}{S})} \mid \boldsymbol{\theta}^{(t)})$$
$$\vdots$$
$$\ge Q(\boldsymbol{\theta}^{(t)} \mid \boldsymbol{\theta}^{(t)})$$

As noted before from Equation (2.1.9) in the Section 2.1.2 that each EM step monotonically increases the likelihood function, we have

$$L(\boldsymbol{\theta}^{(t+1)} \mid \boldsymbol{x}) \geq L(\boldsymbol{\theta}^{(t)} \mid \boldsymbol{x}).$$

We can conclude that the likelihood function monotonically increases in each iteration for the ECM algorithm. Therefore, the ECM algorithm still works like the classical EM algorithm, even faster, more stable.

Now, we are going to show more detail about the implementation of the ECM algorithm. Firstly, we need to decide the number of S CM steps and the order of the parameters that we are going to update in each CM step, but there is no rule to choose it. Normally, S is equal to the number of parameters because ideally we want to update all the parameters once in each iteration. Also, we partition the parameter θ into subvectors $\theta = \{\theta_1, ..., \theta_S\}$ in order to decide the order of the parameters which will be updated in each step. According to the sub-vectors $\theta = \{\theta_1, ..., \theta_S\}$, θ_1 will be updated first by maximization with respect to θ_1 , with all other parameters held fixed and so on and so fort. Basically, $G = \{g_s(\theta), s = 1, ..., S\}$ is just chosen for setting other parameters constant as the constraints in an optimization problem in each CM step. More detail will be in the Example 2.1.3.

Example 2.1.3 (Missing Data in an Univariate Normal Sample (ECM))

In the Example 2.1.2, we showed the steps for the missing data in an univalue normal sample using the classical EM algorithm. Here, we are going to show you the typical steps of the ECM algorithm for this problem and the difference between the classical EM algorithm and the ECM algorithm.

E step of **EM**, **ECM** at iteration t + 1

E step of the ECM algorithm and the EM algorithm are the same. From Equation (2.1.5), we have

$$Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t)}) \coloneqq E[l(\boldsymbol{\theta} \mid \boldsymbol{x}, \boldsymbol{Z}) \mid \boldsymbol{\theta}^{(t)}, \boldsymbol{x}]$$

= $-\frac{m}{2} \ln \sigma^2 - \frac{1}{2\sigma^2} S_2^{(t)} + \frac{\mu}{\sigma^2} S_1^{(t)} - \frac{m\mu^2}{2\sigma^2} + \text{constant},$

where $S_1^{(t)} = \sum_{i=1}^n y_i + (m-n)\mu^{(t)}$ and $S_2^{(t)} = \sum_{i=1}^n y_i^2 + (m-n)(\sigma^{2(t)} + {\mu^{(t)}}^2)$.

M step of EM at iteration t+1

Recalling from Equation (2.1.6), the M-step of the EM algorithm chooses $\boldsymbol{\theta}^{(t+1)} = (\mu^{(t+1)}, \sigma^{2^{(t+1)}})$ that maximizes $Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t)})$. That is

$$\boldsymbol{\theta}^{(t+1)} := \arg \max_{\boldsymbol{\theta}} Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t)}).$$

As we mentioned, we just can solve this optimization problem numerically when there are more than 1 parameters. Therefore, no analytic solution can be found.

CM step of ECM at iteration t+1

Recalling that in normal distribution, there are two parameters with μ and σ^2 . Therefore, we just have two ways to partition them because the order is also taking into account, so that we have case $1 : \theta_1 = \mu$, $\theta_2 = \sigma^2$ and case $2 : \theta_1 = \sigma^2$, $\theta_2 = \mu$. Also, due to S = 2, the M step can be replaced by 2 CM steps.

Case 1 : $\theta_1 = \mu, \ \theta_2 = \sigma^2$

CM-step 1: Due to $\theta_1 = \mu$, the CM-step 1 is to choose $\mu^{(t+\frac{1}{2})}$ by maximizing $Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t)})$ subject to $g_1(\mu, \sigma^2) = g_1(\mu^{(t)}, \sigma^{2^{(t)}})$, where $g_1(\mu, \sigma^2)$ is some selected functions and we will select it a bit later. Therefore, $\mu^{(t+\frac{1}{2})}$ can be denoted as $\mu^{(t+\frac{1}{2})} := \arg \max \ Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t)})$

$$= \arg \max_{\mu} Q(\boldsymbol{\sigma} \mid \boldsymbol{\sigma}^{(1)})$$

= $\arg \max_{\mu} \left(-\frac{m}{2} \ln \sigma^2 - \frac{1}{2\sigma^2} S_2^{(t)} + \frac{\mu}{\sigma^2} S_1^{(t)} - \frac{m\mu^2}{2\sigma^2} + \text{constant} \right)$
subject to $g_1(\mu, \sigma^2) = g_1(\mu^{(t)}, \sigma^{2(t)})$

and other parameters keep constant, so that the "updated" variance is denoted as $\sigma^{2(t+\frac{1}{2})} := \sigma^{2(t)}$.

Now, we would like to choose some constraints $g_1(\mu, \sigma^2) = g_1(\mu^{(t)}, \sigma^{2^{(t)}})$ to make the above optimization can be solved easily, at least by taking the derivatives. The easiest way is to set the parameters other than μ to constant, such that $\sigma^2 = \sigma^{2^{(t)}}$. Formally, $g_1(\mu, \sigma^2) = \sigma^2$ and the conditional optimization problem can be written as

$$\mu^{(t+\frac{1}{2})} := \arg \max_{\mu} \left(-\frac{m}{2} \ln \sigma^{2^{(t)}} - \frac{1}{2\sigma^{2^{(t)}}} S_2^{(t)} + \frac{\mu}{\sigma^{2^{(t)}}} S_1^{(t)} - \frac{m\mu^2}{2\sigma^{2^{(t)}}} + \text{constant} \right)$$

Taking the derivatives with respect to μ and set to 0, we get

$$\mu^{(t+\frac{1}{2})} = \frac{S_1^{(t)}}{m}$$
$$= \frac{\sum_{i=1}^n y_i + (m-n)\mu^{(t)}}{m}.$$

CM-step 2: Similarly, the CM-step 2 is to choose $\sigma^{2^{(t+1)}}$ (or $\sigma^{2^{(t+\frac{2}{2})}}$) by maximizing $Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t+\frac{1}{2})})$ subject to $g_2(\mu, \sigma^2) = g_2(\mu^{(t+\frac{1}{2})}, \sigma^{2^{(t+\frac{1}{2})}})$, where $g_2(\mu, \sigma^2) = \mu$, so that parameters other than σ^2 can keep constant. In this case, $\sigma^{2^{(t+1)}}$ can be defined as

$$\sigma^{2^{(t+1)}} := \arg\max_{\sigma^2} \left(-\frac{m}{2} \ln \sigma^2 - \frac{1}{2\sigma^2} S_2^{(t+\frac{1}{2})} + \frac{\mu^{(t+\frac{1}{2})}}{\sigma^2} S_1^{(t+\frac{1}{2})} - \frac{m(\mu^{(t+\frac{1}{2})})^2}{2\sigma^2} + \text{constant} \right).$$

and other parameters keep constant, so that the "updated" mean is denoted as $\mu^{(t+1)} := \mu^{(t+\frac{1}{2})}$. Taking the derivatives with respect to σ^2 and set to 0, we get

$$\sigma^{2^{(t+1)}} = \frac{S_2^{(t+\frac{1}{2})}}{m} - (\mu^{(t+\frac{1}{2})})^2$$
$$= \frac{\sum_{i=1}^n y_i^2 + (m-n)(\sigma^{2^{(t+\frac{1}{2})}} + (\mu^{(t+\frac{1}{2})})^2)}{m} - (\mu^{(t+\frac{1}{2})})^2.$$

You may notice that the expressions for μ and σ^2 that we found above are the same in the Example 2.1.2, just the notations different. As we mentioned the M steps in that example, fixing the μ to estimate the σ^2 , cannot really maximize $Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}_{old})$ over the parameter space $\boldsymbol{\Theta}$, which is the definition of the M step in the classical EM algorithm. Actually, the method used in the Example 2.1.2 is the ECM algorithm.

For S = 2, we have 2 different orders to update the parameters. This time, we try to update σ^2 first and then μ .

Case 2 : $\theta_1 = \sigma^2, \ \theta_2 = \mu$

The computational steps are exactly the same as the case 1. Just the order is different. So, the steps are skipped here:

CM-step 1:

$$\sigma^{2^{(t+\frac{1}{2})}} := \frac{S_2^{(t)}}{m} - (\mu^{(t)})^2$$
$$= \frac{\sum_{i=1}^n y_i^2 + (m-n)(\sigma^{2^{(t)}} + (\mu^{(t)})^2)}{m} - (\mu^{(t)})^2$$
$$\mu^{(t+\frac{1}{2})} := \mu^{(t)}$$

CM-step 2:

$$\mu^{(t+1)} := \frac{S_1^{(t+\frac{1}{2})}}{m}$$
$$= \frac{\sum_{i=1}^n y_i + (m-n)\mu^{(t+\frac{1}{2})}}{m}$$
$$\sigma^{2^{(t+1)}} := \sigma^{2^{(t+\frac{1}{2})}}$$

In general, assume that there are S parameters in a model. If we just want to update 1 parameter in each CM step, we totally have S! different orders to implement the CM steps. You may be interested in whether the order affects the performance of the parameter estimation. van Dyk and Meng [1997] mentioned that the order that the CM-steps are performed is trivial to change and generally affects how fast the algorithm converges and it is impossible to find an "optimal" order choosing or grouping-choosing rule that will be universally applicable, which means the order of performing CM steps affecting the performance of the parameter estimation.

2. The Multi-cycle Expectation Conditional Expectation (MCECM) Algorithm MCECM algorithm is an extension of ECM algorithm. In the ECM algorithm, E step is just performed once before all the CM steps. But, in the MCECM algorithm, E step is performed before each CM step. Meng and Rubin [1993] mentioned that in some cases, the computation of an E-step may be much cheaper than computation of the CM-steps. Performing an E-step before each CM-step may result in larger increases in likelihood function L per iteration since Q is being updated more often. This is the reason why MCECM may be a better extension of ECM.

Suppose there are S CM steps in each iteration. In MCECM, since E step is performed before each CM step and a cycle is defined by one E step followed by one CM step, where each iteration involves S cycles. This is how multi-cycle Expectation Conditional Expectation is named. More precisely, multi-cycle Expectation Conditional Expectation (MCECM) algorithm at the sth of iteration t + 1 is defined as follows:

E-step: The E step of the MCECM algorithm in the *s*th cycle of iteration t + 1 is to compute the expected value of $l(\boldsymbol{\theta} \mid \boldsymbol{y}_{comp} = (\boldsymbol{x}, \boldsymbol{Z}))$ given the observed data \boldsymbol{x} , and the parameter estimate in the (s - 1)th cycle $\boldsymbol{\theta}^{(t + \frac{s-1}{S})}$:

$$Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t+\frac{s-1}{S})}) \coloneqq E[l(\boldsymbol{\theta} \mid \boldsymbol{x}, \boldsymbol{Z}) \mid \boldsymbol{\theta}^{(t+\frac{s-1}{S})}, \boldsymbol{x}]$$

CM-step: The M step of the MCECM algorithm in the *s*th cycle of iteration t + 1 is to choose $\theta^{(t+\frac{s}{S})}$ to maximize $Q(\theta \mid \theta^{(t+\frac{s-1}{S})})$ which is subject to the constraint $g_s(\theta) = g_s(\theta^{(t+\frac{s-1}{S})})$, or equivalently, the $\theta^{(t+\frac{s}{S})}$ is defined as

$$\begin{split} \boldsymbol{\theta}^{(t+\frac{s}{S})} &:= \arg \max_{\boldsymbol{\theta}} Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t+\frac{s-1}{S})}) \\ &\text{subject to } g_s(\boldsymbol{\theta}) = g_s(\boldsymbol{\theta}^{(t+\frac{s-1}{S})}) \end{split}$$

where $G = \{q_s(\theta), s = 1, ..., S\}$ is a set of S preselected functions. Thus, $\theta^{(t+\frac{s}{S})}$ satisfies

$$Q(\boldsymbol{\theta}^{(t+\frac{s}{S})} \mid \boldsymbol{\theta}^{(t+\frac{s-1}{S})}) \ge Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t+\frac{s-1}{S})}) \text{ for all } \boldsymbol{\theta} \in \{\boldsymbol{\theta} \in \boldsymbol{\Theta} : g_s(\boldsymbol{\theta}) = g_s(\boldsymbol{\theta}^{(t+\frac{s-1}{S})})\}.$$
(2.1.11)

where Θ is parameter space.

In the final Sth cycle of the (t + 1)th iteration, the value $\theta^{(t+\frac{S}{S})} = \theta^{(t+1)}$, is taken to be the input on the (t+2)th iteration.

From Equation (2.1.11), we know that in the sth cycle of MCECM,

$$Q(\boldsymbol{\theta}^{(t+\frac{s}{S})} \mid \boldsymbol{\theta}^{(t+\frac{s-1}{S})}) \ge Q(\boldsymbol{\theta}^{(t+\frac{s-1}{S})} \mid \boldsymbol{\theta}^{(t+\frac{s-1}{S})}).$$

In the Section 2.1.2 from Equation (2.1.9), we have mentioned that EM algorithm monotonically increases the likelihood function, because $Q(\boldsymbol{\theta}_{new} \mid \boldsymbol{\theta}_{old}) \geq Q(\boldsymbol{\theta}_{old} \mid \boldsymbol{\theta}_{old})$ results in $L(\boldsymbol{\theta}_{new} \mid \boldsymbol{x}) \geq L(\boldsymbol{\theta}_{old} \mid \boldsymbol{x})$. Therefore, this inequality leads to

$$L(\boldsymbol{\theta}^{(t+\frac{s}{S})} \mid \boldsymbol{x}) \geq L(\boldsymbol{\theta}^{(t+\frac{s-1}{S})} \mid \boldsymbol{x})$$

Hence, the multi cycle ECM algorithm monotonically increases the likelihood function L, after each cycle. Because each iteration consists of S cycles, likelihood function L also increases in each iteration.

3. The Expectation/Conditional Maximisation Either (ECME) ECME is another extension of ECM algorithm proposed by Liu and Rubin [1994]. They found that this extension is nearly always faster the both the EM and ECM algorithms in terms of the number of iterations. The full name of the ECME algorithm is Expectation/Conditional Maximisation Either. McLachlan and Krishnan [2007] explained that the "either" refers to the fact that, some or all of the CM-steps of the ECM algorithm are replaced by steps that conditionally maximize the actual incomplete-data log likelihood function and not the Q function. Therefore, in the ECME algorithm, each CM step either maximizes the conditional expectation of the complete data log likelihood Q or the actual log likelihood function $l(\theta \mid x)$ from Equation (2.1.1), subject to the same constraints G on θ like in the ECM algorithm.

More precisely, the Expectation/Conditional Maximisation Either (ECME) algorithm at iteration t + 1 is defined as follows:

E-step: The E step in the classic EM algorithm, the ECM algorithm and the ECME algorithm are exactly the same. Compute the expected value of $l(\boldsymbol{\theta} \mid \boldsymbol{y}_{comp} = (\boldsymbol{x}, \boldsymbol{Z}))$ given the observed data \boldsymbol{x} , and the current parameter estimate $\boldsymbol{\theta}^{(t)}$:

$$Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t)}) \coloneqq E[l(\boldsymbol{\theta} \mid \boldsymbol{x}, \boldsymbol{Z}) \mid \boldsymbol{\theta}^{(t)}, \boldsymbol{x}]$$

The M step in the ECME algorithm is replaced by S > 1 steps. Here, we suppose that, in the first (S - 1) steps, the conditional expectation of the complete data log likelihood is maximised and in the last step, the actual log likelihood function $l(\boldsymbol{\theta} \mid \boldsymbol{x})$ from Equation (2.1.1) is maximised.

CM-steps with conditional expectation: At the *s*th CM-step of the (t + 1)th iteration, $\theta^{(t+\frac{s}{S})}$ is chosen to maximize $Q(\theta \mid \theta^{(t)})$ which is subject to the constraint $g_s(\theta) = g_s(\theta^{(t+\frac{s-1}{S})})$, or equivalently, the $\theta^{(t+\frac{s}{S})}$ is defined as $\theta^{(t+\frac{s}{S})} := \arg \max_{\theta} Q(\theta \mid \theta^{(t)})$

$$Q^{(t+\frac{s}{S})} := \arg \max_{\theta} Q(\theta \mid \theta^{(t)})$$

subject to $g_s(\theta) = g_s(\theta^{(t+\frac{s-1}{S})})$

where $G = \{g_s(\theta), s = 1, ..., S - 1\}$ is a set of S - 1 preselected functions. Thus, $\theta^{(t+\frac{s}{S})}$ satisfies

$$Q(\boldsymbol{\theta}^{(t+\frac{s}{S})} \mid \boldsymbol{\theta}^{(t)}) \ge Q(\boldsymbol{\theta} \mid \boldsymbol{\theta}^{(t)}) \text{ for all } \boldsymbol{\theta} \in \{\boldsymbol{\theta} \in \boldsymbol{\Theta} : g_s(\boldsymbol{\theta}) = g_s(\boldsymbol{\theta}^{(t+\frac{s-1}{S})})\}.$$
(2.1.12)

where Θ is parameter space.

After (S-1) CM steps with conditional expectation have been performed, the actual log likelihood function $l(\boldsymbol{\theta} \mid \boldsymbol{x})$ from Equation (2.1.1) will be maximised in the Sth step.

CM-step with actual log likelihood:

$$\begin{split} \boldsymbol{\theta}^{(t+\frac{S}{S})} &:= \arg \max_{\boldsymbol{\theta}} l(\boldsymbol{\theta} \mid x) \\ &\text{subject to } g_{S}(\boldsymbol{\theta}) = g_{S}(\boldsymbol{\theta}^{(t+\frac{S-1}{S})}) \end{split}$$

where g_S is a preselected function. Thus, $\boldsymbol{\theta}^{(t+\frac{s}{S})}$ satisfies

$$l(\boldsymbol{\theta}^{(t+\frac{S}{S})} \mid x) \ge l(\boldsymbol{\theta} \mid x) \text{ for all } \boldsymbol{\theta} \in \{\boldsymbol{\theta} \in \boldsymbol{\Theta} : g_S(\boldsymbol{\theta}) = g_S(\boldsymbol{\theta}^{(t+\frac{S-1}{S})})\}.$$
(2.1.13)

In the final CM-step of the (t + 1)th iteration, the value $\theta^{(t+\frac{S}{S})} = \theta^{(t+1)}$, is taken to be the input on the (t+2)th iteration.

From Equation (2.1.13), we know that

$$l(\boldsymbol{\theta}^{(t+\frac{S}{S})} \mid x) \ge l(\boldsymbol{\theta}^{(t+\frac{S-1}{S})} \mid x)$$
$$\implies L(\boldsymbol{\theta}^{(t+\frac{S}{S})} \mid x) \ge L(\boldsymbol{\theta}^{(t+\frac{S-1}{S})} \mid x).$$
(2.1.14)

From Equation (2.1.12), we know that

$$\implies Q(\boldsymbol{\theta}^{(t+\frac{S-1}{S})} \mid \boldsymbol{\theta}^{(t)}) \ge Q(\boldsymbol{\theta}^{(t+\frac{S-2}{S})} \mid \boldsymbol{\theta}^{(t)})$$
$$\ge Q(\boldsymbol{\theta}^{(t+\frac{S-3}{S})} \mid \boldsymbol{\theta}^{(t)})$$
$$\vdots$$
$$\ge Q(\boldsymbol{\theta}^{(t)} \mid \boldsymbol{\theta}^{(t)})$$

From Equation (2.1.9) in the Section 2.1.2 that each EM step monotonically increases the likelihood function, we have

$$L(\boldsymbol{\theta}^{(t+\frac{S-1}{S})} \mid \boldsymbol{x}) \ge L(\boldsymbol{\theta}^{(t)} \mid \boldsymbol{x}).$$
(2.1.15)

Combining Equation (2.1.14) and (2.1.15), we have

$$L(\boldsymbol{\theta}^{(t+1)} \mid \boldsymbol{x}) \ge L(\boldsymbol{\theta}^{(t)} \mid \boldsymbol{x}).$$

We can conclude that the likelihood function monotonically increases in each iteration for the ECME algorithm.

2.2 Optimization

In this section, the method of Lagrange multipliers for solving optimization problem is introduced. This strategy is needed to solve the analytical solution for mixture model by using EM algorithm in the later section.

For the case of only one constraint and only two choice variables, consider the optimization problem

maximize over
$$(x,y) \in \mathbb{R}^2 : m(x,y)$$

subject to : n(x,y) = 0

The Lagrangian function is defined by

$$\mathcal{L}(x, y, \lambda) = m(x, y) - \lambda n(x, y)$$

where λ is called Lagrange multiplier. The optimization problem can be solved by getting the solution of the equation:

$$\nabla_{x,y,\lambda}\mathcal{L}(x,y,\lambda) = 0$$

Example 2.2.1 (Optimization problem with single constraint)

Suppose we want to maximize m(x, y) = x + y subject to the constraint $x^2 + y^2 = 1$. Formally, the optimization problem can be expressed as

maximize over
$$(x, y) \in \mathbb{R}^2 : x + y$$

subject to
$$: x^2 + y^2 = 1$$

For the method of Lagrange multipliers, the constraint n(x, y) is

$$n(x,y) = x^2 + y^2 - 1 = 0.$$

So, we have the Lagrangian function

$$\mathcal{L}(x, y, \lambda) = m(x, y) - \lambda n(x, y)$$
$$= x + y + \lambda (x^2 + y^2 - 1)$$

Now, we can calculate the gradient:

$$\nabla_{x,y,\lambda} \mathcal{L}(x,y,\lambda) = \left(\frac{\partial \mathcal{L}}{\partial x}, \frac{\partial \mathcal{L}}{\partial y}, \frac{\partial \mathcal{L}}{\partial \lambda}\right)$$
$$= (1 + 2\lambda x, 1 + 2\lambda y, x^2 + y^2 - 1)$$

We set the gradient to 0 and we have a system of equations:

$$\begin{cases} 1 + 2\lambda x = 0\\ 1 + 2\lambda y = 0\\ x^2 + y^2 - 1 = 0 \end{cases}$$

By the first two equations, we have

$$x = y = -\frac{1}{2\lambda}, \ \lambda \neq 0$$

Substituting into the last equation we have:

$$\frac{1}{4\lambda^2} + \frac{1}{4\lambda^2} - 1 = 0$$
$$\implies \lambda = \pm \frac{1}{\sqrt{2}}$$

So, the stationary points, such that function's derivative is zero, of \mathcal{L} are

$$\left(\frac{\sqrt{2}}{2}, \frac{\sqrt{2}}{2}, -\frac{1}{\sqrt{2}}\right)$$
 and $\left(-\frac{\sqrt{2}}{2}, -\frac{\sqrt{2}}{2}, \frac{1}{\sqrt{2}}\right)$.

Plugging in the stationary points to the objective function f(x, y) yields

$$f\left(\frac{\sqrt{2}}{2},\frac{\sqrt{2}}{2}\right) = \sqrt{2}$$
 and $f\left(\frac{-\sqrt{2}}{2},-\frac{\sqrt{2}}{2}\right) = -\sqrt{2}.$

Thus the constrained maximum is $\sqrt{2}$ and the constrained minimum is $-\sqrt{2}$.

In the Figure 2.2.1, this shows you the illustration of the constrained optimization problem. The colourful plane represents function m(x, y) = x + y. The colour near warm colour, like red, the value of m(x, y) is higher. The colour near cold colour, like blue, the value of m(x, y) is lower. Also, the black circle on the ground is the function $x^2 + y^2 = 1$. Therefore, the black circle inside the colour plane shows the value of m(x, y) subject to the constraint $x^2 + y^2 = 1$. When the coordinate is $\left(\frac{\sqrt{2}}{2}, \frac{\sqrt{2}}{2}, -\frac{1}{\sqrt{2}}\right)$, the value of m(x, y) reaches the highest value in the black circle and when the coordinate is $\left(-\frac{\sqrt{2}}{2}, -\frac{\sqrt{2}}{2}, \frac{1}{\sqrt{2}}\right)$, the value of m(x, y) reaches the lowest value in the black circle.

2.3 Copulas

We know that if (Z_1, Z_2) follows a bivariate Gaussian distribution, Z_1 and Z_2 also follow univariate Gaussian distributions. How about the only if situation ? If Z_1 and Z_2 both follow univariate Gaussian distributions, does (Z_1, Z_2) follows a bivariate Gaussian distribution ? The answer is no, because we don't know the dependency between Z_1 and Z_2 . Bivariate Gaussian distribution is a common and symmetric distribution which allows us to have a symmetric dependency, that the value of Z_1 is larger (smaller) and the value of Z_2 has higher chance to get larger (smaller) and vice versa. However, in the real world cases, it is not always reasonable to fit the data with a Gaussian distribution. What if the case of asymmetric dependency or the variables don't follow normal distribution ? Copulas are flexible distribution functions which are more suitable to model data with particular or special distribution, especially for those distributions with asymmetric dependency between variables. Therefore, in this section, we are going to introduce the concept of copulas and vine copulas and the definitions and theories presented will follow Czado [2019]. Also, the related proofs are mostly skipped.



Figure 2.2.1: Illustration of the constrained optimization problem in Example 2.2.1

2.3.1 Concept of Copulas

Copula is a special multivariate distribution function and the marginal distribution of each variable for a Copula is uniform on the interval [0, 1]. Here is the definition of the Copula and its density:

Definition 2.3.1 (Copula and copula density)

A d-dimensional copula C is a multivariate distribution function

$$C: [0,1]^d \to [0,1]$$

with uniformly distributed marginals and it is given by

$$C(u_1, ..., u_d) := P(U_1 \le u_1, ..., U_d \le u_d)$$

where $U_1, U_2, ..., U_d$ are the variables uniformly distributed. The corresponding copula density denote by c can be obtained by partial differentiation, i.e.

$$c(u_1, ..., u_d) := \frac{\partial^d}{\partial u_1 ... \partial u_d} C(u_1, ..., u_d) \text{ for all } \boldsymbol{u} \text{ in } [0, 1]^d$$

Example 2.3.1 (Independence Copula and Comonotonicity copula)

Independence Copula

Let $U_1, U_2, ..., U_d$ be independent and random variables follow uniform distribution on [0, 1]. The independence Copula is

$$C(u_1, u_2, ..., u_d) = P(U_1 \le u_1, ..., U_d \le u_d)$$

= $P(U_1 \le u_1) \cdots P(U_d \le u_d)$
= $u_1 u_2 \cdots u_d$.



Figure 2.3.1: The scatter plot of the 500 data generated from independence copula (left) and comonotonicity copula (right)

Comonotonicity Copula

Let $U_1, U_2, ..., U_d$ be completely dependent that $U_1 = U_2 = \cdots = U_d$. The comonotonicity copula is

$$C(u_1, u_2, ..., u_d) = P(U_1 \le u_1, ..., U_d \le u_d)$$

= $P(U_1 \le u_1, ..., U_1 \le u_d)$
= $P(U_1 \le \min\{u_1, u_2, ..., u_d\})$
= $\min\{u_1, u_2, ..., u_d\}.$

The illustations of the bivariate independence copula and bivariate comonotonicity copula are shown in the Figure 2.3.1.

If we want to model the dependence among random variables using Copula, we need to transform the random variables to uniformly distributed first. For this, we use the probability integral transform:

Definition 2.3.2 (Probability integral transform)

Let X be a continuous random variable following distribution function F and x is an observed value of X, then the transformation u := F(x) is called the Probability integral transform (PIT) at x. Furthermore, the random variable $X \sim F$ after probability integral transform follows uniform distribution, such that U := F(X), since

$$P(U \le u) = P(F(X) \le u) = P(X \le F^{-1}(u)) = F(F^{-1}(u)) = u$$

We suppose that $X = (X_1, ..., X_d)^T$ is a *d*-dimensional continuous random variable and the corresponding marginal distribution functions are F_j , such that $X_j \sim F_j$, j = 1, ..., d. We can construct so called copula data $(U_1, ..., U_d)$ which is uniformly distributed by using probability integral transform, such that

$$(U_1, ..., U_d) = (F_1(X_1), ..., F_d(X_d)).$$

The resulting data is called as in u-scale or copula-scale.

Normally, we don't know the distribution of the data we have. We use the empirical distribution function instead to construct the Copula data:

Definition 2.3.3 (Empirical distribution function)

Let $x_1, ..., x_n$ be an i.i.d. sample from a distribution function F, then the empirical distribution function is defined as

$$\hat{F}(x) := \frac{1}{n+1} \sum_{i=1}^{n} \mathbb{1}_{\{x_i \le x\}}$$
 for all x_i

where \hat{F} is also called estimated marginal distributions.

Assume that we have the data $\{(x_{i1}, x_{i2}, i = 1, ..., n)\}$. We can transform the data to the copula data by using the estimated marginal distributions $\hat{F}_j, j = 1, 2$:

$$(u_{i1}, u_{i2}) := (\hat{F}_1(x_{i1}), \hat{F}_2(x_{i2})) \text{ for } i = 1, \dots, n.$$

$$(2.3.1)$$

Now, we are going to show an important theory called Sklar's Theorem by Sklar [1959] that we can construct the multivariate distribution of $X_1, X_2, ..., X_d$ by the marginal distributions $F_i(x)$ and the copula C.

Definition 2.3.4 (Sklar's Theorem)

Let X be a d-dimensional random vector with joint distribution function F and marginal distribution function F_i , i = 1, ..., d, then the joint distribution function can be expressed as

$$F(x_1, ..., x_d) = C(F_1(x_1), ..., F_d(x_d))$$

with associated density or probability mass function

$$f(x_1, ..., x_d) = c(F_1(x_1), ..., F_d(x_d))f_1(x_1)...f_d(x_d)$$

for some d-dimensional copula C with copula density c. For absolutely continuous distributions, the copula C is unique. The inverse also holds: the copula corresponding to a multivariate distribution function F with marginal distribution functions F_i for i = 1, ..., d can be expressed as

$$C(u_1, ..., u_d) = F(F_1^{-1}(u_1), ..., F_d^{-1}(u_d))$$
(2.3.2)

and its copula density or probability mass function is determined by

$$c(u_1, ..., u_d) = \frac{f(F_1^{-1}(u_1), ..., F_d^{-1}(u_d))}{f_1(F_1^{-1}(u_1)...f_d(F_d^{-1}(u_d))}$$

Using the inverse of Sklar's Theorem (2.3.2), bivariate Gaussian copula and bivariate Student's t copula belonging to elliptical copulas can be constructed easily.

Example 2.3.2 (Bivariate Gaussian copula)

The bivariate Gaussian copula can be constructed by using the bivariate normal distribution with zero mean vector, unit variances, and correlation ρ , and the inverse of Sklar's Theorem (2.3.2). It is given by

$$C(u_1, u_2; \rho) := \Phi_2(\Phi_1^{-1}(u_1), \Phi_1^{-1}(u_2); \rho)$$

where $\Phi(\cdot)$ is the distribution function of a standard normal N(0,1) distribution and $\Phi_2(\cdot,\cdot;\rho)$ is the bivariate normal distribution function with zero means, unit variances, and correlation ρ .

Example 2.3.3 (Bivariate Student's t copula)

The bivariate Student's t copula can be constructed by using the bivariate Student's t distribution with v degrees of freedom, zero mean, and correlation ρ , and the inverse of Sklar's Theorem (2.3.2). It is given by

$$C(u_1, u_2; v, \rho) := T_{2,v}(T_v^{-1}(u_1), T_v^{-1}(u_2); v, \rho)$$

where $T_v(\cdot)$ is the distribution function of Student's t with degrees of freedom v and $T_{2,v}(\cdot, \cdot; v, \rho)$ is the the bivariate Student's t distribution function with degrees of freedom v and correlation ρ .

Now, we would like to introduce other well known copulas called bivariate Archimedean copulas not belonging to elliptical copulas. The copulas are not constructed by using the inverse of Sklar's Theorem (2.3.2) and the definition of bivariate Archimedean copulas is shown in the following:

Definition 2.3.5 (Bivariate Archimedean copulas)

Let Ω be the set of all continuous, strictly monotone decreasing, and convex functions $\varphi : I \to [0, \infty]$ with $\varphi(1) = 0$. Let $\varphi \in \Omega$, then

$$C(u_1, u_2) = \varphi^{[-1]}(\varphi(u_1) + \varphi(u_2))$$

is a copula. C is called a bivariate Archimedean copula with generator φ . Is $\varphi(0) = \infty$, the generator is called strict. Here $\varphi^{[-1]}$ is the pseudo-inverse of φ , which is defined as $\varphi^{[-1]} : [0, \infty] \to [0, 1]$ with

$$\varphi^{[-1]}(t) := \begin{cases} \varphi^{-1}(t) , 0 \le t \le \varphi(0) \\ 0 , \varphi(0) \le t \le \infty \end{cases}$$

By choosing the suitable generator φ , bivariate Archimedean copulas can be structured and some examples of parametric bivariate Archimedean copulas with a single parameter are shown in the Example 2.3.4 For the Archimedean copulas with two parameters, please refer to Czado [2019] for more details.

Example 2.3.4 (Parametric bivariate Archimedean copulas with a single parameter)

Bivariate Clayton copula

The bivariate Clayton copula is given by

$$C(u_1, u_2; \delta) := (u_1^{-\delta} + u_2^{-\delta} - 1)^{-\frac{1}{\delta}}$$

where the parameter of dependence $\delta \in (0, \infty)$ and the generator function φ is given by

$$\varphi(x) = \frac{1}{\delta}(x^{-\delta} - 1),$$

Bivariate Gumbel copula

The bivariate Gumbel copula is given by

$$C(u_1, u_2; \delta) := exp\left\{-\left[(-\log(u_1))^{\delta} + (-\log(u_2))^{\delta}\right]^{\frac{1}{\delta}}\right\}$$

where the parameter of dependence $\delta \in [1, \infty]$ and the generator function φ is given by

$$\varphi(x) = (-\log(x))^{\delta}$$

Bivariate Frank copula

The bivariate Frank copula is given by

$$C(u_1, u_2; \delta) := -\frac{1}{\delta} \log \left(1 + \frac{(exp(-\delta u_1) - 1)(exp(-\delta u_2) - 1)}{exp(-\delta) - 1} \right)$$

where the parameter of dependence $\delta \in \mathbb{R}/\{0\}$ and the generator function φ is given by

$$\varphi(x) = -\log(\frac{exp(-\delta x) - 1}{exp(-\delta) - 1}).$$

Bivariate Joe copula

The bivariate Joe copula is given by

$$C(u_1, u_2; \delta) := 1 - ((1 - u_1)^{\delta} + (1 - u_2)^{\delta} - (1 - u_1)^{\delta} (1 - u_2)^{\delta})^{\frac{1}{\delta}}$$

where the parameter of dependence $\delta \in [1, \infty]$ and the generator function φ is given by

$$\varphi(x) = -\log(1 - (1 - x)^{\delta})$$

You may find that the parameters of dependence in elliptical copulas and Archimedean copulas are different. In the Gaussian copula and Student's t copula, parameters of dependence is correlation $\rho \in [-1, 1]$. But, in the Archimedean copulas, the parameters of dependence are different and they have even different domains. In order to make the parameters of dependence consistent in different copulas for better comparison, we would like to introduce another measure for the dependency between two variables called Kendall rank correlation coefficient or Kendall's tau that we can use this measure only to represent the dependency for each of the copulas.

Definition 2.3.6 (Kendall's tau)

The Kendall's τ between the continuous random variables X_1 and X_2 is defined as

$$\tau(X_1, X_2) = P((X_{11} - X_{21})(X_{12} - X_{22}) > 0) - P((X_{11} - X_{21})(X_{12} - X_{22}) < 0),$$

where (X_{11}, X_{12}) and (X_{21}, X_{22}) are independent and identically distributed copies of (X_1, X_2) .

Definition 2.3.7 (Concordant, discordant and extra pairs)

Let (X_1, X_2) be two continuous random variables and $\{x_{i1}, x_{i2}, i = 1, ..., n\}$ are the corresponding samples. Then, (x_{j1}, x_{j2}) and (x_{k1}, x_{k2}) are called concordant pairs when

$$(x_{j1} - x_{k1})(x_{j2} - x_{k2}) > 0,$$

and called discordant pairs when

$$(x_{j1} - x_{k1})(x_{j2} - x_{k2}) < 0,$$

and called extra x_1 pair or extra x_2 pair when

$$x_{j1} = x_{k1}$$
 or $x_{j2} = x_{k2}$

Definition 2.3.8 (Estimate of Kendall's tau)

Let N_c be the number of concordant pairs, N_d be the number of discordant pairs, N_1 be the number of extra x_1 pairs, and N_2 be the number of extra x_2 pairs of random sample $x_{i1}, x_{i2}, i = 1, ..., n$ from the joint distribution of (X_1, X_2) . Then an estimate of Kendall's ρ is given by

$$\hat{\tau} = \frac{N_c - N_d}{\sqrt{N_c + N_d + N_1} \times \sqrt{N_c + N_d + N_2}}$$

The properties and interpretation way for Pearson correlation ρ and Kendall's τ are quite similar and they are shown in the following: (1) The value of Kendall's τ is between -1 and 1. (2) $\tau = 1$ indicates a perfect positive monotonous relation between two variables. (3) $\tau = 0$ indicates a perfect positive monotonous relation. (4) $\tau = -1$ indicates a perfect negative monotonous relation. (5) If two variables are independent, $\tau = 0$, but the reverse does not always hold.

We have mentioned that Kendall's tau can be used solely to represent the dependency for each of the copulas. Also, Table 2.3.1 shows the relationship between Kendall's τ and copula parameters for different bivariate copula.

Example 2.3.5 (Simulation of the copula data for single parameter copula families)

In the Example 2.3.5, the scatter plots of 500 data simulated from 6 introduced bivariate copulas with single parameter for different dependency levels are shown in Figure 2.3.2. Kendall's tau $\tau = 0.2, \tau = 0.5$ and $\tau = 0.8$ represent the weak, middle and strong dependency level respectively. For the weak dependency level ($\tau = 0.2$), the scatter plots look like independence copula, whose Kendall's tau is 0. As the dependency level increases,

Family	Kendall's $ au$	Range of τ
Gaussian	$ au = \frac{2}{\pi} \arcsin(ho)$	[-1,1]
Student's t	$ au = \frac{2}{\pi} \arcsin(ho)$	[-1,1]
Gumbel	$ au = 1 - rac{1}{\delta}$	[0,1]
Clayton	$ au = rac{\delta}{\delta+2}$	[0,1]
Frank	$\tau = 1 - \frac{4}{\delta} + 4 \frac{D_1(\delta)}{\delta}$ with $D_1(\delta) = \int_o^{\delta} \frac{x/\delta}{e^x - 1} dx$ (Debye function)	[-1,1]
Joe	$\tau = 1 + \left(\frac{-2+2\gamma+2\ln(2)+\psi(\frac{1}{\delta})+\psi(\frac{2+\delta}{2\delta})+\delta}{-2+\psi(\delta)}\right)$ with Euler constant $\gamma = 0.57721$ and digamma function $\psi(x)$	[0,1]

Table 2.3.1: Kendall's τ and copula parameters for different bivariate copula families with single parameter and Student's t copula

the tail dependencies for each copula can be observed easier. We can see that Gaussian, Student's t and Frank are copulas with symmetric tails and Clayton, Gumbel and Joe are copulas with asymmetric tails. For stronger dependency level ($\tau = 0.5$ or 0.8), we can see the tail dependence easier and we will talk about this later after the introduction of tail dependence coefficient.

Definition 2.3.9 (Upper and lower tail dependence coefficient)

Suppose we have continuous random X_1 and X_2 with marginal distributions F_1 and F_2 respectively. The upper tail dependence coefficient of a bivariate distribution with copula C is defined as

$$\lambda^{upper} = \lim_{t \to 1^{-}} P(X_2 > F_2^{-1} | X_1 > F_1^{-1}) = \lim_{t \to 1^{-}} \frac{1 - 2t + C(t, t)}{1 - t}$$

and the lower tail dependence coefficient is

$$\lambda^{lower} = \lim_{t \to 0^+} P(X_2 \le F_2^{-1} | X_1 \le F_1^{-1}) = \lim_{t \to 0^+} \frac{C(t,t)}{t}$$

Tail dependence coefficient is a measure for the dependency level between extreme values of two random variables. We say random variables (X_1, X_2) are upper tail dependent when $\lambda^{upper} > 0$ and upper tail independent when $\lambda^{upper} = 0$. Similarly, $\lambda^{lower} > 0$ means lower tail dependent and $\lambda^{lower} = 0$ means lower tail independent. For example, for strong upper tail dependence between random variables (X_1, X_2) , we can expect that when the value of X_1 is very large, very likely the value of X_2 is very large too.

The Table 2.3.2 shows you the tail dependence coefficients of different bivariate copula families in terms of their copula parameters. We can know that Student's t copula is tail dependent and Clayton copula is upper tail dependent and lower tail independent and so on. The result is quite consistent to the corresponding scatter plots in Figure 2.3.2. For example, for Clayton, there are quite a lot of points around the value 0 of u_1 and u_2 in the lower tail. Compared to the lower tail, there are just a few point around the value 1 in the upper tail. Therefore, Clayton copula is upper tail dependent and not lower tail dependent.

In the Table 2.3.1, you may notice that for some of the copula families, just the positive value of the Kendall's tau is well defined, for example, Gumbel, Clayton and Joe which are asymmetric copulas. Therefore, we would like to define rotated copulas which allow us to model negative correlation between two variables for those asymmetric copulas.

Definition 2.3.10 (Rotated Copulas)

The counterclockwise rotations of the copula density $c(u_1, u_2)$ are given by



(i) Clayton, $\tau{=}0.8$



Figure 2.3.2: Scatter plots of the 500 data simulated from different copulas with single parameter for weak $(\tau=0.2)$, middle $(\tau=0.5)$ and strong $(\tau=0.8)$ dependence.

Family	Upper tail dependence	Lower tail dependence
Gaussian	0	0
Student's t	$2 \operatorname{t}_{v+1} \left(-\sqrt{v+1} \sqrt{\frac{1-\rho}{1+\rho}} \right)$	$2 \operatorname{t}_{v+1} \left(-\sqrt{v+1} \sqrt{\frac{1-\rho}{1+\rho}} \right)$
Gumbel	$2-2^{1/\delta}$	0
Clayton	0	$2^{-1/\delta}$
Frank	0	0
Joe	$2-2^{1/\delta}$	0

Table 2.3.2: Tail dependence coefficients of different bivariate copula families with single parameter

- 90 degrees : $c_{90}(u_1, u_2) = c(1 u_2, u_1)$
- 180 degrees : $c_{180}(u_1, u_2) = c(1 u_1, 1 u_2)$
- 270 degrees : $c_{270}(u_1, u_2) = c(u_2, 1 u_1)$

After defining the rotated copulas, we can for example extend the Clayton copula to a copula with a full range of Kendall's tau values by defining

$$c_{clayton}^{extended}(u_1, u_2; \tau) := \begin{cases} c_{clayton}(u_1, u_2) , \text{ if } \tau > 0\\ c_{clayton,90}(u_1, u_2) \text{ or } c_{clayton,270}(u_1, u_2) , \text{ if } \tau \le 0 \end{cases}$$

Example 2.3.6 (Simulation of the rotated copula data for Clayton)

In the Example 2.3.6, the scatter plots of 500 data simulated from Clayton and its rotated copulas with Kendall's tau ($\tau = 0.8$) are shown in the Figure 2.3.3.

After the introduction of the characteristics of different bivariate copulas, now we would like to talk about the parameter estimation for bivariate copula models. Assume that we have the copula data $\boldsymbol{u} = \{(u_{i1}, u_{i2}), i = 1, ..., n\}$ following bivariate copula C with parameter θ . Then, we can use maximum likelihood estimation (MLE) introduced in Section 2.1.1 to estimate the parameter of the copula C. The likelihood function is given by

$$l(\theta; \boldsymbol{u}) = \prod_{i=1}^{n} c(u_{i1}, u_{i2}; \theta),$$

and the estimated parameter can be found by maximizing the likelihood. The maximum likelihood estimator $\hat{\theta}_{MLE}$ is given by

$$\hat{\theta}_{MLE} = \arg \max_{\theta \in \Theta} \prod_{i=1}^{n} c(u_{i1}, u_{i2}; \theta).$$

where c is the density function of the copula C and Θ denotes the parameter space.

From the Definition (2.3.4) (Sklar's Theorem), we know that we can construct the multivariate distribution by the marginal distributions and the copula and we know how to fit the bivariate copula data by a bivariate copula. Now, the question is whether we can fit the data with three dimension by bivariate copulas, or even higher dimension data. Luckily, a multivariate density function can be decomposed into different pair copulas and marginal distributions. Therefore, the answer is feasible. Now, we will start with some definitions used in the pair copula decomposition in three dimensions.

Definition 2.3.11 (Conditional densities and distribution functions of bivariate distribution)



Figure 2.3.3: Scatter plots of the 500 data simulated from Clayton copula and its rotated copulas for $\tau = 0.8$.

The conditional density and distribution function of X_1 given $X_2 = x_2$ can be rewritten as

$$\begin{split} f_{1|2}(x_1|x_2) &= c_{12}(F_1(x_1),F_2(x_2))f_1(x_1) \\ F_{1|2}(x_1|x_2) &= \frac{\partial}{\partial u_2}C_{12}(F_1(x_1),u_2)|_{u_2=F_2(x_2)} \\ &= \frac{\partial}{\partial F_2(x_2)}C_{12}(F_1(x_1),F_2(x_2)) \\ &= C_{1|2}(F_1(x_1),F_2(x_2)) = C_{1|2}(u_1|u_2) \end{split}$$

where $C_{1|2}(u_1|u_2) = \frac{\partial}{\partial u_2} C_{12}(u_1, u_2).$

Definition 2.3.12 (Simplifying assumption)

When the simplifying assumption is satisfied, the copula density associated with the conditional distribution of (X_1, X_3) given $X_2 = x_2$ is given by for any $x_2 \in \mathbb{R}$

$$c_{1,3;2}(u_1, u_3; x_2) = c_{1,3;2}(u_1, u_3)$$
 for $u_1 \in [0, 1], u_3 \in [0, 1].$

Theorem 2.3.13 (A pair copula construction in three dimensions)

Let X_1, X_2 and X_3 be random variables. The density function and marginal distribution are denoted as f_i and F_i respectively and the conditional distribution function of X_i given X_j is denoted by $F_{i|j}$. Their joint density is given by $f(x_1, x_2, x_3)$. Suppose that the simplifying assumption for copulas in the Definition (2.3.12) holds. A pair copula construction of an arbitrary three dimensional density is given as following recursions:

$$f(x_1, x_2, x_3) = c_{1,3;2}(F_{1|2}(x_1|x_2), F_{3|2}(x_3|x_2)) \times c_{2,3}(F_2(x_2), F_3(x_3)) \times c_{1,2}(F_1(x_1), F_2(x_2))f_3(x_3)f_2(x_2)f_1(x_1)$$

where $c_{1,3;2}$ denotes the copula density associated with the conditional distribution of (X_1, X_3) given $X_2 = x_2$. From this construction we see that the joint three dimensional density can be expressed in terms of bivariate copulas and conditional distribution functions. However this construction is not unique, since

$$f(x_1, x_2, x_3) = c_{1,2;3}(F_{1|3}(x_1|x_3), F_{2|1}(x_2|x_2)) \times c_{1,3}(F_1(x_1), F_3(x_3))$$
$$\times c_{2,3}(F_2(x_2), F_3(x_3))f_3(x_3)f_2(x_2)f_1(x_1)$$

and

$$\begin{aligned} f(x_1, x_2, x_3) &= c_{2,3;1}(F_{2|1}(x_2|x_1), F_{3|1}(x_3|x_1)) \times c_{1,3}(F_1(x_1), F_3(x_3)) \\ & \times c_{1,2}(F_1(x_1), F_2(x_2)) f_3(x_3) f_2(x_2) f_1(x_1) \end{aligned}$$

are two different construction.

Theorem 2.3.14 (Pair copula construction of a joint parametric density in three dimensions)

A parametric pair copula construction in three dimensions specifies a three dimensional density with copula parameter vector $\boldsymbol{\theta} = (\theta_{1,2}, \theta_{2,3}, \theta_{1,2;3})^T$ modelling the dependence and marginal densities and distributions are given by $f_j, F_j, j = 1, 2, 3$ respectively with marginal parameter vector $\boldsymbol{\gamma} = (\gamma_1, \gamma_2, \gamma_3)^T$.

$$f(x_1, x_2, x_3) = c_{1,3;2}(F_{1|2}(x_1|x_2; \gamma_1, \gamma_2, \theta_{1,2}), F_{3|2}(x_3|x_2; \gamma_3, \gamma_2, \theta_{2,3}); \theta_{1,3;2})$$

$$\times c_{2,3}(F_2(x_2; \gamma_2), F_3(x_3; \gamma_3); \theta_{2,3})$$

$$\times c_{1,2}(F_1(x_1; \gamma_1), F_2(x_2; \gamma_2),; \theta_{1,2}) f_3(x_3; \gamma_3) f_2(x_2; \gamma_2) f_1(x_1; \gamma_1)$$

where $c_{1,3;2}(\cdot,\cdot;\theta_{1,3;2}), c_{1,2}(\cdot,\cdot;\theta_{1,2}), c_{2,3}(\cdot,\cdot;\theta_{2,3})$ are arbitrary parametric bivariate copula densities.

2.3.2 Vine Copulas

In the last section, we have introduced some basic concepts of the copula and the characteristics of different bivariate copulas. Also, we have shown that a distribution with 3 variables can be constructed by using bivariate copulas only and importantly, the construction is not unique. Therefore, we would like to introduce vine copulas which use tree structure in graph theory to represent the particular pair copula construction in higher dimensions. [Czado, 2019]

Here shows you the related definitions of the vine copula and the definitions presented follow Czado [2019] :

Definition 2.3.15 (Regular (R-) vine tree sequence)

The set of trees $\mathcal{V} = (T_1, ..., T_{d-1})$ is a regular vine tree sequence on d elements if:

- (1) Each tree $T_j = (N_j, E_j)$ is connected, i.e. for all nodes $a, b \in T_j, j = 1, ..., d-1$, there exists a path $n_1, ..., n_k \subseteq N_j$ with $a = n_1, b = n_k$.
- (2) T_1 is a tree with node set $N_1 = \{1, ..., d\}$ and edge set E_1 .
- (3) For $j \leq 2, T_j$ is a tree with node set $N_j = E_{j-1}$ and edge set E_j .
- (4) For j = 2, ..., d 1 and $\{a, b\} \in E_j$ it must be hold that $|a \cap b| = 1$.

Remark 2.3.16 (Proximity condition)

The property (4) in the Definition (2.3.15) is called the proximity condition. It ensures that if there is an edge e connecting a and b in tree T_j , $j \ge 2$, then a and b (which are edges in T_{j-1}) must share a common node in T_{j-1} .

Definition 2.3.17 (Complete union and conditioned sets)

For any edge $e \in E_i$ define the set

$$A_e := \{ j \in N_1 \mid \exists e_1 \in E_1, ..., e_{i-1} \in E_{i-1} \text{ such that } j \in e_1 \in ... \in e_{i-1} \in e \}.$$

The set A_e is called the complete union of the edge e. The conditioning set D_e of an edge $e = \{a, b\}$ is defined by

$$D_e := A_a \cap A_b$$

and the conditioned sets $C_{e,a}$ and $C_{e,b}$ are given by

$$\mathcal{C}_{e,a} := A_a \setminus D_e, \ \mathcal{C}_{e,b} := A_b \setminus D_e \text{ and } \mathcal{C}_e := \mathcal{C}_{e,a} \cup \mathcal{C}_{e,b}$$

We often abbreviate each edge $e = (\mathcal{C}_{e,a}, \mathcal{C}_{e,b}; D_e)$ in the vine tree sequence by

$$e = (e_a, e_b; D_e).$$

Definition 2.3.18 (Regular vine distribution)

The joint distribution F for the d dimensional random vector $\mathbf{X} = (X_1, ..., X_d)$ has a regular vine distribution, if we can specify a triplet $(\mathcal{F}, \mathcal{V}, \mathcal{B})$ such that:

- (1) Marginal distributions: $\mathcal{F} = (F_1, ..., F_d)$ is a vector of continuous invertible marginal distribution functions, representing the marginal distribution functions of the random variable X_i , i = 1, ..., d.
- (2) Regular vine tree sequence: \mathcal{V} is an R-vine tree sequence on d elements.

- (3) Bivariate copulas: The set $\mathcal{B} = \{C_e | e \in E_i; i = 1, ..., d 1\}$, where is a symmetric bivariate copula with density. Here E_i is the edge set of tree T_i in the R-vine tree sequence \mathcal{V} .
- (4) Relationship between R-vine tree sequence \mathcal{V} and the set \mathcal{B} of bivariate copulas: For each $e \in E_i$, i = 1, ..., d 1, e = [a, b], C_e is the copula associated with the conditional distribution of $X_{\mathcal{C}_{e,a}}$ and $X_{\mathcal{C}_{e,b}}$ given $X_{D_e} = x_{D_e}$. Further $C_e(.,.)$ does not depend on the specific value of x_{D_e} .

Definition 2.3.19 (Pair copula and copula density associated with edge e)

We will denote the copula C_e corresponding to edge e by $C_{\mathcal{C}_{e,a}\mathcal{C}_{e,b};D_e}$ and the corresponding density by $c_{\mathcal{C}_{e,a}\mathcal{C}_{e,b};D_e}$, respectively. This copula is also called a pair copula.

Definition 2.3.20 (Existence of a regular vine distribution)

Assume that $(\mathcal{F}, \mathcal{V}, \mathcal{B})$ satisfy the properties (1)–(3) of the Definition (2.3.18), then there is a unique d dimensional distribution F with density

$$f_{1,..,d}(x_1,...,x_d) = f_1(x_1)\cdots f_d(x_d) \prod_{i=1}^{d-1} \prod_{e \in E_i} c_{\mathcal{C}_{e,a}\mathcal{C}_{e,b};D_e}(F_{\mathcal{C}_{e,a}|D_e}(x_{\mathcal{C}_{e,a}}|\boldsymbol{x}_{D_e}), F_{\mathcal{C}_{e,b}|D_e}(x_{\mathcal{C}_{e,b}}|\boldsymbol{x}_{D_e})),$$

such that for each $e \in E_i$, i = 1, ..., d - 1, with $e = \{a, b\}$ we have for the distribution function of $X_{\mathcal{C}_{e,a}}$ and $X_{\mathcal{C}_{e,b}}$ given $\mathbf{X}_{D_e} = \mathbf{x}_{D_e}$

$$F_{\mathcal{C}_{e,a}\mathcal{C}_{e,b}|D_e}(x_{\mathcal{C}_{e,a}}, x_{\mathcal{C}_{e,b}}|\boldsymbol{x}_{D_e}) = C_e(F_{\mathcal{C}_{e,a}|D_e}(x_{\mathcal{C}_{e,a}}|\boldsymbol{x}_{D_e}), F_{\mathcal{C}_{e,b}|D_e}(x_{\mathcal{C}_{e,b}}|\boldsymbol{x}_{D_e})).$$

Further the one dimensional marginal distributions of F are given by $F_i(x_i), i = 1, ..., d$. The proof is skipped here. For more details, please refer to Bedford and Cooke [2002] and Czado [2019].

Example 2.3.7 (3-dimensional vine copula)

In the Figure 2.3.4, the vine tree sequence \mathcal{V} has two tree level T_1 and T_2 . The node set $N_1 = \{1, 2, 3\}$ and the edge set $E_1 = \{\{2, 1\}, \{1, 3\}\}$ are contained in the first tree T_1 . In the Definition (2.3.15) (3), we know that for second tree T_2 , the node set $N_2 = E_1$. Therefore, in the second tree level, the node set N_2 is $\{\{2, 1\}, \{1, 3\}\}$. According to the Definition (2.3.17), we can know the edge set of the second tree $E_2 = \{\{2, 3; 1\}\}$. The reason is in the following: Since $A_a = \{1, 2\}$ and $A_b = \{1, 3\}$, the conditioning set $D_e = \{1\}$. The conditioned sets $\mathcal{C}_{e,a}$ and $\mathcal{C}_{e,b}$ are $\{2\}, \{3\}$ respectively. Therefore, the edge E_2 can be represent by $\{\{2, 3; 1\}\}$.

In the first tree T_1 , the number of nodes is the number of the variables of the density function. In our case, we have 3 nodes because we construct a 3 dimensional density. Also, the number inside the node represents the corresponding variable, for example, 2 means X_2 and 3 means X_3 . According to the Definition (2.3.18) (4), the edge between two nodes represents the pair copula associated with the corresponding nodes. For example, the edge 2,1 in the first tree represents the copula density $c_{1,2}$ (or $c_{2,1}$). You can observe that the edge set in the vine copula is exactly the same as the subscript of the copula density $c_{2,3;1}$, $c_{1,3}$ and $c_{1,2}$. So, each pair copula construction can be represented by a particular vine tree structure.

By the Definition (2.3.20), the following pair copula construction of a 3 dimensional density

$$\begin{split} f(x_1, x_2, x_3) &= c_{2,3;1}(F_{2|1}(x_2|x_1), F_{3|1}(x_3|x_1)) \times c_{1,3}(F_1(x_1), F_3(x_3)) \\ & \times c_{1,2}(F_1(x_1), F_2(x_2)) f_3(x_3) f_2(x_2) f_1(x_1) \end{split}$$

can be represented by the vine copula tree structure in the Figure 2.3.4.



Figure 2.3.4: An example of 3-dimensional vine copula

2.4 Mixture Models

To ease the explanation, we discuss the ECM algorithm and each CM steps just updates 1 parameter θ by maximization with respect to θ , with all other parameters held fixed. In total, we have S CM steps for each iteration.

In the previous Section 2.3, we have introduced copulas for data modelling with more flexibility. But, including simple Gaussian distribution, they are uni-modal distributions that there is only one peak (single highest value) in the probability density function. But in reality, more often the data that we are trying to model follows multi-modal distribution that there appears more than one peaks in the distribution.

Example 2.4.1 (Distribution of Exam Scores)

Suppose a teacher gives a test to his class. Some of the students studied very hard, while some of them didn't study at all. After marking the exam, the teacher creates a histogram of the test scores. It follows a multi-modal distribution with one peak around low scores for the students who studied a lot and another peak around high scores for the students who didn't study in the Figure 2.4.1. We can clearly see that there are two bell shapes in the histogram. The data can probably be modelled by two normal distributions with different means and variances. We refer to such a model as a **mixture of Gaussian**. In general, it is called **mixture model** if the data can be modelled in terms of a mixture of several components, where each component has a simple parametric form.

2.4.1 Mixture Model Formulation and Parameter Estimation

Formally, a general mixture model with K components for a random vector $\mathbf{X} = (X_1, ..., X_p)^T$ has a density given by

$$f(\boldsymbol{x} \mid \boldsymbol{\eta}) = \sum_{k=1}^{K} \pi_k f_k(\boldsymbol{x} \mid \boldsymbol{\psi}_k)$$
(2.4.1)

with the mixture weights π_k satisfying $\pi_k \geq 0$ for all k and $\sum_{k=1}^K \pi_k = 1$. The probability density function $f_k(\boldsymbol{x} \mid \boldsymbol{\psi}_k)$ with the parameter $\boldsymbol{\psi}_k$ is called the k^{th} mixture component of the mixture density $f(\boldsymbol{x} \mid \boldsymbol{\eta})$, and π_k is called the mixing proportion. The parameters in a general mixture model are denoted as $\boldsymbol{\eta} = \{\boldsymbol{\psi}_1, ..., \boldsymbol{\psi}_K, \pi_1, ..., \pi_K\} \in \boldsymbol{\Theta}$ where $\boldsymbol{\Theta}$ denotes the parameter space.

Example 2.4.2 (Mixture of univariate Gaussian distributions)

The Figure 2.4.2 shows an example of a mixture using univariate Gaussian distribution with 3 components and its probability density function is in the following:

$$f(x \mid \boldsymbol{\eta}) = 0.2 \ f_1(x \mid -2, 0.5) + 0.3 \ f_2(x \mid 1, 2) + 0.5 \ f_3(x \mid 4, 1)$$



Figure 2.4.1: A histogram of test scores. It is clearly a multimodal distribution with two peaks, also called bi-modal distribution.

where $f_k(x \mid \mu_k, \sigma_k^2)$ denotes as a normal density with mean μ_k and variance σ_k^2 and is given by

$$f_k(x \mid \mu_k, \sigma_k^2) = \frac{1}{\sigma_k \sqrt{2\pi}} e^{-(x-\mu_k)^2/2\sigma_k^2}$$
 for $k = 1, 2, 3$.

Intuitively, for each sample, there is a probability with 0.2 generated from a Gaussian with mean -2 and variance 0.5, a probability with 0.3 generated from a Gaussian with mean 1 and variance 2 and a probability with 0.5 generated from a Gaussian with mean 4 and variance 1.

Assuming that $\mathcal{X} = [x_1, x_2, ..., x_n] \in \mathbb{R}^{p \times n}$ is a data matrix with n columns i.i.d. observations generated from the mixture model given in Equation (2.4.1). We are going to estimate the parameters η by the observations x_i using maximum likelihood estimation. The likelihood function of the *n* observations is given by

$$L(\boldsymbol{\eta} \mid \boldsymbol{x}) := \prod_{i=1}^n f(\boldsymbol{x}_i \mid \boldsymbol{\eta})$$

Then, the log-likelihood function can be expressed as

$$l(\boldsymbol{\eta} \mid \boldsymbol{x}) := \ln \left(\prod_{i=1}^{n} f(\boldsymbol{x}_{i} \mid \boldsymbol{\eta}) \right)$$
$$= \sum_{i=1}^{n} \ln \left(\sum_{k=1}^{K} \pi_{k} f_{k}(\boldsymbol{x}_{i} \mid \boldsymbol{\psi}_{k}) \right).$$

The problem of parameter estimation for mixture model can be formulated as

$$\max_{\boldsymbol{\eta}\in\Theta} \quad l(\boldsymbol{\eta} \mid \boldsymbol{x}) = \sum_{i=1}^{n} \ln\left(\sum_{k=1}^{K} \pi_{k} f_{k}(\boldsymbol{x}_{i} \mid \boldsymbol{\psi}_{k})\right)$$

subject to $\quad \pi_{k} \geq 0 \; \forall k \text{ and } \sum_{k=1}^{K} \pi_{k} = 1$ (2.4.2)

The difficulty of solving this optimization problem is to maximize the summation inside the logarithm of objective function. Unfortunately, there is no analytical solution for this optimization problem (2.4.2) by simply taking the derivatives and setting to 0. However, if you don't want to solve it numerically, the good news is that we still may get a closed-form solution by using the EM algorithm.



Figure 2.4.2: An example of a univariate mixture of Gaussian model with 3 components. red dashed line: density of $\mathcal{N}(-2, 0.5)$. Green dashed line: density of $\mathcal{N}(1, 2)$. Blue dashed line: density of $\mathcal{N}(4, 1)$. Black line: density of the Gaussian mixture.

The EM algorithm cannot be applied to solve the mixture model directly. On the contrary, EM algorithm handles this issue by introducing the latent variables. In mixture model, each sample just can be generated by 1 component, but we don't know which component. So, we regard the mixture components as the latent variable z, which is not observable.

Example 2.4.3 (Distribution of exam scores (continued))

In the Example 2.4.1, we know the test scores of all students. However, we cannot randomly pick a score and know whether it is from a hard-working student. So, the students who studied so hard and the students who didn't study at all are latent variables.

Suppose that the number of component K in the mixture model is known and define a random variable $\mathbf{Z} = (Z_1, ..., Z_k)^T \in \mathbb{R}^K$, that each $Z_k \in \{0, 1\}$ and only one of the element is equal to one and the others are zero. Intuitively, \mathbf{Z} is somehow like a state variable to record which component is taken for that sample. For the only component we take, we set it to be 1, while others are set to be 0. The probability of choosing component k is the mixing proportion π_k . Therefore, Z follows a multinomial distribution over K category with the probability of the mixing proportion $\pi = (\pi_1, ..., \pi_K)$ and exactly 1 trail, such that $\mathbf{Z} \sim Multinomial(\pi_1, ..., \pi_K)$. Recall that the probability mass function for multinomial random variable is

$$p(\boldsymbol{z}) = \frac{n!}{z_1! \cdots z_k!} p_1^{z_1} \cdots p_k^{z_k},$$

supported on $\mathbf{z} = (z_1, z_2, ..., z_k)$ where each z_i is a non-negative integer and their sum is n. In our case, n is 1 and $z_k! = 1 \forall k$, where $z_k = \{0, 1\}$. Thus, we can write the probability $P(\mathbf{Z} = \mathbf{z}) = p(\mathbf{z})$ as

$$p(\mathbf{z}) = {\pi_1}^{z_1} \cdots {\pi_k}^{z_k} = \prod_{k=1}^K {\pi_k}^{z_k}$$

And suppose that the conditional density of X at x given the latent vector $Z = e_k \in \mathbb{R}^k$ is $f_k(x \mid \psi_k)$ in Equation (2.4.1), such that

$$p(\boldsymbol{x} \mid \boldsymbol{Z} = \boldsymbol{e}_k) = f_k(\boldsymbol{x} \mid \boldsymbol{\psi}_k)$$
(2.4.3)

where e_k denotes the k^{th} column $K \times K$ identity matrix and Equation (2.4.3) can be also written as

$$p(\boldsymbol{x} \mid \boldsymbol{z}) = \prod_{k=1}^{K} f(\boldsymbol{x} \mid \boldsymbol{\psi}_k)^{\boldsymbol{z}_k}.$$

Thus, the joint probability mass function of X and Z is

$$p(\boldsymbol{x}, \boldsymbol{z}) = p(\boldsymbol{z})p(\boldsymbol{x} \mid \boldsymbol{z}) = \prod_{k=1}^{K} \pi_{k}^{z_{k}} f(\boldsymbol{x} \mid \boldsymbol{\psi}_{k})^{z_{k}}$$

Example 2.4.4 (Mixture of univariate Gaussian distributions (continued))

In Example 2.4.2, the latent variables Z, the random samples X and their distributions can be expressed as in the following expression:

$$\begin{aligned} \mathbf{Z} &= (Z_1, Z_2, Z_3) \sim Multinomial(0.2, 0.3, 0.5) \implies p(\mathbf{z}) = (0.2)^{z_1} (0.3)^{z_2} (0.5)^{z_3} \\ X \mid \mathbf{z} = (1, 0, 0) \sim \mathcal{N}(-2, 0.5) \implies p(x \mid \mathbf{z} = (1, 0, 0)) = f_{\mathcal{N}}(x \mid -2, 0.5) \\ X \mid \mathbf{z} = (0, 1, 0) \sim \mathcal{N}(1, 2) \implies p(x \mid \mathbf{z} = (0, 1, 0)) = f_{\mathcal{N}}(x \mid 1, 2) \\ X \mid \mathbf{z} = (0, 0, 1) \sim \mathcal{N}(4, 1) \implies p(x \mid \mathbf{z} = (0, 0, 1)) = f_{\mathcal{N}}(x \mid 4, 1) \end{aligned}$$

$$p(\boldsymbol{x} \mid \boldsymbol{z}) = (f_{\mathcal{N}}(x \mid -2, 0.5))^{z_1} (f_{\mathcal{N}}(x \mid -2, 0.5))^{z_2} (f_{\mathcal{N}}(x \mid -2, 0.5))^{z_3}.$$

$$p(\boldsymbol{x}, \boldsymbol{z}) = p(\boldsymbol{z}) p(\boldsymbol{x} \mid \boldsymbol{z}) = (0.2 f_{\mathcal{N}}(x \mid -2, 0.5))^{z_1} (0.3 f_{\mathcal{N}}(x \mid -2, 0.5))^{z_2} (0.5 f_{\mathcal{N}}(x \mid -2, 0.5))^{z_3}.$$

where the red, green and blue dashed lines in the Figure 2.4.2 represent component 1, 2 and 3 respectively.

After introduction of the latent variable Z, we now have a new joint distribution for $X_1, ..., X_n$ and their latent variable vector $Z_1, ..., Z_n$. As the latent variable Z is unobservable, the complete data is (z, x) and the incomplete data is x. In the EM algorithm, we want to maximize the log-likelihood function of the complete data. The likelihood of the complete data is given by

$$L(\boldsymbol{\eta} \mid \boldsymbol{x}, \boldsymbol{z}) := \prod_{i=1}^{n} p(\boldsymbol{x}_i, \boldsymbol{z}_i) = \prod_{i=1}^{n} \prod_{k=1}^{K} \pi_k^{z_{ik}} f(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k)^{z_{ik}},$$

where z_{ik} represents the k^{th} element of $\boldsymbol{z}_i = (z_1^i, ..., z_K^i)$. Again the latent variables are collected in a vector $\boldsymbol{z} = (\boldsymbol{z}_1^t, ..., \boldsymbol{z}_n^t)^t$. So that, the log-likelihood can be written as

$$l(\boldsymbol{\eta} \mid \boldsymbol{x}, \boldsymbol{z}) := \sum_{i=1}^{n} \ln(p(\boldsymbol{x}_i, \boldsymbol{z}_i)) = \sum_{i=1}^{n} \sum_{k=1}^{K} z_{ik} \ln(\pi_k f(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k)).$$

After getting the log-likelihood function, the following EM steps can be implemented:

E-step: Computes the expected value of $l(\eta \mid x, Z)$ given the observed data x and the current parameter estimate η_{old} . So that,

$$Q(\boldsymbol{\eta} \mid \boldsymbol{\eta}_{old}) = E[l(\boldsymbol{\eta} \mid \boldsymbol{x}, \boldsymbol{Z}) \mid \boldsymbol{\eta}_{old}, \boldsymbol{x}]$$
$$= \sum_{i=1}^{n} \sum_{k=1}^{K} E[Z_{ik} \mid \boldsymbol{\eta}_{old}, \boldsymbol{x}] \ln(\pi_k f(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k)).$$

where $\eta_{old} = \{\psi_1, ..., \psi_K, \pi_1, ..., \pi_K\}$ is the parameter values at current iteration and $E[\ln(\pi_k f(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k)) \mid \boldsymbol{\eta}_{old}, \boldsymbol{x}] = \ln(\pi_k f(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k))$ since $E[f(X) \mid X] = f(X)$. We denote the expectation of the latent variable given
the samples as r_{ik} and we have

$$E[Z_{ik} \mid \boldsymbol{\eta}_{old}, \boldsymbol{x}] := r_{ik}^{old} = 1 \ p(Z_{ik} = 1 \mid \boldsymbol{x}_i) + 0 \ p(Z_{ik} = 0 \mid \boldsymbol{x}_i) = p(Z_{ik} = 1 \mid \boldsymbol{x}_i)$$

$$= p(Z_i = \boldsymbol{e}_k \mid \boldsymbol{x}_i)$$

$$= \frac{p(Z_i = \boldsymbol{e}_k, \boldsymbol{x}_i)}{p(\boldsymbol{x}_i)}$$

$$= \frac{p(Z_i = \boldsymbol{e}_k)p(\boldsymbol{x}_i \mid \boldsymbol{Z}_i = \boldsymbol{e}_k)}{\sum_{j=1}^{K} p(\boldsymbol{Z} = \boldsymbol{e}_j)p(\boldsymbol{x}_i \mid \boldsymbol{Z}_i = \boldsymbol{e}_j)}$$

$$= \frac{\pi_k^{old}f(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k^{old})}{\sum_{j=1}^{K} \pi_j^{old}f(\boldsymbol{x}_i \mid \boldsymbol{\psi}_j^{old})} \ \text{for } i = 1, ..., n \ \text{and } k = 1, ..., K$$
(2.4.4)

where π^{old}, ψ^{old} are the parameters values at current iteration. Thus,

$$Q(\boldsymbol{\eta} \mid \boldsymbol{\eta}_{old}) = \sum_{i=1}^{n} \sum_{k=1}^{K} r_{ik}^{old} \ln(\pi_k f(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k))$$
(2.4.5)

M-step: The updated parameters η_{new} can be obtained by maximizing $Q(\eta \mid \eta_{old})$ with respect to η :

$$\eta_{new} := \arg \max_{\eta} Q(\eta \mid \eta_{old}).$$
subject to $\pi_k \ge 0 \ \forall k \text{ and } \sum_{k=1}^K \pi_k = 1$

$$(2.4.6)$$

In mixture model, the optimization problem is subject to some constraints with π . To get π_k , the maximization cannot be achieved simply by taking the derivatives with respect to the parameters π_k and then set to zero. Here, we use the method of Lagrange multipliers introduced in the Section 2.2 to solve this optimization problem with constrains.

For the optimization problem (2.4.6), the Lagrangian function is given by

$$\mathcal{L} = \sum_{i=1}^{n} \sum_{k=1}^{K} r_{ik}^{old} \ln(\pi_k f(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k)) + \lambda(\sum_{k=1}^{K} \pi_k - 1)$$

Taking the derivatives with respect to π_k and then set to 0:

$$\frac{\partial \mathcal{L}}{\partial \pi_k} = \frac{\sum_{i=1}^n r_{ik}^{old}}{\pi_k} + \lambda = 0$$

 π_k can be solved as

$$\pi_k = -\frac{\sum_{i=1}^n r_{ik}^{old}}{\lambda}.$$

Then, taking the derivatives with respect to λ and then set to 0:

$$\frac{\partial \mathcal{L}}{\partial \lambda} = \sum_{k=1}^{K} \pi_k - 1 = 0$$
$$\sum_{k=1}^{K} \pi_k = 1$$
(2.4.7)

Plugging in $\pi_k = -\frac{\sum_{i=1}^{N} r_{ik}^{old}}{\lambda}$ in Equation (2.4.7):

$$\sum_{k=1}^{K} \left(-\frac{\sum_{i=1}^{N} r_{ik}^{old}}{\lambda} \right) = 1$$
$$-\sum_{k=1}^{K} \sum_{i=1}^{N} r_{ik}^{old} = \lambda$$
(2.4.8)

Plugging in the expression for r_{ik}^{old} in Equation (2.4.8):

$$\begin{split} &-\sum_{k=1}^{K}\sum_{i=1}^{n}\left(\frac{\pi_{k}^{old}f(\boldsymbol{x}\mid\boldsymbol{\theta}_{k}^{old})}{\sum_{j=1}^{K}\pi_{j}^{old}f(\boldsymbol{x}\mid\boldsymbol{\theta}_{j}^{old})}\right) = \lambda\\ \Longleftrightarrow &-\sum_{i=1}^{N}1 = \lambda\\ \Longleftrightarrow &\lambda = -n \end{split}$$

Thus, we showed that updated mixing proportions are given by

$$\pi_k = \frac{\sum_{i=1}^n r_{ik}^{old}}{n} \quad \text{for} \quad k = 1, ..., K$$
(2.4.9)

About updating the remaining parameters, ψ_t^{new} is the root of the following equation by setting the values of the partial derivatives to zero.

$$\sum_{i=1}^{n} \sum_{k=1}^{K} r_{ik}^{old} \frac{\partial \ln f(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k)}{\partial \boldsymbol{\psi}_t} = 0 \quad \text{for} \quad t = 1, ..., K$$
(2.4.10)

A wonderful characteristic of the EM algorithm is that the solution of Equation (2.4.10) often exists in analytic. Gaussian mixture model is one of the example and it will be demonstrated in detail in the next Section 2.4.2.

Other than the EM or ECM algorithm, we have introduced two more extensions called MCECM and ECME in the Section 2.1.3. Because the steps for the extension are quite similar to the EM or ECM algorithm, in order to avoid content repetition, we will introduce them with the example of GMM in the next Section 2.4.2.

2.4.2 Multivariate Gaussian Mixture Models (GMM)

In this section, we are going to show you the parameter estimation for GMM by the EM algorithm and its extensions. We follow the approaches introduced in Section 2.3.1, since the GMM is just a special case of mixture models and specifying the probability density function $f(\boldsymbol{x} \mid \boldsymbol{\psi}_k)$ in Equation (2.4.1) as multivariate normal density at $\boldsymbol{x} \in \mathbb{R}^p$. As mentioned before, the analytic of GMM exists in the M-step also.

Recalling that the probability density function of multivariate normal distribution with mean $\mu \in \mathbb{R}^p$ and covariance matrix $\Sigma \in \mathbb{R}^{p \times p}$ is:

$$f(\boldsymbol{x} \mid \boldsymbol{\psi}_k) = f_{\mathcal{N}}(\boldsymbol{x} \mid \boldsymbol{\mu}, \boldsymbol{\Sigma}) = \left(\frac{1}{(2\pi)^k |\boldsymbol{\Sigma}|}\right)^{\frac{1}{2}} \exp\left(-\frac{1}{2}(\boldsymbol{x} - \boldsymbol{\mu})^T \boldsymbol{\Sigma}^{-1}(\boldsymbol{x} - \boldsymbol{\mu})\right)$$
(2.4.11)

Based on Equation (2.4.1), a Gaussian mixture model with K components can be written as

$$p(\boldsymbol{x} \mid \boldsymbol{\eta}) = \sum_{k=1}^{K} \pi_k f_{\mathcal{N}}(\boldsymbol{x} \mid \boldsymbol{\mu}_k, \boldsymbol{\Sigma}_k)$$

with the $\pi_k \geq 0$, $\forall k$ and $\sum_{k=1}^{K} \pi_k = 1$. The parameters in the Gaussian mixture model is collected in $\boldsymbol{\eta} = \{\pi_1, ..., \pi_K, \boldsymbol{\mu}_1, ..., \boldsymbol{\mu}_K, \boldsymbol{\Sigma}_1, ..., \boldsymbol{\Sigma}_K\} \in \boldsymbol{\Theta}$ where $\boldsymbol{\Theta}$ denotes the parameter space. Now, we can make use of the derived expressions in E and M steps in the previous section.

E-step: From Equation (2.4.5), we can get the expected value of $l(\boldsymbol{\eta} \mid \boldsymbol{x}, \boldsymbol{Z})$ given the observed data \boldsymbol{x} and the current parameter estimate $\boldsymbol{\eta}_{old}$ denoted as $Q(\boldsymbol{\eta} \mid \boldsymbol{\eta}_{old})$.

$$Q(\boldsymbol{\eta} \mid \boldsymbol{\eta}_{old}) = \sum_{i=1}^{n} \sum_{k=1}^{K} r_{ik}^{old} \ln(\pi_k f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_k, \boldsymbol{\Sigma}_k))$$

where the expectation of the latent variable given the samples denoted as r_{ik}^{old} is given by Equation (2.4.4) and we have

$$r_{ik}^{old} = \frac{\pi_k^{old} f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_k^{old}, \boldsymbol{\Sigma}_k^{old})}{\sum_{j=1}^K \pi_j^{old} f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_j^{old}, \boldsymbol{\Sigma}_j^{old})} \quad \text{for } i = 1, ..., n \text{ and } k = 1, ..., K$$
(2.4.12)

M-step: The updated parameters η_{new} can be obtained by maximizing $Q(\eta \mid \eta_{old})$ with respect to η , where $\eta = {\pi_1, ..., \pi_K, \mu_1, ..., \mu_K, \Sigma_1, ..., \Sigma_K}.$

$$\boldsymbol{\eta}_{new} := \arg \max_{\boldsymbol{\eta}} \left(\sum_{i=1}^{n} \sum_{k=1}^{K} r_{ik}^{old} \ln(\pi_k f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_k, \boldsymbol{\Sigma}_k)) \right).$$
subject to $\pi_k \ge 0 \; \forall k \text{ and } \sum_{k=1}^{K} \pi_k = 1$

$$(2.4.13)$$

The mixing proportion π_k is found in Equation (2.4.9) and it is

$$\pi_k = \frac{\sum_{i=1}^n r_{ik}^{old}}{n} \quad \text{for} \quad k = 1, ..., K$$
(2.4.14)

We can now update the remaining parameters μ_j , j = 1, ..., K by taking the first derivatives of $Q(\eta \mid \eta_{old})$ with respect to μ_j and set to zero. Then, we have

$$\sum_{i=1}^{n} \sum_{k=1}^{K} r_{ik}^{old} \frac{\partial \ln f_{\mathcal{N}}(\boldsymbol{x}_{i} \mid \boldsymbol{\mu}_{k}, \boldsymbol{\Sigma}_{k})}{\partial \boldsymbol{\mu}_{j}} = 0 \iff \sum_{i=1}^{n} r_{ij}^{old} \frac{1}{f_{\mathcal{N}}(\boldsymbol{x}_{i} \mid \boldsymbol{\mu}_{j}, \boldsymbol{\Sigma}_{j})} \frac{\partial f_{\mathcal{N}}(\boldsymbol{x}_{i} \mid \boldsymbol{\mu}_{j}, \boldsymbol{\Sigma}_{j})}{\partial \boldsymbol{\mu}_{j}} = 0 \iff \sum_{i=1}^{n} r_{ij}^{old} \frac{f_{\mathcal{N}}(\boldsymbol{x}_{i} \mid \boldsymbol{\mu}_{j}, \boldsymbol{\Sigma}_{j})}{f_{\mathcal{N}}(\boldsymbol{x}_{i} \mid \boldsymbol{\mu}_{j}, \boldsymbol{\Sigma}_{j})} \frac{\partial (-\frac{1}{2}(\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j})^{T} \boldsymbol{\Sigma}_{j}^{-1}(\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j}))}{\partial \boldsymbol{\mu}_{j}} = 0 \iff \sum_{i=1}^{n} r_{ij}^{old} \frac{\partial ((\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j})^{T} \boldsymbol{\Sigma}_{j}^{-1}(\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j}))}{\partial (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j})} = 0$$

Since $\frac{\partial \boldsymbol{w}^T A \boldsymbol{w}}{\partial \boldsymbol{w}} = 2A \boldsymbol{w}$ if \boldsymbol{w} doesn't depend on A and A is symmetric and $\boldsymbol{\Sigma}_k^{-1}$ is symmetric, we have

$$\sum_{i=1}^{n} r_{ij}^{old} 2\Sigma_{j}^{-1} (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j}) = 0$$

$$\sum_{i=1}^{n} r_{ij}^{old} (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j}) = 0$$

$$\sum_{i=1}^{n} r_{ij}^{old} \boldsymbol{x}_{i} - \sum_{i=1}^{n} r_{ij}^{old} \boldsymbol{\mu}_{j} = 0$$

$$\boldsymbol{\mu}_{j} = \frac{\sum_{i=1}^{n} r_{ij}^{old} \boldsymbol{x}_{i}}{\sum_{i=1}^{n} r_{ij}^{old}} \quad \text{for} \quad j = 1, ..., K \quad (2.4.15)$$

Similarly, for Σ_j ,

$$\sum_{i=1}^{n} \sum_{k=1}^{K} r_{ik}^{old} \ \frac{\partial \ln f_{\mathcal{N}}(\boldsymbol{x}_{i} \mid \boldsymbol{\mu}_{k}, \boldsymbol{\Sigma}_{k})}{\partial \boldsymbol{\Sigma}_{j}} = 0$$
$$\sum_{i=1}^{n} \sum_{k=1}^{K} r_{ik}^{old} \ \frac{\partial}{\partial \boldsymbol{\Sigma}_{j}} \left(-\frac{1}{2} \ln |\boldsymbol{\Sigma}_{k}| - \frac{1}{2} (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{k})^{T} \boldsymbol{\Sigma}_{k}^{-1} (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{k}) \right) = 0$$

Since $\frac{\partial \ln |\mathbf{A}|}{\partial \mathbf{A}} = \mathbf{A}^{-1}$ if \mathbf{A} is a symmetric matrix, $\mathbf{ABC} = tr(\mathbf{BCA})$ if \mathbf{ABC} is a scalar and we know $(\mathbf{x}_i - \mathbf{\mu}_k)^T \mathbf{\Sigma}_k^{-1} (\mathbf{x}_i - \mathbf{\mu}_k)$ is a scalar, we have

$$\sum_{i=1}^{n} r_{ij}^{old} \boldsymbol{\Sigma}_{j}^{-1} + \sum_{i=1}^{n} \sum_{k=1}^{K} r_{ik}^{old} \frac{\partial}{\partial \boldsymbol{\Sigma}_{j}} \left(tr(\boldsymbol{\Sigma}_{k}^{-1} (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{k}) (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{k})^{T} \right) = 0$$

Since $\frac{\partial tr(\boldsymbol{A}^{-1}\boldsymbol{x}\boldsymbol{x}^{T})}{\partial \boldsymbol{A}} = -\boldsymbol{A}^{-1}\boldsymbol{x}\boldsymbol{x}^{T}\boldsymbol{A}^{-1}$ if \boldsymbol{A} is a symmetric matrix, we have

$$\sum_{i=1}^{n} r_{ij}^{old} \Sigma_{j}^{-1} - \sum_{i=1}^{n} r_{ij}^{old} \left(\Sigma_{j}^{-1} (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j}) (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j})^{T} \Sigma_{j}^{-1} \right) = 0$$
$$\sum_{i=1}^{n} r_{ij}^{old} - \sum_{i=1}^{n} r_{ij}^{old} \Sigma_{j}^{-1} (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j}) (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j})^{T} = 0$$
$$\Sigma_{j} = \frac{\sum_{i=1}^{n} r_{ij}^{old} (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j}) (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j})^{T}}{\sum_{i=1}^{n} r_{ij}^{old}} \quad \text{for} \quad j = 1, ..., K$$
(2.4.16)

For simplicity, the steps for the EM or ECM algorithm at a single iteration can be summarised as follows :

EM or ECM E step :

$$r_{ij} = \frac{\pi_j f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_j, \boldsymbol{\Sigma}_j)}{\sum_{J=1}^{K} \pi_J f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_J, \boldsymbol{\Sigma}_J)} \quad \text{for} \quad i = 1, ..., n \text{ and } j = 1, ..., K$$

EM or ECM CM step 1:

$$\pi_j = \frac{\sum_{i=1}^n r_{ij}}{n}$$
 for $j = 1, ..., K$

EM or ECM CM step 2:

$$\mu_j = \frac{\sum_{i=1}^n r_{ij} x_i}{\sum_{i=1}^n r_{ij}}$$
 for $j = 1, ..., K$

EM or ECM CM step 3:

$$\Sigma_{j} = \frac{\sum_{i=1}^{n} r_{ij} (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j}) (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j})^{T}}{\sum_{i=1}^{n} r_{ij}} \quad \text{for} \quad j = 1, ..., K$$

Now, we would like to show you the steps of the parameter estimation for GMM by the MCECM algorithm. Recalling that in MCECM, E step is performed before each CM step and a cycle is defined by one E step followed by one CM step, where each iteration involves S cycles. Because there are total 3 different parameters for each cluster in GMM, each iteration includes 3 cycles. Also, each cycle contains one E step and one M step. More precisely, the steps for the MCECM algorithm at a single iteration are defined as follows (Because of the high similarity, we can mostly make use of the expressions derived in the GMM algorithm for EM algorithm):

Cycle 1

MCECM E step 1: The E step of the MCECM algorithm is the same as the EM or ECM algorithm. From the Equation (2.4.12), the expectation of the latent variable r_{ij} is given by

$$r_{ij} = \frac{\pi_j f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_j, \boldsymbol{\Sigma}_j)}{\sum_{J=1}^{K} \pi_J f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_J, \boldsymbol{\Sigma}_J)} \text{ for } i = 1, ..., n \text{ and } j = 1, ..., K$$

MCECM CM step 1: The CM step 1 of the MCECM algorithm is the same as maximizing $Q(\eta \mid \eta_{old})$ with respect to π_k for the optimization problem (2.4.13). From the Equation (2.4.14), the mixing proportion π_j is given by

$$\pi_j = \frac{\sum_{i=1}^n r_{ij}}{n} \quad \text{for} \quad j = 1, \dots, K$$

Cycle 2

MCECM E step 2: This E step is exactly the same as MCECM E step 1. The only difference is that the mixing proportion π_j used in the calculation of the r_{ij} is updated in the CM step 1.

$$r_{ij} = \frac{\pi_j f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_j, \boldsymbol{\Sigma}_j)}{\sum_{J=1}^{K} \pi_J f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_J, \boldsymbol{\Sigma}_J)} \quad \text{for } i = 1, ..., n \text{ and } j = 1, ..., K$$

MCECM CM step 2: Similarly, the CM step 2 of the MCECM algorithm is the same as maximizing $Q(\boldsymbol{\eta} \mid \boldsymbol{\eta}_{old})$ with respect to μ_j for the optimization problem (2.4.13). From the Equation (2.4.15), the mean μ_j is given by

$$\mu_j = \frac{\sum_{i=1}^{n} r_{ij} x_i}{\sum_{i=1}^{n} r_{ij}}$$
 for $j = 1, ..., K$

Cycle 3

MCECM E step 3: This E step is exactly the same as MCECM E step 2. Similarly, the only difference is that the mean μ_j used in the calculation of the r_{ij} is updated in the CM step 2.

$$r_{ij} = \frac{\pi_j f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_j, \boldsymbol{\Sigma}_j)}{\sum_{J=1}^{K} \pi_J f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_J, \boldsymbol{\Sigma}_J)} \quad \text{for} \quad i = 1, ..., n \text{ and } j = 1, ..., K$$

MCECM CM step 3: Similarly, the CM step 3 of the MCECM algorithm is the same as maximizing $Q(\boldsymbol{\eta} \mid \boldsymbol{\eta}_{old})$ with respect to $\boldsymbol{\Sigma}_j$ for the optimization problem (2.4.13). From the Equation (2.4.16), the mixing proportion $\boldsymbol{\Sigma}_j$ is given by

$$\Sigma_{j} = \frac{\sum_{i=1}^{n} r_{ij} (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j}) (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{j})^{T}}{\sum_{i=1}^{n} r_{ij}} \quad \text{for} \quad j = 1, ..., K$$

Finally, the steps of the parameter estimation for GMM by the ECME algorithm will be introduced here. Recalling that the assumption for the ECME algorithm made in the Section 2.1.3, the difference between ECME and ECM algorithms is that in the last CM step in ECME, the actual log likelihood function $l(\eta \mid x)$ from Equation (2.1.1) is maximised. More precisely, the steps for the ECME algorithm at a single iteration are defined as follows (Because of the high similarity, we can mostly make use of the expressions derived in the GMM algorithm for EM algorithm):

ECME E step : The E step of the ECME algorithm is the same as the EM or ECM algorithm. From the Equation (2.4.12), the expectation of the latent variable r_{ij} is given by

$$r_{ij} = \frac{\pi_j f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_j, \boldsymbol{\Sigma}_j)}{\sum_{J=1}^{K} \pi_J f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_J, \boldsymbol{\Sigma}_J)} \quad \text{for } i = 1, ..., n \text{ and } j = 1, ..., K$$

ECME CM step 1: The CM step 1 of the ECME algorithm is the same as maximizing $Q(\eta \mid \eta_{old})$ with respect to π_j for the optimization problem (2.4.13). From the Equation (2.4.14), the mixing proportion π_j is given by

$$\pi_j = \frac{\sum_{i=1}^n r_{ij}}{n} \quad \text{for} \quad j = 1, \dots, K$$

ECME CM step 2: Similarly, the CM step 2 of the ECME algorithm is the same as maximizing $Q(\eta \mid \eta_{old})$ with respect to μ_j for the optimization problem (2.4.13). From the Equation (2.4.15), the mean μ_j is given by

$$\mu_j = \frac{\sum_{i=1}^n r_{ij} x_i}{\sum_{i=1}^n r_{ij}}$$
 for $j = 1, ..., K$

ECME CM step 3: This step is the only difference with EM or ECM algorithm, because the actual log likelihood function $l(\theta \mid x)$ from Equation (2.1.1) is maximised with respect to Σ_i here, such that

$$\max_{\boldsymbol{\Sigma}_{j}} \quad l(\boldsymbol{\eta} \mid \boldsymbol{x}) = \sum_{i=1}^{n} \ln \left(\sum_{k=1}^{K} \pi_{k} f_{\mathcal{N}}(\boldsymbol{x}_{i} \mid \boldsymbol{\mu}_{k}, \boldsymbol{\Sigma}_{k}) \right) \quad \text{for} \quad j = 1, ..., K$$

Because there is no analytic solution to maximise this actual log likelihood function, we have to get the solution numerically.

2.4.3 Vine Copula Mixture Models (VCMM)

In the Section 2.3, we have shown that copulas are more flexible distribution functions and therefore are more suitable to model complex data with clusters. In this section, we are going to extend the mixture model with the pair copula density and show you the parameter estimation for the vine copula mixture models (VCMM) by the EM algorithm and its extensions. For this section, we mainly follow Sahin and Czado [2021].

For simplicity, we would like to start with the vine copula mixture model formulation for 3 dimensions and 2 components as an example. From Equation (2.4.1), a mixture model with 2 components for a random vector $\boldsymbol{X} = (X_1, X_2, X_3)^T$ has a density given by

$$f(\boldsymbol{x} \mid \boldsymbol{\eta}) = \pi_1 f_1(\boldsymbol{x} \mid \boldsymbol{\psi}_1) + \pi_2 f_2(\boldsymbol{x} \mid \boldsymbol{\psi}_2)$$
(2.4.17)



Figure 2.4.3: Vine copula model of the density functions in the mixture model with two components

with the mixture weights π_1, π_2 satisfying $\pi_1, \pi_2 \ge 0$ and $\pi_1 + \pi_2 = 1$. The parameters in the mixture model are denoted as $\eta = \{\psi_1, \psi_2, \pi_1, \pi_2\}$.

An example taken from Sahin and Czado [2021], suppose that the 3 dimensional density function f_1, f_2 in the mixture model can be constructed by pair copulas and follow the vine copula model in the Figure 2.4.3.

According to the Definition 2.3.26, the pair copula construction of the density of component 1 f_1 with the corresponding copula parameter θ and the marginal parameter γ is given as:

$$f_{1}(\boldsymbol{x} \mid \boldsymbol{\psi}_{1}) = c_{(1)2,3;1}(F_{(1)2|1}(x_{2}|x_{1};\boldsymbol{\gamma}_{2(1)},\boldsymbol{\gamma}_{1(1)},\boldsymbol{\theta}_{(1)1,2}), F_{(1)3|1}(x_{3}|x_{1};\boldsymbol{\gamma}_{3(1)},\boldsymbol{\gamma}_{1(1)},\boldsymbol{\theta}_{(1)1,3}); \boldsymbol{\theta}_{(1)2,3;1}) \\ \times c_{(1)1,2}(F_{1(1)}(x_{1};\boldsymbol{\gamma}_{1(1)}), F_{2(1)}(x_{2};\boldsymbol{\gamma}_{2(1)}); \boldsymbol{\theta}_{(1)1,2}) \\ \times c_{(1)1,3}(F_{1(1)}(x_{1};\boldsymbol{\gamma}_{1(1)}), F_{2(1)}(x_{3};\boldsymbol{\gamma}_{3(1)}); \boldsymbol{\theta}_{(1)1,3}) \\ \times f_{3(1)}(x_{3};\boldsymbol{\gamma}_{3(1)}) f_{2(1)}(x_{2};\boldsymbol{\gamma}_{2(1)}) f_{1(1)}(x_{1};\boldsymbol{\gamma}_{1(1)})$$

$$(2.4.18)$$

where the subscript (1) refers to the second component marker, $\theta_{(1)1,2}, \theta_{(1)1,3}, \theta_{(1)2,3;1}$ are the parameters for the copula density $c_{(1)1,2}, c_{(1)1,3}, c_{(1)2,3;1}$ respectively, $\gamma_{1(1)}, \gamma_{2(1)}, \gamma_{3(1)}$ are the parameters for the marginal density $f_{1(1)}, f_{2(1)}, f_{3(1)}$ and marginal distribution $F_{1(1)}, F_{2(1)}, F_{3(1)}$ respectively. Also, $\psi_1 = (\gamma_1, \theta_1)$, where $\gamma_1 = (\gamma_{1(1)}, \gamma_{2(1)}, \gamma_{3(1)})$ and $\theta_1 = (\theta_{(1)1,2}, \theta_{(1)1,3}, \theta_{(1)2,3;1})$.

Similarly, the pair copula construction of the density of component 2 f_2 with the corresponding copula parameter θ and the marginal parameter γ is given as:

$$f_{2}(\boldsymbol{x} \mid \boldsymbol{\psi}_{2}) = c_{(2)1,2;3}(F_{(2)1|3}(x_{1} \mid x_{3}; \boldsymbol{\gamma}_{1(2)}, \boldsymbol{\gamma}_{3(2)}, \boldsymbol{\theta}_{(2)1,3}), F_{(2)2|3}(x_{2} \mid x_{3}; \boldsymbol{\gamma}_{2(2)}, \boldsymbol{\gamma}_{3(2)}, \boldsymbol{\theta}_{(2)2,3}); \boldsymbol{\theta}_{(2)1,2;3}) \\ \times c_{(2)2,3}(F_{2(2)}(x_{2}; \boldsymbol{\gamma}_{2(2)}), F_{3(2)}(x_{3}; \boldsymbol{\gamma}_{3(2)}); \boldsymbol{\theta}_{(2)2,3}) \\ \times c_{(2)1,3}(F_{1(2)}(x_{1}; \boldsymbol{\gamma}_{1(2)}), F_{3(2)}(x_{3}; \boldsymbol{\gamma}_{3(2)}); \boldsymbol{\theta}_{(2)1,3}) \\ \times f_{3(2)}(x_{3}; \boldsymbol{\gamma}_{3(2)}) f_{2(2)}(x_{2}; \boldsymbol{\gamma}_{2(2)}) f_{1(2)}(x_{1}; \boldsymbol{\gamma}_{1(2)})$$

$$(2.4.19)$$

where the subscript (2) refers to the first component marker, $\theta_{(2)1,3}$, $\theta_{(2)2,3}$, $\theta_{(2)1,2;3}$ are the parameters for the copula density $c_{(2)1,3}$, $c_{(2)2,3}$, $c_{(2)1,2;3}$ respectively, $\gamma_{1(2)}$, $\gamma_{2(2)}$, $\gamma_{3(2)}$ are the parameters for the marginal density $f_{1(2)}$, $f_{2(2)}$, $f_{3(2)}$ and marginal distribution $F_{1(2)}$, $F_{2(2)}$, $F_{3(2)}$ respectively. Also, $\psi_2 = (\gamma_2, \theta_2)$, where $\gamma_2 = (\gamma_{1(2)}, \gamma_{2(2)}, \gamma_{3(2)})$ and $\theta_2 = (\theta_{(2)1,3}, \theta_{(2)2,3}, \theta_{(2)1,2;3})$. Note that although we call θ as copula parameter, we need this to determine conditional distribution function $F_{a|b}$, so the copula parameter is one of the parameters in $F_{a|b}$.

For more details about the formulation for the general vine copula mixture model, please refer to Sahin and Czado [2021].

Now, we need to use EM algorithm and its extension to estimate the following parameters: mixture weights π , copula parameter θ and marginal parameter γ for two components. We start with ECM algorithm and make use of the derived expressions in E and M steps in the previous section.

EM or ECM E-step: From Equation (2.4.5), we can get the expected value of $l(\boldsymbol{\eta} \mid \boldsymbol{x}, \boldsymbol{Z})$, denoted as $Q(\boldsymbol{\eta} \mid \boldsymbol{\eta}_{old})$, given the observed data \boldsymbol{x} and the current parameter estimate $\boldsymbol{\eta}_{old}$.

$$Q(\boldsymbol{\eta} \mid \boldsymbol{\eta}_{old}) = \sum_{i=1}^{n} \sum_{k=1}^{2} r_{ik}^{old} \ln(\pi_k f_k(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k))$$

where the expectation of the latent variable given the samples denoted as r_{ik}^{old} is given by Equation (2.4.4) and we have

$$r_{ik}^{old} = \frac{\pi_k^{old} f_k(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k^{old})}{\sum_{j=1}^2 \pi_j^{old} f_j(\boldsymbol{x}_i \mid \boldsymbol{\psi}_j^{old})} \quad \text{for } i = 1, ..., n \text{ and } k = 1, 2$$
(2.4.20)

For the M step, the updated parameters η_{new} can be obtained by maximizing $Q(\eta \mid \eta_{old})$ with respect to η , where $\eta = \{\psi_1, \psi_2, \pi_1, \pi_2\}, \psi_1 = \{\gamma_1, \theta_1\}$ and $\psi_2 = \{\gamma_2, \theta_2\}$

$$\boldsymbol{\eta}_{new} := \arg \max_{\boldsymbol{\eta}} \left(\sum_{i=1}^{n} \sum_{k=1}^{2} r_{ik}^{old} \ln(\pi_k f_k(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k)) \right).$$
subject to $\pi_1, \pi_2 \ge 0$ and $\pi_1 + \pi_2 = 1$

$$(2.4.21)$$

EM or ECM CM-step 1 (Mixture weights):

The mixing proportion π_k is found in Equation (2.4.9) and it is

$$\pi_k = \frac{\sum_{i=1}^n r_{ik}^{old}}{n} \quad \text{for} \quad k = 1, 2 \tag{2.4.22}$$

EM or ECM CM-step 2 (Pair copula parameters): We now need to update the copula parameters θ_1, θ_2 of the 2 components by maximizing $Q(\eta | \eta_{old})$ with respect to θ_1, θ_2 respectively.

$$\boldsymbol{\theta}_{j} := \arg \max_{\boldsymbol{\theta}_{j}} \left(\sum_{i=1}^{n} \sum_{k=1}^{2} r_{ik}^{old} \ln(\pi_{k} f_{k}(\boldsymbol{x}_{i} \mid \boldsymbol{\gamma}_{k}, \boldsymbol{\theta}_{k}) \right) \text{ for } j = 1, 2$$
(2.4.23)

However, there is no analytic solution for the optimization problem (2.4.23). For this optimization, we can use the R function RVineSeqMLE in the R package VineCopula to estimate the pair-copula parameters θ_1, θ_2 by using maximum likelihood estimation (MLE) with weights r_{ik}^{old} (compare to Equation (2.4.23).

EM or ECM CM-step 3 (Marginal parameters): We now want to update the marginal parameters γ_1, γ_2 of the 2 components by maximizing $Q(\eta \mid \eta_{old})$ with respect to γ_1, γ_2 respectively.

$$\boldsymbol{\gamma}_{j} := \arg \max_{\boldsymbol{\gamma}_{j}} \left(\sum_{i=1}^{n} \sum_{k=1}^{2} r_{ik}^{old} \ln(\pi_{k} f_{k}(\boldsymbol{x}_{i} \mid \boldsymbol{\gamma}_{k}, \boldsymbol{\theta}_{k}) \right) \text{ for } j = 1, 2$$

$$(2.4.24)$$

Because there is no analytic solution for the optimization problem (2.4.24), we have to get the solution numerically.

Now, we would like to show you the steps of the parameter estimation for VCMM by the MCECM algorithm. Recalling that in MCECM, E step is performed before each CM step and a cycle is defined by one E step followed by one CM step.

Cycle 1

MCECM E step 1: The E step of the MCECM algorithm is the same as the EM or ECM algorithm. From the Equation (2.4.20), the expectation of the latent variable r_{ij} is given by

$$r_{ik} = \frac{\pi_k^{old} f_k(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k^{old})}{\sum_{j=1}^2 \pi_j^{old} f_j(\boldsymbol{x}_i \mid \boldsymbol{\psi}_j^{old})} \text{ for } i = 1, ..., n \text{ and } k = 1, 2$$

MCECM CM step 1 (Mixture weights): The CM step 1 of the MCECM algorithm is the same as maximizing $Q(\boldsymbol{\eta} | \boldsymbol{\eta}_{old})$ with respect to π_k for the optimization problem (2.4.21). From the Equation (2.4.22), the mixing proportion π_i is given by

$$\pi_j = \frac{\sum_{i=1}^n r_{ij}}{n} \quad \text{for} \quad j = 1, 2$$

Cycle 2

MCECM E step 2: This E step is exactly the same as MCECM E step 1. The only difference is that the mixing proportion π_i used in the calculation of the r_{ij} is updated in the CM step 1.

$$r_{ij} = \frac{\pi_j f_j(\boldsymbol{x}_i \mid \boldsymbol{\psi}_j)}{\sum_{J=1}^2 \pi_J f_J(\boldsymbol{x}_i \mid \boldsymbol{\psi}_J)} \text{ for } i = 1, ..., n \text{ and } j = 1, 2$$

MCECM CM step 2 (Pair copula parameters): Similarly, the CM step 2 of the MCECM algorithm is the same as maximizing $Q(\eta \mid \eta_{old})$ with respect to θ_1, θ_2 for the optimization problem (2.4.21). From the Equation (2.4.23), the copula parameters θ_1, θ_2 are given by

$$\boldsymbol{\theta}_{j} := \arg \max_{\boldsymbol{\theta}_{j}} \left(\sum_{i=1}^{n} \sum_{k=1}^{2} r_{ik} \ln(\pi_{k} f_{k}(\boldsymbol{x}_{i} \mid \boldsymbol{\gamma}_{k}, \boldsymbol{\theta}_{k}) \right) \text{ for } j = 1, 2$$

$$(2.4.25)$$

Again, there is no analytic solution for the optimization problem (2.4.25). We again use the R function RVineSeqMLE with specified weights in the R package VineCopula to estimate the pair-copula parameters θ_1, θ_2 .

Cycle 3

MCECM E step 3: Similarly, the E step is exactly the same as MCECM E step 2. The only difference is that the copula parameters θ_1, θ_2 used in the calculation of the r_{ij} is updated in the CM step 2.

$$r_{ij} = \frac{\pi_j f_j(\boldsymbol{x}_i \mid \boldsymbol{\psi}_j)}{\sum_{J=1}^2 \pi_J f_J(\boldsymbol{x}_i \mid \boldsymbol{\psi}_J)} \text{ for } i = 1, ..., n \text{ and } j = 1, 2$$

where $\boldsymbol{\psi}_1 = \{\boldsymbol{\gamma}_1, \boldsymbol{\theta}_1\}$ and $\boldsymbol{\psi}_2 = \{\boldsymbol{\gamma}_2, \boldsymbol{\theta}_2\}.$

MCECM CM step 3 (Marginal parameters): Similarly, the CM step 3 of the MCECM algorithm is the same as maximizing $Q(\eta \mid \eta_{old})$ with respect to γ_1, γ_2 for the optimization problem (2.4.21). From the Equation (2.4.24), the marginal parameters γ_1, γ_2 are given by

$$\boldsymbol{\gamma}_{j} := \arg \max_{\boldsymbol{\gamma}_{j}} \left(\sum_{i=1}^{n} \sum_{k=1}^{2} r_{ik} \ln(\pi_{k} f_{k}(\boldsymbol{x}_{i} \mid \boldsymbol{\gamma}_{k}, \boldsymbol{\theta}_{k}) \right) \text{ for } j = 1, 2$$

$$(2.4.26)$$

Because there is no analytic solution for the optimization problem (2.4.26), we have to get the solution numerically.

Finally, the steps of the parameter estimation for VCMM by the ECME algorithm will be introduced here. Recalling that the assumption for the ECME algorithm made in the Section 2.1.3, the difference between ECME and ECM algorithms is that in the last CM step in ECME, the actual log likelihood function $l(\boldsymbol{\theta} \mid \boldsymbol{x})$ from Equation (2.1.1) is maximised.

ECME E step : The E step of the ECME algorithm is the same as the EM or ECM algorithm. From the Equation (2.4.20), the expectation of the latent variable r_{ij} is given by

$$r_{ik} = \frac{\pi_k^{old} f_k(\boldsymbol{x}_i \mid \boldsymbol{\psi}_k^{old})}{\sum_{j=1}^2 \pi_j^{old} f_j(\boldsymbol{x}_i \mid \boldsymbol{\psi}_j^{old})} \text{ for } i = 1, ..., n \text{ and } k = 1, 2$$

ECME CM step 1 (Mixture weights): The CM step 1 of the ECME algorithm is the same as maximizing $Q(\boldsymbol{\eta} \mid \boldsymbol{\eta}_{old})$ with respect to π_j for the optimization problem (2.4.21). From the Equation (2.4.22), the mixing proportion π_j is given by

$$\pi_j = \frac{\sum_{i=1}^n r_{ij}}{n} \quad \text{for} \quad j = 1, 2$$

ECME CM step 2 (Pair copula parameters): Similarly, because there is no analytic solution for the optimization problem (2.4.25), we prefer using the R function RVineSeqMLE to estimate the pair-copula parameters. Theoretically, we want to update the copula parameters θ_1, θ_2 of the 2 components by maximizing $Q(\eta \mid \eta_{old})$ with respect to θ_1, θ_2 respectively.

$$\boldsymbol{\theta}_j := \arg \max_{\boldsymbol{\theta}_j} \left(\sum_{i=1}^n \sum_{k=1}^2 r_{ik} \ln(\pi_k f_k(\boldsymbol{x}_i \mid \boldsymbol{\gamma}_k, \boldsymbol{\theta}_k)) \right) \text{ for } j = 1, 2$$

ECME CM step 3 (Marginal parameters): In the last CM step of the ECME, instead of maximizing $Q(\boldsymbol{\eta} \mid \boldsymbol{\eta}_{old})$, we here maximise the actual log likelihood function $l(\boldsymbol{\eta} \mid \boldsymbol{x})$ from Equation (2.1.1) with respect to the marginal parameters $\boldsymbol{\gamma}_1, \boldsymbol{\gamma}_2$, such that

$$\boldsymbol{\gamma}_j := \max_{\boldsymbol{\gamma}_j} \quad l(\boldsymbol{\eta} \mid \boldsymbol{x}) = \sum_{i=1}^n \ln \left(\sum_{k=1}^2 \pi_k f_k(\boldsymbol{x}_i \mid \boldsymbol{\gamma}_k, \boldsymbol{\theta}_k) \right) \quad \text{for} \quad j = 1, 2$$

Because there is no analytic solution to maximise this actual log likelihood function $l(\eta \mid x)$, we have to get the solution numerically.

2.5 Clustering using Gaussian Mixture Models

In this section, we would like to investigate how the parameters of the Multivariate Gaussian Mixture Model (GMM) determine the shape of generated samples in each cluster in 3 dimensional space, especially the covariance matrix Σ . After that, the characteristics of Gaussian mixture models used in the R package mclust for model-based clustering will be introduced.

2.5.1 Characteristics of clusters in model based Gaussian mixture modelling

In the Section 2.4.2, we know that the parameters of the GMM contain the mixture weight π , mean μ and covariance matrix Σ . The mixture weight π decides the expected number of samples in each cluster, mean μ decides the location of centre of each cluster, where Gaussian distribution is symmetric about the mean, and covariance matrix Σ decides the level of dispersion or shape of the samples. Banfield and Raftery [1993] have considered a parametrization of the covariance matrix in terms of its eigenvalue decomposition based on the distribution, volume, shape and orientation of each component in the mixture model :

$$\boldsymbol{\Sigma}_{k} = \lambda_{k} \boldsymbol{D}_{k} \boldsymbol{A}_{k} \boldsymbol{D}_{k}^{T}, \qquad (2.5.1)$$

where scalar λ_k defines the volume, rotation matrix D_k is an orthogonal matrix which defines its orientation and shape matrix A_k is a diagonal matrix with determinant 1 which defines its shape. In the following, we will illustrate how the values of each parameter affect the shape of the generated samples.

Scalar λ_k

The value of λ_k decides the volume of each cluster. The larger the value is, the bigger the cluster is. The reason behind is quite simple. If we consider λ_1 is 1 and $D_k A_k D_k^T$ is an identity matrix I_3 , the covariance matrix Σ_1 results in an identity matrix I_3 . Therefore, the variance of the generated samples in each coordinate axis is the same with value 1. Because of the same dispersion level of the generated samples in each coordinate axis, this results in a ball shape in the Figure 2.5.1. If we choose the value of λ_2 as 3, the covariance matrix Σ_2 results in $3I_3$ and the variance of the generated samples in each coordinate axis is larger. The generated sample in cluster 2 has higher level of dispersion, so that the volume of cluster 2 is larger. The Figure 2.5.2 with a larger ball, compared to the Figure 2.5.1, proofs our statement.

Shape matrix A_k

As mentioned before, shape matrix A_k is a diagonal matrix with determinant 1 deciding its shape. In order to illustrate the shapes with different values of A_k , suppose that the scalar $\lambda_3 = 1$, rotation matrix is $D_3 = I_3$



Figure 2.5.1: The scatter plot and pairs plot of generated data from multivariate Gaussian distribution with mean $\mu_1 = [0, 0, 0]^T$, covariance matrix $\Sigma_1 = I_3$



Figure 2.5.2: The scatter plot and pairs plot of generated data from multivariate Gaussian distribution with mean $\mu_2 = [0, 0, 0]^T$, covariance matrix $\Sigma_2 = 3I_3$



Figure 2.5.3: The scatter plot and pairs plot of generated data from multivariate Gaussian distribution with mean $\boldsymbol{\mu}_3 = [0, 0, 0]^T$, covariance matrix $\boldsymbol{\Sigma}_3 = \begin{bmatrix} 1 & 0 & 0 \\ 0 & 3 & 0 \\ 0 & 0 & \frac{1}{3} \end{bmatrix}$

and shape matrices A_3 is $\begin{bmatrix} 1 & 0 & 0 \\ 0 & 3 & 0 \\ 0 & 0 & \frac{1}{3} \end{bmatrix}$ respectively. Therefore, Σ_3 is the same as A_3 . From Σ_3 , we can know

that the variance of the generated samples is different for different coordinate axes. The higher the variance, the more the spread of the data. Because the variance for x_1 , x_2 and x_3 axis are 1, 3 and $\frac{1}{3}$ respectively, we can expect that compared to the Figure 2.5.1, the generated samples spread the same level in x_1 axis, more in the x_2 axis and less in x_3 axis. In other words, the shape of the generated data along x_2 axis is longer, along x_3 axis is shorter, so the shape becomes more elliptical, compared to the ball shape in the Figure 2.5.1. The Figure 2.5.3 proofs our statement.

Rotation matrix D_k

The rotation matrix D_k determines the orientation of the corresponding shape. Different dimensional spaces have their own rotation matrix. Because this paper mainly focuses on 3 dimensional space, just the 3 dimensional rotation matrix will be introduced. Each 3D rotation matrix D_k can be decomposed further as

$$\begin{aligned} \boldsymbol{D}_{k} &= \boldsymbol{D}_{k}^{z}(\alpha)\boldsymbol{D}_{k}^{y}(\beta)\boldsymbol{D}_{k}^{x}(\gamma) \\ &= \begin{bmatrix} \cos\alpha & -\sin\alpha & 0\\ \sin\alpha & \cos\alpha & 0\\ 0 & 0 & 1 \end{bmatrix} \begin{bmatrix} \cos\beta & 0 & \sin\beta\\ 0 & 1 & 0\\ -\sin\beta & 0 & \cos\beta \end{bmatrix} \begin{bmatrix} 1 & 0 & 0\\ 0 & \cos\gamma & -\sin\gamma\\ 0 & \sin\gamma & \cos\gamma \end{bmatrix}. \end{aligned}$$

where α, β and γ represent yaw, pitch, and roll angles following the Figure 2.5.4 [Ellis et al., 2014]. Also, 0 degree means no orientation changes, because $D_k^z(0), D_k^y(0), D_k^x(0)$ are just identity matrices. Positive degrees mean the object rotates along the arrow and negative degrees mean the object rotates along the arrow.

Suppose that the scalar $\lambda_4 = 1$, shape matrix \boldsymbol{A}_4 is $\begin{bmatrix} 1 & 0 & 0 \\ 0 & 3 & 0 \\ 0 & 0 & \frac{1}{3} \end{bmatrix}$ and the rotation matrix \boldsymbol{D}_4 is $\boldsymbol{D}_4^{z}(45^\circ)\boldsymbol{D}_4^{y}(0)\boldsymbol{D}_4^{x}(0)$.

Compared to the Figure 2.5.3, we can expect that the volume and the shape are exactly the same, because the scalar λ_4 and the shape matrix A_4 are the same as λ_3 and A_3 respectively. However, the orientation is different because the rotation matrix D_4 is $D_4{}^z(45^\circ)$ and not an identity matrix. So, it is expected that the generated samples will rotate about the yaw axis by 45 degrees anticlockwise, which is shown in the Figure 2.5.5.



Figure 2.5.4: The position of all three axes for describing the angle of its rotations

Model	Expression	Distribution	Volume	Shape	Orientation
EII	λI	Spherical	Equal	Equal	-
VII	$\lambda_k oldsymbol{I}$	Spherical	Variable	Equal	-
EEI	λA	Diagonal	Equal	Equal	Coordinate axes
VEI	$\lambda_k oldsymbol{A}$	Diagonal	Variable	Equal	Coordinate axes
EVI	$\lambda oldsymbol{A}_k$	Diagonal	Equal	Variable	Coordinate axes
VVI	$\lambda_k oldsymbol{A}_k$	Diagonal	Variable	Variable	Coordinate axes
EEE	$\lambda \boldsymbol{D} \boldsymbol{A} \boldsymbol{D}^T$	Ellipsoidal	Equal	Equal	Equal
EVE	$\lambda oldsymbol{D}oldsymbol{A}_koldsymbol{D}^T$	Ellipsoidal	Equal	Variable	Equal
VEE	$\lambda_k \boldsymbol{D} \boldsymbol{A} \boldsymbol{D}^T$	Ellipsoidal	Variable	Equal	Equal
VVE	$\lambda_k \boldsymbol{D} \boldsymbol{A}_k \boldsymbol{D}^T$	Ellipsoidal	Variable	Variable	Equal
EEV	$\lambda oldsymbol{D}_k oldsymbol{A} oldsymbol{D}_k^T$	Ellipsoidal	Equal	Equal	Variable
VEV	$\lambda_k oldsymbol{D}_k oldsymbol{A} oldsymbol{D}_k^T$	Ellipsoidal	Variable	Equal	Variable
EVV	$\lambda oldsymbol{D}_koldsymbol{A}_koldsymbol{D}_k^T$	Ellipsoidal	Equal	Variable	Variable
VVV	$\lambda_k oldsymbol{D}_k oldsymbol{A}_k oldsymbol{D}_k^T$	Ellipsoidal	Variable	Variable	Variable

Table 2.5.1: Table : The characteristics of the clusters in different models defined in the mclust package

After introduction of the relationship between the shape of the generated samples and the parameters in the decomposition of the covariance matrix, we would like to introduce some models describing the difference of clusters in Gaussian mixture model. In the Table 2.5.1 [Scrucca et al., 2016], this shows you different characteristics of the clusters in different models, which is used in the R package mclust for model-based clustering. Because just models EEV, VEV, EVV and VVV will be used in the simulation, we just focus on these 4 models. About the name of the Model, E means "Equal" and U means "Unequal" or "Variable". Also, the first, second and the last alphabets represent Volume λ_k , Shape A_k and Orientation D_k respectively. Suppose that we have a Gaussian mixture model EEV with 2 clusters. Those 2 clusters have the same volume and shape, but different orientation. Therefore, the covariance matrix for cluster 1 and 2 can be represented by $\Sigma_1 = \lambda D_1 A D_1^T$ and $\Sigma_2 = \lambda D_2 A D_2^T$ respectively.

Example 2.5.1 (Visualization of simulated samples from the Gaussian Mixture Model EEV, VEV, EVV and VVV)

In the Figure 2.5.6, scatter plots of the 1000 generated data for EEV, VEV, EVV and VVV of Gaussian mixture

Table 2.5.2: The value of the parameters used in the visualization of simulated samples shown in the Figure 2.5.6.

			Gaussian mix	ture model type	
		(a) EEV	(b) VEV	(c) EVV	(d) VVV
$\mathbf{\Sigma}_1$	Covariance matrix of component 1	$\lambda oldsymbol{D}_1 oldsymbol{A} oldsymbol{D}_1^T$	$\lambda_1 \boldsymbol{D}_1 \boldsymbol{A} \boldsymbol{D}_1^T$	$\lambda oldsymbol{D}_1 oldsymbol{A}_1 oldsymbol{D}_1^T$	$\lambda_1 \boldsymbol{D}_1 \boldsymbol{A}_1 {\boldsymbol{D}_1}^T$
$\mathbf{\Sigma}_2$	Covariance matrix of component 2	$\lambda oldsymbol{D}_2 oldsymbol{A} oldsymbol{D}_2^T$	$\lambda_2 oldsymbol{D}_2 oldsymbol{A} oldsymbol{D}_2^T$	$\lambda oldsymbol{D}_2 oldsymbol{A}_2 oldsymbol{D}_2^T$	$\lambda_2 oldsymbol{D}_2 oldsymbol{A}_2 oldsymbol{D}_2^T$
$\mathbf{\Sigma}_2$	Covariance matrix of component 3	$\lambda oldsymbol{D}_3 oldsymbol{A} oldsymbol{D}_3^T$	$\lambda_3 \boldsymbol{D}_3 \boldsymbol{A} \boldsymbol{D}_3^T$	$\lambda \boldsymbol{D}_3 \boldsymbol{A}_3 {\boldsymbol{D}_3}^T$	$\lambda_3 oldsymbol{D}_3 oldsymbol{A}_3 oldsymbol{D}_3^T$
λ, λ_1	Scalar, Scalar of component 1	1	0.5	1	0.5
λ_2	Scalar of component 2	-	2	-	2
λ_3	Scalar of component 3	-	1	-	1
D_1	Rotation matrix of component 1		$\alpha = 0^{\circ}, \beta$	$=0^{\circ}, \gamma=0^{\circ}$	
$oldsymbol{D}_2$	Rotation matrix of component 2		$\alpha = 45^{\circ}, \beta$	$\beta = 0^{\circ}, \gamma = 0^{\circ}$	
D_3	Rotation matrix of component 3		$\alpha = -45^{\circ},$	$\beta = 0^{\circ}, \gamma = 0^{\circ}$	
$oldsymbol{A},oldsymbol{A}_1$	Shape matrix of component 1	$\left[\begin{array}{rrrr} 4 & 0 & 0 \\ 0 & \frac{1}{4} & 0 \\ 0 & 0 & 1 \end{array}\right]$	$\left[\begin{array}{rrrr} 4 & 0 & 0 \\ 0 & \frac{1}{4} & 0 \\ 0 & 0 & 1 \end{array}\right]$	$\left[\begin{array}{rrrr} 4 & 0 & 0 \\ 0 & \frac{1}{4} & 0 \\ 0 & 0 & 1 \end{array}\right]$	$\left[\begin{array}{rrrr} 4 & 0 & 0 \\ 0 & \frac{1}{4} & 0 \\ 0 & 0 & 1 \end{array}\right]$
$oldsymbol{A}_2$	Shape matrix of component 2	-	-	$\left[\begin{array}{rrrr} 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{array}\right]$	$\left[\begin{array}{rrrr}1 & 0 & 0\\0 & 1 & 0\\0 & 0 & 1\end{array}\right]$
A_3	Shape matrix of component 3	-	-	$\left[\begin{array}{rrrr} 10 & 0 & 0 \\ 0 & \frac{1}{10} & 0 \\ 0 & 0 & 1 \end{array}\right]$	$\left[\begin{array}{rrrr} 10 & 0 & 0 \\ 0 & \frac{1}{10} & 0 \\ 0 & 0 & 1 \end{array}\right]$



Figure 2.5.5: The scatter plot and pairs plot of generated data from multivariate Gaussian distribution with mean $\boldsymbol{\mu}_4 = [0, 0, 0]^T$, scalar $\lambda_4 = 1$, shape matrix $\boldsymbol{A}_4 = \begin{bmatrix} 1 & 0 & 0 \\ 0 & 3 & 0 \\ 0 & 0 & \frac{1}{3} \end{bmatrix}$ and rotation matrix $\boldsymbol{D}_4 = \boldsymbol{D}_4{}^z(45^\circ)$

models are shown and the corresponding parameters used are based on the Table 2.5.2. Also, orange, green and purple cluster represent the Custer 1,2 and 3 respectively. Figure 2.5.6 (a) EEV shows that the volume and shape of those 3 clusters are quite similar, just the orientations are different. In the Figure 2.5.6 (b) VEV, the shape of those clusters basically are the same as EEV, but the volumes are different. In the Figure 2.5.6 (c) EVV, the green cluster becomes a green ball and the purple cluster becomes more elliptical. Therefore, the shapes for those clusters are different. In the Figure 2.5.6 (d) VVV, the shapes are basically the same as EVV, but the volumes are different.



Figure 2.5.6: Scatter plot of the 1000 generated data from the Clustering Gaussian mixture models (EEV, EVV, VEV and VVV) in case of three groups in three dimensions assuming the specifications given in the Table 2.5.2

2.5.2 Issues of singularity and clusters collapse in Gaussian mixture modelling for ECME

After the introduction of how the covariance matrix in the Multivariate Gaussian Mixture Model (GMM) determine the shape of generated samples, in this section, we will try to investigate singularity and cluster collapse issues that leads to error or spurious solutions (non-interesting solutions) after implementation of the

ECME algorithm involving the maximization of the actual log-likelihood function. Firstly, we will introduce the geometric meaning of eigenvalues of covariance matrix. After that, singularity and cluster collapse issues will be introduced. Finally, a solution to avoid these issues will be given.

Geometric interpretation of the eigenvalues of covariance matrix

In the last Section 2.5.1, we have mentioned that the shape of the cluster depends fully on the shape matrix A which is a diagonal matrix with determinant 1. Also, the values on the diagonal determine the variances of the generated samples in different coordinate axes. If shape matrix A is an identity matrix I_3 , we can expect that the shape of the generated samples is a sphere. If the shape matrix A is something else, we can expect that it is like a ellipse or even a French loaf.

Other than the shape matrix A, there is another value showing the shape of the samples, which is eigenvalue. Here is the definition of the eigenvalue : Let B be a square matrix (or linear transformation). A number b is called an eigenvalue of B if there exists a non-zero vector u such that

$$\mathbf{B}\boldsymbol{u} = b\boldsymbol{u} \tag{2.5.2}$$

where the vector \boldsymbol{u} is called an eigenvector associated with this eigenvalue b. Now, we would like to find out the eigenvalue a of a shape matrix \boldsymbol{A} . Equation (2.5.2) can be written as

$$(\boldsymbol{A} - b\boldsymbol{I})\boldsymbol{u} = \boldsymbol{A}\boldsymbol{u} - b\boldsymbol{u} = \boldsymbol{0}, \qquad (2.5.3)$$

and we know that b is an eigenvalue b when Equation (2.5.3) has a non-trivial solution. Therefore, Equation (2.5.3) has a non-trivial solution when

$$\det(\boldsymbol{A} - b\boldsymbol{I}) = 0.$$

In our case, as shape matrix A is a three dimensional diagonal matrix, we have

$$(a_{11} - b)(a_{22} - b)(a_{33} - b) = 0.$$

where a_{11}, a_{22}, a_{33} are the entries of the diagonal of shape matrix A. Solving this equation, we have the eigenvalues b_1, b_2, b_3

$$b_1 = a_{11}$$
 or $b_2 = a_{22}$ or $b_3 = a_{33}$,

which means the eigenvalues of the shape matrix A are the diagonal entries of the shape matrix A. Therefore, the eigenvalues are exactly the variances of the generated samples in different coordinate axes. Furthermore, from Equation (2.5.1), with $\lambda = 1$, we have

$$\boldsymbol{\Sigma} = \boldsymbol{D} \boldsymbol{A} \boldsymbol{D}^T,$$

which is a shape matrix \boldsymbol{A} change after applying a rotation \boldsymbol{D} to it. Not surprisingly, we expect that the eigenvalues of $\boldsymbol{D}\boldsymbol{A}\boldsymbol{D}^T$ keep the same as \boldsymbol{A} because just the orientation changes and the shape remains the same. The proof is in the following: From Equation (2.5.2) and multiplying rotation matrix \boldsymbol{D} on the right, we have

$$DAu = bDu$$

Because $D^{-1}D = I$, we have

$$DAD^{-1}Du = bDu$$

 $DAD^{-1}(Du) = b(Du)$

where the vector Du is the eigenvector associated with the eigenvalue b of DAD^{-1} . Therefore, the eigenvalues remains the same after applying a rotation matrix. If the volume scalar λ is not 1, the new eigenvalue of $\Sigma = \lambda DAD^T$ is λb . Because the volume scalar is applied to all the eigenvalues, the volume becomes bigger if $\lambda > 1$ and smaller if $\lambda < 1$, but the shape remains the same.

Convergence at critical points for mixture modelling

In Section 2.1.3, we have introduced that ECME algorithm are iterative methods to estimate the covariance



Figure 2.5.7: Illustration The global and local maximum (Left) and the saddle point (Right).

by maximizing the actual log-likelihood until convergence. Because there is no analytical solution there, it is required to do optimization numerically. Normally, the optimization algorithm will coverage to the critical point of the log-likelihood function which the derivative of the function is equal to zero. Actually, critical points refer to the maximum and minimum and saddle point. Since the log-likelihood function in general can be non-convex, it is common to have more than one maximum points, often a lot. In the Figure 2.5.7 (Left) [Heal, 2020], it shows that a function can have more than one maximums.

However, some of the local maximums that the algorithm converges to may lead to the spurious solutions which we don't want. Also, the saddle point shown in the Figure 2.5.7 (Right) is a point which is not a local maximum or minimum, but the derivative is zero. Although you may think that saddle point is not a problem for the algorithm, because it seems like it is easy to find another point with higher value of log-likelihood, most of the time, the algorithm is stuck at a saddle point and the value of log-likelihood stays the same. Based on our stopping condition in Equation (2.1.4), the algorithm stops near the saddle point before they get to the right solution.

Singularity issues in mixture modelling

One significant problem associated with the maximum likelihood and leading to error is the singularity issue. In GMM, the singularity issue makes some eigenvalues of the covariance close to 0, such that the determinant of the covariance is also close to 0. Here is the reason why this issue leads to an error: In the ECME algorithm, our purpose is to maximize the actual likelihood or log-likelihood of the density. From Equation (2.4.11), we know that the probability density function of multivariate normal distribution with mean $\mu \in \mathbb{R}^p$ and covariance matrix $\Sigma \in \mathbb{R}^{p \times p}$ is

$$f_{\mathcal{N}}(\boldsymbol{x} \mid \boldsymbol{\mu}, \boldsymbol{\Sigma}) = \left(\frac{1}{(2\pi)^{k} |\boldsymbol{\Sigma}|}\right)^{\frac{1}{2}} \exp\left(-\frac{1}{2}(\boldsymbol{x} - \boldsymbol{\mu})^{T} \boldsymbol{\Sigma}^{-1}(\boldsymbol{x} - \boldsymbol{\mu}),\right)$$

and from the following theory: (Determinant is the product of eigenvalues) Let \boldsymbol{B} be a $n \times n$ matrix and $\lambda_1, ..., \lambda_n$ be the eigenvalues of \boldsymbol{B} , we have

$$\det(\boldsymbol{B}) = \lambda_1 \lambda_2 \cdots \lambda_n. \tag{2.5.4}$$

Obviously, if one of the eigenvalues $\lambda_1, ..., \lambda_n$ tends to 0 for one suitable sample point $x, |\Sigma| \to 0$ too. As we know that a singular square matrix whose determinant is 0 does not have a matrix inverse. Therefore, if $|\Sigma| \to 0$, Σ^{-1} may not exist and this will lead to an error. The Figure 2.5.8 [Bishop, 2006] shows you the illustration of singularity issue in one dimension. We can see that there is a single data point fitted to a



Figure 2.5.8: Illustration of singularity issue happens to Gaussian mixture model

Gaussian and the variance of the associated distribution Gaussian is very close to 0. When the variance is close to 0, the inverse of the variance may not exist and it will lead to an error.

Clusters collapse issue in mixture modelling

Another problem associated with the maximum likelihood and leading to spurious solutions is the cluster collapse issue. For two mixture components in a mixture model, clusters collapse issue is that the EM algorithm converges at a saddle point leading to a non-interesting solution : The volume of one of the cluster shrinks to almost 0, such that the determinant of covariance matrix is almost 0, because at least one of the eigenvalues of the covariance is close to 0.

The Figure 2.5.9 shows you the illustration of the estimated clusters of Gaussian mixture model with the cluster collapse issue. The dark red represents the contour plot of 95% confidence region and the black point represents the mean of the cluster. We can see that the volume of the cluster like a straight line shrinks to almost 0 with the eigenvalues of the covariance matrix 22.4, 1.43 and 8.29×10^{-16} and determinant of the covariance matrix 1.56×10^{-14} , such that no sample points are classified to belong to this cluster. The covariance matrix of the elliptical cluster is actually the covariance of all the sample points, because all the samples belong to this cluster.

Solution for avoiding mentioned issues in simulation

After a brief introduction of singularity and cluster collapse issues, now, we would like to suggest a solution to avoid these problems in simulation and explain the reason why it works.

In our simulation part, our purpose is to get the results from a certain amount of replications without singularity and cluster collapse issues. Therefore, one of the method to solve this problem is to remove replications which might have spurious solutions. [Dang et al., 2017]

The replication is removed when the following conditions are fulfilled:

$$\lambda_l(\Sigma_k^{(t)}) < 10^{-20} \text{ for } l = 1, ..., d, t = 1, 2, 3, ... \text{ and } k = 1, ..., K$$
 (2.5.5)

or

$$\frac{\min_{l=1,\dots,d} \lambda_l(\Sigma_k^{(t)})}{\max_{l=1,\dots,d} \lambda_l(\Sigma_k^{(t)})} < 10^{-10} \text{ for } t = 1, 2, 3, \dots \text{ and } k = 1, \dots, K$$
(2.5.6)

where $\lambda_l(\Sigma_k^{(t)})$ is the *l*th eigenvalues of the estimated covariance of cluster *k* at *t*th iteration of the considered replication. [García-Escudero et al., 2015] [Dang et al., 2017] Intuitively, once one of the eigenvalues of the



Figure 2.5.9: Illustration of cluster collapse issue happens to Gaussian mixture model.

estimated covariance is too small, we remove this replication and start a new one. Because the determinant is the product of the eigenvalues, removal of the replication result based on the above conditions can avoid the estimation of covariance with determinant close to 0. We know that the one of the obvious indicator of singularity and clusters collapse issue happening to GMM is that the determinant of estimated covariance is 0. This is reason why the proposed solution works.

2.6 Performance measures

In this section, we will introduce some measures that we will use in the simulation part for result visualization for mixture model parameter estimation. The replication R below refers to the number of replications after removing the replications with spurious solutions.

Mean number of iterations

It measures the average needed number of iterations for the replications of the experiment. The definition of the iteration for EM algorithm and its extension please refers to the Section 2.4.2. The mean number of iterations is defined as here :

$$\bar{I} := \frac{\sum_{i=1}^{R} I_i}{R}$$

where I_i is the number of iterations needed at replication *i*. In terms of the perspective of the mean iteration number, the lower the needed mean iteration number, the better the algorithm's performance.

However, the mean number of iterations doesn't reflect the actual computation time (seconds). We need to measure the total computation time (seconds) for time efficiency assessment.

Mean computation time (seconds)

It measures the average total needed computation time (seconds) for the replications of the experiment. The mean total computation time (seconds) is defined as here :

$$\bar{T} := \frac{\sum_{i=1}^{R} T_i}{R}$$

where T_i is the total computation time (seconds) in seconds for replication *i*. For time efficiency prospective,

we choose the algorithm with the shortest mean total computation time (seconds).

Other than the time efficiency, we use the following measures for performance assessment about goodness of fit and clustering:

Bayesian information criterion (BIC)

The Bayesian information criterion (BIC) is a criterion for model selection or algorithms selection in our case and the one with the lowest BIC is adopted. The BIC was developed by Schwarz (1978) and it is given by

$$BIC(\boldsymbol{\eta}) = k \ln(n) - 2l(\hat{\boldsymbol{\eta}}|x),$$

where k is the number of parameters estimated, n is the total number of observations and $l(\hat{\eta}|x)$ is the log-likelihood introduced in Equation (3.1.1) with estimated parameter $\hat{\eta}$ which is the output of the Algorithm 1. For GMM parameter estimation, k is determined by $K(d + \frac{(d+1)d}{2}) + (K-1)$, where K is the number of components in GMM and d is the dimension of each sample. In our experiment with 3 dimension and 2 clusters, the total number of parameters estimated is 19. The model with the lowest $BIC(\eta)$ is the preferred model.

Normalised Bayesian information criterion (BIC)

In our experiment setting, the number of samples n generated are 100 or 500. From Equation (3.1.1), we know that the log-likelihood is the summation of n terms. Therefore, it implies that the BIC for 500 samples must be higher than 100 samples. In order to have an appropriate comparison of the BIC for different number of samples n. We prefer to use the normalised BIC and it is defined as follows :

$$\frac{BIC(\boldsymbol{\eta})}{n} = \frac{k \ln(n) - 2l(\hat{\boldsymbol{\eta}}|x)}{n}$$

Classification rate

Firstly, we would like to introduce how generated samples are finally grouped into clusters after parameter estimation from EM and its extension. From Equation (2.4.4), we know that the E step for the algorithm is to compute the expectation of the latent variable given samples for GMM, such that

$$E[Z_{ik} \mid \boldsymbol{\eta}, \boldsymbol{x}] := r_{ik} = 1 \ p(Z_{ik} = 1 \mid \boldsymbol{x}_i) + 0 \ p(Z_{ik} = 0 \mid \boldsymbol{x}_i)$$
$$= p(Z_{ik} = 1 \mid \boldsymbol{x}_i) \text{ for } i = 1, ..., n \text{ and } k = 1, ..., K$$

where $p(Z_{ik} = 1 | \boldsymbol{x}_i)$ is the probability of sample \boldsymbol{x}_i to be in clusters k, k = 1, ..., K. Also, from Equation (2.4.12), the E step, such that

$$r_{ik} = \frac{\pi_k f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_k, \boldsymbol{\Sigma}_k)}{\sum_{j=1}^{K} \pi_j f_{\mathcal{N}}(\boldsymbol{x}_i \mid \boldsymbol{\mu}_j, \boldsymbol{\Sigma}_j)} \text{ for } i = 1, ..., n \text{ and } k = 1, ..., K$$

is actually computing the probability of being each cluster for sample x_i . Therefore, after the algorithm converges, the E step will be additionally implemented using the estimated parameters to do the clustering for each sample. All observations i which have $\max_{\substack{k'=1,...,k}} r_{ik'} = r_{ik}$ will be put in Cluster k.

The classification rate is also called the Rand index by Rand [1971] and it is defined as here : **Rand index :** Given a set of *n* element $S = \{o_1, ..., o_n\}$ and two partitions of *S* to compare, $X = \{X_1, ..., X_r\}$, a partition of *S* into *r* subsets, and $Y = \{Y_1, ..., Y_s\}$, a partition of *S* into *s* subsets, the Rand index is given by

Rand index =
$$\frac{a+b}{\text{Number of unordered pairs in }S} = \frac{a+b}{\binom{n}{2}}$$

where a is the number of pairs of elements in S that are in the same subset in X and in the same subset in Y, b is the number of pairs of elements in S that are in different subset in X and in different subset in Y.

Example 2.6.1 (Calculation of the Rand index)

Suppose we have a data set with 4 elements $S = \{A, B, C, D\}$. The correct cluster is $X = \{1, 1, 1, 2\}$, i.e. cluster 1 consists of $\{A, B, C\}$ and cluster 2 of $\{D\}$. Also, the clustering result from the EM algorithm is $Y = \{1, 1, 2, 2\}$, i.e. cluster $1 = \{A, B\}$ and cluster $2 = \{C, D\}$.

To calculate the Rand index, we need to list out every possible unordered pair in the dataset of four elements: $\{A, B\}, \{A, C\}, \{A, D\}, \{B, C\}, \{B, D\}, \{C, D\}$. There are 6 unordered pairs which is equal to $\binom{4}{2}$.

Now, we need to calculate a, which represents the number of unordered pairs in S that belong to the same cluster in X and in the same subset in Y: {A, B}. In this case, a is 1.

Then, we need to calculate b, which represents the number of unordered pairs in S that belong to different cluster in X and Y: {A,D}, {B,D}. In this case, b is 2.

Therefore, we can calculate the Rand index:

Rand index
$$=$$
 $\frac{1+2}{6} = 0.5$

Also, the classification rate is 0.5.

3 Simulation study

In this chapter, we will use different EM algorithms presented in the Section 2.1 to estimate the parameters for the simulated data from the Gaussian mixture model (**GMM**) and vine copula mixture models (**VCMM**) in order to assess the performance, strengths and restrictions of different EM algorithms.

3.1 EM Algorithm for GMM algorithm

In the Section 2.3, the formulation of Gaussian mixture model and its parameter estimation steps for different EM algorithms are introduced. Now, our main goal is to carry out simulation studies or experiments to compare the performance of different EM algorithms for the Gaussian mixture model. The brief procedure of the experiment is in the following:

Step 1: Generate data for simulation.

Step 2: Decide on the type of EM algorithm and the order of the CM steps for testing.

Step 3: Decide on the starting values and stopping condition for the EM algorithms.

Step 4: Perform the simulation in R.

Step 5: Remove the replication results from spurious solutions.

Step 6: Create visualizations of the performance measures.

Step 7: Analyse and compare the results.

The detail for each step is given in the subsection:

3.1.1 Data simulation and the experiment setup

Step 1: Generate data for simulation.

In this step, we would like to generate some data from the Gaussian mixture model to test different EM algorithms. Because it is possible that the performance of different EM algorithms varies for a Gaussian mixture model with different number of samples, mean and covariance matrix, we would like to set up different scenarios / experiments in order to find out the corresponding characteristics, strengths and restrictions of different EM algorithms.

Code of	Number	Mean	Covariance matrix
Experiment	of samples (n)	of 2 clusters	of 2 clusters
EEV.emu.100	100	Equal	EEV, $\boldsymbol{\Sigma}_k = \lambda \boldsymbol{D}_k \boldsymbol{A} \boldsymbol{D}_k^T$
VEV.emu.100	100	Equal	VEV, $\boldsymbol{\Sigma}_k = \lambda_k \boldsymbol{D}_k \boldsymbol{A} \boldsymbol{D}_k^T$
EVV.emu.100	100	Equal	EVV, $\boldsymbol{\Sigma}_k = \lambda \boldsymbol{D}_k \boldsymbol{A}_k \boldsymbol{D}_k^T$
VVV.emu.100	100	Equal	$VVV, \boldsymbol{\Sigma}_{k} = \lambda_{k} \boldsymbol{D}_{k} \boldsymbol{A}_{k} {\boldsymbol{D}_{k}}^{T}$
EEV.umu.100	100	Unequal	EEV, $\boldsymbol{\Sigma}_k = \lambda \boldsymbol{D}_k \boldsymbol{A} \boldsymbol{D}_k^T$
VEV.umu.100	100	Unequal	VEV, $\boldsymbol{\Sigma}_k = \lambda_k \boldsymbol{D}_k \boldsymbol{A} \boldsymbol{D}_k^T$
EVV.umu.100	100	Unequal	EVV, $\boldsymbol{\Sigma}_k = \lambda \boldsymbol{D}_k \boldsymbol{A}_k \boldsymbol{D}_k^T$
VVV.umu.100	100	Unequal	VVV, $\boldsymbol{\Sigma}_k = \lambda_k \boldsymbol{D}_k \boldsymbol{A}_k {\boldsymbol{D}_k}^T$
EEV.emu.500	500	Equal	EEV, $\boldsymbol{\Sigma}_k = \lambda \boldsymbol{D}_k \boldsymbol{A} \boldsymbol{D}_k^T$
VEV.emu.500	500	Equal	VEV, $\boldsymbol{\Sigma}_k = \lambda_k \boldsymbol{D}_k \boldsymbol{A} \boldsymbol{D}_k^T$
EVV.emu.500	500	Equal	EVV, $\boldsymbol{\Sigma}_k = \lambda \boldsymbol{D}_k \boldsymbol{A}_k {\boldsymbol{D}_k}^T$
VVV.emu.500	500	Equal	VVV, $\boldsymbol{\Sigma}_k = \lambda_k \boldsymbol{D}_k \boldsymbol{A}_k {\boldsymbol{D}_k}^T$
EEV.umu.500	500	Unequal	EEV, $\boldsymbol{\Sigma}_k = \lambda \boldsymbol{D}_k \boldsymbol{A} \boldsymbol{D}_k^T$
VEV.umu.500	500	Unequal	VEV, $\boldsymbol{\Sigma}_k = \lambda_k \boldsymbol{D}_k \boldsymbol{A} \boldsymbol{D}_k^T$
EVV.umu.500	500	Unequal	EVV, $\boldsymbol{\Sigma}_k = \lambda \boldsymbol{D}_k \boldsymbol{A}_k {\boldsymbol{D}_k}^T$
VVV.umu.500	500	Unequal	$VVV, \boldsymbol{\Sigma}_{k} = \lambda_{k} \boldsymbol{D}_{k} \boldsymbol{A}_{k} {\boldsymbol{D}_{k}}^{T}$

Table 3.1.1: The basic set up of Gaussian mixture model in each experiment

In our experiments, the dimension of each sample d is 3 and number of clusters K is 2. The Table 3.1.1 shows the basic set up for different experiments. For example, in experiment EEV.emu.100, 100 samples in total are generated from the Gaussian mixture model with the same mean in each cluster and different covariance matrix following the model type EEV in each cluster, which we introduced in the previous Section 2.5.1. For simplicity, just the models EEV, VEV, EVV and VVV are used in our simulation. Actually, the real data are mostly with different orientations. Therefore, we expect that the simulation result from these four models are more appropriate for the envisioned real data applications.

Other than the basic set up in the Table 3.1.1, we still need to decide the value of the mean vector $\boldsymbol{\mu}_k$, covariance matrix $\boldsymbol{\Sigma}_k$ and the rest of the needed parameters, for example, mixing proportion π_k and so on. Details are shown in Table 3.1.2 and (3.1.3. Also, the correlation matrices of the covariance matrices mentioned in Table 2.4.3) are shown in Table 3.1.4. You can see that different strength levels of correlation are included in each of the correlation matrix.

Example 3.1.1 (The process of data generation and visualization of simulated data from the Gaussian mixture model for different model types)

Let's take the experiment VVV.emu.100 as an example. Assume the number of samples is 100, the means of 2 clusters are equal and the covariance matrices of 2 clusters are following model type VVV. We generate 100 samples and the probability of generating each sample from $\mathcal{N}(\boldsymbol{\mu}_1, \boldsymbol{\Sigma}_1 = \lambda \boldsymbol{D}_1 \boldsymbol{A} \boldsymbol{D}_1^T)$ is π_1 and from $\mathcal{N}(\boldsymbol{\mu}_2, \boldsymbol{\Sigma}_2 = \lambda \boldsymbol{D}_2 \boldsymbol{A} \boldsymbol{D}_2^T)$ is π_2 . We use the values given in the Table 3.1.3 to show the calculation of the needed

Notation	Name	Value				
n	Sample size	1	00 or 5	00		
d	Dimension of each sample		3			
K	Total number of components	2				
π_1	Mixing proportion of component 1	0.4				
π_2	Mixing proportion of component 2	0.6				
	Mean of 2 clusters : Equal					
$oldsymbol{\mu}_1$	Mean vector of component 1	(1.12)	1.63	$(1.95)^T$		
$oldsymbol{\mu}_2$	Mean vector of component 2	(1.12)	1.63	$(1.95)^T$		
	Mean of 2 clusters : Unequ	al				
$oldsymbol{\mu}_1$	Mean vector of component 1	1.63	$(1.95)^T$			
$oldsymbol{\mu}_2$	Mean vector of component 2	(-1.80	$(-1.80 1.50 1.13)^T$			

Table 3.1.2: The parameter values used in the data simulation for Gaussian mixture model

parameters in the following:

$$\begin{split} \pi_1 &= 0.4, \quad \pi_2 = 0.6 \\ \mu_1 &= \mu_2 = \begin{bmatrix} 1.12 & 1.63 & 1.95 \end{bmatrix}^T \\ \mathbf{\hat{\Sigma}}_1 &= \lambda_1 \mathbf{D}_1 \mathbf{A}_1 \mathbf{D}_1^T \approx 1 \begin{bmatrix} 0 & -0.423 & 0.906 \\ -0.5 & 0.785 & 0.366 \\ -0.866 & -0.453 & -0.211 \end{bmatrix} \begin{bmatrix} 1 & 0 & 0 \\ 0 & \frac{1}{6} & 0 \\ 0 & 0 & 6 \end{bmatrix} \begin{bmatrix} 0 & -0.423 & 0.906 \\ -0.5 & 0.785 & 0.366 \\ -0.866 & -0.453 & -0.211 \end{bmatrix}^T \\ &\approx \begin{bmatrix} 4.958 & 1.935 & -1.117 \\ 1.935 & 1.156 & -0.090 \\ -1.117 & -0.090 & 1.052 \end{bmatrix} \\ \text{where } \mathbf{D}_1 &= \begin{bmatrix} \cos 90^\circ & -\sin 90^\circ & 0 \\ \sin 90^\circ & \cos 90^\circ & 0 \\ 0 & 0 & 1 \end{bmatrix} \begin{bmatrix} \cos 120^\circ & 0 & \sin 120^\circ \\ 0 & 1 & 0 \\ -\sin 120^\circ & 0 & \cos 120^\circ \end{bmatrix} \begin{bmatrix} 1 & 0 & 0 \\ 0 & \cos 65^\circ & -\sin 65^\circ \\ 0 & \sin 65^\circ & \cos 65^\circ \end{bmatrix} \\ &\approx \begin{bmatrix} 0 & -0.423 & 0.906 \\ -0.5 & 0.785 & 0.366 \\ -0.866 & -0.453 & -0.211 \end{bmatrix} \\ \mathbf{\hat{\Sigma}}_2 &= \lambda_2 \mathbf{D}_2 \mathbf{A}_2 \mathbf{D}_2^T \approx 2 \begin{bmatrix} 0.25 & -0.276 & 0.928 \\ -0.433 & 0.825 & 0.362 \\ -0.866 & -0.492 & 0.087 \end{bmatrix} \begin{bmatrix} 4 & 0 & 0 \\ 0 & \frac{1}{4} & 0 \\ 0 & 0 & 1 \end{bmatrix} \begin{bmatrix} 0.25 & -0.276 & 0.928 \\ -0.433 & 0.825 & 0.362 \\ -0.866 & -0.492 & 0.087 \end{bmatrix} \\ &\approx \begin{bmatrix} 2.261 & -0.308 & -1.503 \\ -0.308 & 2.103 & 2.860 \\ -1.503 & 2.860 & 6.136 \end{bmatrix} \\ \text{where } \mathbf{D}_2 &= \begin{bmatrix} \cos 120^\circ & -\sin 120^\circ & 0 \\ \sin 120^\circ & -\sin 120^\circ & 0 \\ 0 & 0 & 1 \end{bmatrix} \begin{bmatrix} \cos 120^\circ & 0 & \sin 120^\circ \\ 0 & 1 & 0 \\ -\sin 120^\circ & 0 & \cos 120^\circ \end{bmatrix} \begin{bmatrix} 1 & 0 & 0 \\ 0 & \cos 100^\circ & -\sin 100^\circ \\ 0 & \sin 100^\circ & \cos 100^\circ \end{bmatrix} \\ &\approx \begin{bmatrix} 0.25 & -0.276 & 0.928 \\ -0.433 & 0.825 & 0.362 \\ -0.433 & 0.825 & 0.362 \\ -0.433 & 0.825 & 0.362 \end{bmatrix} \end{bmatrix}$$

For the experiment VVV.emu.100, we generate 100 samples from Cluster 1

$$\mathcal{N}\left(\left[\begin{array}{ccc} 1.12\\ 1.63\\ 1.95\end{array}\right], \left[\begin{array}{ccc} 4.958 & 1.935 & -1.117\\ 1.935 & 1.156 & -0.090\\ -1.117 & -0.090 & 1.052\end{array}\right]\right) \text{ with probability 0.4}$$

Table 3.1.3: The value of the parameters used in the data simulation for Gaussian mixture model (Continued), where α, β and γ represent yaw, pitch, and roll angles following the Figure 2.5.4 (Katherine et al., 2014) in the Section 2.5.

			Gaussian mixt	ure model type	
		EEV	VEV	EVV	VVV
$\mathbf{\Sigma}_1$	Covariance matrix of component 1	$\lambda \boldsymbol{D}_1 \boldsymbol{A} \boldsymbol{D}_1^T$	$\lambda_1 \boldsymbol{D}_1 \boldsymbol{A} {\boldsymbol{D}_1}^T$	$\lambda \boldsymbol{D}_1 \boldsymbol{A}_1 {\boldsymbol{D}_1}^T$	$\lambda_1 oldsymbol{D}_1 oldsymbol{A}_1 oldsymbol{D}_1^T$
$\mathbf{\Sigma}_2$	Covariance matrix of component 2	$\lambda oldsymbol{D}_2 oldsymbol{A} oldsymbol{D}_2^T$	$\lambda_2 \boldsymbol{D}_2 \boldsymbol{A} {\boldsymbol{D}_2}^T$	$\lambda oldsymbol{D}_2 oldsymbol{A}_2 oldsymbol{D}_2^T$	$\lambda_2 oldsymbol{D}_2 oldsymbol{A}_2 oldsymbol{D}_2^T$
λ, λ_1	Scalar, Scalar of component 1	1	1	1	1
λ_2	Scalar of component 2	-	2	-	2
D_1	Rotation matrix of component 1		$\alpha = 90^{\circ}, \beta =$	$120^\circ, \gamma = 65^\circ$	
D_2	Rotation matrix of component 2		$\alpha = 120^{\circ}, \beta =$	$120^\circ, \gamma = 100^\circ$	
$oldsymbol{A},oldsymbol{A}_1$	Shape matrix of component 1	$\left[\begin{array}{rrrr} 1 & 0 & 0 \\ 0 & \frac{1}{6} & 0 \\ 0 & 0 & 6 \end{array}\right]$	$\left[\begin{array}{rrrr} 1 & 0 & 0 \\ 0 & \frac{1}{6} & 0 \\ 0 & 0 & 6 \end{array}\right]$	$\left[\begin{array}{rrrr} 1 & 0 & 0 \\ 0 & \frac{1}{6} & 0 \\ 0 & 0 & 6 \end{array}\right]$	$\left[\begin{array}{rrrr} 1 & 0 & 0 \\ 0 & \frac{1}{6} & 0 \\ 0 & 0 & 6 \end{array}\right]$
$oldsymbol{A}_2$	Shape matrix of component 2	-	$\left[\begin{array}{rrrr} 4 & 0 & 0 \\ 0 & \frac{1}{4} & 0 \\ 0 & 0 & 1 \end{array}\right]$	-	$\left[\begin{array}{rrrr} 4 & 0 & 0 \\ 0 & \frac{1}{4} & 0 \\ 0 & 0 & 1 \end{array}\right]$
Ga	aussian mixture model type	Covari	ance matrix popent 1 Σ_{-}	Covarian of compo	ce matrix nent 2 Σ_2
	EEV	$\approx \begin{bmatrix} 4.958\\ 1.935\\ -1.117 \end{bmatrix}$	$\begin{array}{c} 1.935 & -1.117 \\ 1.156 & -0.090 \\ -0.090 & 1.052 \end{array}$	$\approx \begin{bmatrix} 5.243 & 1 \\ 1.870 & 1 \\ 0.290 & 0 \end{bmatrix}$.870 0.290 .088 0.496 .496 0.836
VEV		$\approx \left[\begin{array}{c} 4.958 \\ 1.935 \\ -1.117 \end{array} \right]$	$\begin{bmatrix} 1.935 & -1.117 \\ 1.156 & -0.090 \\ -0.090 & 1.052 \end{bmatrix}$	$\approx \left[\begin{array}{cc} 10.486 & 3\\ 3.741 & 2\\ 0.579 & 0 \end{array} \right]$	$\begin{array}{ccc} 3.741 & 0.579 \\ 2.176 & 0.992 \\ 0.992 & 1.671 \end{array}$
EVV		$\approx \begin{bmatrix} 4.958 \\ 1.935 \\ -1.117 \end{bmatrix}$	$\begin{bmatrix} 1.935 & -1.117 \\ 1.156 & -0.090 \\ -0.090 & 1.052 \end{bmatrix}$	$\approx \begin{bmatrix} 1.130 & -6 \\ -0.154 & 1 \\ -0.751 & 1 \end{bmatrix}$	$\begin{bmatrix} 0.154 & -0.751 \\ .051 & 1.430 \\ .430 & 3.068 \end{bmatrix}$
VVV		$\approx \begin{bmatrix} 4.958\\ 1.935\\ -1.117 \end{bmatrix}$	$\begin{bmatrix} 1.935 & -1.117 \\ 1.156 & -0.090 \\ -0.090 & 1.052 \end{bmatrix}$	$\approx \begin{bmatrix} 2.261 & -4 \\ -0.308 & 2 \\ -1.503 & 2 \end{bmatrix}$	$\begin{bmatrix} 0.308 & -1.503 \\ .103 & 2.860 \\ .860 & 6.136 \end{bmatrix}$

Table 3.1.4: The corresponding correlation matrices of the covariance matrices mentioned in Table 2.4.3), where the background colour of the text represents different strengths of the correlation, -0.082 is low, -0.489 is medium and 0.808 is high.

Gaussian mixture		Correlation matrix				Correlation matrix			
model type		of com	ponent 1	, \boldsymbol{R}_1		of component 2, \boldsymbol{R}_2			
		1	0.808	-0.489		1	0.783	0.138	
EEV	≈	0.808	1	-0.082	\approx	0.783	1	0.520	
		-0.489	-0.082	1		0.138	0.520	1	
	[1	0.808	-0.489]	1	0.783	0.138	
VEV	\approx	0.808	1	-0.082	\approx	0.783	1	0.520	
		-0.489	-0.082	1		0.138	0.520	1	
		1	0.808	-0.489] [1	-0.141	-0.404	
EVV	\approx	0.808	1	-0.082	\approx	-0.141	1 0.796	0.796	
		-0.489	-0.082	1		-0.404	0.796	1	
		1	0.808	-0.489] [1	-0.141	-0.404	
VVV	\approx	0.808	1	-0.082	\approx	-0.141	1	0.796	
		-0.489	-0.082	1		-0.404	0.796	1	

and Cluster 2

	(1.12		2.261	-0.308	-1.503	$ \rangle$	
\mathcal{N}		1.63	,	-0.308	2.103	2.860		with probability 0.6.
	$\left \right $	1.95		-1.503	2.860	6.136)	

The generated samples are plotted as a 3D scatter plot and pair plot and the orange dots are from Cluster 1 and green dots are from Cluster 2. The plots of all the data simulations are shown in the Figure 3.1.1 and Figure 3.1.2. By comparing the scatter plots, we can see that the overlapping level between 2 clusters is higher for each model type with equal mean. Also, if 2 scatters are not coloured, it is quite hard to observe there are 2 clusters in EEV and VEV. Oppositely, it is clearer to see there are 2 clusters in EVV and VVV. Therefore, we can expect that the result of the performance on classification rate is higher on 1) the unequal mean cases and data settings 2) EVV and VVV.

Step 2: Decide on the EM algorithms and the CM steps order for testing.

In Section 2, we have introduced different EM algorithms for parameter estimation, for example, classical EM, ECM, ECME and MCECM. As we mentioned, it is impossible to find a analytic solution to maximize a function for 2 parameters or more at the same time. Strictly speaking, we don't really use the classical EM algorithm for mixture model parameters estimation. Therefore, we prefer to use the ECM, ECME or MCECM algorithms to update one parameter in each CM step. Because we decide to update each parameter once in each iteration, there are different orders to update the parameters. Although there is no rule which orders reach better performance, for simplicity, we just consider the typical order to update the parameters, such that π , followed by μ and then Σ will be updated accordingly in each iteration.

Apart from the order, the optimization method is also needed to be chosen in the ECME algorithm because different methods affect the performance of parameter estimation when maximizing the actual log-likelihood. In the experiment, two optimization methods, Nelder-Mead and BFGS are used to perform the optimization.

In the Table 3.1.5, the selection of the EM algorithms and CM steps order for testing is shown. Order $= \{\theta_1, \theta_2, \theta_3\}$ means that the order of updating in CM steps is θ_1 , then θ_2 and followed by θ_3 . For the



(m) 3D scatter plot : VVV.emu.100

(n) Pairs plot : VVV.emu.100

(o) 3D scatter plot : VVV.umu.100

(p) Pairs plot : VVV.umu.100

Figure 3.1.1: 3D Scatter plots and the corresponding pairs plots of 100 simulated data generated from a Gaussian Mixture Model with different model types: EEV, VEV, EVV, VVV. The orange and green colour refer to the points in Cluster 1 and 2 respectively.

55



ő

ŋ

(c) 3D scatter plot : EEV.umu.500

x_1

(g) 3D scatter plot : VEV.umu.500



×1×

9

°,







•

•





(e) 3D scatter plot : VEV.emu.500



(i) 3D scatter plot : EVV.emu.500



(f) Pairs plot : VEV.emu.500

(j) Pairs plot : EVV.emu.500

(k) 3D scatter plot : EVV.umu.500

x_1



(d) Pairs plot : EEV.umu.500

x_3

(h) Pairs plot : VEV.umu.500







Figure 3.1.2: 3D Scatter plots and the corresponding pairs plots of 500 simulated data generated from a Gaussian Mixture Model with different model types: EEV, VEV, EVV, VVV. The orange and green colour refer to the points in Cluster 1 and 2 respectively.

Order	Code	ECM	ECME	ECME	MCECM
$= \{\theta_1, \theta_2, \theta_3\}$	of order		(Nelder-Mead)	(BFGS)	
$\{\pi, \boldsymbol{\mu}, \boldsymbol{\Sigma}\}$	123	\checkmark	\checkmark	\checkmark	\checkmark

Table 3.1.5: Selection of the EM algorithms and CM steps order for testing

Table 3.1.6: The selected starting values of the parameters for the EM algorithm used in the data simulation for Gaussian mixture model

Notation	Name	Value
π_1	Mixing proportion of component 1	0.55
π_2	Mixing proportion of component 2	0.45
$oldsymbol{\mu}_1$	Mean vector of component 1	$(2 \ 0 \ 1)^T$
$oldsymbol{\mu}_2$	Mean vector of component 2	$(0 \ 2 \ 1)^T$
$\mathbf{\Sigma}_1$	Covariance matrix of component 1	$\begin{bmatrix} 5.431 & -1.487 & -0.651 \\ -1.488 & 7.870 & 0.524 \\ -0.651 & 0.524 & 6.151 \end{bmatrix}$
$\mathbf{\Sigma}_2$	Covariance matrix of component 2	$\begin{bmatrix} 4.015 & 1.431 & 1.325 \\ 1.431 & 1.476 & 1.593 \\ 1.325 & 1.593 & 2.719 \end{bmatrix}$

convenience of showing the updating order in the graph, the code 1, 2 and 3 represents π , μ and Σ respectively. Therefore, the code of order 123 represents that the updating order in CM step is $\{\pi, \mu, \Sigma\}$.

Step 3: Decide on the starting values and stopping condition for the EM algorithms.

Since the EM algorithm and its extensions are all iterative algorithms, we need to decide the starting values of the parameters that we are going to estimate. Their initial values are denoted as $\boldsymbol{\eta}^{(0)} = \{\pi_1^{(0)}, \pi_2^{(0)}, \boldsymbol{\mu}_1^{(0)}, \boldsymbol{\mu}_2^{(0)}, \boldsymbol{\Sigma}_1^{(0)}, \boldsymbol{\Sigma}_2^{(0)}\}$. In each experiment and simulation, we use the same set of the parameters $\boldsymbol{\pi}$ and $\boldsymbol{\Sigma}$ of the Gaussian mixture model as the starting values and the chosen values should not be too close or too far from the true values. The selected starting values are shown in the Table 3.1.6.

Other than the starting values, the stopping condition is also needed in order to terminate the algorithm when it converges. Theoretically, the log-likelihood increases at each iteration of the EM algorithms. Therefore, the stopping condition 2 from Equation (2.1.4) is used and we select the prespecified ϵ as 0.001 here. Formally, the condition can be expressed as follows:

$$\frac{l(\pmb{\eta}^{(t)}) - l(\pmb{\eta}^{(t-1})}{l(\pmb{\eta}^{(t-1)})} < 0.001, \quad \text{for} \quad t = 1, 2, \dots$$

where the the actual incomplete data log-likelihood for $\boldsymbol{\eta} = (\pi_1, \pi_2, \boldsymbol{\mu}_1, \boldsymbol{\mu}_2, \boldsymbol{\Sigma}_1, \boldsymbol{\Sigma}_2)$ is given by

$$l(\boldsymbol{\eta}) := l(\boldsymbol{\eta} \mid \boldsymbol{x}) = \sum_{i=1}^{n} \ln \left(\sum_{k=1}^{2} \pi_{k} f_{\mathcal{N}}(\boldsymbol{x}_{i} \mid \boldsymbol{\mu}_{k}, \boldsymbol{\Sigma}_{k}) \right)$$
(3.1.1)

and $l(\eta^{(0)})$ represents the value of the actual log-likelihood with the starting values of the parameters and t is number of iterations ranging from 1 to the iteration when it terminates.

Step 4: Perform the simulation in R.

Each experiment shown in the Table 3.1.1 will be repeated for 100 times, which means that the replication

for each experiment R is 100. However, the replication results from spurious solutions will be excluded in the needed 100 replications. More detail will be given in Step 5. The detailed steps of the algorithms for parameter estimation for a single replication is shown in the Algorithm (1).

Algorithm 1 The ECM, ECME and MCECM algorithm for GMM for a single replication Input:

- 1. d-dimensional n samples $\boldsymbol{x}_i = (x_{i,1}, ..., x_{i,d})^T \in \mathbb{R}^d, i = 1, ..., n.$
- 2. Total number of component K.
- 3. Starting values of the parameters $\boldsymbol{\eta}^{(0)} = \{\pi_1^{(0)}, ..., \pi_K^{(0)}, \boldsymbol{\mu}_1^{(0)}, ..., \boldsymbol{\mu}_K^{(0)}, \boldsymbol{\Sigma}_1^{(0)}, ..., \boldsymbol{\Sigma}_K^{(0)}\}$
- 4. The used EM algorithm, denoted as $method \in \{ECM, ECME, MCECM\}$.

Output:

- 1. The estimated parameters $\hat{\boldsymbol{\eta}} = \{\hat{\pi}_1, ..., \hat{\pi}_K, \hat{\boldsymbol{\mu}}_1, ..., \hat{\boldsymbol{\mu}}_K, \hat{\boldsymbol{\Sigma}}_1, ..., \hat{\boldsymbol{\Sigma}}_K\}.$
- 2. A clustering partition of the samples.
- 3. Number of iterations, Total computation time, Log-likelihood, BIC, Classification rate.

```
1: iteration t \leftarrow 0
```

2: while t = 0 or relative change of the log-likelihood $\frac{l(\boldsymbol{\eta}^{(t)}) - l(\boldsymbol{\eta}^{(t-1)})}{l(\boldsymbol{\eta}^{(t-1)})} < 0.001$ do iteration $t \leftarrow t+1$ 3: $\pi_k^{(update)} \leftarrow \pi_k^{(t-1)}, \boldsymbol{\mu}_k^{(update)} \leftarrow \boldsymbol{\mu}_k^{(t-1)}, \boldsymbol{\Sigma}_k^{(update)} \leftarrow \boldsymbol{\Sigma}_k^{(t-1)} \quad \text{for} \quad k = 1, ..., K$ 4: 5: The ECM algorithm 6: if method = ECM then 7: E step (Calculate the expected complete data log-likelihood) 8: $\overline{r_{ik}^{(update)}} \leftarrow \frac{\pi_k^{(update)} \mathcal{N}(\boldsymbol{x}_i | \boldsymbol{\mu}_k^{(update)}, \boldsymbol{\Sigma}_k^{(update)})}{\sum_{j=1}^K \pi_j^{(update)} \mathcal{N}(\boldsymbol{x}_i | \boldsymbol{\mu}_j^{(update)}, \boldsymbol{\Sigma}_j^{(update)})} \text{ for } i = 1, ..., n \text{ and } k = 1, ..., K.$ 9: CM step (Maximize the expected complete data Log-likelihood) $\frac{\text{CM step (INIXIIIIZC UNAL)}}{\pi_k^{(update)} \leftarrow \frac{\sum_{i=1}^n r_{ik}^{(update)}}{n} \quad \text{for } k = 1, ..., K$ $\mu_k^{(update)} \leftarrow \frac{\sum_{i=1}^n r_{ik}^{(update)} x_i}{\sum_{i=1}^n r_{ik}^{(update)}} \quad \text{for } k = 1, ..., K$ $\boldsymbol{\Sigma}_k^{(update)} = \frac{\sum_{i=1}^n r_{ik}^{(update)} (\boldsymbol{x}_i - \boldsymbol{\mu}_k^{(update)}) (\boldsymbol{x}_i - \boldsymbol{\mu}_k^{(update)})^T}{\sum_{i=1}^n r_{ik}^{(update)}} \quad \text{for } k = 1, ..., K$ 10:11: 12: 13:14:The ECME algorithm 15:else if method = ECME then 16:17:E step (Calculate the expected complete data log-likelihood) $\overline{r_{ik}^{(update)}} \leftarrow \frac{\pi_k^{(update)} \mathcal{N}(\boldsymbol{x}_i | \boldsymbol{\mu}_k^{(update)}, \boldsymbol{\Sigma}_k^{(update)})}{\sum_{j=1}^K \pi_j^{(update)} \mathcal{N}(\boldsymbol{x}_i | \boldsymbol{\mu}_j^{(update)}, \boldsymbol{\Sigma}_j^{(update)})} \text{ for } i = 1, ..., n \text{ and } k = 1, ..., K.$ 18: CM step 1 (Maximize the expected complete data Log-likelihood) 19: $\frac{1}{\pi_k^{(update)}} \leftarrow \frac{\sum_{i=1}^n r_{ik}^{(update)}}{n} \quad \text{for} \quad k = 1, ..., K$ $\mu_k^{(update)} \leftarrow \frac{\sum_{i=1}^n r_{ik}^{(update)} x_i}{\sum_{i=1}^n r_{ik}^{(update)}} \quad \text{for} \quad k = 1, ..., K$ CM stor 020:21: CM step 2 (Maximize the actual log-likelihood) 22: $\overline{\boldsymbol{\Sigma}_{k}^{(update)}} \leftarrow \arg \max_{\boldsymbol{\Sigma}_{k}} \sum_{i=1}^{n} \ln \left(\sum_{k=1}^{K} \pi_{k}^{(update)} f_{N}(\boldsymbol{x}_{i} \mid \boldsymbol{\mu}_{k}^{(update)}, \boldsymbol{\Sigma}_{k}^{(update)}) \right) \text{ for } k = 1, ..., K$ 23:24:The MCECM algorithm 25: else if method = MCECM then 26:E step (Calculate the expected complete data log-likelihood) 27: $\frac{\pi_k^{(update)} \mathcal{N}(\boldsymbol{x}_i | \boldsymbol{\mu}_k^{(update)}, \boldsymbol{\Sigma}_k^{(update)})}{\sum_{j=1}^{K} \pi_j^{(update)} \mathcal{N}(\boldsymbol{x}_i | \boldsymbol{\mu}_j^{(update)}, \boldsymbol{\Sigma}_j^{(update)})} \text{ for } i = 1, ..., n \text{ and } k = 1, ..., K.$ 28:CM step (Maximize the expected complete data log-likelihood) 29: $\pi_k^{(update)} \leftarrow \frac{\sum_{i=1}^n r_{ik}^{(update)}}{n} \quad \text{for} \quad k = 1, ..., K$ 30: E step (Calculate the expected complete data log-likelihood) 31: $\overline{r_{ik}^{(update)}} \leftarrow \frac{\pi_k^{(update)} \mathcal{N}(\boldsymbol{x}_i | \boldsymbol{\mu}_k^{(update)}, \boldsymbol{\Sigma}_k^{(update)})}{\sum_{j=1}^K \pi_j^{(update)} \mathcal{N}(\boldsymbol{x}_i | \boldsymbol{\mu}_j^{(update)}, \boldsymbol{\Sigma}_j^{(update)}, \boldsymbol{\Sigma}_j^{(update)})} \quad \text{for } i = 1, ..., n \text{ and } k = 1, ..., K.$ 32: CM step (Maximize the expected complete data log-likelihood) 33: $\overline{\mu_k^{(update)}} \leftarrow rac{\sum_{i=1}^n r_{ik}^{(update)} x_i}{\sum_{i=1}^n r_{ik}^{(update)}}$ for k = 1, ..., K34:



Figure 3.1.3: The number of replications removed from the GMM simulation to reach 100 replications

E step (Calculate the expected complete data log-likelihood)
$\overline{r_{ik}^{(update)}} \leftarrow \frac{\pi_k^{(update)} \mathcal{N}(\boldsymbol{x}_i \boldsymbol{\mu}_k^{(update)}, \boldsymbol{\Sigma}_k^{(update)})}{\sum_{j=1}^K \pi_j^{(update)} \mathcal{N}(\boldsymbol{x}_i \boldsymbol{\mu}_j^{(update)}, \boldsymbol{\Sigma}_j^{(update)})} \text{for } i = 1,, n \text{ and } k = 1,, K.$
CM step (Maximize the expected complete data log-likelihood)
$\overline{\boldsymbol{\Sigma}_{k}^{(update)}} = \frac{\sum_{i=1}^{n} r_{ik}^{(update)} (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{k}^{(update)}) (\boldsymbol{x}_{i} - \boldsymbol{\mu}_{k}^{(update)})^{T}}{\sum_{i=1}^{n} r_{ik}^{(update)}} \text{for} k = 1,, K$
end if
$\pi_k^{(t)} \leftarrow \pi_k^{(update)}, \mu_k^{(t)} \leftarrow \mu_k^{(update)}, \Sigma_k^{(t)} \leftarrow \Sigma_k^{(update)} \text{for} k = 1,, K$
$oldsymbol{\eta}^{(t)} \leftarrow \{ \pi_1^{(t)},, \pi_K^{(t)}, oldsymbol{\mu}_1^{(t)},, oldsymbol{\mu}_K^{(t)}, \Sigma_1^{(t)},, oldsymbol{\Sigma}_K^{(t)} \}$
end while
$\hat{m{\eta}} \leftarrow \! \{ \pi_1^{(t)},, \pi_K^{(t)}, m{\mu}_1^{(t)},, m{\mu}_K^{(t)}, m{\Sigma}_1^{(t)},, m{\Sigma}_K^{(t)} \}$

Step 5: Remove the replication results from spurious solutions

In the Section 2.5.2, we have proposed a solution to tackle the issues of singularity and cluster collapse. When the condition from Equation (2.5.5) or (2.5.6) is fulfilled, the replication is stopped and removed immediately and start a new one, until 100 valid replications are completed.

Figure 3.1.3 is a line plot showing the total number of replications removed from the GMM simulation to reach 100 replications. We can see that, comparing the same number of samples, equal mean experiment setting leads to spurious solutions more than different mean experiment setting. For the equal mean experiment setting, it is more difficult to separate the simulated data into 2 clusters. Therefore, the clusters collapse issue introduced in the Section 2.5.2 happens easily, so that all simulated data tend to be classified into 1 cluster only. Also, comparing the same mean (emu) or different mean (umu), a sample size of n = 500 leads more often spurious results compared to n = 100. Last but not least, for the experiment setting emu.500, there are about 400 spurious solutions that the sample generated from VVV and 800 spurious solutions that the sample generated from EVV.

3.1.2 Result visualisation and the performance

Step 6: Create visualization of the performance measures

The graphs for result visualization are shown in Step 7.

Step 7: Analyse and compare the results.

The output of the algorithm shown above is not only the estimated parameters, but also other performance measures, for example mean number of iterations, mean computation time (seconds), mean log-likelihood, mean BIC, mean classification rate. In order to analyse the result, it is more convenient for us to show the results for each algorithm in a graph, for example, line-plot, box-plot to compare the result with regard to the mean or median. The discussion of the result is in the following:

1. Mean number of iterations

We evaluate the clustering performance about iteration of the selected algorithms by visualizing the mean number of iterations and mean total computation time per simulation replication in line plots. For 100 observations, Figure 3.1.4 (a) and 3.1.4 (b) show that among the selected algorithms, ECME requires iterations for each type of Gaussian clustering models. compared to ECME, ECM and MCECM require more iterations. The experiment result proves the following statement that ECME can have a substantially faster convergence rate than either EM or ECM, measured using the number of iterations [Liu and Rubin, 1994]. Secondly, we expect that the mean number of iteration for MCECM should be lower than ECM, because Meng and Rubin [1993] mentioned that performing an E-step before each CM-step may result in larger increases in likelihood function L per iteration since Q is being updated more often. However, according to our experiment result, MCECM requires similar number of iterations like ECM. Last but not least, in the Figure 3.1.4, we can also see that for each type of the Gaussian clustering models, scenarios with different means require more numbers of iterations than scenarios with same cluster means. The reason behind is that it is quite difficult to separate the clusters in the scenario of same means. The log-likelihood cannot be improved so much by updating the parameters in each step. Therefore, the mean number of iteration in the case of sme means is less than the case of different means. For 500 observations, Figure 3.1.4 (c) and 3.1.4 (d) show that the result is quite similar to 100 observations. However, the needed mean number of iteration for EEV and VEV is almost the same around 3 to 4 times for all algorithms.



Figure 3.1.4: The mean number of iterations for the EM algorithms over 100 replications with n = 100 in the Figure (a) and (b) and n = 500 in the Figure (c) and (d) generated from different types of the Gaussian clustering models.



Figure 3.1.5: Box plots of the number of iterations for the EM algorithms over 100 replications with sample size n = 100 and 500 generated from different types of the Gaussian clustering models.

2. Mean total computation time

About the mean total computation time, Figure 3.1.6 shows that among the selected algorithms, ECM and MCECM require the shortest time, just less than 0.04 seconds and ECME requires much longer time, at least about 25 seconds. It makes sense because ECME needs to implement the optimization numerically and ECM and MCECM have the analytic to maximize the Q in the E step. Also, compared 100 to 500 observations, it is obvious to see that, for ECME, the required mean total computation time for 500 observations is much higher than 100 observations. For ECM and MCECM, the required time remains similar. The reason is basically the same as mentioned before.



Figure 3.1.6: The mean total computation time (seconds) of EM algorithms over 100 replications with n = 100 in the Figure (a) and (b) and n = 500 in the Figure (c) and (d) generated from different types of the Gaussian clustering models.



Figure 3.1.7: Box plots of the **computation time** (seconds) of the EM algorithms over 100 replications with sample size n = 100 and 500 generated from different types of the Gaussian clustering models.

3. Classification rate

The results of the classification rate per simulation are shown by using a box plot. For 100 and 500 observations, Figure 3.1.8 show that the median of classification rate in the scenario of different mean is obviously higher than the scenario of same mean. It makes a lot of sense because in the Figure 3.1.1 and 3.1.2, the level of overlapping between two clusters for the same mean is higher than different mean. Also, we expected that the classification rate is higher in EVV and VVV, but the result here doesn't show that.

For 100 observations, for the scenario of same mean, ECME.Nelder-Mead has the highest median of classification rate in VEV, EVV and VVV, and the second highest median of classification rate in EEV. Oppositely, ECM has the lowest median of classification rate in all 4 Gaussain clustering models. For the scenario of different mean, it seems that ECME.BFGS has the best performance because it leads to the highest classification rate among all 4 algorithms. However, there is not a clear pattern which algorithm is the worst for classification rate, because for each scenario, the worst algorithm is different. Last but not least, except EEV.emu and VEV.emu, the classification rate for 500 observations is higher than for 100 observations. According to the Figure 3.1.1 and 3.1.2, we can see that other than EEV.emu and VEV.emu, there are more samples for 500 observations in non-overlapping area than for 100 observations. Therefore, this is the reason why the classification rate improves in 500 observations.


Figure 3.1.8: Box plots of the **classification rate** for the EM algorithms over 100 replications with sample size n = 100 and 500 generated from different types of the Gaussian clustering models.

3 SIMULATION STUDY

4. Normalised BIC

About the value of the normalised BIC and for 100 observations, the Figure 3.1.9 shows that for the scenario of same mean, the median value of the BIC for ECME.Nelder-Mead is the lowest and for ECM is the highest in all 4 Gaussian clustering models. Theoretically, the model with the lowest BIC or higher classification rate is preferred. Therefore, the BIC value and the classification rate shown in the Figure 3.1.8 and 3.1.9 match this model selection rule basically, that the algorithms with higher classification rate have lower BIC and vice versa. For the scenario of different mean, ECME.Nelder-Mead seems to have the lowest BIC and MCECM has the highest BIC in all 4 Gaussian clustering models.



Figure 3.1.9: Box plots of the **normalised BIC** of the EM algorithms over 100 replications with sample size n = 100 and 500 generated from different types of the Gaussian clustering models.

Ranking of the performance for each EM algorithm

The Figure 3.1.7 shows the ranking of the EM algorithms regarding the performance for GMM algorithm. In this paper, we focus on clustering more, so we regard the higher mean classification rate as better performance. If the mean classification rate is the same, the lower mean normalised BIC and shorter mean computation time are preferred. For each scenario, we can see that the mean classification rate and mean normalised BIC are mostly very similar among different EM algorithms, but the mean computation time varies as we mentioned before. In particular, in EVV.umu.100, VVV.umu.100 and VVV.umu.500, MCECM, ECM and ECM perform the worst respectively and their classification rates are at least 5% less than the third ranking methods. Because we don't know which GMM data being clustered in the real situation, for better or more stable clustering performance, we would suggest using ECME.Nelder.Mead and ECME.BFGS for clustering with GMM algorithm.

Table 3.1.7: The ranking of the EM algorithms regarding the performance for GMM algorithm, where the algorithm with 1) higher mean classification rate, 2) lower mean normalised BIC and 3) shorter mean computation time for 100 replications is considered to be the better performance and the priority is given by (1) > (2) >(3).

Code of	Ranking	Algorithm	Classification	Normalised	Computation time	Code of	Ranking	Algorithm	Classification	Normalised	Computation time
experiment			rate	BIC	(Seconds)	experiment			rate	BIC	(Seconds)
	1	MCECM	0.52	9.4	0.023		1	MCECM	0.52	8.9	0.0226
EEV on 100	2	ECM	0.51	9.4	0.0191	EEV one 500	2	ECM	0.51	8.9	0.0121
LLV.enu.roo	3	ECME.Nelder.Mead	0.51	9.4	28.8	EEV.emu.500	3	ECME.BFGS	0.51	8.9	56.1
	4	ECME.BFGS	0.51	9.4	50.7		4	ECME.Nelder.Mead	0.51	8.9	103
	1	ECM	0.89	10.4	0.0211		1	ECM	0.93	9.9	0.0276
FFV uppu 100	2	MCECM	0.88	10.4	0.0288	FEV	2	MCECM	0.93	9.9	0.0449
111 V.uinu.100	3	ECME.Nelder.Mead	0.88	10.4	29.6	1515 V.umu.500	3	ECME.Nelder.Mead	0.92	9.9	141
	4	ECME.BFGS	0.88	10.4	42.7		4	ECME.BFGS	0.92	9.9	153
	1	ECM	0.51	10.8	0.013		1	ECME.BFGS	0.51	10.3	82.3
VEV onu 100	2	MCECM	0.51	10.8	0.0222	VEV omu 500	2	ECME.Nelder.Mead	0.51	10.3	94.2
VEV.emu.100	3	ECME.Nelder.Mead	0.51	10.8	24.1	v 15 v .emu.500	3	ECM	0.50	10.3	0.012
	4	ECME.BFGS	0.51	10.8	40.9		4	MCECM	0.50	10.3	0.0205
	1	ECM	0.75	11.6	0.027		1	ECME.BFGS	0.87	11.0	158
VEV upper 100	2	ECME.BFGS	0.75	11.6	25.6	VEV 500	2	ECME.Nelder.Mead	0.87	11.0	164
V15V.uiiiu.100	3	ECME.Nelder.Mead	0.74	11.6	31.1	VEV.unu.500	3	ECM	0.86	11.0	0.0374
	4	MCECM	0.72	11.6	0.0315		4	MCECM	0.86	11.0	0.0597
	1	ECME.Nelder.Mead	0.55	9.6	25.9		1	ECME.BFGS	0.58	9.2	115
EVV annu 100	2	MCECM	0.53	9.7	0.0236	EVV on 500	2	ECME.Nelder.Mead	0.58	9.2	136
Evvv.enu.100	3	ECME.BFGS	0.53	9.7	45.9	Evv.enu.500	3	MCECM	0.57	9.2	0.0461
	4	ECM	0.52	9.7	0.0155		4	ECM	0.55	9.2	0.0276
	1	ECM	0.93	10.5	0.0314		1	ECM	0.96	10.0	0.0272
EVV	2	ECME.Nelder.Mead	0.93	10.5	28	EVV	2	ECME.BFGS	0.96	10.0	93.5
Evvv.uiiu.100	3	ECME.BFGS	0.93	10.5	29.7	E v v.unu.500	3	ECME.Nelder.Mead	0.96	10.0	124
	4	MCECM	0.84	10.6	0.032		4	MCECM	0.95	10.0	0.0478
	1	MCECM	0.55	11.0	0.0247		1	ECME.BFGS	0.61	10.5	124
WWW orreg 100	2	ECME.Nelder.Mead	0.55	11.0	26.5	VVV omu 500	2	ECME.Nelder.Mead	0.61	10.5	142
v v v.emu.100	3	ECME.BFGS	0.54	11.0	32.4	v v v .emu.500	3	MCECM	0.60	10.5	0.0454
	4	ECM	0.53	11.0	0.0218		4	ECM	0.58	10.5	0.0247
	1	MCECM	0.74	11.7	0.0389		1	MCECM	0.89	11.1	0.0752
WW uppu 100	2	ECME.BFGS	0.74	11.7	20.4	VVV umu FOO	2	ECME.Nelder.Mead	0.83	11.2	171
v v v.unu.100	3	ECME.Nelder.Mead	0.74	11.7	38.3	v v v.umu.300	3	ECME.BFGS	0.82	11.2	115
	4	ECM	0.69	11.8	0.0339		4	ECM	0.75	11.3	0.0448

3.2 EM Algorithm and initialization strategies for GMM algorithm

In the previous Section 3.1, the simulation is done by a particular and fixed starting value. As log-likelihood function for mixture model is a non-convex optimization problem, the point to which the EM algorithm converges depends on the initial value [Jin et al., 2016]. Therefore, a poor set of starting values for the EM algorithm can significantly impact the the quality of the resulting solution [Shireman et al., 2017]. In this section, we will try to use different initialization strategies to perform our simulation for different GMM models.

The basic set up of the experiment is almost the same as the last simulation, but we just implement EEV.umu.500, VEV.umu.500, EVV.umu.500 and VVV.umu.500 this time, which is the experiment setting - Unequal mean of 2 clusters and n = 500 generated samples. The parameters of the GMM for generating samples is the same as before shown in the Table 3 in the last section.

About the starting value of the EM algorithm, we will try "True" and 5 different initialization strategies. "True" means using the true parameters for generating samples as the starting values, so that we can know what is the best performance we can have through different EM algorithms. For initialization strategies, we use 5 strategies, called "Random", "Kmeans", "Hierarchical", "Kmeans (scale)" and "Hierarchical (scale)", to decide the starting values. Basically, all the initialization strategies separate the generated samples into two groups first. Then, the starting value of mean and covariance of each cluster are the sample mean and the sample covariance of each cluster respectively. "Kmeans" is a well-known K-Means Clustering [Hartigan and Wong, 1979]. "Hierarchical" is an agglomerative hierarchical clustering based on the parameterization of VVV model [Fraley, 1998]. Also, the methods with "(scale)" means that the data is applied to the function scale in R for standardization before application of the initialization strategies. The Table 3.2 summarises the used initialization strategies.

Initialization	R nackage	R function	Brief description
	It package	It function	Differ description
strategies			
Dandam			Separate the samples into 2 groups with the same
Kandom	-	-	number randomly.
Vmoong	atat	line o n a	Minimize a Euclidean distance between the cluster
Kineans	stat	kmeans	center and the samples of the cluster.
			Agglomerative is a "bottom-up" approach: each
Hierorchicol	maluat	boWW	observation starts in its own cluster, and pairs of
merarcincai	mcrust	ncvvv	clusters are merged as one moves up the hierarchy,
			based on the maximum likelihood for a VVV model.
Kmeans	stat	kmeans	Data is applied to the function scale for standardization
(scale)	-	scale	before application of the function kmeans.
Hierarchical	mclust	hcVVV	Data is applied to the function scale for standardization
(scale)	-	scale	before application of the function hcVVV.

Table 3.2.1: The summary of the initialization strategies used.

Example 3.2.1 (Visualization of 500 simulated data set generated from the Gaussian mixture model, the initialization clustering by different initialization strategies)

The Figure 3.2.1 shows the true clustering of the scatter plots for each data setting. As mentioned before, if the scatters are not coloured, it is quite hard to observe there are 2 clusters in EEV and VEV. Oppositely, it is clearer to see there are 2 clusters in EVV and VVV. Therefore, we can divide them into two groups, such



True cluster

Figure 3.2.1: The scatter plots and the corresponding pairs plots of 500 simulated data generated from the Gaussian Mixture Model for model types: EEV, VEV, EVV and VVV. The orange and green colour refer to the points in Cluster 1 and 2 respectively.

that 1. Better separated (EVV and VVV) and 2. Worse separated (EEV and VEV).

Because we use different initialisation strategies in the GMM algorithm, the visualisations of the 500 generated data after initialization clustering by different strategies are shown also in the Figure 3.2.2 and Figure 3.2.3. We can see that all initialisation strategies give us different result of initialisation clustering. Also, from the scatter plots of initializing clustering in the Figure 3.2.2 and Figure 3.2.3, we know that the true cluster is with some level of overlapping, but apart from the "random" case, the scatter plots for all other initialisation strategies do not show any level of the overlapping. It implies that "Kmeans", "Hierarchical" and their "(scale)" strategies cannot classify the overlapping scatters.

EVV.umu.500



Figure 3.2.2: The scatter plots of initializing clustering and final clustering by using different initialization strategies for model type **EEV**, Figure 3.2.1 (a) and **VEV**, Figure 3.2.1 (c). The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. First row : EEV.umu.500 (Initializing clustering), second row : EEV.umu.500 (Final clustering), third row : VEV.umu.500 (Initializing clustering) and fourth row : VEV.umu.500 (Final clustering).

EVV.umu.500



Figure 3.2.3: The scatter plots of initializing clustering and final clustering by using different initialization strategies for model type **EVV**, Figure 3.2.1 (e) and **VVV**, Figure 3.2.1 (g). The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. First row : EVV.umu.500 (Initializing clustering), second row : EVV.umu.500 (Final clustering), third row : VVV.umu.500 (Initializing clustering) and fourth row : VVV.umu.500 (Final clustering).

Discussion of the result

1. Scatter plots after initializing and final clustering

The Figure 3.2.2 and Figure 3.2.3 also show the scatter plots with the classification rate after final clustering. You can see that for a particular initialization strategy, the classification rate increases after final clustering, which means the GMM algorithm improves the clustering performance after using initialization strategy.

2. Number of replications removed for ECME.BFGS

The Table 3.2.2 shows the total number of replications removed of ECME.BFGS algorithm with different initialization strategies. The strategies with "(scale)" lead to obviously lower replications removed than the strategies without "(scale)". Also, for each strategies, EVV leads to the most conditions from Equation (2.5.5) or (2.5.6) obviously and there are over 300 removed replications for EVV and the corresponding strategy "Kmeans" and "Hierarchical". For the algorithm with over 300 removed replications, the simulation cannot be finished, so there is no performance measures information for EVV and "Kmeans" and "Hierarchical". Note that there are no any replications removed for other EM algorithm.

The result of the performance measures, including mean computation time, mean normalised BIC and mean classification rate, for each EM algorithm and its extension with different initialization strategies is shown in the Table 3.2.3. A simple analysis is given in the following:

3. Mean computation time

The result is quite regular. For each initialization strategy, the mean total computation time needed in descending order is in the following : ECME.Nelder.Mead > ECME.BFGS > MCECM > ECM. The reason is in the following : 1) There are analytical solutions in ECM and MCECM and ECME involves solving the optimization problem numerically. So, ECME takes more time. 2) MCECM requires E step before each M step and ECM just require a step in each iteration. So, MCECM takes more time than ECM. 3) ECME.BFGS is a gradient based numerical method and the gradient function is given to the R code. ECME.Nelder.Mead takes longer time than ECME.BFGS.

4. Mean normalised BIC

We can see that the difference of mean normalised BIC among each algorithm is quite small. I think this results from the normalisation of the BIC. However, you can also observe that a small change in normalised BIC lead to a bigger difference in classification rate. Also, recalling that the algorithms with the lowest normalised BIC is adopted. For each initialization strategies and model type, ECME.Nelder algorithm always leads to the lowest normalised BIC.

5. Mean classification rate

"Kmeans" outperforms all other strategies with the highest mean classification rate for EEV, VEV and EVV. Also, in VVV, it is quite unexpected that "Random" outperforms all other strategies for ECME quite a lot. Furthermore, for each initialization strategy, the ECME.Nelder algorithm outperforms the other EM algorithms for each data model type. Last but not least, the strategies Kmeans outperforms Kmeans (scale) and Hierarchical outperforms Hierarchical (scale) for all the data sets. Table 3.2.2: The number of replications removed to reach 50 convergent replications by using ECME.BFGS algorithm with different initialization strategies, where the green and red colour represent the best and the worst initialization strategies for each EM algorithm respectively.

	Starting values		In	itialization	strategies	
	True	Random	Kmeans	Kmeans	Hierarchical	Hierarchical
				(scale)		(scale)
EEV	47	22	120	44	89	48
VEV	34	26	200	68	132	49
EVV	44	182	>300	116	>300	218
VVV	30	101	152	52	214	117

Ranking of the performance for each EM algorithm and initialization strategy

The Table 3.2.4 shows the ranking of the performance of EM algorithms and the initialization strategies by GMM algorithm. The same as before, the algorithm with 1) higher classification rate, 2) lower normalised BIC and 3) shorter computation time is considered to be the better performance and the priority is given by (1) > (2) > (3).

Firstly, apart from data setting EVV, we can see that the mean classification rates vary with different EM algorithms and initialization strategies. Secondly, EM algorithms and the initialization strategies in the top 7 performance can perform better in EVV and VVV than in EEV and VEV. It varies our previous statement, that the data with better separation can be clustered more accurately with a suitable method. Thirdly, some methods always perform better for each data settings which are already highlighted in the table with different colours, for example, ECME.Nelder.Mead (Kmeans), ECME.Nelder.Mead (Hierarchical) and so on. This implies that, there are some clustering methods likely performing better, no matter how the characteristics of the data generated from different Gaussian mixture models.

According to the result from the four data settings, for the best fit, ECME.Nelder.Mead with Kmeans or ECME.Nelder.Mead with Hierarchical are recommended for clustering with GMM algorithm. For shorter computation time and good fit, MCECM with Kmeans is recommended.

			Starting value		In	itialization	strategies	
Data	Performance	Algorithm	True	Random	Kmeans	Kmeans	Hierarchical	Hierarchical
Setting	measures	_				(scale)		(scale)
		ECM	0.018	0.011	0.033	0.037	0.039	0.041
		ECME.BFGS	29	47	87	35	58	47
EEV	Computation time	ECME.Nelder.Mead	62	219	208	167	213	201
		MCECM	0.011	0.020	0.040	0.042	0.035	0.046
		ECM	0.007	0.007	0.036	0.039	0.032	0.015
		ECME.BFGS	41	36	165	74	134	55
VEV	Computation time	ECME.Nelder.Mead	63	117	261	245	261	179
		MCECM	0.011	0.013	0.056	0.045	0.049	0.027
		ECM	0.018	0.031	0.022	0.032	0.026	0.031
		ECME BEGS	30	136	-	40	-	37
EVV	Computation time	ECME Nelder Mead	58	238	155	225	146	233
		MCECM	0.011	0.049	0.030	0.059	0.029	0.046
		ECM	0.000	0.018	0.035	0.000	0.020	0.021
		ECME BEGS	36	132	83	41	56	31
VVV	Computation time	ECME Nelder Mead	59	268	197	203	198	
		MCECM	0.011	0.038	0.044	0.030	0.041	0.040
		DOM	0.011	0.038	0.044	0.050	10.041	0.040
		ECM	9.9	10.3	10.1	10.3	10.2	10.2
EEV	Normalised BIC	ECME.BFGS	9.9	10.3	10.2	10.3	10.3	10.3
		ECME.Nelder.Mead	9.9	10.2	10.0	10.2	10.1	10.1
		MCECM	9.9	10.3	10.1	10.2	10.1	10.2
		ECM	11.0	11.3	11.0	11.2	11.1	11.2
VEV	Normalised BIC	ECME.BFGS	11.0	11.3	11.0	11.2	11.1	11.3
		ECME.Nelder.Mead	11.0	11.2	11.0	11.1	11.0	11.1
		MCECM	11.0	11.3	11.0	11.1	11.0	11.2
		ECM	9.9	10.2	9.9	10.2	9.9	10.1
EVV	Normalised BIC	ECME.BFGS	9.9	10.0	-	10.5	-	10.5
2	Tioninanood Bro	ECME.Nelder.Mead	9.9	10.0	9.9	10.1	9.9	10.0
		MCECM	9.9	10.1	9.9	10.1	9.9	10.1
		ECM	11.1	11.6	11.4	11.5	11.3	11.5
VVV	Normalised BIC	ECME.BFGS	11.1	11.2	11.3	11.6	11.5	11.6
	riormansed Bie	ECME.Nelder.Mead	11.1	11.2	11.2	11.4	11.2	11.4
		MCECM	11.1	11.4	11.3	11.5	11.2	11.4
		ECM	0.94	0.59	0.74	0.58	0.70	0.67
DEV		ECME.BFGS	0.94	0.60	0.68	0.58	0.60	0.55
EEV	Classication rate	ECME.Nelder.Mead	0.94	0.72	0.82	0.64	0.78	0.69
		MCECM	0.94	0.63	0.76	0.60	0.72	0.68
		ECM	0.88	0.56	0.76	0.64	0.73	0.56
1 / 1 3 /	<u> </u>	ECME.BFGS	0.88	0.55	0.75	0.61	0.68	0.56
VEV	Classication rate	ECME.Nelder.Mead	0.88	0.62	0.80	0.72	0.79	0.64
		MCECM	0.88	0.55	0.79	0.69	0.78	0.58
		ECM	0.96	0.84	0.96	0.78	0.96	0.84
	<u>a</u>	ECME.BFGS	0.96	0.96	-	0.59	-	0.58
EVV	Classication rate	ECME.Nelder.Mead	0.97	0.95	0.96	0.88	0.96	0.91
		MCECM	0.97	0.88	0.96	0.82	0.96	0.88
		ECM	0.93	0.69	0.74	0.64	0.82	0.66
		ECME.BFGS	0.93	0.87	0.74	0.56	0.59	0.54
VVV	Classication rate	ECME.Nelder.Mead	0.93	0.92	0.84	0.72	0.89	0.74
		MCECM	0.93	0.76	0.82	0.64	0.86	0.69
	1	1			=			

Table 3.2.3: Mean value of the performance measures for each EM algorithm and its extension with different starting values and initialization strategies for 50 replications for GMM algorithm, where the green and red colour represent the best and the worst initialization strategies for each EM algorithm respectively.

Table 3.2.4: The top 7 performance of EM algorithms and the initialization strategies for GMM algorithm, where the algorithm with 1) higher classification rate, 2) lower normalised BIC and 3) shorter computation time is considered to be the better performance and the priority is given by (1) > (2) > (3). *The separation here refers to the level of separation whether the two clusters can be observed easily if the dots are not coloured or marked.

Data setting	Ranking	Algorithm	Initialization	Classification	Normalised	Computation time
			strategy	rate	BIC	(Seconds)
	1	ECME.Nelder.Mead	Kmeans	0.82	10.0	208
	2	ECME.Nelder.Mead	Hierarchical	0.78	10.1	213
	3	MCECM	Kmeans	0.76	10.1	0.040
EEV (Worse separation)	4	ECM	Kmeans	0.74	10.1	0.33
	5	MCECM	Hierarchical	0.72	10.1	0.035
	6	ECME.Nelder.Mead	Random	0.72	10.2	219
	7	ECM	Hierarchical	0.70	10.2	0.039
	1	ECME.Nelder.Mead	Kmeans	0.80	11.0	261
	2	MCECM	Kmeans	0.79	11.0	0.056
	3	ECME.Nelder.Mead	Hierarchical	0.79	11.0	261
VEV (Worse separation)	4	MCECM	Hierarchical	0.78	11.0	0.049
	5	ECM	Kmeans	0.76	11.0	0.036
	6	ECM	Hierarchical	0.73	11.1	0.032
	7	ECME.Nelder.Mead	Kmeans scale	0.72	11.1	245
	1	ECM	Kmeans	0.96	9.9	0.022
	2	ECM	Hierarchical	0.96	9.9	0.026
	3	MCECM	Hierarchical	0.96	9.9	0.029
EVV (Better separation)	4	MCECM	Kmeans	0.96	9.9	0.030
	5	ECME.Nelder.Mead	Hierarchical	0.96	9.9	146
	6	ECME.Nelder.Mead	Kmeans	0.96	9.9	155
	7	ECME.BFGS	Random	0.96	10.0	136
	1	ECME.Nelder.Mead	Random	0.92	11.2	268
	2	ECME.Nelder.Mead	Hierarchical	0.89	11.2	198
	3	ECME.BFGS	Random	0.87	11.2	132
VVV (Better separation)	4	MCECM	Hierarchical	0.86	11.2	0.041
	5	ECME.Nelder.Mead	Kmeans	0.84	11.2	197
	6	ECM	Hierarchical	0.82	11.3	0.022
	7	MCECM	Kmeans	0.82	11.3	0.044

3.3 EM Algorithm and initialization strategies for VCMM algorithm

In Section 2.4.3, the formulation of the vine copula mixture model (VCMM) and the corresponding steps of EM algorithm and its extension are introduced. After the simulation studies for GMM, in this section, our main goal is to compare the performance of different EM algorithms and initialisation strategies for the algorithm of vine copula mixture model clustering (VCMM) proposed by Sahin and Czado [2021]. In that research paper, ECM algorithm is used for the parameter estimation only. In our simulation studies, basically, we will generate different data sets and then use the algorithm to implement classification. On top of that, the algorithm will be amended a bit for the assessment of different EM algorithms, numerical optimization methods and initialisation strategies, in order to assess the performance under different setting. More details will be given in the following steps:

Step 1: Generate data for simulation.

- Step 2: Decide on the EM algorithms and the CM steps order in the VCMM algorithm.
- Step 3: Decide on the initialization strategies for the VCMM algorithm.
- Step 4: Decide on the marginal distribution and copula families for modelling in the VCMM algorithm.
- Step 5: Perform the simulation in R.
- Step 6: Create visualizations of the performance measures.
- Step 7: Analyse and compare the results.

The detail for each step is given in the following subsection:

3.3.1 Data simulation and the experiment setup

Step 1: Generate data for simulation.

In this step, we would like to generate some data from the vine copula mixture model to test different EM algorithms. Compared to the Gaussian mixture model, the vine copula mixture model allows modelling the data much more flexible, as it allows different univariate marginal distribution and asymmetric dependency. Therefore, we would like to not only generate Gaussian data, but also generate the non-Gaussian data to assess the performance of different EM algorithms and initialization strategies under VCMM.

In our experiments, we generate data from four settings following the ways from Sahin and Czado [2021] in their section of simulation studies. In each of the setting, the dimension of each sample d is 3 and number of clusters K is 2. The Table 3.3.1 shows the basic set up for the data generation. Also, about the characteristics of the generated data set, data sets for setting 1 are generated from the mixture model of vine copulas with Non-Gaussian margins, data sets for setting 2, 8 and 9 are from the mixture model of vine copulas with Non-Gaussian pair copulas and Gaussian or Non-Gaussian margins, data sets for setting 3 are from the mixture model of vine copulas with Gaussian copulas and Gaussian margins and the data sets for setting 4, 5, 6 and 7 are from the mixture model of multivariate skew t distributions with different degrees of freedom. The overview of the nine settings are shown in the Table 3.3.2.

The generated data sets for settings 1, 2, 3, 8 and 9 actually follow the vine copula mixture model introduced in the Section 2.4.3. Recalling from Equation (2.4.17), a mixture model with 2 components and 3 dimensions has a density given by

$$f(\boldsymbol{x} \mid \boldsymbol{\eta}) = 0.4 f_1(\boldsymbol{x} \mid \boldsymbol{\psi}_1) + 0.6 f_2(\boldsymbol{x} \mid \boldsymbol{\psi}_2)$$

where the parameters η in the mixture model are denoted as $\eta = \{\psi_1, \psi_2, \pi_1, \pi_2\}$ and recalling from Equation

Notation	Name	Value
~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	Sample size	500 ( For all settings ) and
n	Sample Size	1000 (For setting 1, 3, 5, 6, 7)
d	Dimension of each sample	3
K	Total number of components	2
$\pi_1$	Mixing proportion of component 1	0.4
$\pi_2$	Mixing proportion of component 2	0.6

Table 3.3.1: The parameter values used in the nine settings of data simulation

Table 3.3.2: The overview of the characteristics of the nine settings of data simulation

	Type of minture model	Marginal distribution	Dependency relationship
	Type of mixture model	Marginar distribution	(Copula family)
Setting 1	VCMM	Non-Gaussian	Non-Gaussian
Setting 2	VCMM	Gaussian and Non-Gaussian	Non-Gaussian
Setting 3	VCMM (= Multivariate Gaussian)	Gaussian	Gaussian
Setting 4	Mixture of multivariate skew t	Skew t	-
Setting 5	Mixture of multivariate skew t	Skew t	-
Setting 6	Mixture of multivariate skew t	Skew t	-
Setting 7	Mixture of multivariate skew t	Skew t	-
Setting 8	VCMM	Gaussian and Non-Gaussian	Non-Gaussian
Setting 9	VCMM	Gaussian and Non-Gaussian	Non-Gaussian

(2.4.18), (2.4.19), the  $f_1(\boldsymbol{x} \mid \boldsymbol{\psi}_1)$  and  $f_2(\boldsymbol{x} \mid \boldsymbol{\psi}_2)$  for each data set are given in the following:

$$\begin{split} f_{1}(\boldsymbol{x} \mid \boldsymbol{\psi}_{2}) &= c_{(2)1,3;2}(F_{(2)1|2}(x_{1} \mid x_{2}; \boldsymbol{\gamma}_{1(2)}, \boldsymbol{\gamma}_{2(2)}, \boldsymbol{\theta}_{(2)1,2}), F_{(2)3|2}(x_{3} \mid x_{2}; \boldsymbol{\gamma}_{3(2)}, \boldsymbol{\gamma}_{2(2)}, \boldsymbol{\theta}_{(2)2,3}); \boldsymbol{\theta}_{(2)1,3;2}) \\ &\times c_{(2)2,3}(F_{2(2)}(x_{2}; \boldsymbol{\gamma}_{2(2)}), F_{3(2)}(x_{3}; \boldsymbol{\gamma}_{3(2)}); \boldsymbol{\theta}_{(2)2,3}) \\ &\times c_{(2)1,2}(F_{1(2)}(x_{1}; \boldsymbol{\gamma}_{1(2)}), F_{2(2)}(x_{2}; \boldsymbol{\gamma}_{2(2)}); \boldsymbol{\theta}_{(2)1,2}) \\ &\times f_{3(2)}(x_{3}; \boldsymbol{\gamma}_{3(2)}) f_{2(2)}(x_{2}; \boldsymbol{\gamma}_{2(2)}) f_{1(2)}(x_{1}; \boldsymbol{\gamma}_{1(2)}) \end{split}$$

#### Setting 1, 8 and 9

 $\begin{aligned} f_{2}(\boldsymbol{x} \mid \boldsymbol{\psi}_{2}) &= c_{(2)1,2;3}(F_{(2)1|3}(x_{1} \mid x_{3}; \boldsymbol{\gamma}_{1(2)}, \boldsymbol{\gamma}_{3(2)}, \boldsymbol{\theta}_{(2)1,3}), F_{(2)2|3}(x_{2} \mid x_{3}; \boldsymbol{\gamma}_{2(2)}, \boldsymbol{\gamma}_{3(2)}, \boldsymbol{\theta}_{(2)2,3}); \boldsymbol{\theta}_{(2)1,2;3}) \\ &\times c_{(2)2,3}(F_{2(2)}(x_{2}; \boldsymbol{\gamma}_{2(2)}), F_{3(2)}(x_{3}; \boldsymbol{\gamma}_{3(2)}); \boldsymbol{\theta}_{(2)2,3}) \\ &\times c_{(2)1,3}(F_{1(2)}(x_{1}; \boldsymbol{\gamma}_{1(2)}), F_{3(2)}(x_{3}; \boldsymbol{\gamma}_{3(2)}); \boldsymbol{\theta}_{(2)1,3}) \end{aligned}$ 

×  $f_{3(2)}(x_3; \gamma_{3(2)}) f_{2(2)}(x_2; \gamma_{2(2)}) f_{1(2)}(x_1; \gamma_{1(2)})$ 

## Setting 2 and 3 $\,$

$$\begin{aligned} f_{2}(\boldsymbol{x} \mid \boldsymbol{\psi}_{2}) &= c_{(2)1,3;2}(F_{(2)1\mid2}(x_{1}\mid x_{2};\boldsymbol{\gamma}_{1(2)},\boldsymbol{\gamma}_{2(2)},\boldsymbol{\theta}_{(2)1,2}), F_{(2)3\mid2}(x_{3}\mid x_{2};\boldsymbol{\gamma}_{3(2)},\boldsymbol{\gamma}_{2(2)},\boldsymbol{\theta}_{(2)2,3}); \boldsymbol{\theta}_{(2)1,3;2}) \\ &\times c_{(2)2,3}(F_{2(2)}(x_{2};\boldsymbol{\gamma}_{2(2)}), F_{3(2)}(x_{3};\boldsymbol{\gamma}_{3(2)}); \boldsymbol{\theta}_{(2)2,3}) \\ &\times c_{(2)1,2}(F_{1(2)}(x_{1};\boldsymbol{\gamma}_{1(2)}), F_{2(2)}(x_{2};\boldsymbol{\gamma}_{2(2)}); \boldsymbol{\theta}_{(2)1,2}) \\ &\times f_{3(2)}(x_{3};\boldsymbol{\gamma}_{3(2)}) f_{2(2)}(x_{2};\boldsymbol{\gamma}_{2(2)}) f_{1(2)}(x_{1};\boldsymbol{\gamma}_{1(2)}) \end{aligned}$$

In particular, for setting 1, 2, 3, 8 and 9 of the marginal distribution F and the associated marginal parameters  $\gamma$  are shown in the Table 3.3.4. Also, the bivariate copula families c used and the associated copula parameters  $\theta$  are shown in the Figure 3.3.1.

The setting 4, 5, 6 and 7 are generated by the mixture of multivariate skew t distributions with two clusters and the density is given by

#### Setting 4, 5, 6 and 7

0.4 
$$ST(\boldsymbol{x} \mid \boldsymbol{\mu}_1, \boldsymbol{\Sigma}_1, \boldsymbol{\lambda}_1, v_1) + 0.6 ST(\boldsymbol{x} \mid \boldsymbol{\mu}_2, \boldsymbol{\Sigma}_2, \boldsymbol{\lambda}_2, v_2).$$

where  $ST(. | \boldsymbol{\mu}, \boldsymbol{\Sigma}, \boldsymbol{\lambda}, v)$  is the density of multivariate skew t distributions,  $\boldsymbol{\mu}_1$  and  $\boldsymbol{\mu}_2$  are location vectors,  $\boldsymbol{\Sigma}_1$ and  $\boldsymbol{\Sigma}_2$  are scale matrices,  $\boldsymbol{\lambda}_1$  and  $\boldsymbol{\lambda}_2$  are skewness vectors and  $v_1$  and  $v_2$  are degrees of freedom for cluster 1 and 2 respectively. Also, the parameters used for data simulation in mixture of multivariate skew t distributions are shown in the Table 3.3.3.

		<u>Cluster 1</u>				<u>Cluster 2</u>		
	$oldsymbol{\mu}_1$	${oldsymbol{\Sigma}}_1$	$oldsymbol{\lambda}_1$	$v_1$	$oldsymbol{\mu}_2$	$\mathbf{\Sigma}_2$	$oldsymbol{\lambda}_2$	$v_2$
Setting 4	$(1, 1, 0)^T$	$\left[\begin{array}{rrrrr} 2 & 1 & 1 \\ 1 & 2 & 1 \\ 1 & 1 & 2 \end{array}\right]$	$(4, -4, 4)^T$	8	$(-2, -2, -2)^T$	$\left[\begin{array}{rrrrr} 2 & 1 & 1 \\ 1 & 2 & 1 \\ 1 & 1 & 2 \end{array}\right]$	$(-4, 4, 4)^T$	10
Setting 5	$(1, 1, 0)^T$	$\left[\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	$(4, -4, 4)^T$	3	$(-2, -2, -2)^T$	$\left[\begin{array}{rrrrr} 2 & 1 & 1 \\ 1 & 2 & 1 \\ 1 & 1 & 2 \end{array}\right]$	$(-4, 4, 4)^T$	3
Setting 6	$(1, 1, 0)^T$	$\left[\begin{array}{rrrrr} 2 & 1 & 1 \\ 1 & 2 & 1 \\ 1 & 1 & 2 \end{array}\right]$	$(4, -4, 4)^T$	3	$(-4.5, 1, 0)^T$	$\left[\begin{array}{rrrrr} 3 & -1 & -1 \\ -1 & 3 & -1 \\ -1 & -1 & 3 \end{array}\right]$	$(-4, 4, 4)^T$	3
Setting 7	$(1, 1, 0)^T$	$\left[\begin{array}{rrrr} 0.5 & 0.1 & 0.1 \\ 0.1 & 0.5 & 0.1 \\ 0.1 & 0.1 & 0.5 \end{array}\right]$	$(4, -4, 4)^T$	3	$(-5, -5, -5)^T$	$\left[\begin{array}{rrrrr} 50 & -1 & -1 \\ -1 & 3 & -1 \\ -1 & -1 & 3 \end{array}\right]$	$(-4, 4, 4)^T$	3

Table 3.3.3: The parameters used for data simulation in the mixture of multivariate skew t distributions for each cluster.

			$\underline{\text{Cluster } 2}$			
	$F_{1(1)}(oldsymbol{\gamma}_{1(1)})$	$F_{2(1)}(\gamma_{2(1)})$	$F_{3(1)}({m \gamma}_{3(1)})$	$F_{1(2)}(\gamma_{1(2)})$	$F_{2(2)}(m{\gamma}_{2(2)})$	$F_{3(2)}(\gamma_{3(2)})$
Sotting 1	llogis(1.5, 1.25)	exp(0.1)	lnorm(0.1, 1.3)	lnorm(2.5, 0.5)	logis(5,3)	exp(0.05)
Setting 1	(6.41, doesn't exist)	(10, 10)	(2.57, 5.41)	(13.80, 7.36)	(5, 9.64)	(20, 20)
Sotting 2	$\mathcal{N}(1,2)$	exp(0.2)	lnorm(0.8, 0.8)	lnorm(1.5, 0.4)	$\mathcal{N}(18,5)$	exp(0.2)
Setting 2	(1,2)	(5,5)	(3.06, 2.90)	(4.85, 2.02)	(18,5)	(5,5)
Sotting 2	$\mathcal{N}(0,2)$	$\mathcal{N}(1,2)$	$\mathcal{N}(1,2)$	$\mathcal{N}(0,2)$	$\mathcal{N}(1,2)$	$\mathcal{N}(-2,2)$
Setting 5	(0,2)	(1,2)	(1,2)	(0,2)	(1,2)	(-2,2)
Sotting 9	$\mathcal{N}(20, 25)$	$\mathcal{N}(15, 10)$	lnorm(2.3, 0.8)	lnorm(3, 0.4)	logis(8, 10)	$\Gamma(0.5,1)$
Setting o	(20, 25)	(15,10)	(13.74, 13.051)	(21.76, 9.06)	(8, 32.15)	(0.5, 0.71)
Sotting 0	$\mathcal{N}(20, 25)$	$\mathcal{N}(15, 10)$	lnorm(0, 0.3)	lnorm(3, 0.4)	logis(8, 10)	$\Gamma(0.5,1)$
Setting 9	(20, 25)	(15,10)	(1.05, 0.32)	(21.76, 9.06)	(8, 32.15)	(0.5, 0.71)

Table 3.3.4: The parameters of the univariate marginal distributions for each dimension and cluster. The abbreviation for the marginal distributions :  $\mathcal{N}(\mu, \sigma)$ : normal distribution with mean  $\mu$  and standard deviation  $\sigma$ .  $exp(\lambda)$ : exponential distribution with rate parameter  $\lambda$ .  $lnorm(\mu, \sigma)$ : log-normal distribution with mean  $\mu$  and standard deviation  $\sigma$  on the logarithmic scale. logis(l, s): logistic distribution with location parameter l and scale parameter s.  $llogis(\alpha, \beta)$ : log-logistic distribution with shape parameter  $\alpha$  and scale parameter  $\beta$ .  $\Gamma(\alpha, \beta)$ : gamma distribution with shape parameter  $\alpha$  and rate parameter  $\beta$ . The mean and the standard deviation of the univariate marginal distributions are given inside the parenthesis (mean, standard deviation) below the abbreviation.

Example 3.3.1 (Visualization of sample data sets from the vine copula mixture model)



Figure 3.3.1: Vine tree structure of simulated data from the vine copula mixture model for **Setting 1, 8 and 9** : (a), (b), **Setting 2**: (c), (d) and **Setting 3**: (e), (f). A capital letter at an edge refers to the bivariate copula family: BB1: BB1 copula, C: Clayton, SC: Survival Clayton, F: Frank, G: Gumbel, SG: Survival Gumbel, J: Joe copula and N : Gaussian copula. The true parameter value and corresponding Kendall's  $\tau$  of the pair copula are given inside the parenthesis (parameter(s)/Kendall's  $\tau$ ).

Table 3.3.5: The characteristics of the clusters in nine settings of data simulation. *The separation here refers to the level of separation whether the two clusters can be observed easily if the dots of each cluster are not coloured or marked. **The overlap here refers whether the two clusters can be separated by a plane almost completely.

		Characte	ristics of clusters	
C	Shape	Separation*	Overlap**	Volume
Setting	(Elliptical / Skew elliptical / Non elliptical)	(Well / Not well)	(Overlapping / Non-overlapping)	(Similar / Different / Very different)
1	Non elliptical	Well separated	Non-overlapping	Different
2	Non elliptical	Not well separated	Overlapping	Similar
3	Elliptical	Well separated	Overlapping (X shape)	Similar
4	Elliptical / Skew elliptical	Not well separated	Overlapping	Similar
5	Skew elliptical	Not well separated	Overlapping	Different
6	Skew elliptical	Well separated	Non-overlapping	Similar
7	Skew elliptical	Well separated	Non-overlapping	Very different
8	Non elliptical	Well separated	Non-overlapping	Different
9	Non elliptical	Well separated	Overlapping (X shape)	Similar

This example will show you the visualisations of each data settings and the scatter plots for 500 and 1000 samples are shown in the Figure 3.3.2, 3.3.3 and 3.3.4. According to the scatter plots for the sample data, we can conclude four different characteristics of the clusters for each data setting, including shapes, separation, overlap and volumes and the conclusion is shown in the Table 3.3.5. Note that the separation here is not exactly the same as overlap. Separation here refers to the level of separation whether the two clusters can be observed easily if the dots are not coloured or marked. Also, the overlap here refers whether the two clusters are heavily overlapping like a "X", but they are still well separated, because we can observe that the data consists of two clusters clearly. The last characteristics volume classifies the size difference between two clusters. In data setting 7, the size difference of the clusters is much bigger than other data settings, so its characteristics volume is classified as "Very different".

Table 3.3.6: Selection of the EM algorithms, optimization methods for marginal parameter  $\gamma$  and their CM steps order for testing.

Algorithm	P	arameter upda	ate
(Optimization method for $\boldsymbol{\gamma}$ )	CM step 1	CM step $2$	CM step $3$
ECM (Nelder-Mead)	π	$\boldsymbol{ heta}$	$\gamma$
ECM (BFGS)	$\pi$	$\boldsymbol{ heta}$	$\gamma$
MCECM (Nelder-Mead)	$\pi$	$\boldsymbol{ heta}$	$\gamma$
MCECM (BFGS)	$\pi$	$\boldsymbol{ heta}$	$\gamma$
ECME (Nelder-Mead)	$\pi$	$\boldsymbol{ heta}$	$\gamma$
ECME (BFGS)	$\pi$	heta	$\gamma$

### Step 2: Decide on the EM algorithms and the CM steps order in the VCMM algorithm

In our experiment, we will use 3 EM algorithms including ECM, MCECM and ECME algorithm to estimate the parameters in the vine copula mixture model. As mentioned in the Section 2.4.3, the parameters that we need to estimate are mixing proportion  $\pi$ , copula parameter  $\theta$  and marginal parameter  $\gamma$ . For mixing proportion  $\pi$ , we can estimate it by using the Equation (2.4.20). Also, we have no analytical solution for the copula parameter  $\theta$ , so we can use the R function RVineSeqMLE in the R package VineCopula to estimate the pair-copula parameters by maximization likelihood estimation (MLE) which uses L-BFGS-B to estimate the copula parameters. For marginal parameter  $\gamma$ , there is no analytical solution for estimation and two numerical optimization methods of which Nelder-Mead is a heuristic search optimization without derivatives and BFGS is gradient based optimization method are used for marginal parameter estimation. In the experiment, each EM



True cluster (n = 500)

Figure 3.3.2: The scatter plots and the corresponding pairs plots of 500 simulated data generated from setting 1 to 5. The orange and green colour refer to the points in Cluster 1 and 2 respectively.



Figure 3.3.3: The scatter plots and the corresponding pairs plots of 500 simulated data generated from setting 6 to 9. The orange and green colour refer to the points in Cluster 1 and 2 respectively.



Figure 3.3.4: The scatter plots and the corresponding pairs plots of 1000 simulated data generated from setting 1, 3, 5, 6 and 7. The orange and green colour refer to the points in Cluster 1 and 2 respectively.

algorithm is used with two different optimization methods to estimate the marginal parameter  $\gamma$ , so we study six algorithms in total. About the order of the CM steps, we just follow the steps shown in the Section 2.4.3. Basically, the parameters update in each CM step follows the order : Mixing proportion  $\pi$  > Copula parameter  $\theta$  > Marginal parameter  $\gamma$ . The overview of the selection of the EM algorithms, optimization methods for marginal parameter and their CM steps in our experiment are shown in the Table 3.3.6.

#### Step 3: Decide on the initialization strategies for the VCMM algorithm

In Sahin and Czado [2021], "Kmeans (scale)" is used for the initialization. Here, we use further initialization strategies which were already introduced in the last Section 3.2. They are "Random", "Kmeans", "Hierarchical", "Kmeans (scale)" and "Hierarchical (scale)".

# Step 4: Decide on the marginal distribution and copula families for modelling in the VCMM algorithm

VCMM algorithm is a very flexible modelling method, because it can model many combinations of marginal distribution, copula family and pair copula construction. Here is a brief introduction for the flexibility of the VCMM algorithm:

#### 1. Pair copula construction (Vine tree structure)

In the Theorem 2.3.18 (A pair copula construction in three dimensions), we have mentioned that there are 3 different ways for copula construction to model different relationship between the variables:

$$\begin{split} f(x_1, x_2, x_3) &= c_{1,3;2}(F_{1|2}(x_1|x_2), F_{3|2}(x_3|x_2)) \times c_{2,3}(F_2(x_2), F_3(x_3)) \\ &\times c_{1,2}(F_1(x_1), F_2(x_2))f_3(x_3)f_2(x_2)f_1(x_1) \\ f(x_1, x_2, x_3) &= c_{1,2;3}(F_{1|3}(x_1|x_3), F_{2|1}(x_2|x_2)) \times c_{1,3}(F_1(x_1), F_3(x_3)) \\ &\times c_{2,3}(F_2(x_2), F_3(x_3))f_3(x_3)f_2(x_2)f_1(x_1) \\ f(x_1, x_2, x_3) &= c_{2,3;1}(F_{2|1}(x_2|x_1), F_{3|1}(x_3|x_1)) \times c_{1,3}(F_1(x_1), F_3(x_3)) \\ &\times c_{1,2}(F_1(x_1), F_2(x_2))f_3(x_3)f_2(x_2)f_1(x_1) \end{split}$$

### **2.** Marginal distribution F

In the VCMM algorithm, we will select 7 different candidates for the marginal distribution: 1. Gaussian distribution, 2. Log-normal distribution, 3. Exponential distribution, 4. Log-logistic distribution, 5. Logistic distribution, 6. Gamma distribution and 7. Student's t distribution with degrees of freedom 3.

#### 3. Copula family c

In the VCMM algorithm, we will use the following copula families :

Gaussian copula, Student's t copula, Gumbel copula, Clayton copula, Frank copula, Joe copula, BB1 copula, BB6 copula and their rotated copulas with 90°, 180°, 270° degrees.

In our simulation studies, we will use 2 different modelling methods which allow the use of different marginal distributions and copula families. The detail is shown in the Table 3.3.7.

Table 3.3.7: The overview of the marginal distribution and copula families used for the modelling in VCMM algorithm. *All refers to the marginal distributions and copula families mentioned above in step 4.

Modelling method	Marginal distribution	Copula family	Pair copula construction
Gaussian VCMM	Gaussian	Gaussian	Determined by the VCMM algorithm
General VCMM	All*	All*	Determined by the VCMM algorithm

## Step 5: Perform the simulation in R

Each experiment shown in the Table 2.1.9 will be repeated for 50 times, which means that the replication for each experiment R is 50. The detail of the VCMM algorithm is not shown here. For more information, please refer to Sahin and Czado [2021] for the VCMM algorithm and the Section 2.4.2 in this thesis for the E and CM steps of the EM algorithms for parameter estimation.

Abbreviations	Setting	Initialization	Modelling	Sample size (n)
		strategies	method	
1.R.Gau.n	1	Random	Gaussian VCMM	500 and 1000
1.K.Gau.n	1	Kemans	Gaussian VCMM	500  and  1000
1.H.Gau.n	1	Hierarchical	Gaussian VCMM	500  and  1000
1.KS.Gau.n	1	Kemans (scale)	Gaussian VCMM	500  and  1000
1.HS.Gau.n	1	Hierarchical (scale)	Gaussian VCMM	500  and  1000
2.R.Gau.n	2	Random	Gaussian VCMM	500
2.K.Gau.n	2	Kemans	Gaussian VCMM	500
2.H.Gau.n	2	Hierarchical	Gaussian VCMM	500
2.KS.Gau.n	2	Kemans (scale)	Gaussian VCMM	500
2.HS.Gau.n	2	Hierarchical (scale)	Gaussian VCMM	500
3.R.Gau.n	3	Random	Gaussian VCMM	500 and $1000$
3.K.Gau.n	3	Kemans	Gaussian VCMM	500 and $1000$
3.H.Gau.n	3	Hierarchical	Gaussian VCMM	500 and $1000$
3.KS.Gau.n	3	Kemans (scale)	Gaussian VCMM	500 and $1000$
3.HS.Gau.n	3	Hierarchical (scale)	Gaussian VCMM	500  and  1000
4.R.Gau.n	4	Random	Gaussian VCMM	500
4.K.Gau.n	4	Kemans	Gaussian VCMM	500
4.H.Gau.n	4	Hierarchical	Gaussian VCMM	500
4.KS.Gau.n	4	Kemans (scale)	Gaussian VCMM	500
4.HS.Gau.n	4	Hierarchical (scale)	Gaussian VCMM	500
5.R.Gau.n	5	Random	Gaussian VCMM	500  and  1000
5.K.Gau.n	5	Kemans	Gaussian VCMM	500  and  1000
5.H.Gau.n	5	Hierarchical	Gaussian VCMM	500  and  1000
5.KS.Gau.n	5	Kemans (scale)	Gaussian VCMM	500 and $1000$
5.HS.Gau.n	5	Hierarchical (scale)	Gaussian VCMM	500  and  1000
6.R.Gau.n	6	Random	Gaussian VCMM	500 and $1000$
6.K.Gau.n	6	Kemans	Gaussian VCMM	500 and $1000$
6.H.Gau.n	6	Hierarchical	Gaussian VCMM	500 and $1000$
6.KS.Gau.n	6	Kemans (scale)	Gaussian VCMM	500 and $1000$
6.HS.Gau.n	6	Hierarchical (scale)	Gaussian VCMM	500  and  1000
7.R.Gau.n	7	Random	Gaussian VCMM	500 and $1000$
7.K.Gau.n	7	Kemans	Gaussian VCMM	500 and $1000$
7.H.Gau.n	7	Hierarchical	Gaussian VCMM	500  and  1000
7.KS.Gau.n	7	Kemans (scale)	Gaussian VCMM	500 and $1000$
7.HS.Gau.n	7	Hierarchical (scale)	Gaussian VCMM	500  and  1000
8.R.Gau.n	8	Random	Gaussian VCM $\overline{\mathbf{M}}$	500
8.K.Gau.n	8	Kemans	Gaussian VCMM	500
8.H.Gau.n	8	Hierarchical	Gaussian VCMM	500
8.KS.Gau.n	8	Kemans (scale)	Gaussian VCMM	500
8.HS.Gau.n	8	Hierarchical (scale)	Gaussian VCMM	500
9.R.Gau.n	9	Random	Gaussian VCMM	500
9.K.Gau.n	9	Kemans	Gaussian VCMM	500
9.H.Gau.n	9	Hierarchical	Gaussian VCMM	500
9.KS.Gau.n	9	Kemans (scale)	Gaussian VCMM	500
9.HS.Gau.n	9	Hierarchical (scale)	Gaussian VCMM	500

Table 3.3.8: Abbreviations for the studied experiments

Abbreviations	Setting	Initialization	Modelling	Sample size (n)
		strategies	method	
1.R.Gen.n	1	Random	General VCMM	500 and 1000
1.K.Gen.n	1	Kemans	General VCMM	500  and  1000
1.H.Gen.n	1	Hierarchical	General VCMM	500  and  1000
1.KS.Gen.n	1	Kemans (scale)	General VCMM	500  and  1000
1.HS.Gen.n	1	Hierarchical (scale)	General VCMM	500  and  1000
2.R.Gen.n	2	Random	General VCMM	500
2.K.Gen.n	2	Kemans	General VCMM	500
2.H.Gen.n	2	Hierarchical	General VCMM	500
2.KS.Gen.n	2	Kemans (scale)	General VCMM	500
2.HS.Gen.n	2	Hierarchical (scale)	General VCMM	500
3.R.Gen.n	3	Random	General VCMM	500 and 1000
3.K.Gen.n	3	Kemans	General VCMM	500  and  1000
3.H.Gen.n	3	Hierarchical	General VCMM	500  and  1000
3.KS.Gen.n	3	Kemans (scale)	General VCMM	500  and  1000
3.HS.Gen.n	3	Hierarchical (scale)	General VCMM	500  and  1000
4.R.Gen.n	4	Random	General VCMM	500
4.K.Gen.n	4	Kemans	General VCMM	500
4.H.Gen.n	4	Hierarchical	General VCMM	500
4.KS.Gen.n	4	Kemans (scale)	General VCMM	500
4.HS.Gen.n	4	Hierarchical (scale)	General VCMM	500
5.R.Gen.n	5	Random	General VCMM	500 and 1000
5.K.Gen.n	5	Kemans	General VCMM	500  and  1000
5.H.Gen.n	5	Hierarchical	General VCMM	500  and  1000
5.KS.Gen.n	5	Kemans (scale)	General VCMM	500  and  1000
5.HS.Gen.n	5	Hierarchical (scale)	General VCMM	500  and  1000
6.R.Gen.n	6	Random	General VCMM	500 and 1000
6.K.Gen.n	6	Kemans	General VCMM	500  and  1000
6.H.Gen.n	6	Hierarchical	General VCMM	500  and  1000
6.KS.Gen.n	6	Kemans (scale)	General VCMM	500  and  1000
6.HS.Gen.n	6	Hierarchical (scale)	General VCMM	500  and  1000
7.R.Gen.n	7	Random	General VCMM	500 and 1000
7.K.Gen.n	7	Kemans	General VCMM	500  and  1000
7.H.Gen.n	7	Hierarchical	General VCMM	500  and  1000
7.KS.Gen.n	7	Kemans (scale)	General VCMM	500  and  1000
7.HS.Gen.n	7	Hierarchical (scale)	General VCMM	500  and  1000
8.R.Gen.n	8	Random	General VCMM	500
8.K.Gen.n	8	Kemans	General VCMM	500
8.H.Gen.n	8	Hierarchical	General VCMM	500
8.KS.Gen.n	8	Kemans (scale)	General VCMM	500
8.HS.Gen.n	8	Hierarchical (scale)	General VCMM	500
9.R.Gen.n	9	Random	General VCMM	500
9.K.Gen.n	9	Kemans	General VCMM	500
9.H.Gen.n	9	Hierarchical	General VCMM	500
9.KS.Gen.n	9	Kemans (scale)	General VCMM	500
9.HS.Gen.n	9	Hierarchical (scale)	General VCMM	500

#### 3.3.2 Result visualisation and the performance

#### Step 6: Create visualization of the performance measures

The graphs for result visualization are shown in Step 7.

#### Step 7: Analyse and compare the results

#### 1. Scatter plots after initializing and final clustering

# Setting 1, n = 500



Figure 3.3.5: The scatter plots of initializing clustering and final clustering by using different initialization strategies for setting 1 with sample size 500. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.





Figure 3.3.6: The scatter plots of initializing clustering and final clustering by using different initialization strategies for setting 2 with sample size 500. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.

Setting 3, n = 500



Figure 3.3.7: The scatter plots of initializing clustering and final clustering by using different initialization strategies for setting 3 with sample size 500. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.

Setting 4, n = 500



Figure 3.3.8: The scatter plots of initializing clustering and final clustering by using different initialization strategies for setting 3 with sample size 500. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.

Setting 5, n = 500



Figure 3.3.9: The scatter plots of initializing clustering and final clustering by using different initialization strategies for setting 5 with sample size 500. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.

Setting 6, n = 500



Figure 3.3.10: The scatter plots of initializing clustering and final clustering by using different initialization strategies for setting 6 with sample size 500. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.





Figure 3.3.11: The scatter plots of initializing clustering and final clustering by using different initialization strategies for Setting 7 with sample size 500. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.

Setting 8, n = 500



Figure 3.3.12: The scatter plots of initializing clustering and final clustering by using different initialization strategies for Setting 8 with sample size 500. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.





Figure 3.3.13: The scatter plots of initializing clustering and final clustering by using different initialization strategies for Setting 9 with sample size 500. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.

Setting 1, n = 1000



Figure 3.3.14: The scatter plots of initializing clustering and final clustering by using different initialization strategies for setting 1 with sample size 1000. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.

Setting 3, n = 1000



Figure 3.3.15: The scatter plots of initializing clustering and final clustering by using different initialization strategies for setting 3 with sample size 1000. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.





Figure 3.3.16: The scatter plots of initializing clustering and final clustering by using different initialization strategies for setting 5 with sample size 1000. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.

Setting 6, n = 1000



Figure 3.3.17: The scatter plots of initializing clustering and final clustering by using different initialization strategies for Setting 6 with sample size 1000. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.
Setting 7, n = 1000



Figure 3.3.18: The scatter plots of initializing clustering and final clustering by using different initialization strategies for Setting 7 with sample size 1000. The blue and pink colour refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. For final clustering, the EM algorithm reaching the highest classification rate is used here. First row : Initializing clustering, second row : Final clustering by General VCMM and third row : Final clustering by Gaussian VCMM.

The Figure 3.3.5 - 3.3.18 show you the scatter plots of initializing clustering and final clustering by using different initialization strategies for each data setting. In the scatter plots 3.3.2, 3.3.3 and 3.3.4, we know that there are totally no overlapping data in data setting 1 and 8. However, after initializing clustering with Kmeans, the data cannot be clustered accurately and Hierarchical can do better than Kmeans. We think that the reason is that Kmeans cannot cluster the non-Gaussian data well, even if the data are not overlapping. Also, for data setting 1, 5, 7 and 8, the Table 3.3.5 shows the volumes of two clusters are not similar. The initializing clustering with Kmeans (scale) for those data settings can cluster better than Kmeans. In particular, for data 7, the volume of the two clusters are very different and it turns out that the classification rate with Kmeans (scale) is much better than Kmeans.

For the result with the classification rate after final clustering with General VCMM, you can observe that for a particular initialization strategy, the classification rate mostly increases after final clustering, which means the General VCMM algorithm improves the clustering performance after using initialization strategy. It proves that the high flexibility of General VCMM can do clustering better than well-known clustering methods.

#### 2. Comparison of the result for different data settings

The result of the performance measures, including mean computation time, mean normalised BIC and mean classification rate, for each EM algorithm with different initialization strategies are shown in the Table 3.3.9, 3.3.10 and 3.3.11 respectively. We can see that for a particular initialization strategy, the normalised BIC and classification rate perform usually similar within different EM algorithms, but there is a great difference in computation time. In the table 3.3.9, the EM algorithms with heuristic based optimization method (Nelder-Maad) is taking more time than with gradient based optimization method (BFGS) in many situations. This could be due to the fact that VCMM might have a nonlinear objective function with many local optima, making heuristic approaches convergence time longer. Also, for General VCMM, for a particular initialization strategy, ECME.BFGS requires more computation time than ECM.BFGS and MCECM.BFGS.

For a particular EM algorithm, the performance can vary a lot by using different initialization strategies. Therefore, regarding to the accuracy of clustering, the way of initialization for EM algorithm is the key factor. Also, we can see that, in many cases, random definitely requires longer computation times than other methods. After initializing clustering with random, we just can get the 50% accuracy of clustering, which is definitely lower than others. Although it cannot improve after initializing clustering, it still tries to find the best partition while taking too much time.

Also, we would like to analyse and discuss the classification rate of different data settings based on the four characteristics mentioned in the Table 3.3.5, for example, shape, separation, overlap and volume. Because our main goal is to assess the performance for data clustering, we will focus on the classification rate here for assessment.

## Separation

For data setting 1, 3, 6, 7, 8 and 9, the two clusters are well separated, where the two clusters can be observed easily even the dots are not marked. You can see that the classification rate with General VCMM and Random can reach over 84%. Although classification rate after initializing cluster is just 50%, the performance after General and Gaussian VCMM turns out to be surprisingly good. Also, in particular, for data setting 1, 8, where the data generated from VCMM model and non-overlapping, although Random performs very well, Hierarchical performs more often better.

#### Overlap

For data setting 3 and 9, the two clusters are overlapping like a "X". In the Figure 3.3.2 (e) and 3.3.3 (g), you can see that the two clusters are heavily overlapping, but well separated. According to the result of classifi-

cation rate of General VCMM, random outperforms all other initialization strategies for the "X" cluster. For the data settings with overlapping clusters and not good separation, such as 2, 4 and 5, Kmeans performs the best.

#### Volume

Recalling in the Example 3.3.1, the volumes of two clusters are not similar for data setting 1, 5, 7 and 8 and the initializing clustering with Kmeans (scale) for those data settings can cluster better than Kmeans. The result in the Table 3.3.11 shows that after final clustering with General VCMM, for data setting 1, 5, 7, the classification rate of Kmeans (scale) is not better than Kmeans. However, for the data setting 7 with very different volumes of clusters, the performances of Kmeans (scale) is better than Kmeans and Hierarchical (scale) is at least as good as Hierarchical.

## Shape

The shape of the data settings can be separated into 3 groups, which are elliptical, skew elliptical and non elliptical. For data setting 4, its characteristics is in between elliptical and skew elliptical, because the degrees of freedoms for each cluster are quite high which are 8 and 10 respectively. For a particular EM algorithm and initialization strategy, we can see that the Gaussian VCMM can perform better than General VCMM for data setting 3 and 4 with elliptical cluster. For data setting 1, 2, 8 and 9 with non elliptical cluster, the General VCMM can perform better than Gaussian VCMM. For data setting 5, 6 and 7 with skew elliptical cluster, the General VCMM can perform mostly better than Gaussian VCMM, except data setting 9 (1000 samples) clustered by ECM or MCECM with Kmeans. Also, for data setting 4, 5, 6 generated from mixture of skew t distribution without very different volumes of clusters, Kmeans outperform other initialization strategies.

### Summary and recommendation

Here is the summary and recommendation, according to the analysis and discussion above:

- For clustering with VCMM algorithm, the process of parameter estimation by different EM algorithms doesn't change the accuracy much.
- The EM algorithms with Neldar-Mead is taking more time than with BFGS in many situations. For general VCMM, ECM.BFGS or MCECM.BFGS are recommended due to shorter required computation time.
- For non elliptical and skew elliptical data, General VCMM for clustering is performing better than Gaussian VCMM as expected.
- In general, random requires longer computation time.
- For the clusters with good separation, random as an initialization strategy is a good choice for initial clustering. More specifically, for the shape of the clusters like a "X", Random is the best choice. Also, for the data without any overlap, Hierarchical is the best choice.
- For the clusters with overlap and not good separation, Kmeans is suggested.
- For the clusters with very different volumes, standardization first before applying General VCMM is suggested.

• For data generated from skew elliptical with similar / small different volumes of clusters, General VCMM with Kmeans is preferred.

Table 3.3.9: Mean value of the computation time for each EM algorithm and its extension with different initialization strategies for 50 replications, where green and red colour represent the best and the worst initialization strategies for each EM algorithm respectively.

- · ·					General VC!	MM				Gaussian VC	<u>MM</u>	
Setting (Semple cize)	Performance	Algorithm	Random	Kmeans	Hierarchical	Kmeans (coolo)	Hierarchical (conlo)	Random	Kmeans	Hierarchical	(conlo)	Hierarchical (conlo)
(Sample size)	measures					(scale)	(scale)				(scale)	(scale)
				VCM	1M (Non e	lliptical)						
		ECM.Nelder.Mead	217	124	92	95	97	144	110	77	102	80
		ECM.BFGS	186	97	84	73	77	134	92	67	89	72
1 (500)	Computation time	ECME.Nelder.Mead	269	139	154	110	145	87	85	67	69	66
- (000)	<i>p</i>	ECME.BFGS	181	85	117	70	96	73	67	54	59	55
		MCECM.Nelder.Mead	210	124	95	93	103	147	106	77	99	83
		MCECM.BFGS ECM Nolder Mond	224	94	120	201	76	137	91	121	91	150
		ECM.BEGS	285	347	113	164	170	235	164	112	175	143
		ECME.Nelder.Mead	370	416	183	209	218	120	124	96	110	106
1(1000)	Computation time	ECME.BFGS	307	298	163	165	170	107	105	88	98	95
		MCECM.Nelder.Mead	296	456	128	173	190	236	177	126	163	153
		MCECM.BFGS	285	341	112	160	158	217	159	115	153	138
		ECM.Nelder.Mead	157	70	71	110	90	143	103	77	94	64
		ECM.BFGS	88	61	74	89	79	108	82	56	74	49
2 (500)	Computation time	ECME.Nelder.Mead	180	102	129	177	137	148	99	71	103	64
		ECME.BFGS	98	88	109	144	02	129	84	62	89	57
		MCECM BEGS	85	61	72	88	79	113	83	57	78	50
		ECM.Nelder.Mead	104	214	59	115	145	64	74	39	70	60
		ECM.BFGS	86	168	47	89	121	53	61	33	59	50
0 (500)	a	ECME.Nelder.Mead	117	279	87	150	196	43	61	36	61	55
8 (500)	Computation time	ECME.BFGS	105	229	73	118	147	38	52	33	53	48
		MCECM.Nelder.Mead	103	224	55	117	143	62	72	37	69	58
		MCECM.BFGS	441	1105	48	101	121	55	62	33	59	50
		ECM.Nelder.Mead	123	156	111	167	140	88	88	72	96	80
		ECMENUL M	98	116	84	118	97	82	75	68	90	74
9 (500)	Computation time	ECME RECS	141	206	173	178	170	52	85	57 52	84	75 66
		MCECM Nelder Mood	117	161	111	160	124	-40	14 84	71	96	76
		MCECM.BFGS	98	115	85	116	100	76	76	61	83	67
	1			0.0.0			• \					
			v	СММ (	Multivaria	te Gauss	sian)					
		ECM.Nelder.Mead	114	93	78	67	66	68	79	63	98	82
		ECM.BFGS	121	94	82	67	62	59	73	58	85	70
3 (500)	Computation time	ECME.Nelder.Mead	118	79	89	70	66	80	83	70	108	93
. ,		ECME.BFGS	121	77	82	73	62	78	83	72	96	83
		MCECM.Neider.Mead	104	88	75	04 68	64	52	73	07 55	90	75 67
		ECM Nelder Mead	191	153	154	111	153	87	144	99	152	132
		ECM.BFGS	194	161	155	117	151	87	145	99	154	132
a (4000)		ECME.Nelder.Mead	185	131	147	119	150	77	137	101	132	123
3 (1000)	Computation time	ECME.BFGS	192	142	148	122	157	76	135	100	137	127
		MCECM.Nelder.Mead	171	155	146	111	152	77	137	95	148	123
		MCECM.BFGS	175	163	148	116	150	76	140	95	150	125
			1	Mixture	of multiva	riate ske	w t					
		ECM Nelder Mead	161	50	50	50	56	127	47	64	51	65
		ECM.BEGS	158	45	42	42	49	116	43	57	47	59
		ECME.Nelder.Mead	177	55	58	58	66	142	60	76	51	74
4 (500)	Computation time	ECME.BFGS	166	46	47	45	54	146	48	64	51	74
		MCECM.Nelder.Mead	166	44	42	44	54	124	42	59	41	65
		MCECM.BFGS	155	42	42	42	50	115	41	60	41	56
		ECM.Nelder.Mead	360	103	100	92	112	43	55	54	54	60
		ECM.BFGS	297	86	86	77	105	42	54	53	52	59
5 (500)	Computation time	ECME.Nelder.Mead	436	113	108	111	127	56	59	60	58	62
		ECME.BFGS	360	100	92	97	120	00	68 59	70	62	72
		MCECM RECS	979	99	90	70	102	20	56	54	51	56
		ECM Nelder Mead	748	159	221	138	217	73	86	86	87	100
		ECM.BFGS	615	138	187	123	206	72	87	85	86	99
F (1000)	Comments in the	ECME.Nelder.Mead	751	166	215	162	220	85	84	87	84	102
5 (1000)	Computation time	ECME.BFGS	636	143	188	144	198	99	95	99	89	111
		MCECM.Nelder.Mead	689	155	206	139	217	69	83	80	84	94
		MCECM.BFGS	523	133	178	118	197	64	79	77	79	89
		ECM.Nelder.Mead	233	49	63	50	101	57	22	28	28	28
		ECMENUL M	204	40	54	42	84	55	21	26	25	26
6 (500)	Computation time	ECME RECS	213	13	93 79	71	119	73	31	34 39	31 29	30 34
		MCECM Nelder Mead	228	50	65	51	101	51	20	97	28	28
		MCECM.BFGS	200	41	53	43	87	50	21	25	23	27
		ECM.Nelder.Mead	495	92	135	112	163	79	32	47	34	43
		ECM.BFGS	364	65	112	85	140	75	30	44	35	43
6 (1000)	Computation time	ECME.Nelder.Mead	345	137	182	161	194	87	40	46	46	46
0 (1000)		ECME.BFGS	294	119	175	137	177	96	43	49	46	49
		MCECM.Nelder.Mead	445	94	141	108	168	70	30	42	33	45
		MCECM.BFGS	361	79	143	101	152	73	38	44	33	42
		ECM.Nelder.Mead	265	125	47	85	45	145	56	18	37	17
		ECME Nelder Mord	214	87	38	09 110	30	128	52 48	17 95	25	14
7 (500)	Computation time	ECME RECS	2.94	146	57	0/	59	79	-40 15	20	36 36	22
		MCECM.Nelder.Mead	249	123	47	34 82	45	141	40 57	18	29	16
		MCECM.BFGS	224	90	38	71	36	128	51	17	25	14
		ECM.Nelder.Mead	465	736	344	545	349	248	1004	87	44	25
		ECM.BFGS	315	206	72	133	61	239	92	31	42	24
7 (1000)	Computation time	ECME.Nelder.Mead	1166	1536	565	176	106	97	66	35	47	28
. (1000)		ECME.BFGS	412	552	448	639	329	259	645	372	45	27
		MCECM.Nelder.Mead	517	679	362	540	278	248	970	247	44	24
	1	MCECM.BFGS	343	258	73	139	71	1014	147	149	155	88

Table 3.3.10: Mean value of the normalised BIC for each EM algorithm and its extension with different initialization strategies for 50 replications, where green and red colour represent the best and the worst initialization strategies for each EM algorithm respectively.

					General VC	MM				Gaussian VC	MM	
Setting	Performance	Algorithm	Random	Kmeans	Hierarchical	Kmeans	Hierarchical	Random	Kmeans	Hierarchical	Kmeans	Hierarchical
(Sample size)	measures					(scale)	(scale)				(scale)	(scale)
				VCN	AM (Non	elliptical	)					
	1		10 -0		10 50	1	/	20.00		10 -1	10.00	10.00
		ECM.Neider.Mead	16.76	17.98	16.59	17.99	17.74	20.03	19.95	19.74	19.90	19.82
		ECME Nelder Mead	16.70	17.99	16.57	18.01	17.70	10.00	19.95	19.74	19.91	19.82
1(500)	Normalised BIC	ECME BEGS	16 74	18.02	16.54	18.02	17.69	19.92	19.94	19.70	19.90	19.81
		MCECM.Nelder.Mead	16.79	17.98	16.59	17.98	17.76	20.01	19.94	19.74	19.90	19.82
		MCECM.BFGS	16.74	17.99	16.57	17.98	17.70	20.01	19.94	19.74	19.90	19.82
		ECM.Nelder.Mead	16.50	17.22	16.50	17.80	17.71	19.93	19.88	19.66	19.86	19.83
		ECM.BFGS	16.50	17.28	16.48	17.72	17.70	19.93	19.88	19.68	19.90	19.86
1 (1000)	Normalised BIC	ECME.Nelder.Mead	16.49	17.53	16.51	17.82	17.78	19.87	19.88	19.66	19.86	19.83
1 (1000)	riormanood Die	ECME.BFGS	16.49	17.51	16.48	17.76	17.74	19.92	19.92	19.66	19.89	19.86
		MCECM.Nelder.Mead	16.53	17.22	16.49	17.82	17.71	19.94	19.88	19.66	19.86	19.84
		ECM Nolder Mond	12.06	17.28	10.48	12.94	12.18	19.90	14.94	19.08	19.88	14.28
		ECM BEGS	13.97	13.16	13.16	13.24	13.17	14.67	14.24	14.31	14.52	14.20
		ECME.Nelder.Mead	13.78	13.13	13.14	13.18	13.19	14.69	14.23	14.31	14.50	14.28
2 (500)	Normalised BIC	ECME.BFGS	13.88	13.12	13.16	13.24	13.17	14.69	14.23	14.31	14.50	14.28
		MCECM.Nelder.Mead	13.98	13.15	13.16	13.24	13.18	14.65	14.24	14.31	14.52	14.28
		MCECM.BFGS	13.98	13.16	13.16	13.24	13.17	14.66	14.24	14.31	14.52	14.28
		ECM.Nelder.Mead	17.67	18.69	17.26	18.92	18.40	19.92	19.92	19.90	19.96	19.95
		ECM.BFGS	17.67	18.68	17.26	18.95	18.36	19.92	19.92	19.90	19.96	19.95
8 (500)	Normalised BIC	ECME.Neider.Mead	17.60	18.72	17.20	18.90	18.30	19.91	19.91	19.90	19.96	19.95
		MCECM.Nelder.Mead	17.73	18.68	17.26	18.91	18.42	19.91	19.92	19.90	19.96	19.95
		MCECM.BFGS	17.73	18.67	17.26	18.89	18.38	19.92	19.92	19.90	19.96	19.95
		ECM.Nelder.Mead	14.65	16.00	15.32	15.46	15.59	16.68	17.35	16.83	16.99	17.05
		ECM.BFGS	14.60	15.99	15.32	15.46	15.62	16.68	17.37	16.83	16.99	17.05
9 (500)	Normalised BIC	ECME.Nelder.Mead	14.57	15.95	15.20	15.67	15.55	16.56	17.28	16.80	16.98	17.00
- ()		ECME.BFGS	14.58	15.95	15.24	15.49	15.50	16.61	17.28	16.80	16.98	17.00
		MCECM.Neider.Mead	14.65	15.95	15.32	15.46	15.59	16.73	17.35	16.83	17.02	17.08
		MCECM.BFG5	14.00	15.99	13.32	15.50	15.01	10.75	17.55	10.00	17.02	17.08
			1	VCMM	(Multivari	ate Gaus	sian)					
		ECM.Nelder.Mead	12.10	12.39	12.30	12.41	12.38	12.01	12.14	12.06	12.16	12.14
		ECM.BFGS	12.10	12.40	12.30	12.41	12.38	12.01	12.14	12.06	12.16	12.14
3 (500)	Normalised BIC	ECME.Nelder.Mead	12.13	12.38	12.29	12.40	12.38	12.06	12.15	12.06	12.15	12.14
0 (000)		ECME.BFGS	12.12	12.38	12.29	12.40	12.38	12.06	12.15	12.06	12.15	12.14
		MCECM.Nelder.Mead	12.08	12.39	12.30	12.41	12.37	12.01	12.14	12.06	12.16	12.14
		MCECM.BFGS ECM Nolder Mond	12.08	12.39	12.30	12.41	12.38	12.01	12.14	12.06	12.16	12.14
		ECM.BFGS	12.01	12.32	12.18	12.31	12.23	11.95	12.00	12.00	12.07	12.02
2 (1000)		ECME.Nelder.Mead	12.04	12.31	12.17	12.30	12.26	11.99	12.08	12.00	12.08	12.02
3 (1000)	Normalised BIC	ECME.BFGS	12.04	12.31	12.17	12.30	12.26	11.99	12.08	12.00	12.08	12.02
		MCECM.Nelder.Mead	12.01	12.32	12.18	12.31	12.27	11.95	12.06	12.00	12.07	12.02
		MCECM.BFGS	12.01	12.32	12.18	12.32	12.27	11.95	12.06	12.00	12.07	12.02
				Mixture	of multiv	ariate sk	ew t					
		ECM Noldor Mond	10.07	10.70	10.81	10.78	10.80	10.84	10.78	10.78	10.77	10.78
		ECM.BFGS	10.97	10.79	10.81	10.79	10.80	10.84	10.77	10.78	10.77	10.78
		ECME.Nelder.Mead	10.97	10.79	10.81	10.79	10.80	10.84	10.77	10.78	10.77	10.78
4 (500)	Normalised BIC	ECME.BFGS	10.97	10.79	10.80	10.79	10.80	10.84	10.77	10.78	10.77	10.78
		MCECM.Nelder.Mead	10.97	10.79	10.81	10.78	10.80	10.84	10.78	10.78	10.77	10.78
		MCECM.BFGS	10.97	10.79	10.81	10.78	10.80	10.84	10.77	10.78	10.77	10.78
		ECM.Nelder.Mead	12.16	12.08	12.10	12.08	12.08	12.12	12.12	12.10	12.12	12.12
		ECM.BFGS	12.17	12.08	12.10	12.08	12.08	12.13	12.12	12.10	12.12	12.12
5(500)	Normalised BIC	ECME BEGS	12.17	12.08	12.10	12.08	12.09	12.12	12.12	12.11	12.12	12.13
		MCECM.Nelder.Mead	12.17	12.08	12.10	12.08	12.08	12.13	12.12	12.10	12.12	12.12
		MCECM.BFGS	12.17	12.08	12.10	12.08	12.09	12.13	12.12	12.10	12.12	12.12
		ECM.Nelder.Mead	12.09	12.02	12.05	12.02	12.01	12.12	12.12	12.12	12.12	12.12
		ECM.BFGS	12.09	12.02	12.05	12.02	12.01	12.12	12.12	12.12	12.12	12.12
5 (1000)	Normalised BIC	ECME.Nelder.Mead	12.08	12.01	12.05	12.02	12.01	12.12	12.12	12.12	12.12	12.12
		ECME.BFGS MCECM Noldor Model	12.08	12.01	12.04	12.01	12.01	12.12	12.12	12.12	12.12	12.12
		MCECM.BFGS	12.10	12.02	12.05	12.02	12.01	12.12	12.12	12.12	12.12	12.12
		ECM.Nelder.Mead	13.49	13.32	13.41	13.38	13.38	14.36	14.35	14.36	14.37	14.42
		ECM.BFGS	13.47	13.32	13.41	13.38	13.38	14.36	14.35	14.36	14.37	14.42
6 (500)	Normalised BIC	ECME.Nelder.Mead	13.47	13.32	13.41	13.38	13.38	14.38	14.35	14.36	14.36	14.42
0 (000)	riormanood Die	ECME.BFGS	13.53	13.32	13.41	13.38	13.38	14.38	14.34	14.36	14.35	14.42
		MCECM.Nelder.Mead	13.48	13.32	13.41	13.38	13.38	14.37	14.35	14.36	14.36	14.42
		MCECM.BFGS	13.47	13.32	13.41	13.38	13.38	14.37	14.35	14.36	14.37	14.42
		ECM.BEGS	13.33	13.26	13.29	13.30	13.30	14.41	14.44	14.47	14.40	14.50
0 (100)		ECME.Nelder.Mead	13.34	13.26	13.29	13.31	13.30	14.41	14.44	14.47	14.46	14.50
6 (1000)	Normalised BIC	ECME.BFGS	13.36	13.26	13.29	13.30	13.30	14.41	14.45	14.47	14.44	14.50
		MCECM.Nelder.Mead	13.36	13.26	13.29	13.31	13.30	14.41	14.44	14.47	14.46	14.50
		MCECM.BFGS	13.35	13.26	13.29	13.31	13.30	14.41	14.46	14.47	14.44	14.50
		ECM.Nelder.Mead	14.11	14.40	13.92	13.92	13.93	14.76	14.80	14.70	14.78	14.70
		ECME Noldon Mag 3	14.07	14.41	13.92	13.92	13.93	14.75	14.80	14.70	14.78	14.70
7 (500)	Normalised BIC	ECME RECS	14.10	14.00	13.92	13.92	13.93	14.74	14.70	14.70	14.74	14.70
		MCECM.Nelder.Mead	14.09	14.38	13.92	13.92	13.93	14.76	14.80	14.70	14.74	14.70
		MCECM.BFGS	14.04	14.41	13.92	13.92	13.92	14.76	14.80	14.70	14.78	14.70
		ECM.Nelder.Mead	13.84	14.06	13.79	13.82	13.79	14.73	14.72	14.78	14.86	14.70
		ECM.BFGS	13.83	14.06	13.80	13.82	13.79	14.73	14.72	14.78	14.82	14.70
7 (1000)	Normalised BIC	ECME.Nelder.Mead	13.84	13.83	13.79	13.82	13.79	14.70	14.72	14.78	14.76	14.70
. /		ECME.BFGS	13.90	13.85	13.80	13.82	13.79	14.67	14.72	14.78	14.76	14.70
		MCECM.Neider.Mead MCECM.BFGS	13.82	14.06	13.79	13.82	13.79	14.73	14.72	14.78	14.85	14.70

Table 3.3.11: Mean value of the classification rate for each EM algorithm and its extension with different initialization strategies for 50 replications, where green and red colour represent the best and the worst initialization strategies for each EM algorithm respectively. Remark : the classification rate close to the best value (within 3%) is highlighted as orange.

					General VC	MM				Gaussian VC	CMM	
Setting	Performance	Algorithm	Random	Kmeans	Hierarchical	l Kmeans	Hierarchical	Random	Kmeans	Hierarchical	Kmeans	Hierarchical
(Sample size)	measures					(scale)	(scale)				(scale)	(scale)
			Data	generation	1 : VCM	M (Non	elliptical)					
		ECM.Nelder.Mead	0.872	0.597	0.934	0.610	0.637	0.592	0.599	0.724	0.589	0.611
		ECM.BFGS	0.885	0.596	0.942	0.610	0.647	0.594	0.598	0.724	0.589	0.612
1 (500)	Classification rate	ECME.Nelder.Mead	0.879	0.602	0.940	0.601	0.628	0.604	0.606	0.730	0.593	0.617
()		ECME.BFGS	0.875	0.584	0.944	0.600	0.647	0.620	0.603	0.725	0.593	0.615
		MCECM.Nelder.Mead	0.868	0.597	0.937	0.611	0.634	0.587	0.598	0.723	0.590	0.613
		ECM Nelder Mead	0.880	0.390	0.945	0.610	0.630	0.582	0.601	0.725	0.590	0.586
		ECM.BFGS	0.930	0.765	0.927	0.644	0.634	0.582	0.601	0.736	0.586	0.584
4 (4000)	<b>C1</b> 10 11 1	ECME.Nelder.Mead	0.930	0.709	0.923	0.628	0.619	0.596	0.605	0.735	0.595	0.590
1 (1000)	Classification rate	ECME.BFGS	0.932	0.709	0.927	0.640	0.623	0.602	0.601	0.735	0.592	0.588
		MCECM.Nelder.Mead	0.923	0.778	0.924	0.626	0.631	0.586	0.601	0.735	0.588	0.586
		MCECM.BFGS	0.935	0.767	0.927	0.644	0.634	0.584	0.601	0.736	0.587	0.583
		ECM.Nelder.Mead	0.627	0.894	0.865	0.861	0.863	0.532	0.811	0.765	0.621	0.777
		ECM.BFGS ECME Nolder Mood	0.625	0.889	0.871	0.863	0.870	0.532	0.811	0.765	0.621	0.777
2 (500)	Classification rate	ECME BEGS	0.645	0.889	0.871	0.863	0.850	0.519	0.803	0.765	0.621	0.775
		MCECM.Nelder.Mead	0.621	0.894	0.866	0.862	0.863	0.546	0.811	0.764	0.616	0.776
		MCECM.BFGS	0.619	0.889	0.871	0.863	0.870	0.544	0.811	0.764	0.617	0.776
		ECM.Nelder.Mead	0.862	0.650	0.991	0.608	0.732	0.907	0.905	0.899	0.892	0.888
		ECM.BFGS	0.861	0.654	0.990	0.602	0.745	0.907	0.905	0.899	0.892	0.888
8 (500)	Classification rate	ECME.Nelder.Mead	0.900	0.648	0.990	0.603	0.752	0.870	0.881	0.893	0.874	0.867
0 (000)		ECME.BFGS	0.912	0.658	0.990	0.611	0.743	0.866	0.882	0.892	0.874	0.862
		MCECM.Neider.Mead	0.843	0.654	0.990	0.607	0.726	0.906	0.905	0.900	0.892	0.887
		ECM Nelder Mead	0.843	0.564	0.990	0.014	0.738	0.906	0.905	0.900	0.892	0.606
		ECM.BFGS	0.860	0.562	0.700	0.756	0.678	0.722	0.575	0.645	0.634	0.606
0 (70-)	<b>a a</b>	ECME.Nelder.Mead	0.877	0.569	0.728	0.731	0.698	0.664	0.580	0.639	0.622	0.608
9 (500)	Classification rate	ECME.BFGS	0.878	0.573	0.720	0.757	0.708	0.690	0.580	0.639	0.620	0.608
		MCECM.Nelder.Mead	0.847	0.569	0.699	0.757	0.685	0.726	0.576	0.645	0.631	0.603
		MCECM.BFGS	0.867	0.561	0.700	0.750	0.679	0.726	0.576	0.637	0.631	0.603
		Da	ta gene	eration : V	CMM (I	Multivari	iate Gaussia	m)				
			0.024	0.888		0.505	0.500		0.886	0.004	0.540	0.850
		ECM.Nelder.Mead	0.821	0.575	0.678	0.537	0.560	0.857	0.756	0.824	0.743	0.759
		ECME Nelder Mead	0.821	0.570	0.678	0.537	0.557	0.820	0.750	0.824	0.745	0.759
3 (500)	Classification rate	ECME.BFGS	0.799	0.572	0.680	0.536	0.557	0.821	0.746	0.824	0.747	0.756
		MCECM.Nelder.Mead	0.835	0.575	0.677	0.537	0.563	0.865	0.756	0.824	0.740	0.760
		MCECM.BFGS	0.835	0.571	0.677	0.537	0.560	0.865	0.756	0.824	0.741	0.760
		ECM.Nelder.Mead	0.850	0.558	0.713	0.528	0.587	0.877	0.787	0.844	0.772	0.825
		ECM.BFGS	0.850	0.559	0.713	0.528	0.587	0.876	0.787	0.844	0.772	0.825
3 (1000)	Classification rate	ECME.Nelder.Mead	0.816	0.550	0.714	0.530	0.593	0.839	0.770	0.844	0.768	0.826
0 (1000)	chaometrion ruce	ECME.BFGS	0.816	0.550	0.714	0.529	0.593	0.839	0.770	0.844	0.768	0.825
		MCECM.Nelder.Mead	0.851	0.559	0.713	0.528	0.588	0.877	0.787	0.844	0.776	0.825
		MCECM.BFG5	0.601	0.009	0.715	0.328	0.387	0.011	0.181	0.844	0.770	0.825
		D	ata gen	eration : 1	Mixture	of multiv	ariate skew	t				
		ECM.Nelder.Mead	0.535	0.826	0.796	0.805	0.774	0.578	0.833	0.818	0.829	0.807
		ECM.BFGS	0.535	0.826	0.797	0.804	0.774	0.578	0.827	0.814	0.825	0.803
4 (500)	Classification rate	ECME.Nelder.Mead	0.538	0.824	0.796	0.803	0.772	0.590	0.826	0.811	0.824	0.794
1 (000)	Chatchiedelion rule	ECME.BFGS	0.541	0.825	0.796	0.804	0.774	0.590	0.826	0.812	0.825	0.793
		MCECM.Nelder.Mead	0.535	0.826	0.796	0.805	0.774	0.568	0.831	0.818	0.829	0.803
		MCECM.BFGS	0.535	0.826	0.796	0.805	0.774	0.568	0.826	0.811	0.825	0.803
		ECM.Neider.Mead	0.557	0.755	0.678	0.746	0.705	0.520	0.521	0.522	0.522	0.521
		ECME.Nelder.Mead	0.560	0.753	0.679	0.746	0.705	0.520	0.522	0.522	0.522	0.523
5(500)	Classification rate	ECME.BFGS	0.563	0.753	0.679	0.746	0.708	0.521	0.522	0.522	0.522	0.523
		MCECM.Nelder.Mead	0.545	0.756	0.677	0.747	0.708	0.520	0.521	0.522	0.522	0.521
		MCECM.BFGS	0.542	0.755	0.677	0.746	0.707	0.520	0.521	0.522	0.522	0.521
		ECM.Nelder.Mead	0.601	0.756	0.664	0.744	0.701	0.520	0.520	0.520	0.520	0.520
		ECM.BFGS	0.602	0.756	0.662	0.744	0.700	0.520	0.520	0.520	0.520	0.520
5 (1000)	Classification rate	ECME DECC	0.624	0.754	0.664	0.742	0.702	0.520	0.521	0.520	0.521	0.520
		MCECM Nelder Mood	0.574	0.754	0.003	0.742	0.703	0.520	0.521	0.520	0.521	0.520
		MCECM.BFGS	0.573	0.756	0.662	0.744	0.700	0.520	0.521	0.520	0.521	0.520
		ECM.Nelder.Mead	0.905	0.987	0.939	0.958	0.960	0.599	0.970	0.856	0.927	0.835
		ECM.BFGS	0.918	0.987	0.939	0.958	0.960	0.599	0.970	0.856	0.927	0.835
6 (500)	Classification rate	ECME.Nelder.Mead	0.918	0.987	0.939	0.958	0.961	0.556	0.969	0.855	0.926	0.832
0 (000)	Classification rate	ECME.BFGS	0.897	0.987	0.939	0.958	0.960	0.546	0.969	0.855	0.925	0.832
		MCECM.Nelder.Mead	0.912	0.987	0.939	0.958	0.960	0.615	0.969	0.855	0.927	0.833
		MCECM.BFGS	0.914	0.987	0.939	0.958	0.960	0.615	0.969	0.856	0.927	0.833
		ECM RECS	0.948	0.988	0.973	0.974	0.974	0.554	0.971	0.876	0.929	0.840
		ECME.Nelder.Mead	0.955	0.988	0.973	0.973	0.974	0.528	0.971	0.881	0.931	0.837
6 (1000)	Classification rate	ECME.BFGS	0.949	0.988	0.973	0.974	0.975	0.528	0.970	0.880	0.929	0.837
		MCECM.Nelder.Mead	0.945	0.988	0.973	0.968	0.974	0.555	0.971	0.882	0.929	0.839
		MCECM.BFGS	0.950	0.988	0.973	0.968	0.974	0.554	0.970	0.881	0.927	0.839
		ECM.Nelder.Mead	0.937	0.834	0.974	0.980	0.971	0.909	0.947	0.954	0.953	0.954
		ECM.BFGS	0.949	0.830	0.974	0.980	0.972	0.909	0.947	0.954	0.953	0.954
7 (500)	Classification rate	ECME.Nelder.Mead	0.946	0.939	0.975	0.980	0.972	0.918	0.939	0.954	0.951	0.954
		EGME.BFGS MCECM Noldor Merd	0.946	0.935	0.975	0.980	0.972	0.918	0.940	0.954	0.951	0.954
		MCECM REGS	0.940	0.840	0.974	0.980	0.972	0.909	0.946	0.954	0.953	0.954
		ECM.Nelder.Mead	0.973	0.898	0,980	0.974	0.980	0.938	0.958	0.950	0.952	0.957
		ECM.BFGS	0.974	0.899	0.978	0.974	0.980	0.938	0.958	0.950	0.951	0.957
7 (1000)	Classifier time and	ECME.Nelder.Mead	0.974	0.969	0.980	0.974	0.980	0.947	0.958	0.950	0.950	0.956
7 (1000)	Classification rate	ECME.BFGS	0.960	0.960	0.978	0.974	0.980	0.947	0.958	0.951	0.950	0.956
		MCECM.Nelder.Mead	0.973	0.898	0.979	0.974	0.980	0.938	0.958	0.950	0.951	0.956
	1	MCECM.BFGS	0.976	0.899	0.978	0.974	0.980	0.938	0.957	0.950	0.950	0.956

#### 3. Paired t test for the comparison between General VCMM and Gaussian VCMM

In the last summary part, we have recommended using General VCMM over Gaussian VCMM for clustering non-elliptical and skew elliptical data. Now, we would like to use paired t test for normalised BIC and classification rate to prove our statement with some statistical evidences and the hypotheses are formally defined below:

$$\begin{split} H_0 &: \Delta \leq 0 \text{ versus } H_1 : \Delta > 0 \\ \text{For classification rate :} \\ \Delta &= \text{General VCMM} \text{ - Gaussian VCMM} \\ \text{For normalised BIC :} \\ \Delta &= \text{Gaussian VCMM} \text{ - General VCMM} \end{split}$$

The Table 3.3.12 shows the p value of the paired t test based on independent replications for the difference in the performance measures normalised BIC, classification rate between General VCMM and Gaussian VCMM. P value close to 0 means General VCMM better than Gaussian VCMM. The value smaller than 0.05 is high-lighted with green colour. Other than data setting 3 and 4 with elliptical clusters, the p values for normalised BIC or classification rate are mostly smaller than 0.05, which means General VCMM is more preferred. This is what we recommended before. However, the p values for classification rate in data setting 8 are a bit weird. Other than Hierarchical, the paired t test with p values close to 1 suggests that there is no difference between General VCMM and Gaussian VCMM. Although we focus more on classification rate for clustering, in general, General VCMM still outperforms Gaussian VCMM in many cases and we cannot know the characteristics of each cluster easily. So, General VCMM is still recommended.

Table 3.3.12: The p-value of the one sided paired t-test comparing the mean of performance measures by using General VCMM and Gaussian VCMM. green indicates the p-value smaller than 0.05 that we can conclude the performance using General VCMM is better than Gaussian VCMM with significance level of 0.05.

Setting (Sample size)	Algorithm	Random	Kmeans	Normalised I Hierarchical	BIC Kmeans (scale)	Hierarchical (scale)	Random	Kmeans	Classification Hierarchical	<u>rate</u> Kmeans (scale)	Hierarchical (scale)
		1	Data g	generation	: VCMM	4 (Non ellip	otical)				
	ECM.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.560	0.000	0.030	0.074
	ECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.551	0.000	0.026	0.030
1 (500)	ECME.Nelder.Mead ECME.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.589 0.909	0.000	0.223 0.255	0.216
	MCECM.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.515	0.000	0.028	0.105
	MCECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.512	0.000	0.028	0.027
	ECM.Neider.Mead ECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.056	0.052
1 (1000)	ECME.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.086	0.106
1 (1000)	ECME.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.008	0.058
	MCECM.Neider.Mead MCECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.054	0.051
	ECM.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
	ECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
2(500)	ECME.Nelder.Mead ECME.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
	MCECM.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
	MCECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
	ECM.Nelder.Mead ECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.937	1.000	0.000	1.000	1.000
0 (500)	ECME.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.128	1.000	0.000	1.000	1.000
8 (000)	ECME.BFGS	0.000	0.000	0.000	0.000	0.000	0.034	1.000	0.000	1.000	1.000
	MCECM.Nelder.Mead MCECM BECS	0.000	0.000	0.000	0.000	0.000	0.980	1.000	0.000	1.000	1.000
	ECM.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.732	0.005	0.000	0.002
	ECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.752	0.004	0.000	0.003
9 (500)	ECME.Nelder.Mead ECME BECS	0.000	0.000	0.000	0.000	0.000	0.000	0.714	0.000	0.000	0.000
	MCECM.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.633	0.005	0.000	0.000
	MCECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.781	0.001	0.000	0.002
		Da	ta gener	ation : VO	CMM (M	Iultivariate	Gaussia	n)			
	ECM.Nelder.Mead	1.000	1.000	1.000	1.000	1.000	0.977	1.000	1.000	1.000	1.000
	ECM.BFGS ECME Nelder Mead	1.000	1.000	1.000	1.000	1.000	0.977	1.000	1.000	1.000	1.000
3 (500)	ECME.BFGS	0.989	1.000	1.000	1.000	1.000	0.833	1.000	1.000	1.000	1.000
	MCECM.Nelder.Mead	1.000	1.000	1.000	1.000	1.000	0.990	1.000	1.000	1.000	1.000
	MCECM.BFGS	1.000	1.000	1.000	1.000	1.000	0.991	1.000	1.000	1.000	1.000
	ECM.BFGS	1.000	1.000	1.000	1.000	1.000	0.977	1.000	1.000	1.000	1.000
3 (1000)	ECME.Nelder.Mead	0.966	1.000	1.000	1.000	1.000	0.841	1.000	1.000	1.000	1.000
0 (1000)	ECME.BFGS	0.966	1.000	1.000	1.000	1.000	0.843	1.000	1.000	1.000	1.000
	MCECM.Neider.Mead MCECM.BFGS	1.000	1.000	1.000	1.000	1.000	0.978	1.000	1.000	1.000	1.000
		D	ata gene	ration : M	lixture of	f multivaria	te skew	t			
	ECM.Nelder.Mead	1.000	0.985	0.997	0.874	0.981	1.000	0.849	0.994	1.000	1.000
	ECM.BFGS	1.000	0.986	0.998	0.910	0.991	1.000	0.555	0.944	0.984	0.994
4 (500)	ECME.Nelder.Mead	1.000	0.967	0.998	0.908	0.992	1.000	0.564	0.911	0.986	0.964
. ,	ECME.BFGS MCECM Nelder Mead	1.000	0.967	0.996	0.922	0.987	1.000	0.559	0.914	0.982	0.950
	MCECM.BFGS	1.000	0.990	0.998	0.853	0.982	1.000	0.506	0.896	0.983	0.993
	ECM.Nelder.Mead	0.997	0.000	0.385	0.001	0.004	0.002	0.000	0.000	0.000	0.000
	ECM.BFGS ECME Nelder Mead	0.991	0.000	0.338	0.000	0.002	0.003	0.000	0.000	0.000	0.000
5 (500)	ECME.BFGS	1.000	0.000	0.309	0.000	0.002	0.000	0.000	0.000	0.000	0.000
	MCECM.Nelder.Mead	0.999	0.000	0.362	0.001	0.002	0.015	0.000	0.000	0.000	0.000
	MCECM.BFGS ECM Noldor Mood	1.000	0.000	0.373	0.001	0.004	0.021	0.000	0.000	0.000	0.000
	ECM.BFGS	0.007	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
5 (1000)	ECME.Nelder.Mead	0.002	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
- ()	ECME.BFGS MCECM Noldor Mond	0.001	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
	MCECM.Neider.Mead MCECM.BFGS	0.135	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
	ECM.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.010	0.000
	ECM.BFGS ECME Nolder Mond	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.010	0.000
6 (500)	ECME.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.008	0.000
	MCECM.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.010	0.000
	MCECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.009	0.000
	ECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.003	0.000
6 (1000)	ECME.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.002	0.000
0 (1000)	ECME.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.002	0.000
	MCECM.Nelder.Mead MCECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.007	0.000
	ECM.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.119	1.000	0.000	0.003	0.000
	ECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.033	1.000	0.000	0.003	0.000
7 (500)	ECME.Nelder.Mead ECME BECS	0.000	0.000	0.000	0.000	0.000	0.065	0.533	0.000	0.002	0.000
	MCECM.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.041	1.000	0.000	0.002	0.000
	MCECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.006	1.000	0.000	0.003	0.000
	ECM.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.998	0.000	0.000	0.000
E /	ECME.Nelder.Mead	0.000	0.000	0.000	0.000	0.000	0.000	0.009	0.000	0.000	0.000
7 (1000)	ECME.BFGS	0.000	0.000	0.000	0.000	0.000	0.055	0.373	0.000	0.000	0.000
	MCECM.Nelder.Mead MCECM.BFGS	0.000	0.000	0.000	0.000	0.000	0.000	0.998 0.998	0.000	0.000	0.000

## 4. Ranking of the performance for each EM algorithm and initialization strategy

The full tables for the ranking of the performance of EM algorithms and the initialization strategies for the nine data set by General VCMM and Gaussian VCMM algorithm are shown in the Appendix B.6. For better and more convenient comparison, four subsets of the ranking table of the data setting with similar characteristics are shown in this part, where the Table 3.3.13 shows the ranking of data setting 4, 5, 6 (Multivariate skew t distribution (Two clusters without big volume difference)), the Table 3.3.14 shows the ranking of data setting 1 and 8 (Well separated with the "X" shape), the Table 3.3.15 shows the ranking of data setting 1 and 8 (Well separated without overlap (Except Gaussian and skew t distribution)) and the last Table 3.3.16 shows the ranking of data setting 7 (Two clusters with big volume difference).

## Data setting 4, 5, 6 (Multivariate skew t distribution (Two clusters without big volume difference)

The Table 3.3.13 shows Kmeans outperforms other initialization strategies with data following skew t distribution. For General VCMM, the performances of different EM algorithms with Kmeans are similarly very good and MCECM.Nelder.Mead with Kmeans reaches the highest classification rate in that 5 simulations. Also, the characteristics of data setting 4 is in between elliptical and skew elliptical. The Gaussian VCMM with ECM.Nelder.Mead and Kemans performs the best with 83.3% classification rate, which is higher than the classification rate 82.6% for General VCMM with the best method.

## Data setting 3 and 9 (Well separated with the "X" shape)

The Table 3.3.14 shows Random outperforms other initialization strategies for those clusters like a "X" shape. Recalling that data setting 3 is Gaussian data. The classification rate for Gaussian VCMM with the best method can be higher than General VCMM. We can see that for both General VCMM and Gaussian VCMM, MCECM.Nelder.Mead and MCECM.BFGS perform the best. For data setting 9 with non-Gaussian data, ECME.BFGS with Random for General VCMM perform the best.

## Data setting 1 and 8 (Well separated without overlap (Except Gaussian and skew t distribution))

The Table 3.3.15 shows the EM algorithms with Random or Hierarchical are the top 12 performance for data setting 1 and 8. In particular, EM algorithms with Hierarchical perform the best data setting 1 with 500 sample sizes and data setting 8. And EM algorithms with Random perform the best data setting 1 with 1000 sample size. Basically, we don't have a best method to cluster all data in this group. However, you can see that the difference of classification rate between different EM algorithms with Random and MCECM.BFGS with Hierarchical in data setting 1 with 1000 sample sizes are smaller than 1% only. So, we will suggest using Hierarchical to cluster data in this group. Comparing the classification rate with different EM algorithms for data setting 1, 8, ECME.BFGS / MCECM.BFGS with Hierarchical should be a good choice.

#### Data setting 7 (Two clusters with big volume difference)

The Table 3.3.16 shows the Kmeans scale and Hierarchical scale perform the best in data setting 7 with sample size 500 and 1000 respectively, which verify our previous statement that standardization first before applying General VCMM is suggested for the clusters with very different volumes. However, the performances between Kmeans scale and Hierarchical scale in data setting 7 with sample size 500 and 1000 are quite different respectively. We have no suggestion which one to use after standardization. Also, Hierarchical without standardization is the second best initialization strategy for both setting. According to the simulation result here, Hierarchical without standardization may be also a good choice.

Table 3.3.13: The top 10 performance of EM algorithms and the initialization strategies for data setting 4, 5, 6, where the algorithm with 1) higher classification rate, 2) lower normalised BIC and 3) shorter computation time is considered to be the better performance and the priority is given by (1) > (2) > (3).

	1		General VCV	M					Cauceian VC	'MM		
Setting	Banking	Algorithm	Initialization	Classification	Normalised	Computation	Banking	Algorithm	Initialization	Classification	Normalised	Computation
(Sample size)	100000		strategy	rate	BIC	time (Seconds)			strategy	rate	BIC	time (Seconds)
		Cha	racteristics : N	lultivariate	skew t dist	ribution (Tw	o cluste	rs without big volu	me difference)			
	1	MCECM.BFGS	Kmeans	0.826	10.79	42	1	ECM.Nelder.Mead	Kmeans	0.833	10.78	47
	2	MCECM.Nelder.Mead	Kmeans	0.826	10.79	44	2	MCECM.Nelder.Mead	Kmeans	0.831	10.78	42
	3	ECM.BFGS	Kmeans	0.826	10.79	45	3	MCECM.Nelder.Mead	Kmeans scale	0.829	10.77	40
	4	ECM.Nelder.Mead	Kmeans	0.826	10.79	50	4	ECM.Nelder.Mead	Kmeans scale	0.829	10.77	51
4 (500)	5	ECME.BFGS	Kmeans	0.825	10.79	46	5	ECM.BFGS	Kmeans	0.827	10.77	43
4 (500)	6	ECME.Nelder.Mead	Kmeans	0.824	10.79	55	6	MCECM.BFGS	Kmeans	0.826	10.77	41
	7	MCECM.BFGS	Kmeans scale	0.805	10.78	42	7	ECME.BFGS	Kmeans	0.826	10.77	48
	8	MCECM.Nelder.Mead	Kmeans scale	0.805	10.78	44	8	ECME.Nelder.Mead	Kmeans	0.826	10.77	60
	9	ECM.Nelder.Mead	Kmeans scale	0.805	10.78	50	9	MCECM.BFGS	Kmeans scale	0.825	10.77	41
	10	ECM.BFGS	Kmeans scale	0.804	10.79	42	10	ECM.BFGS	Kmeans scale	0.825	10.77	47
-	1	ECM.BFGS	Kmeans	0.756	12.08	86	1	ECME.Nelder.Mead	Hierarchical scale	0.523	12.13	62
	2	MCECM.Nelder.Mead	Kmeans	0.756	12.08	99	2	ECME.BFGS	Hierarchical scale	0.523	12.13	72
	3	MCECM.BFGS	Kmeans	0.755	12.08	88	3	MCECM.Nelder.Mead	Hierarchical	0.522	12.10	50
	4	ECM.Nelder.Mead	Kmeans	0.755	12.08	103	4	ECM.BFGS	Hierarchical	0.522	12.10	53
5 (500)	5	ECME.BFGS	Kmeans	0.753	12.08	100	5	ECM.Nelder.Mead	Hierarchical	0.522	12.10	54
5 (500)	6	ECME.Nelder.Mead	Kmeans	0.753	12.08	113	6	MCECM.BFGS	Hierarchical	0.522	12.10	54
	7	MCECM.Nelder.Mead	Kmeans scale	0.747	12.08	87	7	ECME.Nelder.Mead	Hierarchical	0.522	12.11	60
	8	ECM.BFGS	Kmeans scale	0.746	12.08	77	8	ECME.BFGS	Hierarchical	0.522	12.11	70
	9	MCECM.BFGS	Kmeans scale	0.746	12.08	79	9	MCECM.Nelder.Mead	Kmeans scale	0.522	12.12	51
	10	ECM.Nelder.Mead	Kmeans scale	0.746	12.08	92	10	MCECM.BFGS	Kmeans scale	0.522	12.12	51
	1	MCECM.Nelder.Mead	Kmeans	0.757	12.02	155	1	MCECM.BFGS	Kmeans	0.521	12.12	79
	2	MCECM.BFGS	Kmeans	0.756	12.01	133	2	MCECM.BFGS	Kmeans scale	0.521	12.12	79
	3	ECM.BFGS	Kmeans	0.756	12.02	138	3	MCECM.Nelder.Mead	Kmeans	0.521	12.12	83
	4	ECM.Nelder.Mead	Kmeans	0.756	12.02	159	4	ECME.Nelder.Mead	Kmeans	0.521	12.12	84
F (1000)	5	ECME.BFGS	Kmeans	0.754	12.01	143	5	ECME.Nelder.Mead	Kmeans scale	0.521	12.12	84
5 (1000)	6	ECME.Nelder.Mead	Kmeans	0.754	12.01	166	6	MCECM.Nelder.Mead	Kmeans scale	0.521	12.12	84
	7	MCECM.BFGS	Kmeans scale	0.744	12.02	118	7	ECME.BFGS	Kmeans scale	0.521	12.12	89
	8	ECM.BFGS	Kmeans scale	0.744	12.02	123	8	ECME.BFGS	Kmeans	0.521	12.12	95
	9	ECM.Nelder.Mead	Kmeans scale	0.744	12.02	138	9	MCECM.BFGS	Random	0.520	12.12	64
	10	MCECM.Nelder.Mead	Kmeans scale	0.744	12.02	139	10	MCECM.Nelder.Mead	Random	0.520	12.12	69
	1	ECM.BFGS	Kmeans	0.987	13.32	40	1	ECM.BFGS	Kmeans	0.970	14.35	21
	2	MCECM.BFGS	Kmeans	0.987	13.32	41	2	ECM.Nelder.Mead	Kmeans	0.970	14.35	22
	3	ECM.Nelder.Mead	Kmeans	0.987	13.32	49	3	ECME.BFGS	Kmeans	0.969	14.34	28
	4	MCECM.Nelder.Mead	Kmeans	0.987	13.32	50	4	MCECM.BFGS	Kmeans	0.969	14.35	21
6 (500)	5	ECME.BFGS	Kmeans	0.987	13.32	59	5	MCECM.Nelder.Mead	Kmeans	0.969	14.35	23
0 (300)	6	ECME.Nelder.Mead	Kmeans	0.987	13.32	73	6	ECME.Nelder.Mead	Kmeans	0.969	14.35	31
	7	ECME.Nelder.Mead	Hierarchical scale	0.961	13.38	119	7	MCECM.Nelder.Mead	Kmeans scale	0.927	14.36	28
	8	ECM.BFGS	Hierarchical scale	0.960	13.38	84	8	MCECM.BFGS	Kmeans scale	0.927	14.37	23
	9	MCECM.BFGS	Hierarchical scale	0.960	13.38	87	9	ECM.BFGS	Kmeans scale	0.927	14.37	25
	10	ECME.BFGS	Hierarchical scale	0.960	13.38	99	10	ECM.Nelder.Mead	Kmeans scale	0.927	14.37	28
	1	ECM.BFGS	Kmeans	0.988	13.26	65	1	MCECM.Nelder.Mead	Kmeans	0.971	14.44	30
	2	MCECM.BFGS	Kmeans	0.988	13.26	79	2	ECM.Nelder.Mead	Kmeans	0.971	14.44	32
	3	ECM.Nelder.Mead	Kmeans	0.988	13.26	92	3	ECME.Nelder.Mead	Kmeans	0.971	14.44	40
	4	MCECM.Nelder.Mead	Kmeans	0.988	13.26	94	4	ECME.BFGS	Kmeans	0.970	14.45	43
6 (1000)	5	ECME.BFGS	Kmeans	0.988	13.26	119	5	ECM.BFGS	Kmeans	0.970	14.46	30
0 (1000)	6	ECME.Nelder.Mead	Kmeans	0.988	13.26	137	6	MCECM.BFGS	Kmeans	0.970	14.46	38
	7	ECME.BFGS	Hierarchical scale	0.975	13.30	177	7	ECME.Nelder.Mead	Kmeans scale	0.931	14.46	46
	8	ECM.Nelder.Mead	Kmeans scale	0.974	13.30	112	8	ECME.BFGS	Kmeans scale	0.929	14.44	46
	9	ECME.BFGS	Kmeans scale	0.974	13.30	137	9	MCECM.Nelder.Mead	Kmeans scale	0.929	14.46	33
	10	ECM.BFGS	Hierarchical scale	0.974	13.30	140	10	ECM.Nelder.Mead	Kmeans scale	0.929	14.46	34

Table 3.3.14: The top 10 performance of EM algorithms and the initialization strategies for data setting 3 and 9, where the algorithm with 1) higher classification rate, 2) lower normalised BIC and 3) shorter computation time is considered to be the better performance and the priority is given by (1) > (2) > (3).

			General V	CMM					Gaussian V	CMM		
Setting	Ranking	Algorithm	Initialization	Classification	Normalised	Computation	Ranking	Algorithm	Initialization	Classification	Normalised	Computation
(Sample size)			strategy	rate	BIC	time (Seconds)			strategy	rate	BIC	time (Seconds)
				Characte	ristics : W	with the	e "X" shape					
	1	MCECM.Nelder.Mead	Random	0.835	12.08	104	1	MCECM.BFGS	Random	0.865	12.01	53
	2	MCECM.BFGS	Random	0.835	12.08	111	2	MCECM.Nelder.Mead	Random	0.865	12.01	55
	3	ECM.Nelder.Mead	Random	0.821	12.10	114	3	ECM.BFGS	Random	0.857	12.01	59
	4	ECM.BFGS	Random	0.821	12.10	121	4	ECM.Nelder.Mead	Random	0.857	12.01	68
2 (500)	5	ECME.BFGS	Random	0.799	12.12	121	5	MCECM.BFGS	Hierarchical	0.824	12.06	55
3 (300)	6	ECME.Nelder.Mead	Random	0.792	12.13	118	6	MCECM.Nelder.Mead	Hierarchical	0.824	12.06	57
	7	ECME.BFGS	Hierarchical	0.680	12.29	82	7	ECM.BFGS	Hierarchical	0.824	12.06	58
	8	ECME.Nelder.Mead	Hierarchical	0.678	12.29	89	8	ECM.Nelder.Mead	Hierarchical	0.824	12.06	63
	9	ECM.Nelder.Mead	Hierarchical	0.678	12.30	78	9	ECME.Nelder.Mead	Hierarchical	0.824	12.06	70
	10	MCECM.Nelder.Mead	Hierarchical	0.677	12.30	75	10	ECME.BFGS	Hierarchical	0.824	12.06	72
	1	MCECM.Nelder.Mead	Random	0.851	12.01	171	1	MCECM.BFGS	Random	0.877	11.95	76
	2	MCECM.BFGS	Random	0.851	12.01	175	2	MCECM.Nelder.Mead	Random	0.877	11.95	77
	3	ECM.Nelder.Mead	Random	0.850	12.01	191	3	ECM.Nelder.Mead	Random	0.877	11.95	87
	4	ECM.BFGS	Random	0.850	12.01	194	4	ECM.BFGS	Random	0.876	11.95	87
2 (1000)	5	ECME.Nelder.Mead	Random	0.816	12.04	185	5	MCECM.Nelder.Mead	Hierarchical	0.844	12.00	95
3 (1000)	6	ECME.BFGS	Random	0.816	12.04	192	6	MCECM.BFGS	Hierarchical	0.844	12.00	95
	7	ECME.Nelder.Mead	Hierarchical	0.714	12.17	147	7	ECM.Nelder.Mead	Hierarchical	0.844	12.00	99
	8	ECME.BFGS	Hierarchical	0.714	12.17	148	8	ECM.BFGS	Hierarchical	0.844	12.00	99
	9	MCECM.Nelder.Mead	Hierarchical	0.713	12.18	146	9	ECME.BFGS	Hierarchical	0.844	12.00	100
	10	MCECM.BFGS	Hierarchical	0.713	12.18	148	10	ECME.Nelder.Mead	Hierarchical	0.844	12.00	101
	1	ECME.BFGS	Random	0.878	14.58	119	1	MCECM.BFGS	Random	0.726	16.73	76
	2	ECM.Nelder.Mead	Random	0.877	14.57	141	2	MCECM.Nelder.Mead	Random	0.726	16.73	85
	3	MCECM.BFGS	Random	0.867	14.58	98	3	ECM.BFGS	Random	0.722	16.68	82
	4	ECM.BFGS	Random	0.860	14.60	98	4	ECM.Nelder.Mead	Random	0.722	16.68	88
0 (500)	5	MCECM.Nelder.Mead	Random	0.847	14.65	117	5	ECME.BFGS	Random	0.690	16.61	48
9 (000)	6	ECM.Nelder.Mead	Random	0.847	14.65	123	6	ECME.Nelder.Mead	Random	0.664	16.56	52
	7	MCECM.Nelder.Mead	Kmeans scale	0.757	15.46	169	7	ECM.BFGS	Hierarchical	0.645	16.83	68
	8	ECME.BFGS	Kmeans scale	0.757	15.49	137	8	MCECM.Nelder.Mead	Hierarchical	0.645	16.83	71
	9	ECM.BFGS	Kmeans scale	0.756	15.46	118	9	ECM.Nelder.Mead	Hierarchical	0.645	16.83	72
	10	ECM.Nelder.Mead	Kmeans scale	0.756	15.46	167	10	ECME.BFGS	Hierarchical	0.639	16.80	53

Table 3.3.15: The top 12 performance of EM algorithms and the initialization strategies for data setting 1 and 8, where the algorithm with 1) higher classification rate, 2) lower normalised BIC and 3) shorter computation time is considered to be the better performance and the priority is given by (1) > (2) > (3).

			General VO	CMM					Gaussian VC	CMM		
Setting	Ranking	Algorithm	Initialization	Classification	Normalised	Computation	Ranking	Algorithm	Initialization	Classification	Normalised	Computation
(Sample size)			strategy	rate	BIC	time (Seconds)			strategy	rate	BIC	time (Seconds)
-		Ch	montonisting	. Well copp	noted with	aut overlap (i	Event (	oussion and show	t distribution)			
		Clia	acteristics	. wen sepa	lated with	out overlap (	Блері С	aussian and skew	t distribution)			
	1	ECME.BFGS	Hierarchical	0.944	16.54	117	1	ECME.Nelder.Mead	Hierarchical	0.730	19.70	67
	2	MCECM.BFGS	Hierarchical	0.943	16.57	83	2	ECME.BFGS	Hierarchical	0.725	19.70	54
	3	ECM.BFGS	Hierarchical	0.942	16.57	84	3	ECM.BFGS	Hierarchical	0.724	19.74	67
	4	ECME.Nelder.Mead	Hierarchical	0.940	16.57	154	4	ECM.Nelder.Mead	Hierarchical	0.724	19.74	77
1 (500)	5	MCECM.Nelder.Mead	Hierarchical	0.937	16.59	95	5	MCECM.BFGS	Hierarchical	0.723	19.74	65
1 (500)	6	ECM.Nelder.Mead	Hierarchical	0.934	16.59	92	6	MCECM.Nelder.Mead	Hierarchical	0.723	19.74	77
	7	ECM.BFGS	Random	0.885	16.70	186	7	ECME.BFGS	Random	0.620	19.97	73
	8	MCECM.BFGS	Random	0.880	16.74	167	8	ECME.Nelder.Mead	Hierarchical scale	0.617	19.79	66
	9	ECME.Nelder.Mead	Random	0.879	16.74	269	9	ECME.BFGS	Hierarchical scale	0.615	19.81	55
	10	ECME.BFGS	Random	0.875	16.74	181	10	MCECM.BFGS	Hierarchical scale	0.613	19.82	73
	11	ECM.Nelder.Mead	Random	0.872	16.76	217	11	MCECM.Nelder.Mead	Hierarchical scale	0.613	19.82	83
	12	MCECM.Nelder.Mead	Random	0.868	16.79	210	12	ECM.BFGS	Hierarchical scale	0.612	19.82	72
	1	MCECM.BFGS	Random	0.935	16.48	285	1	ECM.BFGS	Hierarchical	0.736	19.68	112
	2	ECME.BFGS	Random	0.932	16.49	307	2	MCECM.BFGS	Hierarchical	0.736	19.68	115
	3	ECM.Nelder.Mead	Random	0.931	16.50	324	3	ECME.BFGS	Hierarchical	0.735	19.66	88
	4	ECME.Nelder.Mead	Random	0.930	16.49	370	4	ECME.Nelder.Mead	Hierarchical	0.735	19.66	96
1 (1000)	5	ECM.BFGS	Random	0.930	16.50	285	5	MCECM.Nelder.Mead	Hierarchical	0.735	19.66	126
1 (1000)	6	MCECM.BFGS	Hierarchical	0.927	16.48	112	6	ECM.Nelder.Mead	Hierarchical	0.735	19.66	131
	7	ECM.BFGS	Hierarchical	0.927	16.48	113	7	ECME.Nelder.Mead	Kmeans	0.605	19.88	124
	8	ECME.BFGS	Hierarchical	0.927	16.48	163	8	ECME.BFGS	Random	0.602	19.92	107
	9	MCECM.Nelder.Mead	Hierarchical	0.924	16.49	128	9	ECM.BFGS	Kmeans	0.601	19.88	164
	10	ECM.Nelder.Mead	Hierarchical	0.924	16.50	130	10	MCECM.Nelder.Mead	Kmeans	0.601	19.88	177
	11	ECME.Nelder.Mead	Hierarchical	0.923	16.51	183	11	ECM.Nelder.Mead	Kmeans	0.601	19.88	186
	12	MCECM.Nelder.Mead	Random	0.923	16.53	296	12	MCECM.BFGS	Kmeans	0.601	19.90	159
	1	ECM.Nelder.Mead	Hierarchical	0.991	17.26	59	1	ECM.BFGS	Random	0.907	19.92	53
	2	ECM.BFGS	Hierarchical	0.990	17.26	47	2	ECM.Nelder.Mead	Random	0.907	19.92	64
	3	MCECM.BFGS	Hierarchical	0.990	17.26	48	3	MCECM.BFGS	Random	0.906	19.92	55
	4	MCECM.Nelder.Mead	Hierarchical	0.990	17.26	55	4	MCECM.Nelder.Mead	Random	0.906	19.92	62
8 (500)	5	ECME.BFGS	Hierarchical	0.990	17.26	73	5	ECM.BFGS	Kmeans	0.905	19.92	61
8 (500)	6	ECME.Nelder.Mead	Hierarchical	0.990	17.26	87	6	MCECM.BFGS	Kmeans	0.905	19.92	62
	7	ECME.BFGS	Random	0.912	17.56	105	7	MCECM.Nelder.Mead	Kmeans	0.905	19.92	72
	8	ECME.Nelder.Mead	Random	0.900	17.60	117	8	ECM.Nelder.Mead	Kmeans	0.905	19.92	74
	9	ECM.Nelder.Mead	Random	0.862	17.67	104	9	MCECM.BFGS	Hierarchical	0.900	19.90	33
	10	ECM.BFGS	Random	0.861	17.67	86	10	MCECM.Nelder.Mead	Hierarchical	0.900	19.90	37
	11	MCECM.Nelder.Mead	Random	0.843	17.73	103	11	ECM.BFGS	Hierarchical	0.899	19.90	33
	12	MCECM.BFGS	Random	0.843	17.73	441	12	ECM.Nelder.Mead	Hierarchical	0.899	19.90	39

Table 3.3.16: The top 20 performance of EM algorithms and the initialization strategies for data setting 7, where the algorithm with 1) higher classification rate, 2) lower normalised BIC and 3) shorter computation time is considered to be the better performance and the priority is given by (1) > (2) > (3).

			General VCM	AM					Gaussian VC	CMM		
Setting	Ranking	Algorithm	Initialization	Classification	Normalised	Computation	Ranking	Algorithm	Initialization	Classification	Normalised	Computation
(Sample size)			strategy	rate	BIC	time (Seconds)			strategy	rate	BIC	time (Seconds)
				Characteris	tics : Two	clusters with	big volu	1me difference				
		DOL DDOG		0.000	10.00		-	DOLDDOG		0.054	1180	
	1	ECM.BFGS	Kmeans scale	0.980	13.92	69	1	ECM.BFGS	Hierarchical scale	0.954	14.70	14
	2	MCECM.BFGS	Kmeans scale	0.980	13.92	71	2	MCECM.BFGS	Hierarchical scale	0.954	14.70	14
	3	MCECM.Nelder.Mead	Kmeans scale	0.980	13.92	82	3	MCECM.Nelder.Mead	Hierarchical scale	0.954	14.70	16
	4	ECM.Nelder.Mead	Kmeans scale	0.980	13.92	85	4	ECM.BFGS	Hierarchical	0.954	14.70	17
	5	ECME.BFGS	Kmeans scale	0.980	13.92	94	5	MCECM.BFGS	Hierarchical	0.954	14.70	17
	6	ECME.Nelder.Mead	Kmeans scale	0.980	13.92	119	6	ECM.Nelder.Mead	Hierarchical scale	0.954	14.70	17
	7	MCECM.BFGS	Hierarchical	0.975	13.92	38	7	ECM.Nelder.Mead	Hierarchical	0.954	14.70	18
	8	ECME.BFGS	Hierarchical	0.975	13.92	57	8	MCECM.Nelder.Mead	Hierarchical	0.954	14.70	18
	9	ECME.Nelder.Mead	Hierarchical	0.975	13.92	72	9	ECME.Nelder.Mead	Hierarchical scale	0.954	14.70	22
7 (500)	10	ECM.BFGS	Hierarchical	0.974	13.92	38	10	ECME.BFGS	Hierarchical scale	0.954	14.70	22
. (000)	11	ECM.Nelder.Mead	Hierarchical	0.974	13.92	47	11	ECME.Nelder.Mead	Hierarchical	0.954	14.70	25
	12	MCECM.Nelder.Mead	Hierarchical	0.974	13.92	47	12	ECME.BFGS	Hierarchical	0.954	14.70	25
	13	MCECM.BFGS	Hierarchical scale	0.972	13.92	36	13	ECM.BFGS	Kmeans scale	0.953	14.78	25
	14	ECM.BFGS	Hierarchical scale	0.972	13.93	36	14	MCECM.BFGS	Kmeans scale	0.953	14.78	25
	15	MCECM.Nelder.Mead	Hierarchical scale	0.972	13.93	45	15	MCECM.Nelder.Mead	Kmeans scale	0.953	14.78	29
	16	ECME.BFGS	Hierarchical scale	0.972	13.93	52	16	ECM.Nelder.Mead	Kmeans scale	0.953	14.78	37
	17	ECME.Nelder.Mead	Hierarchical scale	0.972	13.93	105	17	ECME.Nelder.Mead	Kmeans scale	0.951	14.74	33
	18	ECM.Nelder.Mead	Hierarchical scale	0.971	13.93	45	18	ECME.BFGS	Kmeans scale	0.951	14.74	36
	19	MCECM.BFGS	Random	0.958	14.04	224	19	ECM.BFGS	Kmeans	0.947	14.80	52
	20	ECM.BFGS	Random	0.949	14.07	214	20	ECM.Nelder.Mead	Kmeans	0.947	14.80	56
	1	ECM.BFGS	Hierarchical scale	0.980	13.79	61	1	ECME.Nelder.Mead	Kmeans	0.958	14.72	66
	2	MCECM.BFGS	Hierarchical scale	0.980	13.79	71	2	ECM.BFGS	Kmeans	0.958	14.72	92
	3	ECME.Nelder.Mead	Hierarchical scale	0.980	13.79	106	3	ECME.BFGS	Kmeans	0.958	14.72	645
	4	MCECM.Nelder.Mead	Hierarchical scale	0.980	13.79	278	4	MCECM.Nelder.Mead	Kmeans	0.958	14.72	970
	5	ECME.BFGS	Hierarchical scale	0.980	13.79	329	5	ECM.Nelder.Mead	Kmeans	0.958	14.72	1004
	6	ECM.Nelder.Mead	Hierarchical	0.980	13.79	344	6	ECM.BFGS	Hierarchical scale	0.957	14.70	24
	7	ECM.Nelder.Mead	Hierarchical scale	0.980	13.79	349	7	ECM.Nelder.Mead	Hierarchical scale	0.957	14.70	25
	8	ECME.Nelder.Mead	Hierarchical	0.980	13.79	565	8	MCECM.BFGS	Kmeans	0.957	14.72	147
	9	MCECM.Nelder.Mead	Hierarchical	0.979	13.79	362	9	MCECM.Nelder.Mead	Hierarchical scale	0.956	14.70	24
- (	10	ECM.BFGS	Hierarchical	0.978	13.80	72	10	ECME.BFGS	Hierarchical scale	0.956	14.70	27
7 (1000)	11	MCECM.BFGS	Hierarchical	0.978	13.80	73	11	ECME.Nelder.Mead	Hierarchical scale	0.956	14.70	28
	12	ECME.BFGS	Hierarchical	0.978	13.80	448	12	MCECM.BFGS	Hierarchical scale	0.956	14.70	88
	13	MCECM.BFGS	Random	0.976	13.82	343	13	ECM.Nelder.Mead	Kmeans scale	0.952	14.86	44
	14	ECM.BFGS	Kmeans scale	0.974	13.82	133	14	ECME.BFGS	Hierarchical	0.951	14.78	372
	15	MCECM BEGS	Kmeans scale	0.974	13.82	139	15	ECM.BEGS	Kmeans scale	0.951	14.82	42
	16	ECME. Nelder, Mead	Kmeans scale	0.974	13.82	176	16	MCECM Nelder Mead	Kmeans scale	0.951	14.85	44
	17	MCECM Nelder Mead	Kmeans scale	0.974	13.82	540	17	ECME BEGS	Kmeans scale	0.950	14.76	45
	18	ECM Nelder Mead	Kmeans scale	0.974	13.82	545	18	ECME Nelder Mead	Kmeans scale	0.950	14.76	47
	19	ECME BEGS	Kmeans scale	0.974	13.82	639	19	ECM.BFGS	Hierarchical	0.950	14.78	31
	20	ECM BECS	Random	0.974	13.82	315	20	ECME Nelder Mood	Hierarchicel	0.950	14.78	35
	20	EUM.DFG5	nandom	0.974	19.09	919	20	EX.ME.Iveider.ivlead	merarcincal	0.900	14.10	99

# 5. The summary of the best initialization strategy and EM algorithm for clustering data setting 1 - 9.

According to the results we have from the VCMM simulation, we know that some initialization strategies perform better for the data with some particular characteristics. The best initialization strategy and initialization strategy is summarised as a flow chart in the Figure 3.3.19. The red rectangular box with curved edge is the start of data clustering. Then, you will follow the arrow with a suitable condition to go to other boxes. The orange rectangular boxes refer to the characteristics of the data and the green diamond boxes refer to best initialization strategies and the EM algorithm with the best performance which are the end of the flow chart.



Figure 3.3.19: The summary of the best initialization strategy and EM algorithm for clustering data setting 1 - 9.

## 6. Selection of the EM algorithm and initialization strategy for clustering data without characteristics information

We have mentioned that some initialization strategies perform better for the data with some particular characteristics. However, we cannot know the shape, overlap and volume of each cluster clearly if the clusters are not marked. Therefore, in general, we cannot choose the best initialization strategy for clustering real data set with unknown data characteristics.

We can calculate the BIC value for each initialization strategy and choose the one with lowest BIC and that one probably gives us the best classification rate, but it can probably leads to a very high computation time if the the data is high-dimensional. In order to save some computation time, we would suggest two better initialization strategies according to our simulation result and choose the one with lower BIC. Also, we know that random outperforms other initialization strategies for those clusters like a "X" shape and "X" shape still can be observed easily for three dimensional data.

Based on these two selection's principle, we have come up with two flow charts in the Figure 3.3.20 and 3.3.21 to select the suitable initialization strategy for general VCMM and Gaussian VCMM algorithm respectively. The flow chart is basically similar to the one above. The only difference is that the green rectangular boxes refer to the final decision of the initialization strategy which are the end of the flow chart.

In the Figure 3.3.20 and 3.3.21, we know that if the data has the "X" shape, we recommend random with ECM.Nelder.Mead and MCECM.BFGS for general VCMM and Gaussian VCMM respectively. The result is from the Table 3.3.14 and the EM algorithms topping or almost topping in the list are selected. If the data doesn't look like "X" shape, for general VCMM, we have suggested Hierarchical or Kmeans (scale) with ECM.BFGS / MCECM.BFGS. The selection of the EM algorithms is because of shorter required computation time that we mentioned before. In the Table 3.3.11, except data setting 3,9 with "X" shape, we know that Hierarchical performs quite well in data setting 1,2,4,6-8 and Kmeans (scale) perform quite well in data setting 4-7. More importantly, we know that the clusters in data setting 7 have larger volume or scale difference and real data set usually have this characteristics. So, Hierarchical and Kmeans (scale) is selected. For Gaussian VCMM, EM algorithms with BFGS are selected as we mentioned that it take less time than Nelder.Mead. Also, except data setting 3,9 with "X" shape, Kmeans performs well in data setting 1,2,4-8 and Hierarchical performs quite well in data setting 1,4,5,7 and they don't have worst performance. So, Kmeans and Hierarchical are selected.



Figure 3.3.20: The selection of the initialization strategy for general VCMM algorithm.



Figure 3.3.21: The selection of the initialization strategy for Gaussian VCMM algorithm.

Notation	Name	AIS	BCW
n	Sample size	202	569
d	Dimension of each sample	5	4
	(Variables used)	(LBM, Wt, BMI, EBC and PBF)	(PSE, ES, EC and ECP)
K	Total number of clusters	2	2
$\pi_1$	Mixing proportion of cluster 1	0.495	0.594
	(Name of cluster 1)	(Female)	(Benign)
$\pi_2$	Mixing proportion of cluster 2	0.505	0.406
	(Name of cluster 2)	(Male)	(Malignant)

Table 4.0.1: The overview of the real data sets we used for assessment.

## 4 Real data sets

In Section 3 simulation study, we have generated different data sets to assess the performance of Gaussian mixture model (GMM) and vine copula mixture model (VCMM) algorithm with different EM algorithms and initialization strategies. However, the data for simulation is just 3 dimensions and generated by GMM, VCMM and mixture of multivariate skew t. The real data is probably more complicated, for example, higher dimension, more clusters, following special distributions, more outliers and so on. Therefore, in this section, we just follow the real data sets and the selected variables used in Sahin and Czado [2021] to test the performance of VCMM algorithm : 1. Australian Institute of Sport data set, 2. Breast Cancer Wisconsin Diagnostic data set. The Table 4.0.1 shows you the general overview of two real data sets and more details will be given in the Section 4.1.

## 4.1 Introduction to the real data set

#### Australian Institute of Sport (AIS)

Australian Institute of Sport (AIS) data set contains 13 information of 202 athletes, including the sex of the athletes, the sport of the athletes and 11 different body and blood measurements. There are in total 102 male and 100 female athletes in the data set. We would like to use VCMM algorithm to identify the gender of the athletes, based on 5 selected information of the athletes, which are lean body mass (LBM), weight (Wt), body mass index (BMI), white blood cell count (WBC), and percentage of body fat (PBF). The data is available in the R package DAAG and the data is called **ais**.

The pairwise scatter plots of the subset of AIS data are shown in the Figure 4.1.1. The orange and green dots represent the observation for female and male athletes respectively. From the panels of the Figure 4.1.1 (a), you can see that the two clusters are separated the most between PBF and LBM, which means there are a significant different between male and female for those measurements. In the diagonal of the Figure 4.1.1 (a), the marginal distributions for each cluster are shown. For all variables, the values near the center (median) appear more frequently than values at the tails. Also, the marginal distributions between male and female, it shows a positive skew that tail is on the right. In the Figure 4.1.1 (b), (c) and (d), we can see the dependency between each variable. From the normalized contour plots, we can know that the dependency between the same pair of variables is usually quite similar in females and males and the associated Kendall's tau too, except there are more differences between WBC and PBF. Also, we can see the non-Gaussian dependence patterns in many pair of variables, and asymmetric tail dependence between LBM and Wt.

#### Breast Cancer Wisconsin Diagnostic (BCW)

Breast Cancer Wisconsin Diagnostic (BCW) is a data set from UCI Machine Learning Repository containing 30 features of cell nuclei from breast mass and the classification of breast cancer: malignant (cancer) or benign (not cancer). Ten features, such as size, shape and regularity, are computed for each cell nucleus from a breast mass. The mean, standard error, and extreme value (mean of the three largest values) of each of 10 nuclear parameters is reported for a total of 30 features. There are in total 569 patients, including 212 malignant and 357 benign. With the selection of 4 features, perimeter standard error (PSE), extreme smoothness (ES), extreme concavity (EC), and extreme concave points (ECP), we would like to use VCMM algorithm to classify the status of the breast mass. The data is available in the R package dslabs and the data is called brca.

The pairwise scatter plots of the subset of the BCW data are shown in the Figure 4.1.2. The orange and green dots represent the observation for benign and malignant respectively. Same as the AIS data, for all variables, the values near the center (median) appear more frequently than values at the tails. Apart from ES, the distributions are skewed at a particular level. In particular, EC with malignant and PSE show a positive skew that the tail is on the right. About the dependency, it is quite similar between the same pair of variables in benign and malignant, but some of them the strength are quite different, for example, ES and EC. Also, some of the pairs show the asymmetric tail dependency, for example, the pair of EC and ECP shows the strong asymmetric tail dependency.



(c) Pairs plot for females (Copula data)

(d) Pairs plot for males (Copula data)

Figure 4.1.1: The pairwise scatter plot of the subset of AIS data (top left) with orange dots for observations of females and green dots for males. Pairs plots of all data (top right), females (bottom left) and males (bottom right), where lower: pairs plots of copula data with the values of Kendall's tau, diagonal: histogram of copula margins, upper: normalized contour plots.



(c) Pairs plot for benign (Copula data)

(d) Pairs plot for malignant (Copula data)

Figure 4.1.2: The pairwise scatter plot of the subset of BCW data (top left) with orange dots for observations of benign and green dots for malignant. Pairs plots of all data (top right), benign (bottom left) and malignant (bottom right), where lower: pairs plots of copula data with the values of Kendall's tau, diagonal: histogram of copula margins, upper: normalized contour plots.

## 4.2 Introduction to the modelling methods for clustering

## 1. Modelling methods used for clustering

In this section, four modelling methods will be used for clustering and the overview is shown in the Table 4.2.1. GMM is the Algorithm (1) used in the Section 3.1 and 3.2. General VCMM and Gaussian VCMM are the algorithms used in the Section 3.3. The new modelling method called General VCMM (Copula with single parameter) is used here and the only difference is that just the copula families with one parameter are used for modelling, including Gaussian, Student t, Frank, Clayton, Gumbel and Joe copula and their rotated copulas. And all other factors keep same as General VCMM, for example, marginal distributions used for modelling, CM steps order for EM algorithms.

Table 4.2.1: The overview of the modelling method, the associated marginal distribution and copula families used for clustering. *For marginal distribution, "All" refers to 1. Gaussian distribution, 2. Log-normal distribution, 3. Exponential distribution, 4. Log-logistic distribution, 5. Logistic distribution, 6. Gamma distribution and 7. Student's t distribution with degrees of freedom 3.3. For copula family, "All" refers to Gaussian copula, Student's t copula, Gumbel copula, Clayton copula, Frank copula, Joe copula, BB1 copula, BB6 copula, BB8 copula and their rotated copulas with 90°, 180°, 270° degrees.

Modelling method	Marginal distribution	Copula family	Pair copula construction
GMM	Gaussian	-	-
General VCMM	All*	All*	Determined by the VCMM algorithm
General VCMM	۸ 11*	Copula with single parameter	Determined by the VCMM algorithm
(Copula with single parameter)	All	and Student's t copula	Determined by the VCMM algorithm
Gaussian VCMM	Gaussian	Gaussian	Determined by the VCMM algorithm

# 2. The combination of the modelling methods, EM algorithms and initialisation strategies used for clustering

For each data set, we have in total 110 ways for clustering, including 4 VCMM algorithms, 1 GMM algorithm, 5 initialisation stragties and 6 EM algorithms for VCMM and 4 EM algorithms for GMM. The details of each item are shown in the Table 4.2.2.

Table 4.2.2: The combination of the modelling method, EM algorithm and initialisation strategy used for each real data set clustering.

Modelling method	EM algorithm	Initialisation	Number of
		strategy	combinations
	1. ECM.Nelder.Mead	1. Random	
1. General VCMM	2. ECM.BFGS	2. Kmeans	
2. General VCMM	3. ECME.Nelder.Mead	3. Kmeans scale	00
(Copula with single parameter)	4. ECME.BFGS	4. Hierarchical	90
3. Gaussian VCMM	5. MCECM.Nelder.Mead	5. Hierarchical scale	
	6. MCECM.BFGS		
	1. ECM	1. Random	
	2. ECME.Nelder.Mead	2. Kmeans	
1. GMM	3. ECME.ECME	3. Kmeans scale	20
	4. MCECM	4. Hierarchical	
		5. Hierarchical scale	
		Total :	110

## 4.3 Result and performance of clustering

### 1. Comparison of the result with different methods for clustering

In the Table 4.3.1, it shows the performance of clustering AIS and BCW data, that the mean for 50 replications of the performance measure with different modelling methods, EM algorithms and initialisation strategies. Also, the pair plots and associated classification rates after initialising and final clustering are shown in Appendix C.3 and C.4. From the clustering results for the **AIS data set**, we observe the following:

- For a particular initialization strategy, the normalised BIC and classification rate perform usually similar within different EM algorithms, but there is a great difference in computation time.
- Mostly, the EM algorithms with Neldar-Mead is taking more time than with BFGS.
- For General VCMM and General VCMM (Copula with single parameter), the performance varies quite much with different initialization strategy.
- For each of the initialization strategy, General VCMM (Copula with single parameter) performs mostly better than General VCMM.
- Gaussian VCMM performs better than GMM.
- For each modelling method other than GMM, Hierarchical (scale) usually leads to the best classification rate.
- Apart from Gaussian VCMM, the classification rate after final clustering is always higher or very similar to the classification rate after initialising clustering for a particular initialization strategy.

Also, from the clustering results of **BCW data set**, we can observe the following:

- For a particular initialization strategy, the normalised BIC and classification rate perform usually similar within different EM algorithms, but there is a great difference in computation time.
- Mostly, the EM algorithms with Neldar-Mead is taking more time than with BFGS.
- Random requires longer computation time.
- For General VCMM and General VCMM (Copula with single parameter), the performance varies quite much with different initialization strategy.
- In terms of normalised BIC, General VCMM and General VCMM (Copula with single parameter) performs always better than Gaussian VCMM.
- For General VCMM and General VCMM (Copula with single parameter), Hierarchical (scale) always leads to the best classification rate.
- Apart from Gaussian VCMM, the classification rate after final clustering is always higher or very similar to the classification rate after initialising clustering for a particular initialization strategy.

Combining the observations for AIS and BCW data set above, we can conclude the following:

- Although different EM algorithms mostly don't affect the performance for clustering significantly, there is a significant effect on the running time needed.
- The EM algorithms with Neldar-Mead is taking more time than with BFGS.
- The starting value / initialization strategy for the EM algorithm is the key factor of the performance of the clustering.
- The general VCMM improves the clustering performance after initialization strategy.

- The more complex of the model, it doesn't have to lead to a better result / better fit for clustering.
- For clustering of non-Gaussian data, more flexible method like General VCMM with a suitable initialization strategy is more recommended than Gaussian mixture model.
- For real data set, Hierarchical (scale) is suggested as an initialization strategy.

Table 4.3.1: The mean for 50 replications of the performance measure for each EM algorithm and its extension with different initialization strategies for data set of AIS and BCW, where the green and red colour represent the best and the worst initialization strategies for each EM algorithm respectively and "-" represents not available as the conditions (2.5.5) or (2.5.6) is fulfilled for GMM algorithm.

	AIS											
				General VCM	ſM					Gaussian VC	MM	
Performance	Algorithm	Random	Kmeans	Hierarchical	Kmeans	Hierarchical	Algorithm	Random	Kmeans	Hierarchical	Kmeans	Hierarchical
measures					(scale)	(scale)					(scale)	(scale)
	ECM.Nelder.Mead	402	83	219	393	640	ECM.Nelder.Mead	352	550	227	557	256
	ECM.BFGS	386	53	149	219	303	ECM.BFGS	268	392	155	362	164
Commentation time	ECME.Nelder.Mead	478	218	464	683	914	ECME.Nelder.Mead	787	882	378	950	452
Computation time	ECME.BFGS	502	120	203	359	527	ECME.BFGS	524	632	265	660	356
	MCECM.Nelder.Mead	345	81	232	413	488	MCECM.Nelder.Mead	371	549	210	582	256
	MCECM.BFGS	413	42	149	236	277	MCECM.BFGS	247	348	159	367	151
	ECM.Nelder.Mead	25.60	25.87	24.21	24.70	23.92	ECM.Nelder.Mead	23.06	23.57	23.57	23.57	23.57
	ECM.BFGS	25.81	25.87	24.21	24.70	23.92	ECM.BFGS	23.06	23.57	23.57	23.57	23.57
Normalized BIC	ECME.Nelder.Mead	25.76	26.00	24.21	24.01	23.92	ECME.Nelder.Mead	23.06	23.57	23.57	23.57	23.57
Normanseu Dio	ECME.BFGS	25.86	26.00	24.21	23.52	23.92	ECME.BFGS	23.06	23.57	23.57	23.57	23.57
	MCECM.Nelder.Mead	25.82	25.87	24.21	24.70	23.92	MCECM.Nelder.Mead	23.06	23.57	23.57	23.57	23.57
	MCECM.BFGS	25.81	25.87	24.21	24.70	23.92	MCECM.BFGS	23.06	23.57	23.57	23.57	23.57
	ECM.Nelder.Mead	0.580	0.753	0.914	0.790	0.924	ECM.Nelder.Mead	0.658	0.914	0.905	0.905	0.914
	ECM.BFGS	0.576	0.753	0.914	0.790	0.924	ECM.BFGS	0.658	0.914	0.905	0.905	0.914
Classification rate	ECME.Nelder.Mead	0.572	0.775	0.914	0.905	0.924	ECME.Nelder.Mead	0.658	0.914	0.905	0.905	0.914
Classification rate	ECME.BFGS	0.602	0.775	0.914	0.914	0.924	ECME.BFGS	0.658	0.914	0.905	0.905	0.914
	MCECM.Nelder.Mead	0.602	0.753	0.914	0.790	0.924	MCECM.Nelder.Mead	0.658	0.914	0.905	0.905	0.914
	MCECM.BFGS	0.576	0.753	0.914	0.790	0.924	MCECM.BFGS	0.658	0.914	0.905	0.905	0.914
		Gen	eral VCM	M (Copula with	n single par	ameter)				<u>GMM</u>		
Performance	Algorithm	Random	Kmeans	Hierarchical	Kmeans	Hierarchical	Algorithm	Random	Kmeans	Hierarchical	Kmeans	Hierarchical
measures					(scale)	(scale)					(scale)	(scale)
	ECM.Nelder.Mead	372	104	222	383	538	ECM	0.331	0.034	0.100	0.076	0.263
	ECM.BFGS	599	56	169	213	295	MCECM	0.134	0.047	0.464	0.091	0.268
Computation time	ECME.Nelder.Mead	383	254	452	648	1003	ECME.Nelder.Mead	-	103	263	109	77
1	ECME.BFGS	699	106	301	321	496	ECME.BFGS	-	-	-	-	-
	MCECM.Nelder.Mead	346	78	244	395	593						
	MCECM.BFGS	486	51	154	229	272			_		_	
	ECM.Nelder.Mead	25.75	25.85	23.83	25.03	23.85	ECM	23.56	24.28	23.56	24.28	23.81
	ECM.BFGS	26.44	25.85	23.86	25.03	23.85	MCECM	23.56	24.28	23.56	24.28	23.81
Normalised BIC	ECME.Nelder.Mead	25.70	25.79	23.86	24.13	23.85	ECME.Nelder.Mead	-	24.33	24.14	24.34	24.24
	ECME.BFGS	26.46	25.79	23.86	23.65	23.85	ECME.BFGS	-	-	-	-	-
	MCECM.Nelder.Mead	25.75	25.85	23.86	25.03	23.85						
	MCECM.BFGS	26.46	25.85	23.83	25.03	23.85						
						() ()() (	L ECM	0.652	0.767	0.652	0.760	0.712
	ECM.Nelder.Mead	0.569	0.753	0.924	0.821	0.924	Lom	0.050	0.101	0.002	0.000	0.540
	ECM.Nelder.Mead ECM.BFGS	0.569	0.753	0.924 0.924	0.821	0.924	MCECM	0.652	0.767	0.652	0.767	0.712
Classification rate	ECM.Nelder.Mead ECM.BFGS ECME.Nelder.Mead	0.569 0.616 0.589	0.753 0.753 0.767	0.924 0.924 0.924	0.821 0.821 0.897	0.924 0.924 0.924	MCECM ECME.Nelder.Mead	0.652	0.767	0.652 0.870	0.767 0.699	0.712 0.862
Classification rate	ECM.Nelder.Mead ECM.BFGS ECME.Nelder.Mead ECME.BFGS	0.569 0.616 0.589 0.606	0.753 0.753 0.767 0.767	0.924 0.924 0.924 0.924	0.821 0.821 0.897 0.914	0.924 0.924 0.924 0.924	MCECM ECME.Nelder.Mead ECME.BFGS	0.652	0.767 0.753	0.652	0.767 0.699	0.712
Classification rate	ECM.Nelder.Mead ECM.BFGS ECME.Nelder.Mead ECME.BFGS MCECM.Nelder.Mead	0.569 0.616 0.589 0.606 0.569	0.753 0.753 0.767 0.767 0.753	0.924 0.924 0.924 0.924 0.924	0.821 0.821 0.897 0.914 0.821	0.924 0.924 0.924 0.924 0.924	MCECM ECME.Nelder.Mead ECME.BFGS	0.652	0.767	0.652	0.767 0.699	0.712
Classification rate	ECM.Nelder.Mead ECM.BFGS ECME.Nelder.Mead ECME.BFGS MCECM.Nelder.Mead MCECM.BFGS	0.569 0.616 0.589 0.606 0.569 0.606	0.753 0.753 0.767 0.767 0.753 0.753	0.924 0.924 0.924 0.924 0.924 0.924	0.821 0.821 0.897 0.914 0.821 0.821	0.924 0.924 0.924 0.924 0.924 0.924	MCECM ECME.Nelder.Mead ECME.BFGS	0.652 -	0.767	0.652	0.767 0.699	0.712 0.862

				General VCM	M					Gaussian VCM	IM	
Performance	Algorithm	Random	Kmeans	Hierarchical	Kmeans	Hierarchical	Algorithm	Random	Kmeans	Hierarchical	Kmeans	Hierarchical
measures	0				(scale)	(scale)	0				(scale)	(scale)
	ECM.Nelder.Mead	255	111	384	55	112	ECM.Nelder.Mead	327	124	267	102	101
	ECM.BFGS	420	66	232	61	74	ECM.BFGS	286	109	237	85	86
	ECME.Nelder.Mead	532	304	260	106	165	ECME.Nelder.Mead	355	143	307	153	148
Computation time	ECME.BFGS	302	125	155	73	108	ECME.BFGS	326	134	294	122	133
	MCECM.Nelder.Mead	635	153	367	55	111	MCECM.Nelder.Mead	336	118	266	97	108
	MCECM.BFGS	349	66	249	52	91	MCECM.BFGS	296	104	246	84	92
	ECM.Nelder.Mead	-6.98	-7.07	-7.16	-7.21	-7.24	ECM.Nelder.Mead	-6.61	-6.26	-6.61	-6.60	-6.61
	ECM.BFGS	-6.90	-7.07	-7.11	-7.21	-7.24	ECM.BFGS	-6.61	-6.26	-6.61	-6.60	-6.61
	ECME.Nelder.Mead	-7.02	-7.10	-7.15	-7.21	-7.24	ECME.Nelder.Mead	-6.61	-6.26	-6.61	-6.60	-6.61
Normalised BIC	ECME.BFGS	-6.98	-7.09	-7.15	-7.21	-7.24	ECME.BFGS	-6.61	-6.26	-6.61	-6.60	-6.61
	MCECM.Nelder.Mead	-6.87	-7.07	-7.16	-7.21	-7.24	MCECM.Nelder.Mead	-6.61	-6.26	-6.61	-6.60	-6.61
	MCECM.BFGS	-6.87	-7.07	-7.15	-7.21	-7.24	MCECM.BFGS	-6.61	-6.26	-6.61	-6.60	-6.61
	ECM.Nelder.Mead	0.499	0.621	0.814	0.851	0.869	ECM.Nelder.Mead	0.784	0.685	0.776	0.797	0.784
	ECM.BFGS	0.499	0.621	0.814	0.851	0.869	ECM.BFGS	0.784	0.685	0.776	0.797	0.784
	ECME.Nelder.Mead	0.504	0.636	0.822	0.851	0.869	ECME.Nelder.Mead	0.784	0.685	0.776	0.797	0.784
Classification rate	ECME.BFGS	0.499	0.636	0.822	0.851	0.869	ECME.BFGS	0.784	0.685	0.779	0.797	0.784
	MCECM.Nelder.Mead	0.500	0.621	0.814	0.851	0.869	MCECM.Nelder.Mead	0.784	0.685	0.771	0.797	0.784
	MCECM.BFGS	0.499	0.621	0.822	0.851	0.869	MCECM.BFGS	0.784	0.685	0.776	0.797	0.784
		Gen	eral VCM	A (Copula with	single par	ameter)				GMM		
Performance	Algorithm	Random	Kmeans	Hierarchical	Kmeans	Hierarchical	Algorithm	Random	Kmeans	Hierarchical	Kmeans	Hierarchical
measures	_				(scale)	(scale)	-				(scale)	(scale)
	ECM.Nelder.Mead	604	131	241	218	162	ECM	0.080	0.088	0.073	0.095	0.039
	ECM.BFGS	428	67	138	147	135	MCECM	0.104	0.098	0.099	0.125	0.060
<i>a</i>	ECME.Nelder.Mead	267	267	183	281	214	ECME.Nelder.Mead	-	-	-	-	-
Computation time	ECME.BFGS	545	117	98	199	143	ECME.BFGS	-	-	-	-	-
	MCECM.Nelder.Mead	517	141	300	154	147						
	MCECM.BFGS	389	73	159	138	128						
	ECM.Nelder.Mead	-7.04	-7.07	-7.09	-7.10	-7.08	ECM	-6.10	-6.10	-6.44	-6.44	-6.44
	ECM.BFGS	-6.98	-7.08	-7.09	-7.12	-7.14	MCECM	-6.10	-6.10	-6.44	-6.44	-6.44
N P IDIO	ECME.Nelder.Mead	-7.00	-7.10	-7.10	-7.11	-7.08	ECME.Nelder.Mead	-	-	-	-	-
Normalised BIC	ECME.BFGS	-7.06	-7.08	-7.10	-7.12	-7.12	ECME.BFGS	-	-	-	-	-
	MCECM.Nelder.Mead	-7.04	-7.07	-7.09	-7.14	-7.08						
	MCECM.BFGS	-7.00	-7.08	-7.09	-7.12	-7.11						
	ECM.Nelder.Mead	0.510	0.618	0.701	0.837	0.878	ECM	0.681	0.681	0.781	0.784	0.784
	ECM.BFGS	0.511	0.618	0.699	0.828	0.869	MCECM	0.681	0.681	0.781	0.784	0.784
C1 10 11	ECME.Nelder.Mead	0.507	0.630	0.748	0.828	0.878	ECME.Nelder.Mead	-	-	-		-
Classification rate	ECME.BFGS	0.508	0.628	0.748	0.828	0.866	ECME.BFGS	-	-	-	-	-
	MCECM.Nelder.Mead	0.509	0.618	0.703	0.839	0.875						
		0.844	0.010	0.001	0.000	0.000						

Table 4.3.2: The best estimated vine copula model for the data set AIS and BCW with the highest classification rate and smallest normalised BIC. The variable encoding is given as follows: (AIS) 1: LBM, 2: Wt, 3: BMI, 4: WBC, 5: PBF and (BCW) 1: PSE, 2: ES, 3: EC and 4: ECP. For marginal distributions (Left column), the estimated marginal distributions and parameters for each cluster are shown. For vine tree structure (Right column), the first and second tree level of the estimated vine copula models are shown here. The number 1,5 represents the edge of the tree level, letter N is the abbreviation of the copula and the true parameter value and corresponding Kendall's  $\tau$  of the pair copula are given inside the parenthesis (parameter(s)/Kendall's  $\tau$ ) near the letter. The meaning of the abbreviation for marginal distribution and copula families is shown in the appendix B.1 and B.2.

	Marginal distrib	utions	Vine tree structure							
17	Cluster 1	Classie 0		Cl			Cluster 2			
Variable Cluster I Cluster		Cluster 2	Tree 1 Tree 2			Tree 1		Tree 2		
				AIS						
		General	VCM	M (Copula with sing	gle para	meter) (Hierarchical)				
		Cl	assifica	ation rate : 92.4%	Normali	sed BIC : 23.83				
1	llogis(12.42, 55.34)	lnorm(4.3, 0.13)	2,5	C(1.53/0.43)	1,5;2	R90G(-3.38/-0.7)	2,1	SG(9.87/0.9)	3,1;2	R90G(-1.32/-0.24)
2	$\mathcal{N}(68.59, 12.29)$	lnorm(4.4, 0.14)	2,1	T(0.94, 6.26/0.78)	3,1;2	N(-0.25/-0.16)	3,2	N(0.83/0.62)	5,2;3	C(0.2/0.09)
3	lnorm(3.09, 0.12)	llogis(18.22, 23.46)	3,2	N(0.87/0.67)	4,2;3	R270C(0/0)	5,3	T(0.64, 11.61/0.45)	4,3;5	R270C(-0.17/-0.08)
4	lnorm(1.93, 0.24)	lnorm(1.93, 0.26)	4,3	F(1.56/0.17)			5,4	F(2.53/0.26)		
5	$\Gamma(11.04, 0.62)$	snorm(8.81, 2.28, 4.3)								
BCW										
General VCMM (Copula with single parameter) (Hierarchical scale)										
		Cl	assific	ation rate : 87.8%	Normali	sed BIC : -7.08	-			
1	snorm(2.01, 0.74, 2.4)	sstd(4.45, 2.5, 4.36, 2.79)	4,1	F(0.59/0.07)	2,1;4	R270C(-0.19/-0.08)	2,1	N(-0.3/-0.19)	3,1;2	SC(0.13/0.06)
2	$\Gamma(39.12, 313.49)$	llogis(12.5, 0.14)	4,2	SG(1.27/0.22)	3,2;4	R270G(-1.05/-0.04)	3,2	SG(1.44/0.31)	4,2;3	SJ(1.11/0.06)
3	snorm(0.16, 0.12, 190.82)	sstd(0.48, 0.19, 7.18, 1.94)	4,3	T(0.77, 30/0.56)			4,3	SG(1.82/0.45)		
4	$\mathcal{N}(0.07, 0.03)$	sstd(0.19, 0.04, 9.8, 1.41)								

## 2. The best estimated vine copula model for VCMM algorithm

The best estimated vine copula model with the highest classification rate and smallest normalised BIC for AIS and BCW is shown in the Table 4.3.2. For AIS data set, General VCMM (Copula with single parameter) with Hierarchical performs the best with 92.4% classification rate. As the same as expected before, the variable WBC for both clusters is fitted by the same distribution, which is log-normal distribution with the same mean and very similar standard deviation. For PBE female data with a positive skew, it is fitted by skew normal distribution. About the estimated vine tree structure pairs with large Kendall's tau, the pair between LBM and Wt for male is fitted by survival Gumbel with asymmetric tail. However, the female data is finally fitted by symmetric t copula, where the asymmetry level of tail for the female data is not as high as male. For the pair between Wt and BMI, the female and male data are both fitted by Gaussian copula.

Also, for BCW data set, General VCMM (Copula with single parameter) with Hierarchical scale performs the best with 87.8% classification rate. Due to the skewness of variables PSE and EC, the estimated marginal densities for PSE and EC are finally fitted by skew normal distribution and skew Student's t distribution respectively. About the strong dependency with asymmetric tail between EC and ECP, female data is fitted by survival Gumbel with asymmetric tail.

For other VCMM models and initialisation strategies, the estimated vine copula models are shown in the Appendix C.2.

## 3. Vuong test for the comparisons between VCMM model and Gaussian model

In the last part, we have concluded that for clustering non-Gaussian data, it is more recommended to use General VCMM. We still would like to determine whether there is enough statistical evidence to support our suggestion. Therefore, Voung test with BIC correction [Vuong, 1989] [Desmarais and Harden, 2013] is used

Table 4.3.3: The z statistics of the the Vuong test with BIC correction, where the colour yellow, orange, green, blue represent  $z \leq -1.65, -1.65 < z \leq 0, 0 < z \leq 1.65$  and 1.65 < z respectively. (-1.65 / 1.65 is the z statistics at the significance level of 0.05.) Also, "-" represents not available as the conditions (2.5.5) or (2.5.6) is fulfilled for GMM algorithm.

		In	itialization stra	ategies		Initialization strategies						
Algorithm	Random	Kmeans	Hierarchical	Kmeans	Hierarchical	Random	Kmeans	Hierarchical	Kmeans	Hierarchical		
				(scale)	(scale)			(scale)	(scale)			
			AIS			BCW						
				Gen	eral VCMM an	d Gaussian	d Gaussian VCMM					
ECM.Nelder.Mead	12.012	10.720	2.931	5.212	1.806	-3.681	-6.212	-5.423	-5.664	-6.083		
ECM.BFGS	12.812	10.720	2.932	5.212	1.806	-2.904	-6.212	-4.738	-5.665	-6.077		
ECME.Nelder.Mead	12.606	10.288	2.931	1.899	1.806	-4.365	-6.452	-5.393	-5.663	-6.124		
ECME.BFGS	13.911	10.288	2.932	-0.253	1.806	-3.678	-6.625	-5.357	-5.664	-6.082		
MCECM.Nelder.Mead	14.442	10.720	2.931	5.212	1.806	-2.680	-6.212	-5.453	-5.665	-6.082		
MCECM.BFGS	12.812	10.720	2.932	5.212	1.806	-2.684	-6.212	-5.440	-5.664	-6.082		
			General V	/CMM (Co	pula with singl	e parameter) and Gaussian VCMM						
ECM.Nelder.Mead	12.649	10.602	1.251	6.249	1.425	-4.538	-6.595	-5.072	-5.467	-5.123		
ECM.BFGS	15.929	10.602	1.422	6.249	1.425	-3.782	-6.582	-5.069	-5.672	-5.708		
ECME.Nelder.Mead	12.152	9.464	1.422	2.578	1.425	-4.179	-6.674	-4.992	-5.548	-5.123		
ECME.BFGS	16.296	9.464	1.422	0.343	1.425	-4.643	-6.745	-4.952	-5.669	-5.548		
MCECM.Nelder.Mead	12.650	10.601	1.422	6.249	1.425	-4.421	-6.595	-5.150	-5.775	-5.127		
MCECM.BFGS	16.308	10.602	1.251	6.248	1.425	-4.023	-6.582	-5.070	-5.671	-5.669		
	General VCMM						MM and GMM					
ECM	9.681	8.989	2.828	2.279	0.438	-6.953	-7.563	-6.880	-6.950	-7.439		
MCECM	10.528	8.972	2.829	2.283	0.438	-6.686	-7.558	-6.871	-6.948	-7.426		
ECME.Nelder.Mead	-	8.014	0.330	-1.567	-1.642	-	-	-	-	-		
ECME.BFGS	-	-	-	-	-	-	-	-	-	-		

here to determine whether the model fits the data statistically significantly better than the another one.

We follow the formula in the paper "Testing for zero inflation in count models: Bias correction for the Vuong test" [Desmarais and Harden, 2013] to calculate the z statistics of the the Vuong test with BIC correction. The Table 3.2.2 shows the z statistics of the the Vuong test between three pairs of models: 1. General VCMM and Gaussian VCMM, 2. General VCMM (Copula with single parameter) and Gaussian VCMM and 3. General VCMM and GMM. The interpretation of the z statistics is in the following:

- $z \leq -1.65$ : The first model is preferred with significance level of 0.05.
- $-1.65 < z \le 0$ : The first model is preferred, but not significantly.
- $0 < z \le 1.65$ : The second model is preferred, but not significantly.
- 1.65 < z: The second model is preferred with significance level of 0.05.

For data BCW, all the z statistics are smaller than -1.65, so we can conclude that General VCMM / General VCMM (Copula with single parameter) is preferred significantly over Gaussian VCMM or GMM with significance level of 0.05. This is what we can expect because the high flexibility of VCMM algorithm should fit the BCW data with non-Gaussian data better. However, For data AIS, most algorithms are with z statistics larger 1.65, which means the Gaussian VCMM or GMM are preferred. Note that the BIC value tends to select a model that reasonably approximates the density. However, the model does not always give a good clustering partition. Such model examples can be found in Punzo and McNicholas [2016] and Scrucca et al. [2016]. Although the result in Vuong test with BIC correction is out of our expectation, referring to the Table 4.3.1, the classification rate of general VCMM and General VCMM (Copula with single parameter) with Hierarchical (scale) is still higher than Gaussian VCMM. For clustering non-Gaussian data, general VCMM is still more recommended.

## 4. Ranking of the performance for each modelling method

The Table 4.3.4 shows the ranking of the performance of EM algorithms and the initialization strategies for AIS and BCW data set by VCMM and GMM algorithm. As we mentioned before, based on the result from real data sets, we recommend using Hierarchical scale as the initialization strategy. With regard to classification rate, the methods with top ranking here are using Hierarchical scale. Also, among the EM algorithms with Hierarchical scale, you can see that needed computation time of MCECM.BFGS is mostly the least. Therefore, we would recommend using general VCMM with MCECM.BFGS and Hierarchical scale for clustering non Gaussian data.

Table 4.3.4: The top 8 and bottom 4 performance of EM algorithms and the initialization strategies for the data set of AIS and BCW by using VCMM and GMM algorithm, where the algorithm with 1) higher classification rate, 2) lower normalised BIC and 3) shorter computation time is considered to be the better performance and the priority is given by (1) > (2) > (3).

General VSM         General VSM         General VSM         General VSM         Comparation Resultance Configuration Resultance		AIS										
Jamia         Classifiant         Normal         Response of the section of			General VCM	IM					Gaussian VCN	4M		
ora         oral         oral <th< td=""><td>Ranking</td><td>Algorithm</td><td>Initialization</td><td>Classification</td><td>Normalised</td><td>Computation</td><td>Ranking</td><td>Algorithm</td><td>Initialization</td><td>Classification</td><td>Normalised</td><td>Computation</td></th<>	Ranking	Algorithm	Initialization	Classification	Normalised	Computation	Ranking	Algorithm	Initialization	Classification	Normalised	Computation
1         MCRCALBUCS         Heardwall and M         0.044         2.027         1         MCRCALBUCS         Heardwall and M         0.044         2.027         101           3         MCRCALBUCS         Heardwall and M         0.024         2.029         4.8         3         MCRCALBUCS         Heardwall and M         0.024         2.029         4.8         3         MCRCALBUCS         Heardwall and M         0.024         2.127         2.04           4         REXISTENCIS         Heardwall and M         0.024         2.129         4.9         6         REXISTENCIA Model And M         2.021         3.0           5         MCRCALBUCS         More and M         0.024         2.129         9.0         6         MCRCALBUCS         More and M         0.014         2.217         2.00           5         MCRCALBUCS         More and M         0.024         2.03         9.0         FCALBUCS         Reama         0.004         2.237         3.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00         7.00        <	0	0	strategy	rate	BIC	time (Seconds)		0	strategy	rate	BIC	time (Seconds)
1     IDEX LINENS     Hermicial and Networks     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94     9.94 <td>1</td> <td>MCECM.BFGS</td> <td>Hierarchical scale</td> <td>0.924</td> <td>23.92</td> <td>277</td> <td>1</td> <td>MCECM.BFGS</td> <td>Hierarchical scale</td> <td>0.914</td> <td>23.57</td> <td>151</td>	1	MCECM.BFGS	Hierarchical scale	0.924	23.92	277	1	MCECM.BFGS	Hierarchical scale	0.914	23.57	151
3     MCCM.Mode.Mod     International and Mathematical Mathmatical Mathmatical Mathmatical Mathematical Mathmatimatical Mathm	2	ECM.BFGS	Hierarchical scale	0.924	23.92	303	2	ECM.BFGS	Hierarchical scale	0.914	23.57	164
4     ICCM ENERS     International and Antiper Ant	3	MCECM.Nelder.Mead	Hierarchical scale	0.924	23.92	488	3	ECM.Nelder.Mead	Hierarchical scale	0.914	23.57	256
5     EXCLAPECA     Herenkia Land     949     949     95     MCCLAPECS     Herenkia Land     944     913     933       6     EXM.RNAMM     Herenkia Land     6044     23.7     330     7     EXM.RNAM     Herenkia Land     6044     23.7     330       7     EXM.RNAM     Herenkia Land     Herenkia Land     Herenkia Land     Herenkia Land     6044     23.7     430       3     EXM.RNAM     Herenkia Land     Herenkia Land     Herenkia Land     Herenkia Land     6048     20.6     54.1       4     EXM.RNAM     Herenkia Land     Her	4	ECME.BFGS	Hierarchical scale	0.924	23.92	527	4	MCECM.Nelder.Mead	Hierarchical scale	0.914	23.57	256
6     RNAREAGE     Networksing of an angle of a set of a	5	ECM.Nelder.Mead	Hierarchical scale	0.924	23.92	640	5	MCECM.BFGS	Kmeans	0.914	23.57	348
7     EXALL PICS     Kurams     00.14     2.37     309     7     EXALLPICS     Kurams     0.01.1     2.37     302       27     EXALMPCS     Readom     0.38     2.50     402     27     MCEXLMode_Med     Readom     0.65     2.08     522       28     EXALMPCS     Readom     0.57     2.51     84     20     EXALMPCS     Readom     0.65     2.08     523       29     MCEXLMPCS     Readom     0.67     2.51     14     20     EXALMPCS     Readom     0.65     2.08     523       30     MCEXLMPCS     Rescription     Constitution     Readom     0.65     2.08     524       4     Mariniu     Rescription     Constitution     Readom     0.67     2.14     2.05       1     MCEXLMPCS     Rescription     0.02     2.35     2.25     2.04     4.04     Readom     0.680     2.14     2.05       3     MCEXLMPCS     Rescription     0.024     2.35     2.25     2.04     MCEXLMPCM     Rescription     0.037     2.42     0.031       1     MCEXLMPCS     Rescription     Rescription     Rescription     Rescription     Rescription     0.037     2.42     0.031	6	ECME.Nelder.Mead	Hierarchical scale	0.924	23.92	914	6	ECME.BFGS	Hierarchical scale	0.914	23.57	356
s     ECCURINCS     Herrarbial     0.014     2.121     140     8     ECRENANderAled     Encodem     0.053     2.036     520       28     ECMLBYCS     Randem     0.570     2.5.1     846     2.0     ECMLANderAled     Randem     0.656     2.0.8     571       30     ECMLANDERS     Randem     0.571     2.5.1     444     2.0     ECMLANDERS     Randem     0.656     2.0.8     571       7     ECMLANDERS     Randem     0.571     2.5.1     774     30     ECMLANDERS     Randem     0.656     2.0.8     771       Randem     Casar Values     Casar Values     Casar Values     Casar Values     Randem     0.657     2.0.8     771       10     ECRLANDDERS     Herarchical acid     0.034     2.3.5     2.02     1     ECMLANDARAD     Herarchical acid     0.037     2.1.8     0.014       11     ECRLANDDERS     Herarchical acid     0.034     2.3.5     0.010     Herarchical acid     0.037     2.1.8     0.017       12     ECRLANDDERS     Herarchical acid     0.031     2.3.5     0.010     Herarchical acid     0.077     2.1.8     0.017       13     MCEMLBYCS     Herarchical acid     0.021     2.3.5 <td>7</td> <td>ECME.BFGS</td> <td>Kmeans scale</td> <td>0.914</td> <td>23.52</td> <td>359</td> <td>7</td> <td>ECM.BFGS</td> <td>Kmeans</td> <td>0.914</td> <td>23.57</td> <td>392</td>	7	ECME.BFGS	Kmeans scale	0.914	23.52	359	7	ECM.BFGS	Kmeans	0.914	23.57	392
121     ECM_Medic_Made     Random     0.08     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9.00     9	8	ECM.BFGS	Hierarchical	0.914	24.21	149	8	ECME.Nelder.Mead	Hierarchical scale	0.914	23.57	452
28         RCM. BFCS         Random         0.757         25.81         41.44         29         RCME.BFCS         Random         0.058         23.04         721           RCME.NLBFCS         Random         10.57         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0         27.0 <td< td=""><td>27</td><td>ECM.Nelder.Mead</td><td>Random</td><td>0.580</td><td>25.60</td><td>402</td><td>27</td><td>MCECM.Nelder.Mead</td><td>Random</td><td>0.658</td><td>23.06</td><td>352</td></td<>	27	ECM.Nelder.Mead	Random	0.580	25.60	402	27	MCECM.Nelder.Mead	Random	0.658	23.06	352
9         MCRCM BPG         Random         0.578         25.78         25.78         47.84         290         RCMERAING         Random         0.068         21.06         75.74           Commany Brain Landom         Commany Brain Landom <th< td=""><td>28</td><td>ECM.BFGS</td><td>Random</td><td>0.576</td><td>25.81</td><td>386</td><td>28</td><td>ECM.Nelder.Mead</td><td>Random</td><td>0.658</td><td>23.06</td><td>371</td></th<>	28	ECM.BFGS	Random	0.576	25.81	386	28	ECM.Nelder.Mead	Random	0.658	23.06	371
10         EXABLE-Mode/ Mode         Random         6.0.57         20.70         FXOR ENded/ Mode         Random         6.0.58         2.0.90         FXOR           Random         Algerithm         Initialization         Classification         Normalised         Computation         Random         Algerithm         Initialization         Normalised         Computation           1         MCREALING         Hierarchical         0.024         2.2.8         1         EXMLE-Noder-Mode         Hierarchical         0.8.70         2.1.4         2.0.2.8           3         MCREALING         Hierarchical         0.0.21         2.2.5         2.5.2         3         BCM         Konnons         0.0.77         2.1.28         0.0031           4         MCREALING         Hierarchical         0.0.21         2.2.5         0.033         6         MCREAN         Konnons eads         0.0.77         2.1.28         0.0017           5         MCREANAde/ Mode         Hierarchical and         0.0.21         2.5.5         0.303         6         MCREAN         Konnons eads         0.0.77         2.1.28         0.0017           7         MCREANAde/ Mode         Random         0.0.569         2.5.75         3.46         MCREANAde/ Mode         Normalised<	29	MCECM.BFGS	Random	0.576	25.81	414	29	ECME.BFGS	Random	0.658	23.06	524
Initialization of cassingui a variante de la seriante de la s	30	ECME.Nelder.Mead	Random	0.572	25.76	478	30	ECME.Nelder.Mead	Random	0.658	23.06	787
Ranking         Agerithm         Initialization         Classification         Normalised         Computation         Radgerithm         Initialization         Classification         Classification <th< td=""><td></td><td>General</td><td>VCMM (Copula with</td><td>single paramet</td><td>er)</td><td></td><td></td><td></td><td>GMM</td><td></td><td></td><td></td></th<>		General	VCMM (Copula with	single paramet	er)				GMM			
	Ranking	Algorithm	Initialization	Classification	Normalised	Computation	Ranking	Algorithm	Initialization	Classification	Normalised	Computation
1         MCECM.BFC3         Hierarchical         0.024         23.83         1.24         ECML Noble-Model         Hierarchical         0.0302         24.14         283           3         MCECM.BFC3         Hierarchical ando         0.024         2.85         2.72         3         ECML Noble-Model         Hierarchical ando         0.024         2.35         0.034         MCECM         Kneenas         0.077         2.42         0.017           6         ECMLBROK Moder Model         Hierarchical ando         0.244         2.85         5.63         MCECM         Kneenas scale         0.077         2.42         0.0176           7         MCECM.Moder Model         Hierarchical ando         0.244         2.85         5.63         Kneenas         0.076         2.42         0.0176           8         ECMLBORS         Random         0.59         2.57         3.46         6.099         Kneenas         Kneenas         Kneenas         Kneenas         Kneenas         2.81         Kneenas			strategy	rate	BIC	time (Seconds)			strategy	rate	BIC	time (Seconds)
2         EXM. Noder. March         Interarchial and Ord         0.024         21.85         27.22         3         ECML Sectures         Interarchial and Ord         0.07.7         21.28         0.037           4         IDEXA BICSS         Interarchial and Ord         0.024         23.85         29.55         44         MCECM         Kneenus         0.07.70         21.28         0.037           6         IDEXA BICSS         Interarchial and Ord         0.024         23.85         5.58         6         ECM. McECM         Kneenus         0.07.70         21.28         0.07.60           6         IDEXA Noblez-Model         Interarchial and Ord         0.024         23.85         10.03          1.2	1	MCECM.BFGS	Hierarchical	0.924	23.83	154	1	ECME.Nelder.Mead	Hierarchical	0.870	24.14	263
3         MCECM_MERS         Interactional social (0.924         22.85         272         3         KCM         Kneaum         0.767         24.28         0.034           5         IPCAMERES         Interactional social (0.924         22.85         646         5         MCECM         Kneaum scale         0.767         24.28         0.091           6         IPCAMERES         Interactional social (0.924         22.85         0.034         22.85         0.034           7         MCECM Neder Mend         Interactional social (0.924         22.85         0.034         22.85         0.034           27         ECME_BFCS         Random         0.059         22.77         3.46              SCCMAN-Meder Mend         Random         0.569         22.75         3.46           28         ECME_BFCS         Random         0.569         22.75         3.46                MCECM_Meder Mend         Random         0.569         2.77         3.46           Compatibion              Compatibion            MCECM_MECM_Meder	2	ECM.Nelder.Mead	Hierarchical	0.924	23.83	222	2	ECME.Nelder.Mead	Hierarchical scale	0.862	24.24	77
4         ICXALERCS         Interactional solar         0.034         2.2.85         2.95         4         MCCM         Knowams         0.767         2.1.28         0.047           6         Interactional solar         0.034         2.2.85         5.38         6         ECAL         Knowam scale         0.767         2.1.28         0.076           8         Interactional solar         0.924         2.2.85         5.036         6         ECAL         Knowam scale         0.767         2.1.28         0.076           8         IECME-Robined         Bineractional solar         0.924         2.2.85         0.030                 Non-Non-Non-Non-Non-Non-Non-Non-Non-Non-	3	MCECM.BFGS	Hierarchical scale	0.924	23.85	272	3	ECM	Kmeans	0.767	24.28	0.034
b         IDEALL IPENS         Interactional scale         0.94         2.8.8         0.96         5         MECAM         Kunnan scale         0.767         2.1.28         0.0076           7         MCEAM Neder Mond         Hierarchical scale         0.924         2.8.8         0.034            2.1.28         0.076         2.1.28         0.076         2.1.28         0.076           77         ECME. IPICS         Random         0.59         25.70         3.84                2.1.8         1.0.8              2.1.8         1.0.8                  2.2.8         0.076         3.3.4                 3.4.8         1.0.8         1.0.8            2.8.8         0.8.8         0.0.7         1.0.8.8         Computation         Intidization         Normalies         Computation         Intidization         Normalies         Computation         Intidization         Normalies         Normalies         Computation	4	ECM.BFGS	Hierarchical scale	0.924	23.85	295	4	MCECM	Kmeans	0.767	24.28	0.047
0         Interactional seta         0.924         2.8.8         5.88         0         D.C.M         Knears scale         0.,00         24.2.8         00.00           8         ICCNERVedor Mear         Hierarchian seta         0.924         2.8.8         1003         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -<	5	ECME.BFGS	Hierarchical scale	0.924	23.85	496	5	MCECM	Kmeans scale	0.767	24.28	0.091
aMACE/AL Notifier and is all interactional seale0.9242.2.8.8.93627ECALE_RECSRandom0.06620.4.6.09028ECALE_RECSRandom0.5.8022.7.3.36329MCE/CL/Note/AL-MedRandom0.50022.7.3.36420ECAL-REVISRandom0.50022.7.3.36420MCE/CL/Note/AL-MedRandom0.50022.7.3.364Intellization of the second secon	6	ECM.Nelder.Mead	Hierarchical scale	0.924	23.85	538	6	ECM	Kmeans scale	0.760	24.28	0.076
a         background and on background and on background and on background and background and	1	MCECM.Nelder.Mead	Hierarchical scale	0.924	23.85	593						
21         EX.NE.ONS         Rundom         0.000         2.07         383           28         EXEN.Neder.Med         Random         0.509         2.57         383           29         MCECN.Neder.Med         Random         0.509         2.57         372           20         EXEN.Neder.Med         Random         0.509         2.57         372           EXENT         EXENT           Scatter State         Scatter State           Scatter State	07	ECME.Neider.Mead	nierarchical scale	0.924	20.00	1003						
10         10         10         0.00         2.01         0.00           20         MCRCM.Neder.Meal         Random         0.00         2.57         340           30         ECM.Neder.Meal         Random         0.00         2.57         372           30         ECM.Neder.Meal         Random         0.000         2.57         372           30         Common Mathematical Mathmathematical Mathematical Mathmathmatical Mathematical Mathemati	21	ECME.BrG5	Random	0.580	20.40	282						
and CRAMAde/MedAnalom0.0000.0000.0000.0000.0000.0000.000ECM.Nedde/MedRandom0.0000.0000.0000.0000.0000.0000.000CRAMAde/MedRandomCancerCancerCancerCancerCancerCancerCancerCRAMENGAgorithmInitializationCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerCancerC	20	MCECM Nelder Mead	Bandom	0.569	25.75	346						
Both Mathematican         Name         Ones         Data         Interpretation           Ranking         General VCMM         Initialization         Classification         Normalised         Computation         Ranking         Algorithm         Initialization         Classification         Normalised         Computation           1         ECM.EPGS         Hierarchical scale         0.869         -7.24         91         2         ECM.BFGS         Kmeans scale         0.797         -6.60         84           2         MCECM.BFGS         Hierarchical scale         0.869         -7.24         108         3         MCECM.MFGS         Kmeans scale         0.797         -6.60         85           3         ECMLEMGGS         Hierarchical scale         0.869         -7.24         108         3         MCECM.Meder.Mead         Kmeans scale         0.797         -6.60         122           6         RCMLNeder.Mead         Hierarchical scale         0.869         -7.24         112         5         ECMLEPGS         Kmeans scale         0.797         -6.60         122           6         ECMLEMGERS         Kmeans scale         0.851         -7.21         52         F         MCECM.BFGS         Hierarchical scale         0.784         -6	30	ECM Nelder Mead	Random	0.569	25.75	372						
Herein Unitalization Energi VersionComputation Ranking AlgorithmAlgorithmInitialization InitializationComputation Ranking MescumberAlgorithmInitialization InitializationComputation images1ECMLBFOSHierarchical scale0.869-7.24912ECMLBFOSKmeans scale0.797-6.60-853MCECM.Neder MeadHierarchical scale0.869-7.241012ECMLBFOSKmeans scale0.797-6.60-975ECMLBFOSHierarchical scale0.869-7.241014ECMLNeder.MeadKmeans scale0.797-6.601025ECMLMAder.MeadHierarchical scale0.869-7.241115ECMLBFOSKmeans scale0.797-6.601126ECMLNeder.MeadHierarchical scale0.869-7.241115ECMLBFOSKmeans scale0.797-6.601536ECMLNeder.MeadHierarchical scale0.851-7.21527ECMLBFOSHierarchical scale0.784-6.61-868ECMLNeder.MeadRmeans scale0.851-7.21527MCECM.Neder.MeadKmeans0.685-6.261182ECMLNeder.MeadRmeans0.851-7.21527MCECM.Neder.MeadKmeans0.685-6.261183MCECM.Neder.MeadRmeans0.851-7.21		Benniterationa	Tuntom	0.000	20110	012						
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$						BC	ŚW					
Ranking         Algorithm         Initialization         Initialization         Classification         Normalise         Computation           Image: Computation         Farte         Ric         BC         BC         image: Computation         strateg         Image: Computation			G 1.1/01/									
interpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinterpretationinter	Banking		General VCM	IM					Gaussian VCN	<u>4M</u>		
1     ECM.BFGS     Hierarchical scale     0.869     -7.24     74     1     MCECM.BFGS     Kmeans scale     0.797     -6.60     85       3     ECME.BFGS     Hierarchical scale     0.869     -7.24     108     33     MCECM.Nelder.Mead     Kmeans scale     0.797     -6.60     97       4     MCECM.Nelder.Mead     Hierarchical scale     0.869     -7.24     111     4     ECM.Nelder.Mead     0.797     -6.60     122       6     ECME.Nelder.Mead     Hierarchical scale     0.869     -7.24     165     6     ECME.Nelder.Mead     0.797     -6.60     122       6     ECM.Nelder.Mead     Hierarchical scale     0.869     -7.24     165     6     ECME.Nelder.Mead     0.797     -6.60     122       7     MCECM.BFGS     Kmeans scale     0.869     -7.24     165     8     MCECM.BFGS     Hierarchical scale     0.784     -6.61     92       27     MCECM.BFGS     Kmeans     Random     0.499     -6.98     320     28     ECM.ENGHCM.Mend     Kmeans     0.685     -6.26     124       29     ECM.BFGS     Random     0.499     -6.98     329     28     ECM.ENGHCM.Mend     Kmeans     0.685     -6.26     124	100000	Algorithm	General VCM Initialization	Classification	Normalised	Computation	Ranking	Algorithm	Gaussian VCM Initialization	<u>IM</u> Classification	Normalised	Computation
2         MCECM.BFGS         Hierarchical scale         0.869         -7.24         91         2         ECM.BFGS         Kmeans scale         0.797         -6.60         95           3         IECME.BFGS         Hierarchical scale         0.869         -7.24         111         44         ECM.Nelder.Mead         Kmeans scale         0.797         -6.60         102           5         IECME.Nelder.Mead         Hierarchical scale         0.869         -7.24         112         5         ECM.ENdeler.Mead         0.797         -6.60         153           6         CCME.Nelder.Mead         Hierarchical scale         0.869         -7.24         152         5         ECM.BelGer.Mead         0.797         -6.60         153           7         MCECM.BFGS         Kmeans scale         0.851         -7.21         52         7         ECM.BelGS         Hierarchical scale         0.784         -6.61         92           7         ECM.Nelder.Mead         Kmeans scale         0.499         -6.98         332         28         ECM.Nelder.Mead         Kmeans         0.685         -6.26         134           20         ECM.Nelder.Mead         Initialization         0.499         -6.87         349         30         ECM.ENdele	- tunning	Algorithm	General VCM Initialization strategy	M Classification rate	Normalised BIC	Computation time (Seconds)	Ranking	Algorithm	<u>Gaussian VCM</u> Initialization strategy	1 <u>M</u> Classification rate	Normalised BIC	Computation time (Seconds)
3     ECCM. BelGer.     Hierarchical seale     0.869     -7.24     111     44     CCM.Nelder.Mead     Hierarchical seale     0.869     -7.24     111     44     ECM.Nelder.Mead     Kneans scale     0.797     -6.60     102       6     ECM.Nelder.Mead     Hierarchical seale     0.869     -7.24     112     55     ECM.ENder.Mead     Kneans scale     0.797     -6.60     102       6     ECM.Nelder.Mead     Hierarchical seale     0.869     -7.24     112     55     6     ECM.ENder.Mead     Hierarchical seale     0.797     -6.60     153       7     MCECM.Nelder.Mead     Kmeans scale     0.851     -7.21     55     8     MCECM.Nelder.Mead     Hierarchical seale     0.784     -6.60     153       7     ECM.Nelder.Mead     Kmeans scale     0.851     -7.21     55     8     MCECM.Nelder.Mead     Kmeans     0.685     -6.26     118       28     ECM.Nelder.Mead     Random     0.499     -6.90     392     28     ECM.Nelder.Mead     Kmeans     0.685     -6.26     118       29     ECM.BFGS     Random     0.499     -6.90     392     28     ECM.ENder.Mead     Kmeans     0.685     -6.26     118       20     ECM.BFGS <td>1</td> <td>Algorithm ECM.BFGS</td> <td>General VCM Initialization strategy Hierarchical scale</td> <td>Classification rate 0.869</td> <td>Normalised BIC -7.24</td> <td>Computation time (Seconds) 74</td> <td>Ranking 1</td> <td>Algorithm MCECM.BFGS</td> <td>Gaussian VCM Initialization strategy Kmeans scale</td> <td>1M Classification rate 0.797</td> <td>Normalised BIC -6.60</td> <td>Computation time (Seconds) 84</td>	1	Algorithm ECM.BFGS	General VCM Initialization strategy Hierarchical scale	Classification rate 0.869	Normalised BIC -7.24	Computation time (Seconds) 74	Ranking 1	Algorithm MCECM.BFGS	Gaussian VCM Initialization strategy Kmeans scale	1M Classification rate 0.797	Normalised BIC -6.60	Computation time (Seconds) 84
4         MCECM.Nelder.Mead         Hierarchical scale         0.869 $-7.24$ 112         5         ECMLBelder.Mead         Minerabical scale         0.869 $-7.24$ 112         5         ECMLBelder.Mead         Minerabical scale         0.797 $-6.60$ 122           7         MCECM.BFCS         Kmeans scale         0.869 $-7.24$ 165         6         ECMLBelGes         Kmeans scale         0.797 $-6.60$ 122           7         MCECM.BFCS         Kmeans scale         0.851 $-7.21$ 52         7         ECMLBelGes         Hierarchical scale         0.784 $-6.61$ 92           27         ECM.Nelder.Mead         Kmeans scale         0.851 $-7.21$ 55         8         MCECM.Nelder.Mead         Kmeans         0.685 $-6.26$ 118           28         ECM.Nelder.Mead         Kmeans         0.685 $-6.26$ 124         14           30         MCECM.BFCS         Random         0.499 $-6.97$ 349         30         ECM.EBrCs         Kmeans         0.685 $-6.26$ 144           30         MCECM.BFCS         Random         0.499 $-6.47$ </td <td>1 2</td> <td>Algorithm ECM.BFGS MCECM.BFGS</td> <td>General VCM Initialization strategy Hierarchical scale Hierarchical scale</td> <td>Classification rate 0.869 0.869</td> <td>Normalised BIC -7.24 -7.24</td> <td>Computation time (Seconds) 74 91</td> <td>Ranking 1 2</td> <td>Algorithm MCECM.BFGS ECM.BFGS</td> <td>Gaussian VCM Initialization strategy Kmeans scale Kmeans scale</td> <td><u>4M</u> Classification rate 0.797 0.797</td> <td>Normalised BIC -6.60 -6.60</td> <td>Computation time (Seconds) 84 85</td>	1 2	Algorithm ECM.BFGS MCECM.BFGS	General VCM Initialization strategy Hierarchical scale Hierarchical scale	Classification rate 0.869 0.869	Normalised BIC -7.24 -7.24	Computation time (Seconds) 74 91	Ranking 1 2	Algorithm MCECM.BFGS ECM.BFGS	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale	<u>4M</u> Classification rate 0.797 0.797	Normalised BIC -6.60 -6.60	Computation time (Seconds) 84 85
bCM.Nedder.Mead       Hierarchical scale       0.869       -7.24       162       5       ECM.E.BefCS       Kmeans scale       0.797       -6.60       122         6       DCM.Neder.Mead       Hierarchical scale       0.869       -7.24       165       6       ECM.ENder.Mead       Hierarchical scale       0.797       -6.60       152         7       MCECM.BFGS       Kmeans scale       0.851       -7.21       52       8       MCECM.BFGS       Hierarchical scale       0.784       -6.61       92         7       ECM.Nelder.Mead       Kmeans       Random       0.499       -6.98       322       27       MCECM.Nelder.Mead       Kmeans       0.685       -6.26       118         20       ECM.EBFGS       Random       0.499       -6.98       329       20       ECM.Nelder.Mead       Kmeans       0.685       -6.26       118         20       ECM.BFGS       Random       0.499       -6.90       20       20       ECM.Nelder.Mead       Kmeans       0.685       -6.26       134         20       ECM.BFGS       Random       0.499       -6.90       Computation       Ranking       Algorithm       Initialization       Normalised       Computation       Ranking       Al	1 2 3	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS	Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale	<u>M</u> Classification rate 0.869 0.869 0.869	Normalised BIC -7.24 -7.24 -7.24	Computation time (Seconds) 74 91 108	Ranking 1 2 3	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale	<u>1M</u> Classification rate 0.797 0.797 0.797	Normalised BIC -6.60 -6.60 -6.60	Computation time (Seconds) 84 85 97
6 6 1EXME.Neider.MeadInterarchical scale0.889 $-7.24$ 1656 6EVM.E.Neider.MeadKmeans scale0.797 $-6.60$ $153$ 7MCECM.BFGSKmeans scale0.851 $-7.21$ 558MCECM.BFGSHierarchical scale0.784 $-6.61$ 9227ECM.Nelder.MeadRandom0.499 $-6.98$ 25527MCECM.Nelder.MeadKmeans0.685 $-6.26$ 11828ECM.BFGSRandom0.499 $-6.98$ 25527MCECM.Nelder.MeadKmeans0.685 $-6.26$ 12430MCECM.BFGSRandom0.499 $-6.98$ 20920ECM.EBFGSKmeans0.685 $-6.26$ 13430MCECM.BFGSRandom0.499 $-6.97$ 34930ECME.Nelder.MeadKmeans0.685 $-6.26$ 14330MCECM.BFGSRandom0.499 $-6.97$ 34930ECME.Nelder.MeadKmeans0.685 $-6.26$ 14330MCECM.BFGSRandom0.499 $-6.97$ 34930ECME.Nelder.MeadKmeans $0.685$ $-6.26$ 14330MCECM.BFGSRandom0.499 $-6.97$ 34930ECME.Nelder.MeadKmeans $0.685$ $-6.26$ 14331MCECM.BFGSHierarchical scale0.899 $6.87$ $7.08$ RankingRalkingAlgorithmInitializationClassificationNormalise $0.099$ 31MCECM.Nelder.Mead<	1 2 3 4	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead	Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale	Image: Classification rate           0.869           0.869           0.869           0.869           0.869           0.869	Normalised BIC -7.24 -7.24 -7.24 -7.24	Computation time (Seconds) 74 91 108 111	Ranking 1 2 3 4	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale	IM           Classification           rate           0.797           0.797           0.797           0.797           0.797	Normalised BIC -6.60 -6.60 -6.60 -6.60	Computation time (Seconds) 84 85 97 102
7       MCBCM.BFGS       Hierarchical scale $0.784$ $-6.61$ $86$ 8       ECM.Nelder.Mead       Random $0.499$ $-6.98$ $255$ $8$ MCECM.BFGS       Hierarchical scale $0.784$ $-6.61$ $92$ 27       ECM.Nelder.Mead       Random $0.499$ $-6.98$ $392$ $28$ ECM.Nelder.Mead       Kneans $0.685$ $-6.26$ $114$ 29       ECM.BFGS       Random $0.499$ $-6.9$ $392$ $28$ ECME.Nelder.Mead       Kneans $0.685$ $-6.26$ $124$ 29       BCM.BFGS       Random $0.499$ $-6.9$ $349$ $30$ ECME.Nelder.Mead       Kneans $0.685$ $-6.26$ $143$ CMM (Copula with singe parameter)       ECME.Nelder.Mead       Kneans $0.685$ $-6.26$ $143$ CMM (Copula with singe parameter)       ECME.Nelder.Mead       Kneans $0.685$ $-6.26$ $143$ CMM (Copula with singe parameter)       ECM.Nelder.Mead       Kneans $0.685$ $-6.26$ $143$ Hierarchical scale $0.878$	1 2 3 4 5	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead	Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale	M Classification rate 0.869 0.869 0.869 0.869 0.869	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24	Computation time (Seconds) 74 91 108 111 112	Ranking 1 2 3 4 5	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECME.BFGS	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale	1M           Classification           rate           0.797           0.797           0.797           0.797           0.797           0.797           0.797	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60	Computation time (Seconds) 84 85 97 102 122
8         ECM.Nelder.Mead         Kmeans scale         0.85         -7.21         55         8         MCECM.MBder.S         Herarchical scale         0.784         -6.01         92           27         ECM.Nelder.Mead         Random         0.499         -6.98         392         28         ECM.Nelder.Mead         Kmeans         0.685         -6.26         114           28         ECM.ERGS         Random         0.499         -6.90         420         29         ECME.Nelder.Mead         Kmeans         0.685         -6.26         134           30         MCECM.BFGS         Random         0.499         -6.90         420         29         ECME.Nelder.Mead         Kmeans         0.685         -6.26         134           30         MCECM.BFGS         Random         0.499         -6.90         420         29         ECME.Nelder.Mead         Kmeans         0.685         -6.26         134           30         ECMENELGENENK         Cemeral/VCM/Vinuel/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/Vinue/V	1 2 3 4 5 6	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead	Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale	M Classification rate 0.869 0.869 0.869 0.869 0.869 0.869 0.869	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24	Computation time (Seconds) 74 91 108 111 112 165	Ranking 1 2 3 4 5 6	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECME.BFGS ECME.Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale	1M           Classification           rate           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.60	Computation time (Seconds) 84 85 97 102 122 153
24ECM. Neder. MeadRandom0.439-0.3825324NLCECM.Neder. MeadKmeans0.085-0.2011828ECM. BEGSRandom0.499-6.9042029ECME.BFGSKmeans0.685-6.2612429ECM.BFGSRandom0.499-6.9042029ECME.BFGSKmeans0.685-6.2613430MCECM.BFGSRandom0.499-6.8734930ECME.Nelder.MeadKmeans0.685-6.26143RankingAlgorithmInitializationClossificationNormalisedComputationRankingAlgorithmInitializationClassificationNormalisedComputation1ECM.Nelder.MeadHierarchical scale0.878-7.081621ECMHierarchical scale0.784-6.440.0392ECM.BFGSHierarchical scale0.875-7.081473ECMHierarchical scale0.784-6.440.0954MCECM.BFGSHierarchical scale0.875-7.081473ECMKmeans scale0.784-6.440.0954MCECM.BFGSHierarchical scale0.876-7.141355ECMHierarchical0.784-6.440.0734MCECM.BFGSHierarchical scale0.869-7.141355ECMHierarchical0.784-6.440.0735ECM.BFGSHierarchical scale0.869-7.14<	1 2 3 4 5 6 7	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead MCECM.BFGS DEMM.Nelder.Mead MCECM.BFGS	General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kineans scale	M Classification rate 0.869 0.869 0.869 0.869 0.869 0.869 0.869 0.869 0.851	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21	Computation time (Seconds) 74 91 108 111 112 165 52 77	Ranking 1 2 3 4 5 6 7 7	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.EBFGS ECME.Nelder.Mead ECM.EBFGS	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale	4M           Classification           rate           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.61	Computation time (Seconds) 84 85 97 102 122 153 86 86
1010100.457 $-0.53$ $0.52$ $25$ $-0.10440$ $-0.024$ $0.0050$ $-0.20$ $124$ 29ECM.ENGSRandom0.499 $-6.87$ $349$ $20$ ECM.ENGSKmeans $0.685$ $-6.26$ $134$ 30MCECM.BFGSRandom $0.499$ $-6.87$ $349$ $30$ ECME.Nelder.MeadKmeans $0.685$ $-6.26$ $134$ RankingAlgorithmInitializationClassificationNormaliseComputationRankingAlgorithmInitializationClassificationNormaliseComputation1ECM.Nelder.MeadHierarchical scale $0.878$ $-7.08$ $162$ $1$ ECMHierarchical scale $0.784$ $-6.44$ $0.039$ 2ECM.ENded.MeadHierarchical scale $0.878$ $-7.08$ $162$ $1$ ECMHierarchical scale $0.784$ $-6.44$ $0.039$ 3MCECM.Nelder.MeadHierarchical scale $0.875$ $-7.08$ $143$ $2$ $MCECMHierarchical scale0.784-6.440.0953MCECM.Nelder.MeadHierarchical scale0.869-7.141355ECM.MeanMean0.784-6.440.0734MCECM.Nelder.MeadHierarchical scale0.869-7.141355ECM.MeanMean0.681-6.440.0735ECM.ENGSHierarchical scale0.869-7.141356MCECM$	1 2 3 4 5 6 7 8	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead	General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale	<u>M</u> Classification rate 0.869 0.869 0.869 0.869 0.869 0.859 0.851 0.851	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -6.00	Computation time (Seconds) 74 91 108 111 112 165 52 55 255	Ranking 1 2 3 4 5 6 7 8 27	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.EBFGS ECME.Nelder.Mead ECM.BFGS MCECM.MEGS	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale	IM           Classification           rate           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.784           0.684	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.61	Computation time (Seconds) 84 85 97 102 122 153 86 92 112
25DCMEPGSRandom0.499 $-0.59$ $-0.59$ $-0.29$ DCMEDGSRandom0.095 $-0.29$ 14730MCECM-BFCSRandom0.499 $-6.57$ $-349$ 30ECME.Nelder.MeadKmeans0.085 $-6.26$ 143Compatibility implementationCompatibility implementationClassificationNormalisedComputationRankingAlgorithmInitializationClassificationNormalisedComputationAlgorithmInitializationStrategyrateBICtime (Seconds)1ECM.Nelder.MeadHerarchical scale0.878 $-7.08$ 1021ECMHierarchical scale0.784 $-6.44$ 0.0392ECME.Nelder.MeadHierarchical scale0.878 $-7.08$ 1473ECMKmeans scale0.784 $-6.44$ 0.0063MCECM.Nelder.MeadHierarchical scale0.872 $-7.11$ 1284MCECMKmeans scale0.784 $-6.44$ 0.0954MCECM.BFGSHierarchical scale0.866 $-7.12$ 1436MCECMHierarchical0.781 $-6.44$ 0.01255ECM.BFGSHierarchical scale0.866 $-7.12$ 1436MCECMHierarchical0.781 $-6.44$ 0.0736ECM.BFGSHierarchical scale0.866 $-7.12$ 1436MCECMHierarchical0.781 $-6.44$ 0.1257MCECM.Nelder.Mead <td< td=""><td>1 2 3 4 5 6 7 8 27 28</td><td>Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECME.Nelder.Mead MCECCM.BFGS ECM.Nelder.Mead ECME.VELder.Mead ECM.Nelder.Mead</td><td>General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Random Beacdom</td><td>M Classification rate 0.869 0.869 0.869 0.869 0.869 0.869 0.851 0.851 0.499</td><td>Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -6.98 6.08</td><td>Computation time (Seconds) 74 91 108 111 112 165 52 55 255 209</td><td>Ranking 1 2 3 4 5 6 7 8 27 28</td><td>Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.BFGS ECME.Nelder.Mead ECM.BFGS MCECM.BFGS MCECM.Nelder.Mead ECM Nelder.Mead</td><td>Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans</td><td>IM           Classification           rate           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.784           0.685           0.685</td><td>Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 6.26</td><td>Computation time (Seconds) 84 85 97 102 122 153 86 92 118 118</td></td<>	1 2 3 4 5 6 7 8 27 28	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECME.Nelder.Mead MCECCM.BFGS ECM.Nelder.Mead ECME.VELder.Mead ECM.Nelder.Mead	General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Random Beacdom	M Classification rate 0.869 0.869 0.869 0.869 0.869 0.869 0.851 0.851 0.499	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -6.98 6.08	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 209	Ranking 1 2 3 4 5 6 7 8 27 28	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.BFGS ECME.Nelder.Mead ECM.BFGS MCECM.BFGS MCECM.Nelder.Mead ECM Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans	IM           Classification           rate           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.784           0.685           0.685	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 6.26	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 118
30Interaction0.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.4030.403	1 2 3 4 5 6 7 8 27 28 20	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead	General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Random Random Random	M Classification rate 0.869 0.869 0.869 0.869 0.869 0.869 0.851 0.851 0.851 0.851 0.499 0.499	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -6.98 -6.98 -6.98 -6.90	Computation time (Seconds) 74 91 108 101 112 165 52 55 255 255 2255 392 420	Ranking 1 2 3 4 5 6 7 8 27 28 20	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS MCECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans Kmeans Kmeans	IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.7984         0.685           0.685         0.685	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 124
Concrete Version (Colspan=Ware-Displace Parameter)ComputationRankingAlgorithmInitializationComputationRankingAlgorithmInitializationComputationComputationComputationComputationComputationRankingAlgorithmInitializationClassificationNormalisedComputation1ECM.Nelder.MeadHierarchical scale0.878-7.081621ECMHierarchical scale0.784-6.440.0392ECME.Nelder.MeadHierarchical scale0.875-7.082142MCECMHierarchical scale0.784-6.440.0954MCECM.BFGSHierarchical scale0.869-7.141284MCECMKmeans scale0.784-6.440.01255ECMLBFGSHierarchical scale0.866-7.121436MCECMHierarchical0.781-6.440.01256ECM.BFGSHierarchical scale0.866-7.141555ECMHierarchical0.781-6.440.1257MCECM.Nelder.MeadKmeans scale0.839-7.141546ECMLBFGSHierarchical scale0.866-7.121436MCECMHierarchical0.781-6.44 <t< td=""><td>1 2 3 4 5 6 7 8 27 28 29 30</td><td>Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.EBFGS ECM.BFGS ECM.BFGS MCECM BFGS</td><td>General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Random Random Random</td><td>M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.851           0.499           0.499           0.499</td><td>Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -6.98 -6.98 -6.90 -6.87</td><td>Computation time (Seconds) 74 91 108 111 112 165 52 55 255 392 420 340</td><td>Ranking 1 2 3 4 5 6 7 8 27 28 29 30</td><td>Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Relder.Mead ECME.BFGS ECME.Nelder.Mead ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Relder.Mead ECME.BFGS</td><td>Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans</td><td>IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.784           0.784         0.685           0.685         0.685</td><td>Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26</td><td>Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 124 134 143</td></t<>	1 2 3 4 5 6 7 8 27 28 29 30	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.EBFGS ECM.BFGS ECM.BFGS MCECM BFGS	General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Random Random Random	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.851           0.499           0.499           0.499	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -6.98 -6.98 -6.90 -6.87	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 392 420 340	Ranking 1 2 3 4 5 6 7 8 27 28 29 30	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Relder.Mead ECME.BFGS ECME.Nelder.Mead ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Relder.Mead ECME.BFGS	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans	IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.784           0.784         0.685           0.685         0.685	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 124 134 143
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	1 2 3 4 5 6 7 8 27 28 29 30	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECME.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead ECME.BFGS ECM.BFGS MCECM.BFGS MCECM.BFGS	General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Generation of the scale Hierarchical scale Kmeans scale Kmeans scale Random Random Random Random WCMM (Consula with	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -6.98 -6.98 -6.98 -6.87 -787	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 392 420 349	Ranking 1 2 3 4 5 6 7 8 27 28 29 30	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.EBFGS ECME.Nelder.Mead ECM.BFGS MCECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECM.EBFGS ECME.BFGS	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans	IM           Classification           rate           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.798           0.784           0.685           0.685           0.685           0.685	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26 -6.26	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 118 124 134 134
ECM.Nelder.Mead         Herarchical scale         0.878         -7.08         162         1         ECM         Hierarchical scale         0.784         -6.44         0.039           2         ECME.Nelder.Mead         Hierarchical scale         0.878         -7.08         14         2         MCECM         Hierarchical scale         0.784         -6.44         0.039           3         MCECM.Nelder.Mead         Hierarchical scale         0.875         -7.08         147         3         ECM         Hierarchical scale         0.784         -6.44         0.060           3         MCECM.BFGS         Hierarchical scale         0.875         -7.08         147         3         ECM         Kmeans scale         0.784         -6.44         0.095           5         ECM.BFGS         Hierarchical scale         0.872         -7.11         128         4         MCECM         Kmeans scale         0.784         -6.44         0.073           6         ECM.BFGS         Hierarchical scale         0.869         -7.14         135         5         ECM         Hierarchical         0.781         -6.44         0.125           6         ECM.Nelder.Mead         Kmeans scale         0.839         -7.14         154         5	1 2 3 4 5 6 7 8 27 28 29 30 Banking	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECME.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead ECME.BFGS ECM.BFGS MCECM.BFGS General Algorithm	General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Random Random Random Random Nandom unitialization	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499           0.499           0.499           0.499	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -6.98 -6.98 -6.98 -6.98 -6.90 -6.87 er] Normalised	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 392 420 349	Ranking 1 1 2 3 4 5 6 7 8 27 28 29 30	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.BFGS ECME.Nelder.Mead ECM.BFGS MCECM.BFGS MCECM.Nelder.Mead ECM.BFGS ECME.BFGS ECME.Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans Kmeans Mmeans Kmeans	IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.7984         0.685           0.685         0.6855           0.685         0.685           0.685         0.685	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26 -6.26	Computation time (Seconds) 84 85 97 102 122 153 86 92 92 118 124 134 134 143
2         ECME.Nelder.Mead         Hierarchical scale $0.878$ $-7.08$ $214$ 2         MCECM         Hierarchical scale $0.674$ $-6.44$ $0.000$ 3         MCECM.Nelder.Mead         Hierarchical scale $0.875$ $-7.08$ $147$ 3         ECM         Kmeans scale $0.784$ $-6.44$ $0.095$ 4         MCECM.BFGS         Hierarchical scale $0.872$ $-7.11$ $128$ 4         MCECM         Kmeans scale $0.784$ $-6.44$ $0.095$ 5         ECM.BFGS         Hierarchical scale $0.869$ $-7.14$ $135$ $5$ ECM         Hierarchical $0.781$ $-6.44$ $0.073$ 6         ECM.BFGS         Hierarchical scale $0.866$ $-7.12$ $143$ $6$ MCECM         Hierarchical $0.781$ $-6.44$ $0.073$ 7         MCECM.Nelder.Mead         Kmeans scale $0.837$ $-7.10$ $218$ $-6.44$ $0.125$ 28         MCECM.Nelder.Mead         Random $0.509$ $-7.04$ $604$ $-6.44$	1 2 3 4 5 6 7 8 27 28 29 30 Ranking	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead Algorithm	General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Random Random Random Random VCMM (Copula with Initialization stratev	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499           0.499           0.asification           rate	Normalised BIC           -7.24           -7.24           -7.24           -7.24           -7.24           -7.24           -7.21           -6.98           -6.99           -6.98           -6.87           xr)           Normalised           BIC	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 392 420 349 Computation time (Seconds)	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECME.Nelder.Mead ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECME.BFGS ECME.Relder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans	M         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.7984         0.685           0.685         0.685           0.685         0.685           0.685         0.685	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 Normalised BIC	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 134 134 134 134 134 134 134 134
3         MCECM.Nelder.Mead         Hierarchical scale $0.875$ $-7.08$ $147$ 3         ECM         Kmeans scale $0.784$ $-6.44$ $0.095$ 4         MCECM.BFGS         Hierarchical scale $0.872$ $-7.11$ $128$ 4         MCECM         Kmeans scale $0.784$ $-6.44$ $0.095$ 5         ECM.BFGS         Hierarchical scale $0.869$ $-7.14$ $135$ $5$ ECM         Hierarchical $0.781$ $-6.44$ $0.073$ 6         ECM.BFGS         Hierarchical scale $0.866$ $-7.14$ $135$ $5$ ECM         Hierarchical $0.781$ $-6.44$ $0.073$ 7         MCECM.Nelder.Mead         Kmeans scale $0.866$ $-7.10$ $218$ $-6.44$ $0.125$ 27         ECM.Nelder.Mead         Random $0.510$ $-7.04$ $604$ $-7.04$ $-7.04$ $-7.04$ $-7.04$ $-7.04$ $-7.04$ $-7.04$ $-7.04$ $-7.04$ $-7.04$ $-7.04$ $-7.04$ $-7.04$ $-7.04$ $-7.04$	1 2 3 4 5 6 7 8 27 28 29 30 Ranking	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS ECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCEC	General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Random Random Random Random Standom Stategy Hierarchical scale	M Classification rate 0.869 0.869 0.869 0.869 0.869 0.869 0.851 0.851 0.499 0.499 0.499 0.499 0.499 0.499 0.499 0.499 0.499 0.578	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -6.98 -6.98 -6.98 -6.99 -6.87 <b>x</b> ) Normalised BIC	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 255 255 392 420 349 Computation time (Seconds) 162	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1	Algorithm MCECM.BFGS ECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.EBFGS ECME.Nelder.Mead ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Stategy Hierarchical scale	IM           Classification           rate           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.784           0.685           0.685           Classification           rate           0.784	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.26 -6.26 -6.26 -6.26 -6.26 Normalised BIC	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 134 143 143 Computation time (Seconds) 0.039
4         MCECM.BFGS         Hierarchical scale         0.872         -7.11         128         4         MCECM         Kmeans scale         0.784         -6.44         0.125           5         ECM.BFGS         Hierarchical scale         0.869         -7.14         135         5         ECM         Hierarchical         0.781         -6.44         0.073           6         ECM.BFGS         Hierarchical scale         0.866         -7.12         143         6         MCECM         Hierarchical         0.781         -6.44         0.073           7         MCECM.Nelder.Mead         Kmeans scale         0.863         -7.14         154           0.781         -6.44         0.125           8         ECM.Nelder.Mead         Kmeans scale         0.839         -7.10         218	1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.EBFGS ECM.BFGS ECM.BFGS General Algorithm ECM.Nelder.Mead ECM.Nelder.Mead	General VCM Initialization strategy Hierarchical scale Hierarchical scale Generation of the scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Random Random Random Random Stategy VCMM (Copula with Initialization strategy Hierarchical scale	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.878	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -6.98 -6.90 -6.90 -6.90 -6.90 -6.90 -7.08	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 392 420 349 Computation time (Seconds) 162 214	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 1 2 2 2 2 2 2 3 3 2 2	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.BFGS ECME.BFGS ECME.BFGS MCECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECM.EBFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Strategy Hierarchical scale	IM           Classification           rate           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.797           0.784           0.685           0.685           0.685           Classification           rate           0.784           0.784	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.26 -6.26 -6.26 -6.26 -6.26 Normalised BIC -6.44	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 134 134 143 143 Computation time (Seconds) 0.039 0.060
5         ECM.BFGS         Hierarchical scale         0.869         -7.14         135         5         ECM         Hierarchical         0.781         -6.44         0.073           6         ECME.BFGS         Hierarchical scale         0.866         -7.12         143         6         MCECM         Hierarchical         0.781         -6.44         0.073           7         MCECM.Nelder.Mead         Kmeans scale         0.837         -7.10         154                     0.781         -6.44         0.125            7         MCECM.Nelder.Mead         Kmeans scale         0.837         -7.10         218	1 2 3 4 5 6 7 8 27 28 29 30	Algorithm ECM.BFGS MCECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead ECME.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.Meder.Mead ECME.Nelder.Mead MCECM.Nelder.Mead	General VCM           Initialization           strategy           Hierarchical scale           Kmeans scale           Kmeans scale           Random           Hierarchical scale	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499           0.499           0.499           0.878           0.878           0.875	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -6.98 -6.98 -6.90 -6.87 -7.08 BIC -7.08 -7.08	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 392 420 349 Computation time (Seconds) 162 214 147	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 2 3 3 4 5 6 7 8 29 30 30 8 29 30 30 29 30 20 30 20 30 20 30 20 30 20 30 20 30 20 30 20 30 20 30 20 30 20 30 3	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.EBFGS ECME.Nelder.Mead ECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans	IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.7984         0.685           0.685         0.685           0.685         0.685           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 Normalised BIC -6.44 -6.44	Computation time (Seconds) 84 85 97 102 122 153 86 92 92 118 124 134 143 143 Computation time (Seconds) 0.039 0.060 0.095
6         ECME.BFGS         Hierarchical scale         0.866         -7.12         143         6         MCECM         Hierarchical         0.781         -6.44         0.125           7         MCECM.Nelder.Mead         Kmeans scale         0.839         -7.14         154         -         -         -         -         -         -         -         -         -         -         -         -         0.125           8         ECM.Nelder.Mead         Kmeans scale         0.837         -7.10         218         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         - <td>1 2 3 4 5 6 6 7 8 27 28 29 30 Ranking 1 2 3 4</td> <td>Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS</td> <td>General VCM           Initialization           strategy           Hierarchical scale           Kmeans scale           Kmeans scale           Random           Riterarchical scale           Hierarchical scale           Hierarchical scale           Hierarchical scale           Hierarchical scale           Hierarchical scale</td> <td>M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499           0.878           0.878           0.875           0.875           0.875</td> <td>Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.20 -7.24 -7.21 -7.21 -7.21 -7.21 -7.20 -7.20 -7.20 -7.21 -7.21 -7.21 -7.20 -7.20 -7.20 -7.21 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.</td> <td>Computation time (Seconds) 74 91 108 111 112 165 55 255 392 420 349 Computation time (Seconds) 162 214 147 128</td> <td>Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 4 5 6 7 8 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 20 29 30 20 29 30 20 29 30 20 29 30 20 29 30 20 29 30 20 29 30 20 29 30 20 20 20 20 30 20 20</td> <td>Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS ECME.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.E.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead</td> <td>Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale GMM Initialization strategy Hierarchical scale Kmeans scale Kmeans scale</td> <td>IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.7984         0.685           0.685         0.685           0.685         0.685           0.685         0.685           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784</td> <td>Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26 Normalised BIC -6.44 -6.44 -6.44</td> <td>Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 134 134 134 143 Computation time (Seconds) 0.039 0.060 0.095 0.125</td>	1 2 3 4 5 6 6 7 8 27 28 29 30 Ranking 1 2 3 4	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS	General VCM           Initialization           strategy           Hierarchical scale           Kmeans scale           Kmeans scale           Random           Riterarchical scale           Hierarchical scale           Hierarchical scale           Hierarchical scale           Hierarchical scale           Hierarchical scale	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499           0.878           0.878           0.875           0.875           0.875	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.20 -7.24 -7.21 -7.21 -7.21 -7.21 -7.20 -7.20 -7.20 -7.21 -7.21 -7.21 -7.20 -7.20 -7.20 -7.21 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.	Computation time (Seconds) 74 91 108 111 112 165 55 255 392 420 349 Computation time (Seconds) 162 214 147 128	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 4 5 6 7 8 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 20 29 30 20 29 30 20 29 30 20 29 30 20 29 30 20 29 30 20 29 30 20 29 30 20 20 20 20 30 20 20	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS ECME.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.E.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale GMM Initialization strategy Hierarchical scale Kmeans scale Kmeans scale	IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.7984         0.685           0.685         0.685           0.685         0.685           0.685         0.685           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26 Normalised BIC -6.44 -6.44 -6.44	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 134 134 134 143 Computation time (Seconds) 0.039 0.060 0.095 0.125
7         MCECM.Nelder.Mead         Kmeans scale         0.839         -7.14         154           8         ECM.Nelder.Mead         Kmeans scale         0.837         -7.10         218           27         ECM.Nelder.Mead         Random         0.510         -7.04         604           28         MCECM.Nelder.Mead         Random         0.509         -7.04         604           29         ECME.BFGS         Random         0.508         -7.06         545           30         ECME.Nelder.Mead         Random         0.507         -7.00         267	1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 5	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS ECM.CHART.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS ECM.BFGS MCECM.BFGS General Algorithm ECM.Nelder.Mead MCECM.Nelder.Mead MCECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead	General VCM           Initialization strategy           Hierarchical scale           Kmeans scale           Kmeans scale           Random           Random           Random           Random           Random           Random           Ritarchical scale           Hierarchical scale	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.851           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.878           0.878           0.875           0.872           0.869	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -6.98 -6.98 -6.99 -6.87 e.99 -6.87 millsed BIC string -7.08 -7.08 -7.08 -7.08 -7.11 -7.14	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 392 420 349 Computation time (Seconds) 162 214 147 128 135	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 5 6 7 8 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 29 30 20 29 30 20 29 30 20 29 30 20 20 20 30 20 20 30 20 2	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Velder.Mead ECM.EBFGS ECME.Nelder.Mead ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.Melder.Mead ECME.Nelder.Mead ECME.Nelder.Mead ECME.Nelder.Mead ECME.Nelder.Mead ECME.Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale	IM         Classification           classification         rate           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.7984         0.685           0.685         0.685           0.685         0.685           Classification         rate           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.60 -6.61 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.44 -6.44 -6.44	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 134 134 134 134 134 134 134 0.009 0.009 0.005 0.125 0.073
8         ECM.Nelder.Mead         Kmeans scale         0.837         -7.10         218           27         ECM.Nelder.Mead         Random         0.510         -7.04         604           28         MCECM.Nelder.Mead         Random         0.509         -7.04         517           29         ECME.BFGS         Random         0.508         -7.06         545           30         ECME.Nelder.Mead         Random         0.507         -7.00         267	1           2           3           4           5           6           7           8           27           28           29           30           1           2           30           1           2           30           1           2           3           4           5           6	Algorithm ECM.BFGS CCME.BFGS CCME.BFGS CCME.BFGS CCME.Nelder.Mead ECM.Nelder.Mead CECM.Nelder.Mead ECM.Nelder.Mead ECME.BFGS ECM.NEGS CCM.BFGS CCM.BFGS CCM_BFGS CM_BFGS CM_BFGS CCM_BFGS CM_BFGS	General VCM Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.490           0.878           0.875           0.872           0.869           0.866	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -6.98 -6.90 -6.87 -6.90 -6.87 ef. Normalised BIC -7.08 -7.08 -7.08 -7.08 -7.08 -7.08 -7.11 -7.12	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 255 255 255 255 255 255 255	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 5 6 6 7 8 27 28 29 30 6 7 8 29 30 6 7 8 29 30 6 7 8 29 30 6 7 8 29 30 6 7 8 29 30 6 7 8 29 30 7 8 8 8 8 8 9 30 7 8 8 8 8 8 9 30 7 8 8 8 8 8 8 9 30 7 8 8 8 8 8 8 8 8 8	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Eder.Mead ECM.EGS MCECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS ECME.BFGS ECME.BFGS ECME.Nelder.Mead ECM.Eder.Mead ECM.Eder.Mead ECM.Eder.Mead ECM.Eder.Mead ECM.Eder.Mead ECM.Eder.Mead ECM.ECM MCECM	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans K	IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.795         0.797           0.784         0.685           0.685         0.685           0.685         0.685           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.781         0.781	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.61 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.44 -6.44 -6.44 -6.44 -6.44	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 134 143 143 Computation time (Seconds) 0.039 0.060 0.095 0.125 0.073 0.125
27         ECM.Nelder.Mead         Random         0.510         -7.04         604           28         MCECM.Nelder.Mead         Random         0.509         -7.04         517           29         ECME.BFGS         Random         0.508         -7.06         545           30         ECME.Nelder.Mead         Random         0.507         -7.00         267	1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 5 6 7	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECME.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECME.Nelder.Mead MCECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS MCECM.Nelder.Mead	VCMM (Copula with Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kmeans de kmeans Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Random Ran	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499           0.878           0.875           0.875           0.872           0.866           0.839	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -6.98 -6.90 -6.87 er] Normalised BIC -7.08 -7.08 -7.08 -7.11 -7.14	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 392 420 349 Computation time (Seconds) 162 214 147 128 135 143 154	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 5 6 8 29 30 8 29 30 8 29 30 8 29 30 8 29 30 8 29 30 8 8 29 30 8 8 29 30 8 8 29 30 8 8 8 29 30 8 8 8 8 8 8 8 8 8	Algorithm MCECM.BFGS ECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECM.BFGS MCECM.BFGS MCECM.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECM.EARCA ECM MCECM	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale GMM Initialization strategy Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans scale	IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.7984         0.685           0.685         0.685           0.685         0.685           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.781         0.781	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.24 -6.44 -6.44 -6.44 -6.44 -6.44	Computation time (Seconds) 84 85 97 102 122 153 86 92 92 118 124 134 143 Computation time (Seconds) 0.039 0.060 0.095 0.125 0.073 0.125
28         MCECM.Nelder.Mead         Random         0.509         -7.04         517           29         ECME.BFGS         Random         0.508         -7.06         545           30         ECME.Nelder.Mead         Random         0.507         -7.00         267	1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 5 6 7 8	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS ECME.Meder.Mead ECME.Nelder.Mead ECME.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS ECM.Nelder.Mead Algorithm ECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead MCECM.BFGS ECM.Nelder.Mead MCECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS	VCMM (Copula with Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Word (Copula with Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499           0.499           0.878           0.878           0.878           0.872           0.866           0.839           0.837	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.20 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.20 -7.21 -7.20 -7.21 -7.20 -7.21 -7.20 -7.20 -7.21 -7.20 -7.20 -7.21 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.20 -7.08 -7.08 -7.10 -7.11 -7.11 -7.11 -7.11 -7.11 -7.11 -7.11 -7.11 -7.11 -7.11 -7.11 -7.11 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11	Computation time (Seconds) 74 91 108 111 112 165 55 255 392 420 349 Computation time (Seconds) 162 214 147 128 135 143 135 143 143 154 218	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 5 6 7 8 29 30 8 8 29 30 8 8 8 29 30 8 8 8 8 8 8 8 8 8	Algorithm MCECM.BFGS ECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS ECME.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.E.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead Algorithm	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale GMM Initialization strategy Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale	IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.7984         0.685           0.685         0.685           0.685         0.685           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.781         0.781	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26 BIC -6.44 -6.44 -6.44 -6.44 -6.44	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 134 134 143 Computation time (Seconds) 0.039 0.060 0.095 0.125 0.073 0.125
29         ECME.BFGS         Random         0.508         -7.06         545           30         ECME.Nelder.Mead         Random         0.507         -7.00         267	1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 5 6 7 8 27	Algorithm ECM.BFGS CCME.BFGS CCME.BFGS CCME.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS ECM.BFGS CCM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead	VCMM (Copula with Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Random Random Random Random Random Random Random Random Random Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.851           0.851           0.499           0.499           0.499           0.499           0.499           0.878           0.878           0.875           0.872           0.837           0.510	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -6.98 -6.98 -6.99 -6.87 mormalised BIC -7.08 -7.08 -7.08 -7.08 -7.08 -7.08 -7.08 -7.08 -7.14 -7.14 -7.14 -7.14 -7.14 -7.10 -7.04	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 392 420 349 Computation time (Seconds) 162 214 147 128 135 143 154 218 604	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 5 6 6 7 8 29 30 8 1 2 3 4 5 6 7 8 29 30 8 8 29 30 8 8 8 8 8 8 8 9 9 9	Algorithm MCECM.BFGS ECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.EBFGS ECME.Nelder.Mead ECM.BFGS MCECM.Nelder.Mead ECM.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.Medder.Mead ECME.Nelder.Mead ECME.Nelder.Mead ECME.Nelder.Mead ECME.Nelder.Mead ECME.Nelder.Mead	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Strategy Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Hierarchical Kmeans scale	IM         Classification           classification         rate           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.784         0.685           0.685         0.685           Classification         rate           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.781         0.781	Normalised BIC -6.60 -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.24 -6.44 -6.44 -6.44 -6.44	Computation time (Seconds) 84 85 97 102 122 153 86 92 118 124 134 143 143 Computation time (Seconds) 0.039 0.060 0.095 0.125 0.073 0.125
30 ECME.Nelder.Mead Random 0.507 -7.00 267	1           2           3           4           5           6           7           8           27           28           29           30           1           2           30           1           2           30           1           2           3           4           5           6           7           8           27           28	Algorithm ECM.BFGS COM.BFGS ECM.BFGS COM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead	Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kmeans de kmeans Random Random Random Random Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.499           0.878           0.875           0.872           0.866           0.837           0.510           0.509	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.20 -7.21 -7.20 -7.21 -7.20 -7.08 -7.08 -7.08 -7.08 -7.11 -7.12 -7.11 -7.12 -7.08 -7.08 -7.11 -7.12 -7.11 -7.12 -7.12 -7.12 -7.12 -7.12 -7.08 -7.08 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.11 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.12 -7.14 -7.10 -7.04 -7.04 -7.04 -7.04	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 255 255 255 255 255 255 255	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 5 6 6 7 8 29 30 8 8 29 30 8 8 8 29 30 8 8 8 8 8 8 8 8 8	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Eder.Mead ECME.BFGS ECME.BFGS MCECM.BFGS MCECM.BFGS MCECM.BFGS ECME.BFGS ECME.BFGS ECME.Nelder.Mead ECM.EdFGS ECME.Nelder.Mead ECM.EdEr.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.CME.BFGS ECME.Nelder.Mead Algorithm	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Km	IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.795         0.685           0.685         0.685           0.685         0.685           0.685         0.685           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.781         0.781	Normalised BIC -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 -6.44 -6.44 -6.44 -6.44 -6.44 -6.44	Computation time (Seconds) 84 85 97 102 122 153 86 92 92 118 124 134 143 143 Computation time (Seconds) 0.039 0.060 0.095 0.125 0.073 0.125
	1           2           3           4           5           6           7           8           27           28           29           30           Ranking           1           2           3           4           5           6           7           8           27           8           27           8           27           8           27           28           29	Algorithm ECM.BFGS MCECM.BFGS ECME.BFGS MCECM.Nelder.Mead ECME.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.BFGS ECM.BFGS MCECM.BFGS MCECM.Nelder.Mead ECME.Nelder.Mead ECME.Nelder.Mead MCECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.BFGS ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead ECM.Nelder.Mead	Initialization strategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans de scale Random Random Random Random Random Random Random Karategy Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Kmeans scale Kandom Random	M           Classification           rate           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.869           0.851           0.499           0.499           0.499           0.499           0.499           0.878           0.878           0.875           0.872           0.866           0.839           0.837           0.510           0.509	Normalised BIC -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.24 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.21 -7.20 -7.08 -7.08 -7.08 -7.11 -7.14 -7.14 -7.14 -7.14 -7.14 -7.14 -7.14 -7.14 -7.14 -7.14 -7.14 -7.14 -7.14 -7.14 -7.14 -7.14 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.04 -7.06 -7.04 -7.06 -7.04 -7.04 -7.06 -7.04 -7.06 -7.06 -7.04 -7.06 -7.04 -7.06 -7.04 -7.06 -7.04 -7.06 -7.04 -7.06 -7.04 -7.06 -7.04 -7.06 -7.04 -7.06 -7.04 -7.06 -7.06 -7.04 -7.06 -7.04 -7.06 -7.06 -7.04 -7.06 -7.06 -7.04 -7.06 -7.06 -7.06 -7.04 -7.06 -7.06 -7.06 -7.06 -7.04 -7.06 -7.06 -7.06 -7.04 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.06 -7.	Computation time (Seconds) 74 91 108 111 112 165 52 55 255 392 420 349 Computation time (Seconds) 162 214 147 128 135 143 154 154 218	Ranking 1 2 3 4 5 6 7 8 27 28 29 30 Ranking 1 2 3 4 5 6 8 29 30 Ranking 1 2 3 4 2 2 3 4 2 2 3 3 4 2 2 3 3 4 2 2 3 3 4 2 2 3 3 4 2 2 3 3 4 2 2 3 3 4 5 6 8 2 2 3 3 4 5 6 8 2 9 30 8 8 8 8 8 8 8 8 8	Algorithm MCECM.BFGS ECM.BFGS MCECM.Nelder.Mead ECM.Eder.Mead ECME.BFGS ECME.Nelder.Mead ECM.BFGS MCECM.BFGS MCECM.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead ECME.BFGS ECME.Nelder.Mead Algorithm ECM MCECM ECM MCECM ECM MCECM	Gaussian VCM Initialization strategy Kmeans scale Kmeans scale Kmeans scale Kmeans scale Kmeans scale Hierarchical scale Hierarchical scale Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Kmeans Stategy Hierarchical scale Hierarchical scale Kmeans scale Kmeans scale Hierarchical Kmeans scale	IM         Classification           rate         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.797         0.797           0.7984         0.685           0.685         0.685           0.685         0.685           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.784           0.784         0.781           0.781         0.781	Normalised BIC -6.60 -6.60 -6.60 -6.61 -6.61 -6.26 -6.26 -6.26 -6.26 -6.26 -6.26 Mormalised BIC -6.44 -6.44 -6.44 -6.44 -6.44	Computation time (Seconds) 84 85 97 102 122 153 86 92 92 118 124 124 124 143 Computation time (Seconds) 0.039 0.060 0.095 0.125 0.073 0.125

## 5. The performance of the selected models in the flow chart 3.3.20

In the flow chart 3.3.20, we have came up with a way to select the suitable initialisation strategies for data clustering without characteristics information for general VCMM algorithm. We try to apply the method there to see how the performances for the real data set AIS and BCW are.

The real data set is not three dimensional and we haven't defined what "X" shape in higher dimensional data is, so we select the path that the data has no "X" shape. Then, for general VCMM algorithm, we should compare the BIC value of Hierarchical and Kmeans (scale) with ECM.BFGS or MCECM.BFGS and select the lower BIC. In order to save time, we don't try both and just take ECM.BFGS instead. Actually, from our simulation reuslt, the times needed for ECM.BFGS and MCECM.BFGS are quite similar.

According to the Table 4.3.1, the initialisation strategies with lower BIC for ECM.BFGS is Hierarchical and Kmeans (scale) for AIS and BCW respectively. The performances of the selected model and the best ranking model are shown in the Table 4.3.5. Although the classification rate of the selected model is not as good as the best ranking model, the difference of the classification rate is just less than 2% for both real data set AIS and BCW. Therefore, we think that by the flow chart proposed in the Figure 3.3.20, we still can choose a suitable initialisation strategy and EM algorithm for data clustering with general VCMM algorithm.

Table 4.3.5: The performances of the selected model and the best ranking model for general VCMM. *The model is selected by the lower BIC value, so the total computation time is equal to the summation of the computation time of the two models.

	<u>A</u>	IS
	Selected model	Best ranking model
Initialisation strategy	Hierarchical	Hierarchical (scale)
EM algorithm	ECM.BFGS	MCECM.BFGS
Total computation time (seconds)	$149 (+219) = 368^*$	277
Normalised BIC	24.21	23.92
Classification rate	0.914	0.924
	BC	<u>CW</u>
	BC Selected model	<b>CW</b> Best ranking model
Initialisation strategy	BC Selected model Kmeans (scale)	<b>CW</b> Best ranking model Hierarchical (scale)
Initialisation strategy EM algorithm	BC Selected model Kmeans (scale) ECM.BFGS	<b>CW</b> Best ranking model Hierarchical (scale) ECM.BFGS
Initialisation strategy EM algorithm Total computation time (seconds)	BC Selected model Kmeans (scale) ECM.BFGS 61 (+232) = 293*	<b>CW</b> Best ranking model Hierarchical (scale) ECM.BFGS 74
Initialisation strategy EM algorithm Total computation time (seconds) Normalised BIC	$\begin{array}{r} \underline{BC} \\ \hline \\ Selected model \\ \hline \\ Kmeans (scale) \\ ECM.BFGS \\ \hline 61 (+232) = 293^* \\ -7.21 \end{array}$	<b>CW</b> Best ranking model Hierarchical (scale) ECM.BFGS 74 -7.24

# 5 Conclusion

In this master thesis, we aim at assessing the performance of the iterative algorithm - Gaussian mixture model (GMM) algorithm and vine copula mixture model (VCMM) algorithm for clustering simulated data with different characteristics and real data. In particular, we focus on the influence on the performance by using different versions of EM algorithms for parameter estimation of the GMM and VCMM algorithms with various initialisation strategies and optimization.

The simulation studies show that a particular initialisation strategy, using different expectation and maximization (EM) algorithms for parameter estimation of VCMM doesn't impact the classification rate significantly, but this is not the case for GMM algorithm. For different sets of Gaussian simulated data, GMM algorithm with maximisation either (ECME) algorithm and the optimization method Nelder-Mead perform mostly the best with regard to classification rate. Using the best and the worst EM algorithm in GMM algorithm can lead to almost 30% difference in classification rate. However, the difference is much smaller in VCMM algorithm and it is just from 0% to 5%.

Although there is no significant effect on the performance regarding to classification rate for VCMM with different EM algorithms and optimization, it does affect the computation time. We found that generally, heuristic based optimization method (Nelder-Maad) takes more time than with gradient based optimization method (BFGS). Also, different initialisation strategies do have the strong effect on clustering accuracy. We have found that there is usually a strategy outperforming others on the classification rate for simulated data with different characteristics and we can summarise it as a flow chart, for example, the best strategy for clustering data with the clusters like a "X" is "Random", for data with large volume difference of clusters is standardization first and so on. However, the flow chart cannot be used directly for data clustering without characteristics information. So, we proposed a way to select a the initialization strategy by comparing the BIC value of two suggested strategies and the one with lower BIC is selected.

Lastly, we have tried the selected model for clustering two sets of real data with general VCMM algorithm and the clustering accuracy is almost as good as the best ranking model where the difference of the classification rate is just less than 2%.

## A Appendices

## A Appendix for Section 3.2 GMM Algorithm

## A.1 Box plots of the classification rate



Figure A.1: Box plots of the **classification rate** for the EM algorithms for different initialization strategies over 50 replications with n = 500 generated from different types of the Gaussian clustering models





Figure A.2: Box plots of the **normalised BIC** of the EM algorithms for different initialization strategies over 50 replications with n = 500 generated from different types of the Gaussian clustering models

### A.3 Box plots of the computation time



Figure A.3: Box plots of the total omputation time of the EM algorithms for different initialization strategies over 50 replications with n = 500 generated from different types of the Gaussian clustering models

## B Appendix for Section 3.3 VCMM Algorithm

## B.1 Abbreviation for marginal distributions

 $\mathcal{N}(\mu, \sigma)$ : normal distribution with mean  $\mu$  and standard deviation  $\sigma$ .

 $exp(\lambda)$ : exponential distribution with rate parameter  $\lambda$ .

 $lnorm(\mu, \sigma)$ : log-normal distribution with mean  $\mu$  and standard deviation  $\sigma$  on the logarithmic scale.

logis(l, s): logistic distribution with location parameter l and scale parameter s.

 $llogis(\alpha, \beta)$ : log-logistic distribution with shape parameter  $\alpha$  and scale parameter  $\beta$ .

 $\Gamma(\alpha,\beta)$ : gamma distribution with shape parameter  $\alpha$  and rate parameter  $\beta$ .

 $t_3(\mu, \sigma)$ : Student's t distribution with mean parameter  $\mu$ , standard deviation parameter  $\sigma$ , and degrees of freedom 3.

 $snorm(\mu, \sigma, \lambda)$ : Skewed normal distribution with with location parameter  $\mu$ , scale parameter  $\sigma$ , skewness parameter  $\lambda$ .

 $sstd(\mu, \sigma, \alpha, \lambda)$ : Skewed student's t distribution with location parameter  $\mu$ , scale parameter  $\sigma$ , shape parameter  $\alpha$  (Degrees of freedom), skewness parameter  $\lambda$ .

## B.2 Abbreviation for copula families

Table B.1: The abbreviation for copula families with different degrees of rotation.

Copula	Degrees of rotation (Anticlockwise)								
Copula	0°	90°	180°	$270^{\circ}$					
Copulas with symmetric tail									
Gaussian	N	-	-	-					
Student t	Т	-	-	-					
Frank	F	-	-	-					
Copulas with asymmetric tail									
Clayton	С	R90C	R180C	R270C					
Gumbel	G	R90G	R180G	R270G					
Joe	J	R90J	R180J	R270J					
BB1	BB1	R90BB1	R180BB1	R270BB1					
BB6	BB6	R90BB6	R180BB6	R270BB6					
BB8	BB8	R90BB8	R180BB8	R270BB8					

## B.3 Box plots of the classification rate



Figure B.1: Box plots of the **classification rate** for the EM algorithms using different initialization strategies for 50 replications from settings 1 and 2.



Figure B.2: Box plots of the **classification rate** for the EM algorithms using different initialization strategies for 50 replications from settings 3.



Figure B.3: Box plots of the **classification rate** for the EM algorithms using different initialization strategies for 50 replications from settings 4 and 5.


Figure B.4: Box plots of the **classification rate** for the EM algorithms using different initialization strategies for 50 replications from settings 6 and 7.



Figure B.5: Box plots of the **classification rate** for the EM algorithms using different initialization strategies for 50 replications from settings 7, 8 and 9.



Figure B.6: Box plots of the **normalised BIC** for the EM algorithms using different initialization strategies for 50 replications from settings 1 and 2.

13.0





Figure B.7: Box plots of the **normalised BIC** for the EM algorithms using different initialization strategies for 50 replications from settings 3.



Figure B.8: Box plots of the **normalised BIC** for the EM algorithms using different initialization strategies for 50 replications from settings 4 and 5.



Figure B.9: Box plots of the **normalised BIC** for the EM algorithms using different initialization strategies for 50 replications from settings 6 and 7.



Figure B.10: Box plots of the **normalised BIC** for the EM algorithms using different initialization strategies for 50 replications from settings 7, 8 and 9.

### B.5 Box plots of the computation time



Figure B.11: Box plots of the **computation time** for the EM algorithms using different initialization strategies for 50 replications from settings 1 and 2.



Figure B.12: Box plots of the **computation time** for the EM algorithms using different initialization strategies for 50 replications from settings 3.



Figure B.13: Box plots of the **computation time** for the EM algorithms using different initialization strategies for 50 replications from settings 4 and 5.



Figure B.14: Box plots of the **computation time** for the EM algorithms using different initialization strategies for 50 replications from settings 6 and 7.



Figure B.15: Box plots of the **computation time** for the EM algorithms using different initialization strategies for 50 replications from settings 7, 8 and 9.

Table B.2: The ranking of the performance of EM algorithms and initialization strategies for VCMM algorithm, where the algorithm with 1) higher classification rate, 2) lower normalised BIC and 3) shorter computation time is considered to be the better performance and the priority is given by (1) > (2) > (3).

			General VC!	MM					Gaussian VC	MM		
Setting	Ranking	Algorithm	Initialization	Classification	Normalised	Computation	Ranking	Algorithm	Initialization	Classification	Normalised	Computation
(Sample size)	_	-	strategy	rate	BIC	time (Seconds)	-	_	strategy	rate	BIC	time (Seconds)
(	1	ECME.BFGS	Hierarchical	0.944	16.54	117	1	ECME.Nelder.Mead	Hierarchical	0.730	19.70	67
	2	MCECM.BFGS	Hierarchical	0.943	16.57	83	2	ECME.BFGS	Hierarchical	0.725	19.70	54
	3	ECMBEGS	Hierarchical	0.942	16.57	84	3	ECM BEGS	Hierarchical	0.724	19.74	67
	4	ECME.Nelder.Mead	Hierarchical	0.940	16.57	154	4	ECM Nelder Mead	Hierarchical	0.724	19.74	77
	5	MCECM Nelder Mead	Hierarchical	0.937	16.59	95	5	MCECM BEGS	Hierarchical	0.723	19.74	65
	6	ECM Nelder Mead	Hierarchical	0.934	16.59	02	6	MCECM Nelder Mead	Hierarchical	0.723	10.71	77
	7	ECM BECS	Bandom	0.885	16.70	186	7	FCME BECS	Bandom	0.620	10.07	73
		MCECM RECS	Random	0.880	16.74	167		ECME Noldor Mond	Higrarghigal coolo	0.617	10.70	66
	ů	ECME Noldor Mond	Random	0.870	16.74	260	0	ECME DECS	Hierarchical scale	0.615	10.91	55
	10	ECME DECE	Dendem	0.875	10.74	209	10	MCECM DECC	Hierarchical scale	0.013	10.82	79
	10	ECM Noldor Mood	Random	0.873	16.74	217	10	MCECM Nolder Mood	Hierarchical scale	0.612	10.82	13
	10	MCECM Noldor Mood	Random	0.872	16.70	217	10	ECM DECS	Hierarchical scale	0.612	10.82	79
	12	MCECM DECC	Handom	0.808	17.70	210	12	ECM Nelder Meed	Hierarchical scale	0.012	10.82	12
	10	MCECM.BrG5	Hierarchical scale	0.649	17.70	10	10	ECMENTINE M	nierarcinicai scale	0.011	19.82	80
	14	ECME.BFGS	Hierarchical scale	0.647	17.69	96	14	ECME.Neider.Mead	Kmeans	0.606	19.93	85
1 (500)	15	ECM.BFGS	Hierarchical scale	0.647	17.70	11	10	ECME.Neider.Mead	Kandom	0.604	19.92	87
	10	ECM.Neider.Mead	Hierarchical scale	0.637	17.74	97	10	ECME.BFGS	Kmeans	0.603	19.94	67
	17	MCECM.Neider.Mead	Hierarchical scale	0.634	17.76	103	11	ECM.Neider.Mead	Kmeans	0.599	19.95	110
	18	ECME.Nelder.Mead	Hierarchical scale	0.628	17.77	145	18	MCECM.Nelder.Mead	Kmeans	0.598	19.94	106
	19	MCECM.Nelder.Mead	Kmeans scale	0.611	17.98	93	19	ECM.BFGS	Kmeans	0.598	19.95	92
	20	MCECM.BFGS	Kmeans scale	0.610	17.98	71	20	MCECM.BFGS	Kmeans	0.597	19.94	91
	21	ECM.BFGS	Kmeans scale	0.610	17.99	73	21	ECM.BFGS	Random	0.594	20.00	134
	22	ECM.Nelder.Mead	Kmeans scale	0.610	17.99	95	22	ECME.BFGS	Kmeans scale	0.593	19.90	59
	23	ECME.Nelder.Mead	Kmeans	0.602	17.97	139	23	ECME.Nelder.Mead	Kmeans scale	0.593	19.90	69
	24	ECME.Nelder.Mead	Kmeans scale	0.601	18.01	110	24	ECM.Nelder.Mead	Random	0.592	20.03	144
	25	ECME.BFGS	Kmeans scale	0.600	18.02	69	25	MCECM.BFGS	Kmeans scale	0.590	19.90	91
	26	ECM.Nelder.Mead	Kmeans	0.597	17.98	124	26	MCECM.Nelder.Mead	Kmeans scale	0.590	19.90	99
	27	MCECM.Nelder.Mead	Kmeans	0.597	17.98	124	27	ECM.Nelder.Mead	Kmeans scale	0.589	19.90	102
	28	MCECM.BFGS	Kmeans	0.596	17.99	94	28	ECM.BFGS	Kmeans scale	0.589	19.91	89
	29	ECM.BFGS	Kmeans	0.596	17.99	97	29	MCECM.BFGS	Random	0.588	20.01	137
	30	ECME.BFGS	Kmeans	0.584	18.02	85	30	MCECM.Nelder.Mead	Random	0.587	20.01	147
	1	MCECM.BFGS	Random	0.935	16.48	285	1	ECM.BFGS	Hierarchical	0.736	19.68	112
	2	ECME.BFGS	Random	0.932	16.49	307	2	MCECM.BFGS	Hierarchical	0.736	19.68	115
	3	ECM.Nelder.Mead	Random	0.931	16.50	324	3	ECME.BFGS	Hierarchical	0.735	19.66	88
	4	ECME.Nelder.Mead	Random	0.930	16.49	370	4	ECME.Nelder.Mead	Hierarchical	0.735	19.66	96
	5	ECM.BFGS	Random	0.930	16.50	285	5	MCECM.Nelder.Mead	Hierarchical	0.735	19.66	126
	6	MCECM.BFGS	Hierarchical	0.927	16.48	112	6	ECM.Nelder.Mead	Hierarchical	0.735	19.66	131
	7	ECM.BFGS	Hierarchical	0.927	16.48	113	7	ECME.Nelder.Mead	Kmeans	0.605	19.88	124
	8	ECME.BFGS	Hierarchical	0.927	16.48	163	8	ECME.BFGS	Random	0.602	19.92	107
	9	MCECM.Nelder.Mead	Hierarchical	0.924	16.49	128	9	ECM.BFGS	Kmeans	0.601	19.88	164
	10	ECM.Nelder.Mead	Hierarchical	0.924	16.50	130	10	MCECM.Nelder.Mead	Kmeans	0.601	19.88	177
	11	ECME.Nelder.Mead	Hierarchical	0.923	16.51	183	11	ECM.Nelder.Mead	Kmeans	0.601	19.88	186
	12	MCECM.Nelder.Mead	Random	0.923	16.53	296	12	MCECM.BFGS	Kmeans	0.601	19.90	159
	13	ECM.Nelder.Mead	Kmeans	0.778	17.22	453	13	ECME.BFGS	Kmeans	0.601	19.92	105
	14	MCECM.Nelder.Mead	Kmeans	0.778	17.22	456	14	ECME.Nelder.Mead	Random	0.596	19.87	120
1 (1000)	15	MCECM.BFGS	Kmeans	0.767	17.28	341	15	ECME.Nelder.Mead	Kmeans scale	0.595	19.86	110
1 (1000)	16	ECM.BFGS	Kmeans	0.765	17.28	347	16	ECME.BFGS	Kmeans scale	0.592	19.89	98
	17	ECME.BFGS	Kmeans	0.709	17.51	298	17	ECME.Nelder.Mead	Hierarchical scale	0.590	19.83	106
	18	ECME.Nelder.Mead	Kmeans	0.709	17.53	416	18	ECM.Nelder.Mead	Kmeans scale	0.589	19.86	175
	19	MCECM.BFGS	Kmeans scale	0.644	17.72	160	19	ECME.BFGS	Hierarchical scale	0.588	19.86	95
	20	ECM.BFGS	Kmeans scale	0.644	17.72	164	20	MCECM.Nelder.Mead	Kmeans scale	0.588	19.86	163
	21	ECME.BFGS	Kmeans scale	0.640	17.76	165	21	MCECM.BFGS	Kmeans scale	0.587	19.88	153
	22	MCECM.BFGS	Hierarchical scale	0.634	17.70	158	22	ECM.Nelder.Mead	Hierarchical scale	0.586	19.83	159
	23	ECM.BFGS	Hierarchical scale	0.634	17.70	170	23	MCECM.Nelder.Mead	Hierarchical scale	0.586	19.84	153
	24	MCECM.Nelder.Mead	Hierarchical scale	0.631	17.71	190	24	ECM.BFGS	Kmeans scale	0.586	19.90	155
	25	ECM.Nelder.Mead	Hierarchical scale	0.630	17.71	201	25	MCECM.Nelder.Mead	Random	0.586	19.94	236
	26	ECM.Nelder.Mead	Kmeans scale	0.628	17.80	201	26	ECM.BFGS	Hierarchical scale	0.584	19.86	143
	27	ECME.Nelder.Mead	Kmeans scale	0.628	17.82	209	27	MCECM.BFGS	Random	0.584	19.96	217
	28	MCECM.Nelder.Mead	Kmeans scale	0.626	17.82	173	28	MCECM.BFGS	Hierarchical scale	0.583	19.88	138
	29	ECME.BFGS	Hierarchical scale	0.623	17.74	170	29	ECM.BFGS	Random	0.582	19.93	211
	30	ECME.Nelder.Mead	Hierarchical scale	0.619	17.78	218	30	ECM.Nelder.Mead	Random	0.582	19.93	235

94

78

95

146

113

108

143 129

148

53

55

59

68

55

57

58

63

70

72

78

80

67

75

70

82

70

73

73

79

83

93

96

108

83

83

85

98

81

90

0.621

0.617

 $0.616 \\ 0.546$ 

0.544

0.532

0.532

0.519

0.519

0.865

0.865

0.857

0.857

0.824

0.824

0.824

0.824

0.824

0.824

0.821

0.820

0.760

0.760

0.759

0.759

0.756

0.756

0.756

0.756

0.756

0.756

0.747

0.747

0.746

0.746

0.743

0.743

0.741

0.740

14.52

14.52

14.52

14.65

14.66

14.67

14.67

14.69

14.69

12.01

12.01

12.01

12.06

12.06

12.06

12.06

12.06

12.06

12.06

12.06

12.14

12.14

12.14

12.14

12.14

12.14

12.14

12.14

12.14

12.14

12.15

12.15

12.15

12.15

12.16

12.16

12.16

12.16

Kmeans scale

Kmeans scale

Kmeans scale

Random

Hierarchical

Hierarchical

Hierarchical

Hierarchical

Hierarchical

Hierarchical

Random

Random

Hierarchical scale

Hierarchical scale

Hierarchical scale

Hierarchical scale

Kmeans

Kmeans

Kmeans

Kmeans

Hierarchical scale

Hierarchical scale

Kmeans scale

Kmeans scale

Kmeans

Kmeans

Kmeans scale

Kmeans scale

Kmeans scale

Kmeans scale

		General VC	MM					Gaussian VC	MM		
Ranking	Algorithm	Initialization	Classification	Normalised BIC	Computation time (Seconds)	Ranking	Algorithm	Initialization	Classification	Normalised	Computation
1	ECM Nelder Mead	Kmeane	0.894	13.15	70	1	FCM BEGS	Kmeans	0.811	14.24	82
2	MCECM Nelder Mead	Kmeans	0.894	13.15	73	2	MCECM BEGS	Kmeans	0.811	14.24	83
3	ECME BEGS	Kmeans	0.889	13.12	88	3	MCECM Nelder Mead	Kmeans	0.811	14.24	97
4	ECME Nelder Mead	Kmeans	0.889	13.13	102	4	ECM.Nelder.Mead	Kmeans	0.811	14.24	103
5	ECM.BEGS	Kmeans	0.889	13.16	61	5	ECME Nelder Mead	Kmeans	0.808	14.23	99
6	MCECM.BFGS	Kmeans	0.889	13.16	61	6	ECME.BFGS	Kmeans	0.807	14.23	84
7	MCECM.BFGS	Hierarchical	0.871	13.16	72	7	ECM.BFGS	Hierarchical scale	0.777	14.28	49
8	ECM.BFGS	Hierarchical	0.871	13.16	74	8	ECM.Nelder.Mead	Hierarchical scale	0.777	14.28	64
9	ECME.BFGS	Hierarchical	0.871	13.16	109	9	MCECM.BFGS	Hierarchical scale	0.776	14.28	50
10	ECME.Nelder.Mead	Hierarchical	0.870	13.14	129	10	MCECM.Nelder.Mead	Hierarchical scale	0.776	14.28	60
11	ECM.BFGS	Hierarchical scale	0.870	13.17	79	11	ECME.BFGS	Hierarchical scale	0.775	14.28	57
12	MCECM.BFGS	Hierarchical scale	0.870	13.17	79	12	ECME.Nelder.Mead	Hierarchical scale	0.775	14.28	64
13	ECME.Nelder.Mead	Kmeans scale	0.870	13.18	177	13	ECM.BFGS	Hierarchical	0.765	14.31	56
14	ECME.BFGS	Hierarchical scale	0.869	13.17	114	14	ECME.BFGS	Hierarchical	0.765	14.31	62
15	MCECM.Nelder.Mead	Hierarchical	0.866	13.16	72	15	ECM.Nelder.Mead	Hierarchical	0.765	14.31	77
16	ECM.Nelder.Mead	Hierarchical	0.865	13.17	71	16	MCECM.BFGS	Hierarchical	0.764	14.31	57
17	MCECM.Nelder.Mead	Hierarchical scale	0.863	13.18	83	17	MCECM.Nelder.Mead	Hierarchical	0.764	14.31	68
18	ECM.Nelder.Mead	Hierarchical scale	0.863	13.18	90	18	ECME.Nelder.Mead	Hierarchical	0.764	14.31	71
19	MCECM.BFGS	Kmeans scale	0.863	13.24	88	19	ECME.BFGS	Kmeans scale	0.621	14.50	89
20	ECM.BFGS	Kmeans scale	0.863	13.24	89	20	ECME.Nelder.Mead	Kmeans scale	0.621	14.50	103
21	ECME.BFGS	Kmeans scale	0.863	13.24	144	21	ECM.BFGS	Kmeans scale	0.621	14.52	74

107

110

137 180

98

157

88 130

85

104

111

114

121

118

82

89 78

75

80

82

88

93

78 77

93

94

67

64

66

62

62

66

64

67 67

68

73

22

23

 $\frac{24}{25}$ 

26 27

28 29

30

2

3

4

5

6

7

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

 24 

25

26

27

28

29

ECM.Nelder.Mead MCECM.BFGS

MCECM.Nelder.Mead MCECM.Nelder.Mead

MCECM.BFGS

ECM.BFGS

ECM.Nelder.Mead

ECME.BFGS

ECME Nelder Mead

MCECM.BFGS

MCECM.Nelder.Mead

ECM.BFGS

ECM.Nelder.Mead

MCECM.BFGS

MCECM.Nelder.Mead

ECM.BFGS

ECM.Nelder.Mead ECME.Nelder.Mead

ECME.BFGS

ECME.BFGS

ECME.Nelder.Mead

MCECM.BFGS

MCECM.Nelder.Mead

ECM.BFGS

ECM.Nelder.Mead MCECM.BFGS

ECM.BFGS

MCECM.Nelder.Mead

ECM.Nelder.Mead

ECME.BFGS

ECME.Nelder.Mead

ECME.BFGS

ECME.Nelder.Mead

ECME.Nelder.Mead

ECME.BFGS

ECM.BFGS

ECM.Nelder.Mead

MCECM.BFGS

ICECM.Nelder.Me

Setting

(Sample size)

2(500)

22

23

24

25

26

27

28 29

30

2

3

4

 $\mathbf{5}$ 

6

7

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

3 (500)

MCECM.Nelder.Mead

ECM.Nelder.Mead

ECME.Nelder.Mead

ECME.Nelder.Mead

ECME BEGS

ECM.Nelder.Mead

ECM.BFGS MCECM.Nelder.Mead

MCECM.BFGS

MCECM.Nelder.Mea

MCECM.BFGS

ECM.Nelder.Mead

ECM BEGS

ECME.BFGS

ECME.Nelder.Mead

ECME.BFGS

ECME.Nelder.Mead ECM.Nelder.Mead

MCECM.Nelder.Mead

MCECM.BFGS

ECM.BFGS

MCECM.Nelder.Mead

ECM.Nelder.Mead

ECME.Nelder.Mead

ECME.BFGS

MCECM.BFGS

ECM.BFGS

MCECM.Nelder.Mead

MCECM.BFGS

ECM.Nelder.Mead

ECM.BFGS

ECME.BFGS

ECME.Nelder.Mead

MCECM.Nelder.Mead

ECM.Nelder.Mead

ECM.BFGS

MCECM.BFGS

ECME.BFGS

ECME.Nelder.Mead

0.862

0.861

0.856

0.659

0.645

0.627

0.625

0.621

0.619

0.835

0.821

0.821

0.799

0.792

0.680

0.678

0.678

0.677

0.677

0.677

0.575

0.575

0.573

0.572

0.571

0.570

0.563

0.560

0.560

0.558

0.557

0.557

0.537

0.537

0.537

0.537

0.536

0.535

13.24

13.24

13.19

13.78

13.88

13.96

13.97

13.98

13.98

12.08

12.08

12.10

12.10

12.12

12.13

12.29

12.29

12.30

12.30

12.30

12.30

12.39

12.39

12.38

12.38

12.39

12.40

12.37

12.38

12.38

12.38

12.38

12.38

12.41

12.41

12.41

12.41

12.40

12.40

Kmeans scale

Kmeans scale

Hierarchical scal

Random

Hierarchical

Hierarchical

Hierarchical

Hierarchical

Hierarchical

Hierarchical

Kmeans

Kmeans

Kmeans

Kmeans

Kmeans

Kmeans

Hierarchical scale

Hierarchical scale

Hierarchical scale

Hierarchical scale

Hierarchical scale

Hierarchical scale

Kmeans scale

Kmeans scale

Kmeans scale

Kmeans scale

Kmeans scale

Kmeans scale

			General VC	JM					Caussian VC	MM		
C-441	Denking	A1	Tuiti-limition	Classification	Name	Commutation	Development	A 1i+1	Gaussian ve		N	Commutation
(Complemine)	Ranking	Algorithm	mitialization	Classification	DIC	Computation	Ranking	Algorithm	muanzation	Classification	DIC	Computation
(Sample size)	1	MODOM N 11 M 1	strategy	rate	10.01	time (Seconds)	1	MORGNERROG	strategy	rate	11.05	time (Seconds)
		MCECM.Neider.Mead	Random	0.851	12.01	171	1	MCECM Nelder Mer d	Random	0.877	11.95	76
	2	MCECM.BrG5	Random	0.851	12.01	175	2	ECM N 11 M	Random	0.877	11.95	
	3	ECM.Neider.Mead	Random	0.850	12.01	191	3	ECM.Neider.Mead	Random	0.877	11.95	87
	4	ECM.BFGS	Random	0.850	12.01	194	4	ECM.BFGS	Random	0.876	11.95	87
	5	ECME.Neider.Mead	Random	0.816	12.04	185	6	MCECM.Neider.Mead	Hierarchical	0.844	12.00	95
	0	ECME.BFGS	Random	0.816	12.04	192	0	MCECM.BFGS	Hierarchical	0.844	12.00	95
		ECME.Neider.Mead	Hierarchical	0.714	12.17	147	( 0	ECM.Neider.Mead	Hierarchical	0.844	12.00	99
	8	ECME.BFGS	Hierarchical	0.714	12.17	148	8	ECM.BFGS	Hierarchical	0.844	12.00	99
	9	MCECM.Neider.Mead	Hierarchical	0.713	12.18	140	9	ECME.BFGS	Hierarchical	0.844	12.00	100
	10	MCECM.BrG5	Hierarchical	0.713	12.18	148	10	ECME.Neider.Mead	nierarchicai	0.844	12.00	101
	10	ECM DECS	Hierarchical	0.713	12.18	104	10	ECME Nelder Meed	Random	0.839	11.99	70
	12	ECM.BFGS	Hierarchical	0.713	12.18	155	12	ECME.Neider.Mead	Random	0.839	11.99	102
	10	ECME DECS	Hierarchical scale	0.595	12.20	150	1.0	MCECM Nelder Meed	Hierarchical scale	0.820	12.02	120
	14	ECME.DFG5	Hierarchical scale	0.595	12.20	157	14	MCECM.Neider.Mead	Hierarchical scale	0.825	12.02	125
3(1000)	10	MCECM.Neider.Mead	Hierarchical scale	0.588	12.27	152	10	ECME DECS	Hierarchical scale	0.825	12.02	120
	10	ECM RECS	Hierarchical scale	0.587	12.27	150	10	ECML.DFG5	Hierarchical scale	0.825	12.02	127
	19	ECM.br65	Hierarchical scale	0.587	12.27	151	19	ECM DECS	Hierarchical scale	0.825	12.02	132
	10	MCECM Nelder Meed	Hierarchical scale	0.587	12.28	155	10	ECM.DFG5	Hierarchical scale	0.825	12.02	132
	19	ECM DECS	Kineans	0.559	12.32	161	19	MCECM.Neider.Mead	Kineans	0.787	12.00	137
	20	MCECM PECS	Kmeans	0.559	12.32	161	20	ECM Noldor Mood	Kmeans	0.787	12.06	140
	21	FCM Nolder Mood	Kmeans	0.559	12.32	152	21	ECM DECS	Kmeans	0.787	12.00	1.45
	22	ECME Nolder Mood	Kmeans	0.558	12.32	100	22	MCECM Nolder Mood	Kmoong goolo	0.776	12.00	140
	20	ECME DECS	Kineans	0.550	12.31	149	2.0	MCECM DECC	Kineans scale	0.776	12.07	140
	24	ECME.brG5	Kineans Kmoong gaolo	0.550	12.31	142	24	ECM Noldor Mood	Kineans scale	0.779	12.07	150
	20	ECME DECS	Kmeens scale	0.530	12.30	119	20	ECM DECS	Kmeens seele	0.772	12.07	154
	20	ECME.DFG5	Kineans scale	0.529	12.30	122	20	ECME DECE	Kineans scale	0.772	12.07	104
	21	ECM DECS	Kineans scale	0.528	12.31	111	21	ECME Nelder Meed	Kineans	0.770	12.08	133
	20	ECM.BrG5	Kmeans scale	0.528	12.31	117	20	ECME.Neider.Mead	Kineans Kmoona aaalo	0.776	12.08	137
	30	MCECM BEGS	Kmeans scale	0.528	12.32	116	30	ECME BECS	Kmeans scale	0.768	12.00	132
	1	MCECM BEGS	Kmeane	0.826	10.79	42	1	ECM Nelder Mead	Kmeans	0.833	10.78	47
	2	MCECM Nelder Mead	Kmeans	0.826	10.79	44	2	MCECM Nelder Mead	Kmeans	0.831	10.78	42
	3	FCM BEGS	Kmeans	0.826	10.79	45	2	MCECM Nelder Mead	Kmeans scale	0.820	10.77	40
	4	ECM Nelder Mead	Kmeans	0.826	10.79	50	4	ECM Nelder Mead	Kmeans scale	0.829	10.77	51
	5	ECME BEGS	Kmeans	0.825	10.79	46	5	ECM BEGS	Kmeans	0.827	10.77	43
	6	ECME Nelder Mead	Kmeans	0.824	10.79	55	6	MCECM BEGS	Kmeans	0.826	10.77	41
	7	MCECM BEGS	Kmeans scale	0.805	10.78	42	7	ECME BEGS	Kmeans	0.826	10.77	48
	8	MCECM.Nelder.Mead	Kmeans scale	0.805	10.78	44	8	ECME Nelder Mead	Kmeans	0.826	10.77	60
	9	ECM Nelder Mead	Kmeans scale	0.805	10.78	50	9	MCECM BEGS	Kmeans scale	0.825	10.77	41
	10	ECM BEGS	Kmeans scale	0.804	10.79	42	10	ECM BEGS	Kmeans scale	0.825	10.77	47
	11	ECME BEGS	Kmeans scale	0.804	10.79	45	11	ECME BEGS	Kmeans scale	0.825	10.77	51
	12	ECME.Nelder.Mead	Kmeans scale	0.803	10.79	58	12	ECME Nelder Mead	Kmeans scale	0.824	10.77	51
	13	ECM BEGS	Hierarchical	0.797	10.81	42	13	MCECM Nelder Mead	Hierarchical	0.818	10.78	59
	14	ECME BEGS	Hierarchical	0.796	10.80	47	14	ECM Nelder Mead	Hierarchical	0.818	10.78	64
	15	MCECM Nelder Mead	Hierarchical	0.796	10.81	42	15	ECM BEGS	Hierarchical	0.814	10.78	57
4(500)	16	MCECM BEGS	Hierarchical	0.796	10.81	42	16	ECME.BEGS	Hierarchical	0.812	10.78	64
	17	ECM.Nelder.Mead	Hierarchical	0.796	10.81	50	17	MCECM.BFGS	Hierarchical	0.811	10.78	59
	18	ECME.Nelder.Mead	Hierarchical	0.796	10.81	58	18	ECME Nelder Mead	Hierarchical	0.811	10.78	76
	19	ECM.BFGS	Hierarchical scale	0.774	10.80	49	19	ECM.Nelder.Mead	Hierarchical scale	0.807	10.78	65
	20	MCECM.BFGS	Hierarchical scale	0.774	10.80	50	20	MCECM.BFGS	Hierarchical scale	0.803	10.78	56
	21	ECME.BFGS	Hierarchical scale	0.774	10.80	54	21	ECM.BFGS	Hierarchical scale	0.803	10.78	59
	22	MCECM.Nelder.Mead	Hierarchical scale	0.774	10.80	54	22	MCECM.Nelder.Mead	Hierarchical scale	0.803	10.78	65
	23	ECM.Nelder.Mead	Hierarchical scale	0.774	10.80	56	23	ECME.Nelder.Mead	Hierarchical scale	0.794	10.78	74
	24	ECME.Nelder.Mead	Hierarchical scale	0.772	10.80	66	24	ECME.BFGS	Hierarchical scale	0.793	10.78	74
	25	ECME.BFGS	Random	0.541	10.97	166	25	ECME.Nelder.Mead	Random	0.590	10.84	142
	26	ECME.Nelder.Mead	Random	0.538	10.97	177	26	ECME.BFGS	Random	0.590	10.84	146
	27	MCECM.BFGS	Random	0.535	10.97	155	27	ECM.BFGS	Random	0.578	10.84	116
	28	ECM.BFGS	Random	0.535	10.97	158	28	ECM.Nelder.Mead	Random	0.578	10.84	127
	29	ECM.Nelder.Mead	Random	0.535	10.97	161	29	MCECM.BFGS	Random	0.568	10.84	115
	30	MCECM.Nelder.Mead	Random	0.535	10.97	166	30	MCECM.Nelder.Mead	Random	0.568	10.84	124

			Conorol VC	-m					Coursion VC	MM		
Setting	Banking	Algorithm	Initialization	Classification	Normalized	Computation	Banking	Algorithm	Initialization	Classification	Normalized	Computation
(Sample size)	Ranking	Algorithm	strategy	rate	BIC	time (Seconds)	Ranking	mgormini	strategy	rate	BIC	time (Seconds)
(	1	ECM.BEGS	Kmeans	0.756	12.08	86	1	ECME Nelder Mead	Hierarchical scale	0.523	12.13	62
	2	MCECM.Nelder.Mead	Kmeans	0.756	12.08	99	2	ECME.BFGS	Hierarchical scale	0.523	12.13	72
	3	MCECM.BFGS	Kmeans	0.755	12.08	88	3	MCECM.Nelder.Mead	Hierarchical	0.522	12.10	50
	4	ECM.Nelder.Mead	Kmeans	0.755	12.08	103	4	ECM.BFGS	Hierarchical	0.522	12.10	53
	5	ECME.BFGS	Kmeans	0.753	12.08	100	5	ECM.Nelder.Mead	Hierarchical	0.522	12.10	54
	6	ECME.Nelder.Mead	Kmeans	0.753	12.08	113	6	MCECM.BFGS	Hierarchical	0.522	12.10	54
	7	MCECM.Nelder.Mead	Kmeans scale	0.747	12.08	87	7	ECME.Nelder.Mead	Hierarchical	0.522	12.11	60
	8	ECM.BFGS	Kmeans scale	0.746	12.08	77	8	ECME.BFGS	Hierarchical	0.522	12.11	70
	9	MCECM.BFGS	Kmeans scale	0.746	12.08	79	9	MCECM.Nelder.Mead	Kmeans scale	0.522	12.12	51
	10	ECM.Nelder.Mead	Kmeans scale	0.746	12.08	92	10	MCECM.BFGS	Kmeans scale	0.522	12.12	51
	11	ECME.BFGS	Kmeans scale	0.746	12.08	97	11	ECM.BFGS	Kmeans scale	0.522	12.12	52
	12	ECME.Nelder.Mead	Kmeans scale	0.746	12.08	111	12	ECM.BFGS	Kmeans	0.522	12.12	54
	13	MCECM.Nelder.Mead	Hierarchical scale	0.708	12.08	113	13	ECM.Nelder.Mead	Kmeans scale	0.522	12.12	54
	14	ECME.BFGS	Hierarchical scale	0.708	12.08	120	14	ECME.Nelder.Mead	Kmeans scale	0.522	12.12	58
5 (500)	15	ECM.BFGS	Hierarchical scale	0.707	12.08	105	15	ECME.Nelder.Mead	Kmeans	0.522	12.12	59
. ()	16	MCECM.BFGS	Hierarchical scale	0.707	12.09	103	16	ECME.BFGS	Kmeans scale	0.522	12.12	62
	17	ECM.Nelder.Mead	Hierarchical scale	0.705	12.08	112	17	ECME.BFGS	Kmeans	0.522	12.12	68
	18	ECME.Nelder.Mead	Hierarchical scale	0.705	12.09	127	18	MCECM.Nelder.Mead	Kmeans	0.521	12.12	52
	19	ECME.BFGS	Hierarchical	0.679	12.10	92	19	ECM.Nelder.Mead	Kmeans	0.521	12.12	55
	20	ECME.Nelder.Mead	Hierarchical	0.679	12.10	108	20	ECME.Nelder.Mead	Random	0.521	12.12	56
	21	ECM.BFGS	Hierarchical	0.678	12.10	80	21	MCECM.BFGS	Kmeans	0.521	12.12	50
	22	MCECM.BFGS	Hierarchical	0.677	12.10	80	22	MCECM.Neider.Mead	Hierarchical scale	0.521	12.12	50
	23	ECM Nelder Meed	Hierarchical	0.677	12.10	98	23	MCECM.BFGS	Hierarchical scale	0.521	12.12	50
	24	ECME DECS	Random	0.562	12.10	260	24	ECM.BrG5	Hierarchical scale	0.521	12.12	
	20	ECME Noldor Mond	Random	0.505	12.17	426	20	ECME DECS	Pandom	0.521	12.12	66
	20	ECM RECS	Random	0.557	12.17	430	20	ECM Nelder Mond	Random	0.521	12.12	42
	21	ECM Nelder Mead	Random	0.556	12.11	360	28	MCECM Nelder Mead	Bandom	0.520	12.12	38
	20	MCECM Nelder Mead	Random	0.545	12.10	324	20	MCECM BEGS	Bandom	0.520	12.13	38
	30	MCECM.BFGS	Random	0.542	12.17	272	30	ECM.BFGS	Random	0.520	12.13	42
	1	MCECM.Nelder.Mead	Kmeans	0.757	12.02	155	1	MCECM.BFGS	Kmeans	0.521	12.12	79
	2	MCECM.BFGS	Kmeans	0.756	12.01	133	2	MCECM.BFGS	Kmeans scale	0.521	12.12	79
	3	ECM.BFGS	Kmeans	0.756	12.02	138	3	MCECM.Nelder.Mead	Kmeans	0.521	12.12	83
	4	ECM.Nelder.Mead	Kmeans	0.756	12.02	159	4	ECME.Nelder.Mead	Kmeans	0.521	12.12	84
	5	ECME.BFGS	Kmeans	0.754	12.01	143	5	ECME.Nelder.Mead	Kmeans scale	0.521	12.12	84
	6	ECME.Nelder.Mead	Kmeans	0.754	12.01	166	6	MCECM.Nelder.Mead	Kmeans scale	0.521	12.12	84
	7	MCECM.BFGS	Kmeans scale	0.744	12.02	118	7	ECME.BFGS	Kmeans scale	0.521	12.12	89
	8	ECM.BFGS	Kmeans scale	0.744	12.02	123	8	ECME.BFGS	Kmeans	0.521	12.12	95
	9	ECM.Nelder.Mead	Kmeans scale	0.744	12.02	138	9	MCECM.BFGS	Random	0.520	12.12	64
	10	MCECM.Nelder.Mead	Kmeans scale	0.744	12.02	139	10	MCECM.Nelder.Mead	Random	0.520	12.12	69
	11	ECME.BFGS	Kmeans scale	0.742	12.01	144	11	ECM.BFGS	Random	0.520	12.12	72
	12	ECME.Nelder.Mead	Kmeans scale	0.742	12.02	162	12	ECM.Nelder.Mead	Random	0.520	12.12	73
	13	ECME.BFGS	Hierarchical scale	0.703	12.01	198	13	MCECM.BFGS	Hierarchical	0.520	12.12	77
	14	ECME.Nelder.Mead	Hierarchical scale	0.702	12.01	220	14	MCECM.Nelder.Mead	Hierarchical	0.520	12.12	80
5 (1000)	15	ECM.Nelder.Mead	Hierarchical scale	0.701	12.01	217	15	ECME.Nelder.Mead	Random	0.520	12.12	85
	10	MCECM.Neider.Mead	Hierarchical scale	0.701	12.01	217	10	ECM.BFGS	Hierarchical	0.520	12.12	80
	17	MCECM.BFGS	Hierarchical scale	0.700	12.01	197	10	ECM.Nelder.Mead	Kmeans	0.520	12.12	80
	10	ECMEN-Ider Mand	Hierarchical scale	0.700	12.01	200	10	ECM.Neider.Mead	Hierarchical	0.520	12.12	80
	20	ECM Nolder Mood	Hierarchical	0.664	12.05	213	20	ECM DECS	Kineans scale	0.520	12.12	80
	20	ECME BECS	Hierarchical	0.004	12.05	188	20	ECME Nelder Mood	Hierarchical	0.520	12.12	87
	21	MCECM BECS	Hierarchical	0.003	12.04	178	21	ECM Nelder Mood	Kmeane scale	0.520	12.12	87
	23	ECM BEGS	Hierarchical	0.662	12.05	187	23	MCECM BECS	Hierarchical scale	0.520	12.12	89
	24	MCECM.Nelder.Mead	Hierarchical	0.662	12.05	206	24	MCECM.Nelder.Mead	Hierarchical scale	0.520	12.12	94
	25	ECME BFGS	Random	0.626	12.08	636	25	ECME BFGS	Random	0.520	12.12	99
	26	ECME.Nelder.Mead	Random	0.624	12.08	751	26	ECME.BFGS	Hierarchical	0.520	12.12	99
	27	ECM.BFGS	Random	0.602	12.09	615	27	ECM.BFGS	Hierarchical scale	0.520	12.12	99
	28	ECM.Nelder.Mead	Random	0.601	12.09	748	28	ECM.Nelder.Mead	Hierarchical scale	0.520	12.12	100
	29	MCECM.Nelder.Mead	Random	0.574	12.10	689	29	ECME.Nelder.Mead	Hierarchical scale	0.520	12.12	102
	30	MCECM.BFGS	Random	0.573	12.10	523	30	ECME.BFGS	Hierarchical scale	0.520	12.12	111

Stup Indiang Agerian International states Computation Reaker Algerian International states Computational state   1 ECM.MPCS Kussus 0.87 3.32 40 1 ECM.Marks Kussus 0.070 14.35 21   3 ECM.Molex Madel Kussus 0.887 3.32 40 3 ECM.Molex Madel Kussus 0.690 14.35 22   3 ECM.Molex Madel Kussus 0.887 13.32 50 4 MCCM.Molex Madel Kussus 0.690 14.35 23   5 ECM.FERCE Kussus 0.887 13.32 7.0 6 ECM.Molex Madel Kussus 0.090 14.35 33   7 ECM.Molex Madel Kussus 0.890 13.8 10 1 MCCM.Molex Madel Kussus 0.097 14.37 25   10 ECM.Molex Madel Interactional cacle 0.900 13.8 10 11 ECM.Molex Madel Kussus scale				General VC	MM					Gaussian VC	MM		
	Setting	Banking	Algorithm	Initialization	Classification	Normalised	Computation	Ranking	Algorithm	Initialization	Classification	Normalised	Computation
b I EXMLPDS Kassis 0.087 1.3.2 40 1 EXLMPCS Kassis 0.070 14.35 24 2 EXLMAde_Med Kassis 0.070 14.35 22   3 ECMN.bder.Mead Kassis 0.087 1.3.22 49 3 EXLMPCS Kassis 0.990 14.35 22   5 ECME.BRCS Kassis 0.087 1.3.22 59 5 MCEXLMAder.Mead Kassis 0.999 14.35 23   6 ECME.BRCS Kassis 0.087 1.3.22 59 5 MCEXLMAder.Mead Kassis 0.999 14.35 23   7 FCME.Meder.Mead Emarchica scale 0.061 1.3.8 84 MCEXLMAder.Mead Kassis 0.0277 14.37 28   9 ECME.Meder.Mead Emarchica scale 0.0690 1.3.8 89 10 ECME.Meder.Mead Kassis scale 0.0277 14.37 28   10 ECMERERCS Kineans scale 0.058	(Sample size)			strategy	rate	BIC	time (Seconds)	0	0	strategy	rate	BIC	time (Seconds)
6 (00) 10 11.32 41 2 EXALEPTOS Kenoas 0.070 14.35 22   4 MCEON Noise Anda Kassas 0.097 13.32 49 3 EXALEPTOS Kenoas 0.090 14.35 23   4 MCEON Noise Anda Kassas 0.097 13.32 59 54 MCECM Noise Kenoas 0.090 14.35 23   6 ECMENNoise Model Eventricial scale 0.091 13.38 13.38 13.38 13.38 14 8 MCECM Noise Mean differentricial scale 0.090 13.38 84 8 MCECM Noise Mean differentricial scale 0.090 13.38 10 EXIMPOS Kenoans scale 0.027 14.37 25   10 EXIMPOS Bierarchical scale 0.090 13.38 10 10 EXIMPOS Kenoans scale 0.027 14.37 25   11 SCMEPICS Kenoans scale 0.098 13.38 10 BCMNPOS Kenoans scale 0.027 14.37 25 <	(	1	ECM BEGS	Kmeans	0.987	13.32	40	1	ECM BEGS	Kmeans	0.970	14.35	21
6 (190) 3 CMANder Med Kunnen 0.937 13.32 40 3 MCCLABRYS Kunnas 0.990 11.43 28   5 ECME FRGS Kunnas 0.937 13.32 50 5 MCCLABRYS Kunnas 0.990 14.35 23   7 ECME Folder-Meid Hierarchical acide 0.961 13.38 119 7 MCCCALMolder Meid Kunnas 0.927 14.46 28   9 MCCCMLEPCS Hierarchical acide 0.960 13.38 81 9 10.02 Kunnas scale 0.927 14.47 28   9 MCCCMLEPCS Hierarchical acide 0.990 13.38 101 11 ECME.Nider Maid Kunnas scale 0.928 14.36 37   11 RCLN.Nider.Maid Kunnas scale 0.938 13.38 101 11 ECME.Pider Maid Kunnas scale 0.938 13.38 101 12 ECME.Pider Maid Nick 13.38 13 ECME.Pider Maid Nick 14.36		2	MCECM.BFGS	Kmeans	0.987	13.32	41	2	ECM.Nelder.Mead	Kmeans	0.970	14.35	22
6 (60) 4 (10) MCCLMAder.Med Kumas 0.97 1.3.2 50 4 (10) MCCLAINERS Kumas 0.99 1.1.5 21   6 (10) ECME.Nder.Med Herrchical sola 0.987 1.3.2 73 6 ECME.Nder.Med Herrchical sola 0.987 1.3.2 73 6 ECME.Nder.Med Herrchical sola 0.990 1.4.3 31   7 FCME.Nder.Med Herrchical sola 0.990 1.3.8 84 8 ECME.Nder.Med Herrchical sola 0.990 1.3.8 84 8 ECME.Nder.Med Herrchical sola 0.990 1.3.8 84 8 MCECM.HURS Kumass sola 0.927 1.4.3 32   10 ECME.NdeR.Med Herrchical sola 0.960 1.3.8 90 10 ECM.Nder.Med Kumass sola 0.920 1.4.5 33 33 101 11 BCMENNEYS 1.4.5 33 34 14 MCEM.NdeR.Med Kumass sola 0.956 1.3.8 101 15 ECM.NdeR.Med Kumass sola 0.956 1.3.8 101 15 ECM.NdeR.Med Kumass sola 0.957 <t< td=""><td></td><td>3</td><td>ECM.Nelder.Mead</td><td>Kmeans</td><td>0.987</td><td>13.32</td><td>49</td><td>3</td><td>ECME.BFGS</td><td>Kmeans</td><td>0.969</td><td>14.34</td><td>28</td></t<>		3	ECM.Nelder.Mead	Kmeans	0.987	13.32	49	3	ECME.BFGS	Kmeans	0.969	14.34	28
6 EXRE ERGS Known 0.87 13.22 59 5 MCCCM Nolder Meed Known 0.990 14.35 31   7 ECME Nolder Meed Herrachical scale 0.061 13.38 119 7 MCCCM Nolder Meed Known scale 0.927 14.35 31   9 MCECM EVGES Herrachical scale 0.060 13.38 81 8 Kensens scale 0.927 14.37 285   9 MCECM Nolder Meel Herrachical scale 0.060 13.38 817 9 ECM.Nolder Meel Kensens scale 0.927 14.37 285   11 ECM.Nolder Meel Herrachical scale 0.050 13.38 101 11 ECME BYCES Kensens scale 0.928 14.35 387   13 ECM.Nolder Meel Kensens scale 0.958 13.38 15 ECM.BYCES Herrachical 0.556 14.38 325   14 MYECM Meel Kensens scale 0.958 13.38 15 ECME BYCE Meel Kenchical 0.556		4	MCECM.Nelder.Mead	Kmeans	0.987	13.32	50	4	MCECM.BFGS	Kmeans	0.969	14.35	21
6 ECME-Noder-Mand Kenons 0.097 13.22 7.3 0 ECME-Noder-Mand Mean Market 0.090 14.35 3   8 ECME-Noder-Mand Herarchical scale 0.900 13.38 814 8 MCECM IPICS Kureans scale 0.927 14.37 252   10 ECME-Refer-Mard Herarchical scale 0.900 13.38 90 10 ECMI-Noder-Mand Kureans scale 0.927 14.37 255   11 ECME-Refer-Mand Herarchical scale 0.900 13.38 101 11 ECMI-Noder-Mand Kureans scale 0.925 14.35 383   12 MCECM-Noder-Mand Herarchical scale 0.988 13.38 101 11 ECMI-Noder-Mand Kureans scale 0.958 13.38 101 11 ECMI-Noder-Mand Herarchical 0.955 14.36 28   14 MCECM-Noder-Mand Kureans scale 0.958 13.38 101 16 MCECM-Noder-Mand Herarchical 0.955 14.36		5	ECME.BFGS	Kmeans	0.987	13.32	59	5	MCECM.Nelder.Mead	Kmeans	0.969	14.35	23
6 (600) 7 MCCM.Neider.Mead 0.921 1.4.3 109 7. MCCM.Neider.Mead 0.927 1.4.37 23   9 MCECM.IPICS Birearchical acial 0.900 1.3.38 87 9 MCECM.IPICS Kireans scale 0.927 1.4.37 25   11 ECM.Neider.Med Birearchical acial 0.900 1.3.38 101 11 ECM.Neider.Med Kineans scale 0.927 1.4.37 25   13 ECM.Neider.Med Birearchical acial 0.900 1.3.38 101 11 ECM.Neider.Med Kineans scale 0.926 1.4.36 35   14 MCECM.INFGIS Kineans scale 0.988 1.3.38 101 11 ECM.Neider.Med Kineans scale 0.988 1.3.38 50 15 ECM.Neider.Med Kineans scale 0.988 1.3.38 50 15 ECM.Neider.Med Hirarchical 0.55 1.4.36 22   15 ECM.Neider.Med Kineans scale 0.988 1.3.38 50 15		6	ECME.Nelder.Mead	Kmeans	0.987	13.32	73	6	ECME.Nelder.Mead	Kmeans	0.969	14.35	31
8 ECM.IPEGS Hierarchical acide 0.000 13.38 84 8 MECCALIPEGS Kinoma sole 0.027 14.37 25   10 BCME.RIPGS Hierarchical acide 0.900 13.38 99 10 ECM.Nelder.Med 0.927 14.37 25   11 ECM.Nelder.Med Hierarchical acide 0.900 13.38 101 11 ECM.Nelder.Med Kuncaus acide 0.927 14.35 383   12 MCECM.Nelder.Med Hierarchical acide 0.900 13.38 101 11 ECM.Nelder.Med Kuncaus acide 0.925 14.35 383   13 BCM.RPGS Kunsaus scale 0.988 13.38 14 ECM.Nelder.Med Hierarchical 0.855 14.36 225 14.36 225 14.36 226 14.36 226 14.36 226 14.36 236 14.36 226 14.36 236 14.36 236 14.36 236 14.36 14.36 236 14.36 14.36		7	ECME.Nelder.Mead	Hierarchical scale	0.961	13.38	119	7	MCECM.Nelder.Mead	Kmeans scale	0.927	14.36	28
9 MCEVALIPEG Hierarchical sode 0.900 13.38 87 9 EXALIPCS Koneas sode 0.97 14.37 25   10 ECMLNder Med Hierarchical sode 0.900 13.38 101 11 ECMLNder/Med Hierarchical sode 0.900 13.38 101 11 ECMLNder/Med Hierarchical 0.856 14.36 37   13 ECMLPGS Kineans sode 0.958 13.38 42 13 MCECM/DFCS Hierarchical 0.856 14.36 26   14 MCECM/DFCS Kineans sode 0.958 13.38 50 15 ECMLNder/Med Hierarchical 0.855 14.36 27   17 ECMLPGH Med Kineans sode 0.958 13.38 50 17 ECMLPGS Hierarchical 0.855 14.36 32   18 ECMLNder/Med Kineans sode 0.958 13.38 50 16 MCECM/Med/Med Hierarchical 0.855 14.36 32   17 <t< td=""><td></td><td>8</td><td>ECM BEGS</td><td>Hierarchical scale</td><td>0.960</td><td>13.38</td><td>84</td><td>8</td><td>MCECM BEGS</td><td>Kmeans scale</td><td>0.927</td><td>14.37</td><td>23</td></t<>		8	ECM BEGS	Hierarchical scale	0.960	13.38	84	8	MCECM BEGS	Kmeans scale	0.927	14.37	23
0 ECM.BFOS Herarchical sole 0.900 13.38 90 10 ECM.Neder Med Herarchical sole 0.901 13.48 101 11 ECM.Neder Med Herarchical sole 0.901 13.38 101 12 ECM.ENFCS Kussus sole 0.925 1.435 837   12 MCECM.Medred Med Herarchical 0.958 13.38 4.21 MCECM.PGS Herarchical 0.856 1.436 845   14 MCECM.Medred Means sone 0.958 13.38 50 15 ECM.NedreMed Herarchical 0.856 1.436 856 1.436 856 1.436 856 1.436 856 1.436 856 1.436 856 1.436 856 1.436 855 1.436 838 1.42 856 1.436 835 1.442 856 1.436 835 1.442 848 1.62 ECM.Pedder.Med Herarchical 0.939 1.341 651 1.206 Herarchical 8.35 1.442 848 1.62 <td></td> <td>9</td> <td>MCECM BEGS</td> <td>Hierarchical scale</td> <td>0.960</td> <td>13.38</td> <td>87</td> <td>9</td> <td>ECM BEGS</td> <td>Kmeans scale</td> <td>0.927</td> <td>14.37</td> <td>25</td>		9	MCECM BEGS	Hierarchical scale	0.960	13.38	87	9	ECM BEGS	Kmeans scale	0.927	14.37	25
6 (50) 11 ECML Nobler Madt Herarchical sede 0.90 13.38 101 11 12 ECML BFGS Kussas sede 0.925 14.36 93   13 ECMLBGS Kussas sede 0.958 13.38 42 13 MCCM BFGS Hierarchical 0.856 14.36 95   15 ECMLNder Medt Kussas seade 0.958 13.38 50 15 ECMLNder Medt Hierarchical 0.855 14.36 97   16 MCCM Medre Medt Kussas seade 0.958 13.38 50 15 ECML Medre Medt Hierarchical 0.855 14.36 97   17 ECME Medre Medt Kussas seade 0.958 13.38 71 18 ECMLNder Medt Hierarchical 0.855 14.36 97   18 ECMLBFGS Hierarchical 0.939 13.41 63 21 ECMLNder Medt Hierarchical 0.853 14.42 23   19 MCECM Medra Medt Hierarchical 0.39 13.		10	ECME.BFGS	Hierarchical scale	0.960	13.38	99	10	ECM.Nelder.Mead	Kmeans scale	0.927	14.37	28
6 (500) 12 MCECM Noder Meet Hierarchiel scale 0.969 13.38 101 12 ECMEBFGS Krassa scale 0.925 14.35 95   6 (500) 15 ECM.Nder Meet Kinssa scale 0.958 13.38 43 14 ECM.Nder.Meet 0.856 14.36 25   15 ECM.Nder Meet Kinssan scale 0.958 13.38 50 15 ECM.Nder Meet 0.856 14.36 28   16 MCECM.Nder Meet Kinssan scale 0.658 13.38 50 15 ECM.EDFCS Hierarchial 0.855 14.36 32   17 ECMERGS Hierarchial 0.658 13.38 50 15 ECM.EDFCS Hierarchial 0.855 14.36 32   19 MCECM.BFCS Hierarchial 0.698 13.41 63 21 MCECM.BFCS Hierarchial scale 0.835 14.42 28   20 ECME.BFCS Hierarchial 0.398 13.41 72 23 ECM.BFCS		11	ECM Nelder Mead	Hierarchical scale	0.960	13.38	101	11	ECME Nelder Mead	Kmeans scale	0.926	14.36	37
		12	MCECM.Nelder.Mead	Hierarchical scale	0.960	13.38	101	12	ECME.BFGS	Kmeans scale	0.925	14.35	38
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		13	ECM.BFGS	Kmeans scale	0.958	13.38	42	13	MCECM.BFGS	Hierarchical	0.856	14.36	25
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		14	MCECM.BFGS	Kmeans scale	0.958	13.38	43	14	ECM.BFGS	Hierarchical	0.856	14.36	26
6 (300) 16 MCECM.Neder.Mead Kneans seale 0.988 13.38 51 16 MCECM.Neder.Mead Bierarchical 0.855 14.36 27   17 ECME.BFGS Kneans seale 0.988 13.38 71 ECME.BFGS Hierarchical 0.855 14.36 32   18 ECME.BFGS Hierarchical 0.939 13.41 53 19 ECM.Neder.Mead Hierarchical seale 0.835 14.42 28   21 ECM.Neder.Mead Hierarchical 0.939 13.41 65 21 MCECM.Neder.Mead Hierarchical seale 0.833 14.42 28   23 ECME.BFGS Hierarchical 0.939 13.41 72 23 ECME.Neder.Mead Hierarchical seale 0.832 14.42 28   24 ECME.Neder.Meder.Meder Mead 0.939 13.41 72 23 ECME.Neder.Mead Hierarchical seale 0.832 14.42 25   25 ECM.BFGS Ruadom 0.918 13.47	- ()	15	ECM.Nelder.Mead	Kmeans scale	0.958	13.38	50	15	ECM.Nelder.Mead	Hierarchical	0.856	14.36	28
17 ECML BFCS Kneans seale 0.98 13.38 59 17 ECML BFCS Hierarchical 0.855 14.36 34   19 MCECM BFGS Hierarchical 0.939 13.41 53 19 ECML BFGS Hierarchical 0.835 14.42 28   20 ECM Meder Mead Hierarchical 0.939 13.41 63 21 MCECM Meder Mead Hierarchical scale 0.835 14.42 28   21 ECM Meder Mead Hierarchical 0.939 13.41 63 21 MCECM Meder Mead Hierarchical scale 0.833 14.42 28   22 MCECM Meder Mead Hierarchical 0.939 13.41 72 23 ECME Meder Mead Hierarchical scale 0.832 14.42 38   24 ECM Meder Mead Hierarchical 0.939 13.41 72 24 ECM Meder Mead Hierarchical scale 0.852 14.42 35   25 ECM Brods Random 0.918 13.47 215 </td <td>6 (500)</td> <td>16</td> <td>MCECM.Nelder.Mead</td> <td>Kmeans scale</td> <td>0.958</td> <td>13.38</td> <td>51</td> <td>16</td> <td>MCECM.Nelder.Mead</td> <td>Hierarchical</td> <td>0.855</td> <td>14.36</td> <td>27</td>	6 (500)	16	MCECM.Nelder.Mead	Kmeans scale	0.958	13.38	51	16	MCECM.Nelder.Mead	Hierarchical	0.855	14.36	27
18 ECME.Neder.Mead Kneenss scale 0.958 13.38 7.1 18 ECM.BFGS Hierarchical 0.855 14.36 34   19 MCECM.BFGS Hierarchical 0.939 13.41 53 12 ECM.BFGS Hierarchical 0.835 14.42 28   21 ECM.Nelder.Mead Hierarchical 0.399 13.41 65 22 MCECM.MedR-Mead Hierarchical scale 0.833 14.42 28   23 ECM.ENCM.Nelder.Mead Hierarchical 0.399 13.41 72 23 ECME.BFGS Hierarchical scale 0.832 14.42 24   24 ECM.ENGCS Hierarchical 0.399 13.41 72 23 ECME.BFGS Random 0.615 14.37 50   25 ECM.BFGS Random 0.918 13.47 204 27 ECM.ENGCM.Mead Random 0.615 14.37 51   26 NCMECM.NedReMad Random 0.911 13.48 238 28		17	ECME.BFGS	Kmeans scale	0.958	13.38	59	17	ECME.BFGS	Hierarchical	0.855	14.36	32
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		18	ECME.Nelder.Mead	Kmeans scale	0.958	13.38	71	18	ECME.Nelder.Mead	Hierarchical	0.855	14.36	34
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		19	MCECM.BFGS	Hierarchical	0.939	13.41	53	19	ECM.BFGS	Hierarchical scale	0.835	14.42	26
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		20	ECM.BFGS	Hierarchical	0.939	13.41	54	20	ECM.Nelder.Mead	Hierarchical scale	0.835	14.42	28
22 MCECM.Nelder.Mead Hierarchical 0.939 13.41 65 22 MCECM.Nelder.Mead Hierarchical scale 0.832 14.42 28   23 ECME.BFGS Hierarchical 0.939 13.41 72 23 ECME.BFGS Hierarchical scale 0.832 14.42 34   24 ECME.Nelder.Mead Hierarchical 0.939 13.41 73 24 ECME.Nelder.Mead 0.832 14.42 35   25 ECM.BFGS Random 0.918 13.47 204 25 MCECM.Nelder.Mead Random 0.615 14.37 50   26 MCECM.Nelder.Mead Random 0.914 13.47 200 27 ECM.BFGS Random 0.59 14.36 57   29 ECM.Nelder.Mead Random 0.912 13.48 228 28 ECM.Nelder.Mead Random 0.556 14.38 77   30 ECM.BFGS Random 0.897 13.53 187 30 ECM.EFGS Random		21	ECM.Nelder.Mead	Hierarchical	0.939	13.41	63	21	MCECM.BFGS	Hierarchical scale	0.833	14.42	27
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		22	MCECM.Nelder.Mead	Hierarchical	0.939	13.41	65	22	MCECM.Nelder.Mead	Hierarchical scale	0.833	14.42	28
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		23	ECME.BFGS	Hierarchical	0.939	13.41	72	23	ECME.BFGS	Hierarchical scale	0.832	14.42	34
25 ECM.BFGS Random 0.918 13.47 204 25 MCECM.BFGS Random 0.615 14.37 50   26 ECME.Nelder.Mead Random 0.918 13.47 215 26 MCECM.Nelder.Mead Random 0.615 14.37 51   27 MCECM.Nelder.Mead Random 0.912 13.48 228 28 ECM.Nelder.Mead Random 0.599 14.36 557   28 MCECM.Nelder.Mead Random 0.912 13.48 228 28 ECME.Nelder.Mead Random 0.556 14.38 73   30 ECME.BFCS Kmeans 0.988 13.26 65 1 MCECM.Nelder.Mead Kmeans 0.971 14.44 30   2 MCECM.Nelder.Mead Kmeans 0.988 13.26 92 3 ECM.Nelder.Mead Kmeans 0.971 14.44 30   4 MCECM.Nelder.Mead Kmeans 0.970 14.45 43 30 111 46		24	ECME.Nelder.Mead	Hierarchical	0.939	13.41	93	24	ECME.Nelder.Mead	Hierarchical scale	0.832	14.42	35
26 ECME.Nelder.Mead Random 0.918 13.47 215 26 MCECM.Nelder.Mead Random 0.615 14.37 51   27 MCECM.BFGS Random 0.914 13.47 200 27 ECM.BFGS Random 0.599 14.36 557   29 ECM.Nelder.Mead Random 0.905 13.49 233 29 ECME.Nelder.Mead Random 0.556 14.38 777   30 ECME.BFGS Random 0.905 13.49 233 29 ECME.Nelder.Mead Random 0.556 14.38 777   30 ECME.BFGS Kmeans 0.988 13.26 65 1 MCECM.Nelder.Mead Kmeans 0.971 14.44 30   4 MCECM.Nelder.Mead Kmeans 0.988 13.26 94 4 ECME.Nelder.Mead Kmeans 0.971 14.44 40   4 ECME.BFGS Kmeans 0.988 13.26 119 5 ECME.Nelder.Mead Kmeans		25	ECM.BFGS	Random	0.918	13.47	204	25	MCECM.BFGS	Random	0.615	14.37	50
27 MCECM.BFGS Random 0.914 13.47 200 27 ECM.BFGS Random 0.599 14.36 55   28 MCECM.Neider.Mead Random 0.912 13.48 228 28 ECM.Neider.Mead Random 0.559 14.36 57   29 ECM.Neider.Mead Random 0.905 13.53 187 30 ECM.EBFGS Random 0.556 14.38 73   30 ECM.EBFGS Random 0.987 13.53 187 30 ECME.BFGS Random 0.546 14.38 73   2 MCECM.BFGS Kmeans 0.988 13.26 79 2 ECME.Meder.Mead Kmeans 0.971 14.44 30   4 MCECM.Meder.Mead Kmeans 0.988 13.26 94 4 ECME.BFGS Kmeans 0.971 14.46 30   5 ECME.BFGS Kmeans 0.988 13.26 137 6 MCECM.BFGS Kmeans 0.970		26	ECME.Nelder.Mead	Random	0.918	13.47	215	26	MCECM.Nelder.Mead	Random	0.615	14.37	51
28 MCECM.Nelder.Mead Random 0.912 13.48 228 28 ECM.Nelder.Mead Random 0.599 14.36 57   30 ECME.BFGS Random 0.905 13.49 233 29 ECME.Nelder.Mead Random 0.556 14.38 77   30 ECM.EFGS Random 0.897 13.53 187 30 ECME.BFCS Random 0.566 1.438 773   2 MCECM.BFGS Kmeans 0.988 13.26 65 1 MCECM.Nelder.Mead Kmeans 0.971 14.44 30   3 ECM.Nelder.Mead Kmeans 0.988 13.26 92 3 ECM.Nelder.Mead Kmeans 0.970 14.45 43   5 ECME.BFGS Kmeans 0.988 13.26 117 7 ECM.EBFGS Kmeans 0.970 14.46 30   6 ECME.BFGS Hierarchical scale 0.975 13.30 177 7 ECM.EMeder.Mead Kmeans scale		27	MCECM.BFGS	Random	0.914	13.47	200	27	ECM.BFGS	Random	0.599	14.36	55
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		28	MCECM Nelder Mead	Random	0.912	13.48	228	28	ECM Nelder Mead	Bandom	0.599	14.36	57
30 ECME.BFGS Random 0.897 13.53 187 30 ECME.BFGS Random 0.546 14.38 73   1 ECM.DFGS Kmeans 0.988 13.26 65 1 MCECM.Nelder.Mead Kmeans 0.971 14.44 30   2 MCECM.BFGS Kmeans 0.988 13.26 92 3 ECM.Nelder.Mead Kmeans 0.971 14.44 30   4 MCECM.Nelder.Mead Kmeans 0.988 13.26 94 4 ECME.Nelder.Mead Kmeans 0.971 14.44 40   5 ECME.MEder.Mead Kmeans 0.988 13.26 119 5 ECM.BFGS Kmeans 0.970 14.46 30   6 ECME.BFGS Hierarchical scale 0.975 13.30 177 7 ECME.Meder.Mead Kmeans scale 0.931 14.46 46   8 ECM.Nelder.Mead Kmeans scale 0.974 13.30 112 8 ECME.Meder.Mead Kmeans scale<		29	ECM.Nelder.Mead	Random	0.905	13.49	233	29	ECME.Nelder.Mead	Random	0.556	14.38	77
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		30	ECME.BFGS	Random	0.897	13.53	187	30	ECME.BFGS	Random	0.546	14.38	73
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		1	ECM.BFGS	Kmeans	0.988	13.26	65	1	MCECM.Nelder.Mead	Kmeans	0.971	14.44	30
$ 6 (1000) \begin{tabular}{ c c c c c c c c c c c c c c c c c c c$		2	MCECM.BFGS	Kmeans	0.988	13.26	79	2	ECM.Nelder.Mead	Kmeans	0.971	14.44	32
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		3	ECM.Nelder.Mead	Kmeans	0.988	13.26	92	3	ECME.Nelder.Mead	Kmeans	0.971	14.44	40
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		4	MCECM.Nelder.Mead	Kmeans	0.988	13.26	94	4	ECME.BFGS	Kmeans	0.970	14.45	43
$ 6 \  \  \  \  \  \  \  \  \  \  \  \  \$		5	ECME.BFGS	Kmeans	0.988	13.26	119	5	ECM.BFGS	Kmeans	0.970	14.46	30
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		6	ECME.Nelder.Mead	Kmeans	0.988	13.26	137	6	MCECM.BFGS	Kmeans	0.970	14.46	38
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		7	ECME.BFGS	Hierarchical scale	0.975	13.30	177	7	ECME.Nelder.Mead	Kmeans scale	0.931	14.46	46
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		8	ECM.Nelder.Mead	Kmeans scale	0.974	13.30	112	8	ECME.BFGS	Kmeans scale	0.929	14.44	46
$ 6 (1000) \begin{tabular}{ c c c c c c c c c c c c c c c c c c c$		9	ECME.BFGS	Kmeans scale	0.974	13.30	137	9	MCECM.Nelder.Mead	Kmeans scale	0.929	14.46	33
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		10	ECM.BFGS	Hierarchical scale	0.974	13.30	140	10	ECM.Nelder.Mead	Kmeans scale	0.929	14.46	34
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		11	MCECM.BFGS	Hierarchical scale	0.974	13.30	152	11	MCECM.BFGS	Kmeans scale	0.927	14.44	33
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		12	ECM.Nelder.Mead	Hierarchical scale	0.974	13.30	163	12	ECM.BFGS	Kmeans scale	0.927	14.44	35
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		13	MCECM.Nelder.Mead	Hierarchical scale	0.974	13.30	168	13	MCECM.Nelder.Mead	Hierarchical	0.882	14.47	42
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		14	ECME.Nelder.Mead	Hierarchical scale	0.974	13.30	194	14	MCECM.BFGS	Hierarchical	0.881	14.47	44
6 (1000) 16 ECM.Nelder.Mead Hierarchical 0.973 13.29 135 16 ECM.BFGS Hierarchical 0.880 14.47 49   17 MCECM.Nelder.Mead Hierarchical 0.973 13.29 141 17 ECM.BFGS Hierarchical 0.876 14.47 44   18 MCECM.BFGS Hierarchical 0.973 13.29 143 18 ECM.Nelder.Mead Hierarchical 0.876 14.47 44   19 ECME.BFGS Hierarchical 0.973 13.29 175 19 ECM.Nelder.Mead Hierarchical 0.876 14.47 47   20 ECME.BFGS Hierarchical 0.973 13.29 175 19 ECM.Nelder.Mead Hierarchical 0.840 14.50 43   20 ECME.Nelder.Mead Hierarchical 0.973 13.21 161 21 MCECM.BFGS Hierarchical 0.840 14.50 42   21 ECME.Nelder.Mead Hierarchical 0.839 14.50		15	ECM.BFGS	Hierarchical	0.973	13.29	112	15	ECME.Nelder.Mead	Hierarchical	0.881	14.47	46
17 MCECM.Nelder.Mead Hierarchical 0.973 13.29 141 17 ECM.BFGS Hierarchical 0.876 14.47 44   18 MCECM.BFGS Hierarchical 0.973 13.29 143 18 ECM.Nelder.Mead Hierarchical 0.876 14.47 47   19 ECME.BFGS Hierarchical 0.973 13.29 175 19 ECM.Nelder.Mead Hierarchical scale 0.840 14.50 43   20 ECME.Nelder.Mead Hierarchical 0.973 13.29 182 20 ECM.BFGS Hierarchical scale 0.840 14.50 43   21 ECME.Nelder.Mead Hierarchical 0.973 13.31 161 21 MCECMBFGS Hierarchical scale 0.839 14.50 42	6 (1000)	16	ECM.Nelder.Mead	Hierarchical	0.973	13.29	135	16	ECME.BFGS	Hierarchical	0.880	14.47	49
18 MCECM.BFGS Hierarchical 0.973 13.29 143 18 ECM.Nelder.Mead Hierarchical 0.876 14.47 47   19 ECME.BFGS Hierarchical 0.973 13.29 175 19 ECM.Nelder.Mead Hierarchical scale 0.840 14.50 43   20 ECME.Nelder.Mead Hierarchical 0.973 13.29 182 20 ECM.BFGS Hierarchical scale 0.840 14.50 43   21 ECME.Nelder.Mead Hierarchical 0.973 13.21 161 21 MCECMBFGS Hierarchical scale 0.839 14.50 43		17	MCECM.Nelder.Mead	Hierarchical	0.973	13.29	141	17	ECM.BFGS	Hierarchical	0.876	14.47	44
19 ECME.BFGS Hierarchical 0.973 13.29 175 19 ECM.Nelder.Mead Hierarchical scale 0.840 14.50 43   20 ECME.Nelder.Mead Hierarchical 0.973 13.29 182 20 ECM.BFGS Hierarchical scale 0.840 14.50 43   21 ECME.Nelder.Mead Keenas scale 0.973 13.31 161 21 MCECM.BFGS Hierarchical scale 0.839 14.50 42		18	MCECM.BFGS	Hierarchical	0.973	13.29	143	18	ECM.Nelder.Mead	Hierarchical	0.876	14.47	47
20 ECME.Nelder.Mead Hierarchical 0.973 13.29 182 20 ECM.BFGS Hierarchical scale 0.840 14.50 43   21 ECME.Nelder.Mead Kmeans scale 0.973 13.31 161 21 MCECM.BFGS Hierarchical scale 0.839 14.50 42		19	ECME BEGS	Hierarchical	0.973	13.29	175	19	ECM Nelder Mead	Hierarchical scale	0.840	14.50	43
21 ECME.Nelder.Mead Kmeans scale 0.973 13.31 161 21 MCECM.BEGS Hierarchical scale 0.839 14.50 42		20	ECME.Nelder.Mead	Hierarchical	0.973	13.29	182	20	ECM.BFGS	Hierarchical scale	0.840	14.50	43
		21	ECME.Nelder.Mead	Kmeans scale	0.973	13.31	161	21	MCECM.BFGS	Hierarchical scale	0.839	14.50	42
22 ECM_BFGS Kmeans scale 0.968 13.31 85 22 MCECM_Nelder_Mead Hierarchical scale 0.839 14.50 45		22	ECM.BFGS	Kmeans scale	0.968	13.31	85	22	MCECM.Nelder.Mead	Hierarchical scale	0.839	14.50	45
23 MCECM.BFGS Kmeans scale 0.968 13.31 101 23 ECME.Nelder.Mead Hierarchical scale 0.837 14.50 46		23	MCECM.BFGS	Kmeans scale	0.968	13.31	101	23	ECME.Nelder.Mead	Hierarchical scale	0.837	14.50	46
24 MCECM.Nelder.Mead Kmeans scale 0.968 13.31 108 24 ECME.BFGS Hierarchical scale 0.837 14.50 49		24	MCECM.Nelder.Mead	Kmeans scale	0.968	13.31	101	24	ECME.BFGS	Hierarchical scale	0.837	14.50	49
25 ECMBFGS Random 0.959 13.33 364 25 MCECM Nelder Mead Random 0.555 14.441 70		25	ECM BFGS	Random	0,959	13.33	364	25	MCECM.Nelder Mead	Random	0,555	14.41	70
26 ECME Nelder Mead Bandom 0.955 113.34 345 26 MCECM REGS Bandom 0.554 14.41 73		26	ECME.Nelder.Mead	Random	0.955	13.34	345	26	MCECM BEGS	Random	0.554	14.41	73
27 MCECMBEGS Random 0.950 13.35 361 27 ECMBEGS Random 0.554 14.441 75		27	MCECM.BFGS	Random	0.950	13.35	361	27	ECM.BFGS	Random	0.554	14.41	75
28 ECME.BFGS Random 0.949 13.36 294 28 ECM.Nelder Mead Random 0.554 14.41 79		28	ECME BFGS	Random	0,949	13.36	294	28	ECM.Nelder.Mead	Random	0.554	14.41	79
29 FCM Neiler Mead Random 0.948 13.35 405 29 FCM Neiler Mead Random 0.578 14.41 87		29	ECM.Nelder Mead	Random	0.948	13 35	495	29	ECME Nelder Mead	Random	0.528	14 41	87
30 MCECM.Nelder.Mead Random 0.945 13.36 445 30 ECME.BFGS Random 0.528 14.41 96		30	MCECM.Nelder.Mead	Random	0.945	13.36	445	30	ECME.BFGS	Random	0.528	14.41	96

247

155

259

97

239

248

248

1014

			Ceneral VC	MM					Caussian VC	MM		
Setting	Banking	Algorithm	Initialization	Classification	Normalised	Computation	Banking	Algorithm	Initialization	Classification	Normalised	Computation
(Sample size)			strategy	rate	BIC	time (Seconds)	0	0	strategy	rate	BIC	time (Seconds)
	1	ECM.BFGS	Kmeans scale	0.980	13.92	69	1	ECM.BFGS	Hierarchical scale	0.954	14.70	14
	2	MCECM.BFGS	Kmeans scale	0.980	13.92	71	2	MCECM.BFGS	Hierarchical scale	0.954	14.70	14
	3	MCECM.Nelder.Mead	Kmeans scale	0.980	13.92	82	3	MCECM.Nelder.Mead	Hierarchical scale	0.954	14.70	16
	4	ECM.Nelder.Mead	Kmeans scale	0.980	13.92	85	4	ECM.BFGS	Hierarchical	0.954	14.70	17
	5	ECME.BFGS	Kmeans scale	0.980	13.92	94	5	MCECM.BFGS	Hierarchical	0.954	14.70	17
	6	ECME.Nelder.Mead	Kmeans scale	0.980	13.92	119	6	ECM.Nelder.Mead	Hierarchical scale	0.954	14.70	17
	7	MCECM.BFGS	Hierarchical	0.975	13.92	38	7	ECM.Nelder.Mead	Hierarchical	0.954	14.70	18
	8	ECME.BFGS	Hierarchical	0.975	13.92	57	8	MCECM.Nelder.Mead	Hierarchical	0.954	14.70	18
	9	ECME.Nelder.Mead	Hierarchical	0.975	13.92	72	9	ECME.Nelder.Mead	Hierarchical scale	0.954	14.70	22
	10	ECM.BFGS	Hierarchical	0.974	13.92	38	10	ECME.BFGS	Hierarchical scale	0.954	14.70	22
	11	ECM.Nelder.Mead	Hierarchical	0.974	13.92	47	11	ECME.Nelder.Mead	Hierarchical	0.954	14.70	25
	12	MCECM.Nelder.Mead	Hierarchical	0.974	13.92	47	12	ECME.BFGS	Hierarchical	0.954	14.70	25
	13	MCECM.BFGS	Hierarchical scale	0.972	13.92	36	13	ECM.BFGS	Kmeans scale	0.953	14.78	25
	14	ECM.BFGS	Hierarchical scale	0.972	13.93	36	14	MCECM.BFGS	Kmeans scale	0.953	14.78	25
7 (500)	15	MCECM.Nelder.Mead	Hierarchical scale	0.972	13.93	45	15	MCECM.Nelder.Mead	Kmeans scale	0.953	14.78	29
	16	ECME.BFGS	Hierarchical scale	0.972	13.93	52	16	ECM.Nelder.Mead	Kmeans scale	0.953	14.78	37
	17	ECME.Nelder.Mead	Hierarchical scale	0.972	13.93	105	17	ECME.Nelder.Mead	Kmeans scale	0.951	14.74	33
	18	ECM.Nelder.Mead	Hierarchical scale	0.971	13.93	45	18	ECME.BFGS	Kmeans scale	0.951	14.74	36
	19	MCECM.BFGS	Random	0.958	14.04	224	19	ECM.BFGS	Kmeans	0.947	14.80	52
	20	ECM.BFGS	Random	0.949	14.07	214	20	ECM.Neider.Mead	Kmeans	0.947	14.80	56
	21	ECME Noldor Mond	Random	0.940	14.09	209	21	MCECM Nolder Mood	Kmeens	0.940	14.80	57
	22	ECME BECS	Random	0.940	14.10	234	22	FCME BECS	Kmeans	0.940	14.80	45
	20	ECME Nelder Mead	Kmeans	0.930	14.06	185	20	ECME Nelder Mead	Kmeans	0.030	14.74	48
	25	ECM Nelder Mead	Random	0.937	14.11	265	25	ECME.BFGS	Random	0.918	14.74	72
	26	ECME BEGS	Kmeans	0.935	14.06	146	26	ECME Nelder Mead	Bandom	0.918	14.74	75
	27	MCECM.Nelder.Mead	Kmeans	0.840	14.38	123	27	ECM.BFGS	Random	0.909	14.75	128
	28	ECM.Nelder.Mead	Kmeans	0.834	14.40	125	28	MCECM.BFGS	Random	0.909	14.76	128
	29	ECM.BFGS	Kmeans	0.830	14.41	87	29	MCECM.Nelder.Mead	Random	0.909	14.76	141
	30	MCECM.BFGS	Kmeans	0.830	14.41	90	30	ECM.Nelder.Mead	Random	0.909	14.76	145
	1	ECM.BFGS	Hierarchical scale	0.980	13.79	61	1	ECME.Nelder.Mead	Kmeans	0.958	14.72	66
	2	MCECM.BFGS	Hierarchical scale	0.980	13.79	71	2	ECM.BFGS	Kmeans	0.958	14.72	92
	3	ECME.Nelder.Mead	Hierarchical scale	0.980	13.79	106	3	ECME.BFGS	Kmeans	0.958	14.72	645
	4	MCECM.Nelder.Mead	Hierarchical scale	0.980	13.79	278	4	MCECM.Nelder.Mead	Kmeans	0.958	14.72	970
	5	ECME.BFGS	Hierarchical scale	0.980	13.79	329	5	ECM.Nelder.Mead	Kmeans	0.958	14.72	1004
	6	ECM.Nelder.Mead	Hierarchical	0.980	13.79	344	6	ECM.BFGS	Hierarchical scale	0.957	14.70	24
	7	ECM.Nelder.Mead	Hierarchical scale	0.980	13.79	349	7	ECM.Nelder.Mead	Hierarchical scale	0.957	14.70	25
	8	ECME.Nelder.Mead	Hierarchical	0.980	13.79	565	8	MCECM.BFGS	Kmeans	0.957	14.72	147
	9	MCECM.Nelder.Mead	Hierarchical	0.979	13.79	362	9	MCECM.Nelder.Mead	Hierarchical scale	0.956	14.70	24
	10	ECM.BFGS	Hierarchical	0.978	13.80	72	10	ECME.BFGS	Hierarchical scale	0.956	14.70	27
	11	MCECM.BFGS	Hierarchical	0.978	13.80	73	11	ECME.Nelder.Mead	Hierarchical scale	0.956	14.70	28
	12	ECME.BFGS	Hierarchical	0.978	13.80	448	12	MCECM.BFGS	Hierarchical scale	0.956	14.70	88

MCECM.BFGS

ECM.BFGS

MCECM.BFGS

ECME.Nelder.Mead MCECM.Nelder.Mead

ECM.Nelder.Mead

ECME.BFGS

ECM.BFGS

ECME.Nelder.Mead

ECM.Nelder.Mead

MCECM.Nelder.Mead

ECME.Nelder.Mead ECME.BFGS

ECME.BFGS ECM.BFGS

MCECM.BFGS

MCECM.Nelder.Mead

ECM.Nelder.Mead

Random

Kmeans scale

Kmeans scale

Kmeans scale

Kmeans scale

Kmeans scale

Kmeans scale

Random

Random

Random

Random

Kmeans

Kmeans

Random

Kmeans

Kmeans

Kmeans

Kmeans

0.976

0.974

0.974

0.974 0.974

0.974

0.974

0.974

0.974

0.973

0.973

0.969

0.960

0.960 0.899

0.899

0.898

0.898

13.82

13.82

13.82

13.82

13.82

13.82

13.82

13.83

13.84

13.84

13.85

13.83

13.85

13.90

14.06

14.06

14.06

14.06

343

133

139

176

540

545

639

315

1166

465

517

1536

552

412

206

258

679

736

13

14

15

16 17

18 19

20

21

22

23 24 25

26 27

28

29

30

ECM.Nelder.Mead

ECME.BFGS

ECM.BFGS

MCECM.Nelder.Mead ECME.BFGS

ECME.Nelder.Mead

ECM.BFGS

ECM.BFGS ECME.Nelder.Mead ECM.Nelder.Mead

MCECM.BFGS

MCECM.Nelder.Mead

MCECM.BFGS ECME.BFGS

ECME.Nelder.Mead ECM.BFGS

ECM.Nelder.Mead MCECM.Nelder.Mead

MCECM.BFGS

0.952

0.951

0.951

0.951

0.950

0.950

0.950

0.950

0.950

0.950

0.950

0.950

0.947

0.947

0.938

0.938

0.938

0.938

14.86

14.78

14.82

14.85

14.76

14.76

14.78

14.78

14.78

14.78

14.78

14.82

14.67

14.70

14.73

14.73

14.73

14.73

Kmeans scale

Hierarchical

Kmeans scale

Kmeans scale

Kmeans scale

Kmeans scale

Hierarchical

Hierarchical

Hierarchical

Hierarchical

Hierarchical

Kmeans scale

Random

Random

Random

Random

Random

Random

13

14

15

 $16 \\ 17$ 

18

19

20

21

22

23

24 25

26 27

28

29

30

7 (1000)

			General VC!	MM					Gaussian VC	MM		
Setting	Ranking	Algorithm	Initialization	Classification	Normalised	Computation	Ranking	Algorithm	Initialization	Classification	Normalised	Computation
(Sample size)	Ĭ		strategy	rate	BIC	time (Seconds)	0	Ŭ	strategy	rate	BIC	time (Seconds)
	1	ECM.Nelder.Mead	Hierarchical	0.991	17.26	59	1	ECM.BFGS	Random	0.907	19.92	53
	2	ECM.BFGS	Hierarchical	0.990	17.26	47	2	ECM.Nelder.Mead	Random	0.907	19.92	64
	3	MCECM.BFGS	Hierarchical	0.990	17.26	48	3	MCECM.BFGS	Random	0.906	19.92	55
	4	MCECM.Nelder.Mead	Hierarchical	0.990	17.26	55	4	MCECM.Nelder.Mead	Random	0.906	19.92	62
	5	ECME.BFGS	Hierarchical	0.990	17.26	73	5	ECM.BFGS	Kmeans	0.905	19.92	61
	6	ECME.Nelder.Mead	Hierarchical	0.990	17.26	87	6	MCECM.BFGS	Kmeans	0.905	19.92	62
	7	ECME.BFGS	Random	0.912	17.56	105	7	MCECM.Nelder.Mead	Kmeans	0.905	19.92	72
	8	ECME.Nelder.Mead	Random	0.900	17.60	117	8	ECM.Nelder.Mead	Kmeans	0.905	19.92	74
	9	ECM.Nelder.Mead	Random	0.862	17.67	104	9	MCECM.BFGS	Hierarchical	0.900	19.90	33
	10	ECM.BFGS	Random	0.861	17.67	86	10	MCECM.Nelder.Mead	Hierarchical	0.900	19.90	37
	11	MCECM.Nelder.Mead	Random	0.843	17.73	103	11	ECM.BFGS	Hierarchical	0.899	19.90	33
	12	MCECM.BFGS	Random	0.843	17.73	441	12	ECM.Nelder.Mead	Hierarchical	0.899	19.90	39
	13	ECME.Nelder.Mead	Hierarchical scale	0.752	18.36	196	13	ECME.Nelder.Mead	Hierarchical	0.893	19.90	36
	14	ECM.BFGS	Hierarchical scale	0.745	18.36	121	14	ECME.BFGS	Hierarchical	0.892	19.90	33
8 (500)	15	ECME.BFGS	Hierarchical scale	0.743	18.41	147	15	ECM.BFGS	Kmeans scale	0.892	19.96	59
8 (500)	16	MCECM.BFGS	Hierarchical scale	0.738	18.38	121	16	MCECM.BFGS	Kmeans scale	0.892	19.96	59
	17	ECM.Nelder.Mead	Hierarchical scale	0.732	18.40	145	17	MCECM.Nelder.Mead	Kmeans scale	0.892	19.96	69
	18	MCECM.Nelder.Mead	Hierarchical scale	0.726	18.42	143	18	ECM.Nelder.Mead	Kmeans scale	0.892	19.96	71
	19	ECME.BFGS	Kmeans	0.658	18.65	229	19	ECM.BFGS	Hierarchical scale	0.888	19.95	50
	20	MCECM.BFGS	Kmeans	0.656	18.67	1105	20	ECM.Nelder.Mead	Hierarchical scale	0.888	19.95	60
	21	ECM.BFGS	Kmeans	0.654	18.68	168	21	MCECM.BFGS	Hierarchical scale	0.887	19.95	50
	22	MCECM.Nelder.Mead	Kmeans	0.654	18.68	224	22	MCECM.Nelder.Mead	Hierarchical scale	0.887	19.95	58
	23	ECM.Nelder.Mead	Kmeans	0.650	18.69	214	23	ECME.BFGS	Kmeans	0.882	19.91	52
	24	ECME.Nelder.Mead	Kmeans	0.648	18.72	279	24	ECME.Nelder.Mead	Kmeans	0.881	19.91	61
	25	MCECM.BFGS	Kmeans scale	0.614	18.89	101	25	ECME.BFGS	Kmeans scale	0.874	19.96	53
	26	ECME.BFGS	Kmeans scale	0.611	18.92	118	26	ECME.Nelder.Mead	Kmeans scale	0.874	19.96	61
	27	ECM.Nelder.Mead	Kmeans scale	0.608	18.92	115	27	ECME.Nelder.Mead	Random	0.870	19.91	43
	28	MCECM.Nelder.Mead	Kmeans scale	0.607	18.91	117	28	ECME.Nelder.Mead	Hierarchical scale	0.867	19.95	55
	29	ECME.Nelder.Mead	Kmeans scale	0.603	18.96	150	29	ECME.BFGS	Random	0.866	19.91	38
	30	ECM.BFGS	Kmeans scale	0.602	18.95	89	30	ECME.BFGS	Hierarchical scale	0.862	19.96	48
	1	ECME.BFGS	Random	0.878	14.58	119	1	MCECM.BFGS	Random	0.726	16.73	76
	2	ECME.Nelder.Mead	Random	0.877	14.57	141	2	MCECM.Nelder.Mead	Random	0.726	16.73	85
	3	MCECM.BFGS	Random	0.867	14.58	98	3	ECM.BFGS	Random	0.722	16.68	82
	4	ECM.BFGS	Random	0.860	14.60	98	4	ECM.Nelder.Mead	Random	0.722	16.68	88
	5	MCECM.Nelder.Mead	Random	0.847	14.65	117	5	ECME.BFGS	Random	0.690	16.61	48
	6	ECM.Nelder.Mead	Random	0.847	14.65	123	6	ECME.Nelder.Mead	Random	0.664	16.56	52
	7	MCECM.Nelder.Mead	Kmeans scale	0.757	15.46	169	7	ECM.BFGS	Hierarchical	0.645	16.83	68
	8	ECME.BFGS	Kmeans scale	0.757	15.49	137	8	MCECM.Nelder.Mead	Hierarchical	0.645	16.83	71
	9	ECM.BFGS	Kmeans scale	0.756	15.46	118	9	ECM.Nelder.Mead	Hierarchical	0.645	16.83	72
	10	ECM.Nelder.Mead	Kmeans scale	0.756	15.46	167	10	ECME.BFGS	Hierarchical	0.639	16.80	53
	11	MCECM.BFGS	Kmeans scale	0.750	15.50	116	11	ECME.Nelder.Mead	Hierarchical	0.639	16.80	57
	12	ECME.Nelder.Mead	Kmeans scale	0.731	15.67	178	12	MCECM.BFGS	Hierarchical	0.637	16.86	61
	13	ECME.Nelder.Mead	Hierarchical	0.728	15.20	173	13	ECM.BFGS	Kmeans scale	0.634	16.99	90
	14	ECME.BFGS	Hierarchical	0.720	15.24	128	14	ECM.Nelder.Mead	Kmeans scale	0.634	16.99	96
9 (500)	15	ECME.BFGS	Hierarchical scale	0.708	15.50	124	15	MCECM.BFGS	Kmeans scale	0.631	17.02	83
0 (000)	16	ECM.BFGS	Hierarchical	0.700	15.32	84	16	MCECM.Nelder.Mead	Kmeans scale	0.631	17.02	96
	17	MCECM.BFGS	Hierarchical	0.700	15.32	85	17	ECME.Nelder.Mead	Kmeans scale	0.622	16.98	84
	18	ECM.Nelder.Mead	Hierarchical	0.699	15.32	111	18	ECME.BFGS	Kmeans scale	0.620	16.98	75
	19	MCECM.Nelder.Mead	Hierarchical	0.699	15.32	111	19	ECME.BFGS	Hierarchical scale	0.608	17.00	66
	20	ECME.Nelder.Mead	Hierarchical scale	0.698	15.55	170	20	ECME.Nelder.Mead	Hierarchical scale	0.608	17.00	75
	21	MCECM.Nelder.Mead	Hierarchical scale	0.685	15.59	139	21	ECM.BFGS	Hierarchical scale	0.606	17.05	74
	22	ECM.Nelder.Mead	Hierarchical scale	0.685	15.59	140	22	ECM.Nelder.Mead	Hierarchical scale	0.606	17.05	80
	23	MCECM.BFGS	Hierarchical scale	0.679	15.61	100	23	MCECM.BFGS	Hierarchical scale	0.603	17.08	67
	24	ECM.BFGS	Hierarchical scale	0.678	15.62	97	24	MCECM.Nelder.Mead	Hierarchical scale	0.603	17.08	76
	25	ECME.BFGS	Kmeans	0.573	15.95	150	25	ECME.BFGS	Kmeans	0.580	17.28	74
	26	MCECM.Nelder.Mead	Kmeans	0.569	15.95	161	26	ECME.Nelder.Mead	Kmeans	0.580	17.28	85
	27	ECME.Nelder.Mead	Kmeans	0.569	15.95	206	27	MCECM.BFGS	Kmeans	0.576	17.35	76
	28	ECM.Nelder.Mead	Kmeans	0.564	16.00	156	28	MCECM.Nelder.Mead	Kmeans	0.576	17.35	84
	29	ECM.BFGS	Kmeans	0.562	15.99	116	29	ECM.Nelder.Mead	Kmeans	0.576	17.35	88
	30	MCECM.BFGS	Kmeans	0.561	15.99	115	30	ECM.BFGS	Kmeans	0.575	17.37	75

# C Appendix for Section 4 Real data sets

## C.1 Estimated vine copula model for AIS data

Table C.1: The result of the fitted vine copula model for the AIS data by using different VCMM algorithms and initialization strategies. The variable encoding is given as follows: (AIS) 1: LBM, 2: Wt, 3: BMI, 4: WBC, 5: PBF. For marginal distributions (Left column), the estimated marginal distributions and parameters for each cluster are shown. For vine tree structure (Right column), the first and second tree level of the estimated vine copula models are shown here. The number 1,5 represents the edge of the tree level, letter N is the abbreviation of the copula and the true parameter value and corresponding Kendall's  $\tau$  of the pair copula are given inside the parenthesis (parameter(s)/Kendall's  $\tau$ ) near the letter. The meaning of the abbreviation for marginal distribution and copula families is shown in the appendix B.1 and B.2.

	Marginal distrib	itions				Vine tre	e stru	cture		
Variable	Cluster 1	Cluster 2		C	luster 1	<b>m</b> a		Clus	ter 2	
				Tree 1		Tree 2		Tree 1		Tree 2
				AIS						
				General VCMM (	Randon	<u>n)</u>				
1	(70 70 11 05 0 (1)		ssifica	tion rate : 60.2% N	ormalise	ed BIC : 25.82	0.1	DD1(1.5.0.10/0.54)	910	N( 0.10 / 0.10)
1	snorm(70.79, 11.85, 0.61)	llogis(9.28, 59.85)	1,5	R90J(-3.35/-0.56)	2,5;1	SBB8(4.43,0.82/0.51)	2,1	BB1(1.5,2.16/0.74)	3,1;2	N(-0.18/-0.12)
2	$\mathcal{N}(79.54, 10.06)$ $\Gamma(182.87, 7.07)$	llogis(8.58, 10.85)	2,1	N(0.99/0.91) N(0.70/0.58)	3,1;2	R90G(-1.14/-0.12) R00C(-0.07/-0.02)	3,2	N(0.89/0.69) E(1.40/0.16)	4,2;3	SC(0.04/0.02)
4	1(103.01, 1.91)	$\Gamma(16 14 2 26)$	1 3,2	I(1.26/0.13)	4,2,3	1900(-0.07/-0.03)	5.2	F(1.49/0.10) F(2.45/0.26)	5,4,5	5G(1.09/0.08)
5	snorm(11.98.4.4.142.37)	morm(14, 54, 6, 73, 268, 28)	4,0	5(1.20/0.15)			0,0	r (2.45/ 0.20)		
	Shorm(11.50, 4.4, 142.07)	General	VCM	fM (Copula with sin	l gle para	meter) (Random)				
		Cla	ssifica	tion rate : 61.6% N	ormalise	ed BIC : 26.44				
1	snorm(70.17, 11.75, 0.65)	llogis(9.1, 59.53)	1,5	R90J(-3.52/-0.57)	2,5;1	F(-0.08/-0.01)	2,1	SG(4.02/0.75)	3,1;2	N(-0.27/-0.17)
2	snorm(77.51, 11.35, 2.97)	$\Gamma(21.87, 0.3)$	2,1	F(23.02/0.84)	3,1;2	SC(0.27/0.12)	3,2	N(0.9/0.71)	5,2;3	F(-0.79/-0.09)
3	$\mathcal{N}(22.93, 1.78)$	llogis(12.35, 22.59)	3,2	N(0.76/0.54)	4,2;3	SC(0/0)	3,4	F(1.36/0.15)	5,4;3	SG(1.08/0.07)
4	llogis(7.32, 6.8)	$\Gamma(16.21, 2.27)$	4,3	G(1.2/0.17)			5,3	F(2.74/0.28)		
5	sstd(11.23, 4.93, 4.27, 146.43)	snorm(14.88, 6.99, 121.58)								
		Cla		Gaussian VCMM	(Rando	n) d DIC - 22.06				
1	M(67.60, 12.27)	V(61 27 11 71)	ssinca	N(0.88/0.68)	ormanse	N(0.16/0.1)	15	N(024/022)	9.5.1	N(0.00/0.01)
2	$\mathcal{N}(07.09, 13.37)$	$\mathcal{N}(01.37, 11.71)$ $\mathcal{N}(76.08, 13.26)$	2,3	N(1/0.04)	5.2.1	N(0.07/0.83)	1,0	N(-0.34/-0.22) N(0.05/0.81)	2,0,1	N(0.99/0.91) N(-0.19/-0.12)
2	$\mathcal{N}(22, 38, 2, 46)$	N(23.68, 3.13)	5.1	N(0.33/0.22)	1 1.5	N(0.57/0.05)	2,1	N(0.85/0.65)	4 2.3	N(0.05/0.03)
4	$\mathcal{N}(7.03, 1.84)$	N(7.2.1.73)	5.4	N(0.23/0.15)	4,1,0	11(0.14/0.00)	13	N(0.27/0.18)	4,2,0	11(0.00/0.00)
5	$\mathcal{N}(8.8, 1.97)$	$\mathcal{N}(19.36, 4.35)$	0,1	11(0120/0110)			1,0	11(0121/0110)		
				General VCMM (	Kmean	5)				
		Cla	ssifica	tion rate : 77.5% N	ormalise	ed BIC : 26.00				
1	snorm(56.44, 7.22, 0.6)	snorm(78.08, 7.62, 83.9)	3,4	SC(0.24/0.11)	2,4;3	R90C(-0.02/-0.01)	2,1	G(3.36/0.7)	3,1;2	N(-0.31/-0.2)
2	sstd(66.86, 9.67, 8.29, 0.75)	snorm(87.38, 10.26, 160.5)	2,1	SJ(3.77/0.6)	3,1;2	N(-0.2/-0.13)	3,2	BB8(4.39, 0.85/0.53)	5,2;3	N(0.04/0.03)
3	lnorm(3.08, 0.1)	sstd(24.82, 3.22, 3.03, 1.59)	3,2	N(0.83/0.62)	5,2;3	SC(0.25/0.11)	5,3	G(1.71/0.42)	4,3;5	R270C(-0.26/-0.12)
4	lnorm(1.9, 0.24)	lnorm(1.98, 0.26)	5,3	SC(0.97/0.33)			5,4	SBB8(2.58,0.79/0.28)		
	snorm(15.11, 7.16, 164.59)	sstd(10.39, 4.85, 3.2, 176.82)		04/0 1 11 i	ļ					
		Genera	I VCN	1M (Copula with sin	gle para	meter) (Kmeans)				
1	snorm(56.35, 7.17, 0.58)	enorm(78 41 7 87 02 25)	3 A	SC(0.27/0.12)	2 4.3	E(0.47/0.05)	9.1	C(3 23/0.60)	3 1.9	F(2.45/0.26)
2	snorm(66.64, 9.15, 0.67)	snorm(87.47, 10.33, 133.42)	21	SI(3 79/0.6)	3 1.2	P(-0.47/-0.05) B90C(-0.25/-0.11)	3.2	G(1.99/0.5)	5.2.3	P(-2.40/-0.20) B90C(-0.01/-0.01)
2	lnorm(3.07, 0.1)	sstd(24 84 3 18 3 09 1 58)	3.2	N(0.83/0.62)	5 2.3	SC(0 27/0 12)	5.3	F(4.87/0.45)	4 3.5	B270G(-1.06/-0.06)
4	lnorm(1.9, 0.24)	lnorm(1.97, 0.26)	5.3	SC(0.92/0.31)	0,2,0	00(0121/0112)	5.4	C(0.35/0.15)	1,0,0	1021003(1100) 0100)
5	snorm(15.03, 7.1, 231.11)	sstd(10.84, 7.86, 2.42, 118.25)						- ()		
	, , , , , , , , , , , , , , , , , , , ,			Gaussian VCMM	(Kmear	s)				
		Cla	ssifica	tion rate : 91.4% N	ormalise	ed BIC : 23.57				
1	$\mathcal{N}(56.33, 8.89)$	$\mathcal{N}(74.68, 9.72)$	2,5	N(0.61/0.42)	1,5;2	N(-0.98/-0.87)	2,1	N(0.99/0.9)	3,1;2	N(-0.28/-0.18)
2	$\mathcal{N}(69.02, 12.71)$	$\mathcal{N}(81.89, 11.86)$	2,1	N(0.94/0.78)	3,1;2	N(-0.14/-0.09)	3,2	N(0.83/0.62)	5,2;3	N(0.23/0.15)
3	$\mathcal{N}(22.32, 3.02)$	$\mathcal{N}(23.69, 2.46)$	3,2	N(0.87/0.67)	4,2;3	N(-0.01/0)	5,3	N(0.65/0.45)	4,3;5	N(-0.15/-0.1)
4	$\mathcal{N}(7.1, 1.69)$	$\mathcal{N}(7.12, 1.91)$	4,3	N(0.23/0.15)			5,4	N(0.36/0.23)		
5	$\mathcal{N}(17.79, 5.28)$	$\mathcal{N}(8.59, 2.16)$								

	Marginal dis	tributions				Vine tree	struct	ure		
Variable	Cluster 1	Cluster 2		Clus	ter 1			Clus	ster 2	
				Tree 1		Tree 2		Tree 1		Tree 2
				AIS	5					
				General VCMM	(Hierarc	hical)				
			Cla	ssification rate : 91.4%	Normal	ised BIC : 24.21				
1	$t_3(55.42, 6.12)$	lnorm(4.3, 0.13)	2,1	N(0.94/0.77)	5,1;2	N(-0.86/-0.66)	2,1	N(0.99/0.9)	3,1;2	F(-2.13/-0.23)
2	llogis(9.14, 68.41)	lnorm(4.39, 0.14)	3,4	F(1.57/0.17)	2,4;3	F(-0.41/-0.05)	3,2	N(0.81/0.6)	5,2;3	C(0.2/0.09)
3	llogis(13.2, 22.04) lnorm(1.93.0.24)	llogis(19.93, 23.36) lnorm(1.03, 0.26)	2,3	N(0.88/0.68) SBB8(3.78.0.88/0.5)	5,3;2	N(0.18/0.11)	5,3	N(0.58/0.39) F(2.67/0.28)	4,3;5	R270C(-0.17/-0.08)
5	$\Gamma(11.31, 0.63)$	snorm(8.63, 2.14, 4.08)	0,2	55550(3.10,0.00/0.3)			0,4	r (2.01/0.20)		
		Ger	neral '	VCMM (Copula with sin	igle para	ameter) (Hierarchical)				
			Cla	ssification rate : $92.4\%$	Normal	ised BIC : 23.83				
1	llogis(12.42, 55.34)	lnorm(4.3, 0.13)	2,5	C(1.53/0.43)	1,5;2	R90G(-3.38/-0.7)	2,1	SG(9.87/0.9)	3,1;2	R90G(-1.32/-0.24)
2	$\mathcal{N}(68.59, 12.29)$	lnorm(4.4, 0.14)	2,1	T(0.94,6.26/0.78)	3,1;2	N(-0.25/-0.16)	3,2	N(0.83/0.62)	5,2;3	C(0.2/0.09)
3	lnorm(3.09, 0.12)	llogis(18.22, 23.46)	3,2	N(0.87/0.67) F(1.56/0.17)	4,2;3	R270C(0/0)	5,3	T(0.64,11.61/0.45) F(2.52/0.26)	4,3;5	R270C(-0.17/-0.08)
4 5	$\Gamma(11.04, 0.62)$	snorm(8.81, 2.28, 4.3)	4,5	F(1.50/0.17)			0,4	F (2.55/0.20)		
	- ()			Gaussian VCMM	(Hierar	chical)				
			Cla	ssification rate : 90.5%	Normal	ised BIC : 23.57				
1	$\mathcal{N}(56.27, 8.9)$	$\mathcal{N}(74.56, 9.74)$	2,5	N(0.62/0.42)	1,5;2	N(-0.98/-0.86)	2,1	N(0.99/0.9)	3,1;2	N(-0.28/-0.18)
2	$\mathcal{N}(69, 12.77)$	$\mathcal{N}(81.77, 11.85)$	2,1	N(0.94/0.79)	$^{3,1;2}$	N(-0.14/-0.09)	3,2	N(0.83/0.63)	5,2;3	N(0.2/0.13)
3	$\mathcal{N}(22.31, 3.04)$	$\mathcal{N}(23.68, 2.45)$	3,2	N(0.87/0.67)	4,2;3	N(-0.01/-0.01)	5,3	N(0.63/0.44)	4,3;5	N(-0.16/-0.1)
4	$\mathcal{N}(7.08, 1.69)$	$\mathcal{N}(7.14, 1.91)$ $\mathcal{N}(8.62, 2.17)$	4,3	N(0.23/0.15)			5,4	N(0.37/0.24)		
	N (11.65, 5.27)	N (8.02, 2.17)		General VCMM (	Kmeans	scale)				
			Cla	ssification rate : 91.4%	Normal	ised BIC : 23.52				
1	logis(55.22, 3.77)	lnorm(4.32, 0.12)	2,5	F(6.06/0.52)	1,5;2	N(-0.99/-0.9)	2,1	SG(6.35/0.84)	3,1;2	R90C(-0.42/-0.17)
2	$\mathcal{N}(66.92, 10.53)$	snorm(83.75, 11.59, 1.72)	2,1	SG(3.56/0.72)	3,1;2	N(-0.25/-0.16)	3,2	BB8(4.87, 0.87/0.58)	5,2;3	N(0.23/0.15)
3	lnorm(3.08, 0.11)	sstd(24.17, 2.81, 4.67, 1.95)	3,2	N(0.85/0.65)	4,2;3	R270C(0/0)	5,3	$BB8 (4.07,\!0.82/0.48)$	4,3;5	R90J(-1.16/-0.08)
4	lnorm(1.92, 0.24)	lnorm(1.95, 0.26)	4,3	F(1.16/0.13)			5,4	SBB8(6, 0.42/0.3)		
5	$\mathcal{N}(17.18, 5.82)$	sstd(9.46, 3.66, 3.8, 5.55)		COND. (Character del 1911)	1	(IZ	Ĺ			
		Gen	Cla	CMM (Copula with sin esification rate : 01.4%	gle para Normal	meter) (Kmeans scale isod BIC : 23.65	<u>)</u>			
1	logis(55.22, 3.77)	lnorm(4.32, 0.12)	2.5	F(6.06/0.52)	1.5:2	N(-0.99/-0.9)	2.1	SG(6.35/0.84)	3.1:2	R90C(-0.48/-0.19)
2	$\mathcal{N}(66.92, 10.53)$	snorm(83.75, 11.59, 1.72)	2,1	SG(3.56/0.72)	3,1;2	N(-0.25/-0.16)	3,2	G(2.24/0.55)	5,2;3	C(0.25/0.11)
3	lnorm(3.08, 0.11)	sstd(24.17, 2.81, 4.67, 1.95)	3,2	N(0.85/0.65)	$^{4,2;3}$	R270C(0/0)	5,3	G(1.79/0.44)	4,3;5	N(-0.07/-0.05)
4	lnorm(1.92, 0.24)	lnorm(1.95, 0.26)	4,3	F(1.16/0.13)			5,4	T(0.41, 30/0.27)		
5	$\mathcal{N}(17.18, 5.82)$	sstd(9.46, 3.66, 3.8, 5.55)								
			Cla	Gaussian VCMM seification rate : 90.5%	(Kmean Normal	$\frac{s \text{ scale}}{100 \text{ scale}}$				
1	N(56.27, 8.9)	$\mathcal{N}(74.56, 9.74)$	2.5	N(0.62/0.42)	1.5:2	N(-0.98/-0.86)	2.1	N(0.99/0.9)	3.1:2	N(-0.28/-0.18)
2	$\mathcal{N}(69, 12.77)$	$\mathcal{N}(81.77, 11.85)$	2,1	N(0.94/0.79)	3,1;2	N(-0.14/-0.09)	3.2	N(0.83/0.63)	5.2;3	N(0.2/0.13)
3	$\mathcal{N}(22.31, 3.04)$	$\mathcal{N}(23.68, 2.45)$	3,2	N(0.87/0.67)	4,2;3	N(-0.01/-0.01)	5,3	N(0.63/0.44)	4,3;5	N(-0.16/-0.1)
4	$\mathcal{N}(7.08, 1.69)$	$\mathcal{N}(7.14, 1.91)$	4,3	N(0.23/0.15)			5,4	N(0.37/0.24)		
5	$\mathcal{N}(17.85, 5.27)$	$\mathcal{N}(8.62, 2.17)$		a 110001/11						
			Cl-	General VCMM (H	Normal	isod BIC - 22.02				
1	llogis(12.36, 55 59)	lnorm(4.31.0.13)	2.5	SBB8(4.07.0.82/0.48)	1.5:2	T(-0.92.3.56/-0.74)	2.1	SG(10.7/0.91)	3,1:2	R270G(-1.27/-0.21)
2	$\mathcal{N}(68.68, 12.13)$	lnorm(4.4, 0.14)	2,1	BB1(1.05,2.81/0.77)	3,1;2	N(-0.24/-0.15)	3,2	N(0.83/0.63)	5,2;3	N(0.26/0.17)
3	llogis(14.4, 21.97)	llogis(17.89, 23.49)	3,2	N(0.87/0.67)	4,2;3	F(-0.27/-0.03)	5,3	G(1.78/0.44)	4,3;5	R270C(-0.16/-0.08)
4	$\Gamma(18, 2.54)$	lnorm(1.93, 0.26)	4,3	F(1.55/0.17)			5,4	F(2.58/0.27)		
5	$\Gamma(10.82, 0.61)$	snorm(8.74, 2.23, 4.24)								
		Gener	al VC	MM (Copula with single	e param	eter) (Hierarchical sca	le)			
1	llogis(12 36 55 59)	<i>lnorm</i> (4.31, 0.13)	2 1	T(0.94.5.09/0.77)	Normal 5 1.2	B270G(-3.87/-0.74)	21	SG(10.7/0.91)	3 1.9	B270G(-1.27/-0.21)
2	$\mathcal{N}(68.68, 12.13)$	lnorm(4.4, 0.14)	3.4	F(1.55/0.17)	2,4:3	F(-0.27/-0.03)	3.2	N(0.83/0.63)	5,2;3	N(0.26/0.17)
3	llogis(14.4, 21.97)	llogis(17.89, 23.49)	2,3	N(0.87/0.67)	5,3;2	SC(0.39/0.16)	5,3	G(1.78/0.44)	4,3;5	R270C(-0.16/-0.08)
4	$\Gamma(18, 2.54)$	lnorm(1.93, 0.26)	5,2	C(1.4/0.41)			5,4	F(2.58/0.27)		
5	$\Gamma(10.82, 0.61)$	snorm(8.74, 2.23, 4.24)		~						
			CL	Gaussian VCMM (E	lierarchi	cal scale)				
1	N (56 33 8 89)	N(74.68.0.72)	2.5	N(0.61/0.42)	1 5.9	N(-0.98/-0.87)	2.1	N(0.99/0.9)	3 1.9	N(-0.28/-0.18)
2	$\mathcal{N}(69.02, 12.71)$	$\mathcal{N}(81.89, 11.86)$	2,5	N(0.94/0.78)	3.1:2	N(-0.14/-0.09)	3.2	N(0.83/0.62)	5,2:3	N(0.23/0.15)
3	$\mathcal{N}(22.32, 3.02)$	$\mathcal{N}(23.69, 2.46)$	3,2	N(0.87/0.67)	4,2;3	N(-0.01/0)	5,3	N(0.65/0.45)	4,3;5	N(-0.15/-0.1)
4	$\mathcal{N}(7.1, 1.69)$	$\mathcal{N}(7.12, 1.91)$	4,3	N(0.23/0.15)			5,4	N(0.36/0.23)		
5	$\mathcal{N}(17.79, 5.28)$	$\mathcal{N}(8.59, 2.16)$								

## C.2 Estimated vine copula model for BCW data

Table C.2: The result of the fitted vine copula model for BCW data by using different VCMM algorithms and initialization strategies. The variable encoding is given as follows: (BCW) 1: PSE, 2: ES, 3: EC and 4: ECP. For marginal distributions (Left column), the estimated marginal distributions and parameters for each cluster are shown. For vine tree structure (Right column), the first and second tree level of the estimated vine copula models are shown here. The number 1,5 represents the edge of the tree level, letter N is the abbreviation of the copula and the true parameter value and corresponding Kendall's  $\tau$  of the pair copula are given inside the parenthesis (parameter(s)/Kendall's  $\tau$ ) near the letter. The meaning of the abbreviation for marginal distribution and copula families is shown in the appendix B.1 and B.2.

	Marginal distrib	outions				Vine tree	struct	ure		
Variable	Cluster 1	Cluster 2		Clus	ter 1			Clus	ster 2	
Turiubic	Cluster 1	citator 2		Tree 1		Tree 2		Tree 1		Tree 2
				BCW						
				General VCMM (Ra	ndom)					
		Cla	assific	ation rate : 50.4% Norn	nalised	BIC : -7.02				
1	lnorm(1.04, 0.59)	sstd(2.4, 1.47, 3.42, 4.25)	4,2	T(0.41,8.73/0.27)	1,2;4	F(0.69/0.08)	3,2	F(4.21/0.4)	4,2;3	SJ(1.21/0.11)
2	$\Gamma(42.86, 358.15)$	lnorm(-1.95, 0.14)	4,1	BB8(3.45, 0.96/0.53)	3,1;4	R270G(-1.15/-0.13)	4,1	BB8 (3.61, 0.93/0.52)	3,1;4	F(-0.67/-0.07)
3	snorm(0.23, 0.17, 296.73)	snorm(0.32, 0.22, 315.76)	4,3	SBB8(6, 0.88/0.65)			4,3	F(13.63/0.74)		
4	snorm(0.09, 0.06, 1.9)	snorm(0.13, 0.06, 1.85)								
		Genera	d VCN	MM (Copula with single	parame	ter) (Random)				
		Cla	assific	ation rate : 51.1% Norn	nalised	BIC : -7.00				
1	sstd(3.44, 2.85, 2.98, 5.46)	snorm(2.14, 0.85, 2.33)	3,2	T(0.42, 22.14/0.28)	4,2;3	SJ(1.09/0.05)	4,2	T(0.67, 9.46/0.47)	1,2;4	R90C(-0.22/-0.1)
2	$\Gamma(55.5, 416.91)$	snorm(0.13, 0.03, 1.69)	4,1	SC(1.29/0.39)	3,1;4	SC(0.08/0.04)	4,1	G(1.46/0.31)	3,1;4	F(0.69/0.08)
3	snorm(0.25, 0.19, 192.62)	snorm(0.31, 0.23, 8.48)	4,3	F(22.07/0.83)			4,3	F(17.45/0.79)		
4	snorm(0.12, 0.07, 1.41)	snorm(0.1, 0.06, 2.19)								
				Gaussian VCMM (Ra	andom)					
		Cla	assific	ation rate : 78.4% Norn	nalised	BIC : -6.61				
1	N(2, 0.69)	$\mathcal{N}(4.69, 2.61)$	4,2	N(0.45/0.3)	1,2;4	N(-0.07/-0.05)	2,1	N(-0.24/-0.16)	3,1;2	N(-0.01/-0.01)
2	$\mathcal{N}(0.13, 0.02)$	$\mathcal{N}(0.14, 0.02)$	4,1	N(0.15/0.1)	3,1;4	N(-0.01/-0.01)	3,2	N(0.44/0.29)	4,2;3	N(0.27/0.18)
3	$\mathcal{N}(0.17, 0.11)$	$\mathcal{N}(0.49, 0.2)$	4,3	N(0.84/0.63)			4,3	N(0.55/0.37)		
4	$\mathcal{N}(0.08, 0.04)$	$\mathcal{N}(0.19, 0.05)$								
				General VCMM (Kn	neans)					
		Cla	assific	ation rate : 63.6% Norn	nalised	BIC : -7.10				
1	sstd(6.18, 2.76, 3.28, 114.46)	snorm(2.15, 0.8, 2.16)	3,2	SG(1.69/0.41)	4,2;3	C(0.27/0.12)	4,2	T(0.58,14.54/0.39)	1,2;4	R270J(-1.09/-0.05)
2	$\mathcal{N}(0.14, 0.02)$	lnorm(-2.04, 0.18)	4,1	F(4.03/0.39)	3,1;4	N(-0.01/-0.01)	4,1	SC(0.78/0.28)	3,1;4	R270C(-0.06/-0.03)
3	logis(0.4, 0.1)	sstd(0.24, 0.19, 15.92, 244.39)	4,3	SBB8(5.36,0.92/0.65)			4,3	SBB8(6,0.91/0.67)		
4	$\mathcal{N}(0.18, 0.06)$	snorm(0.1, 0.06, 1.62)								
		Genera	al VCI	MM (Copula with single	parame	eter) (Kmeans)				
	()(0.50, 0.50, 0.00, 104, 50)	(2.22.0.27.2.22)	assific	ation rate : 63.0% Norn	nalised	BIC : -7.10	10	TP (0, F0, 1, T, 00, (0, 00))	1.0.1	Domo I ( 1 00 ( 0 05)
1	sstd(6.56, 2.76, 3.29, 124.58)	snorm(2.22, 0.87, 2.32)	3,2	SG(1.74/0.42)	4,2;3	SJ(1.19/0.1)	4,2	T(0.58,17.02/0.39)	1,2;4	R270J(-1.09/-0.05)
2	$\mathcal{N}(0.13, 0.02)$	1(32.13, 243.51)	4,1	F(3.83/0.37)	3,1;4	C(0.14/0.06)	4,1	SC(0.79/0.28)	3,1;4	R270C(-0.07/-0.04)
3	snorm(0.42, 0.18, 1.3)	sstd(0.25, 0.2, 10.23, 317.63)	4,3	SG(2.63/0.62)			4,3	F(12.2/0.72)		
4	logis(0.19, 0.03)	snorm(0.1, 0.06, 1.63)			L					
		Cla	secifio	Gaussian VCMM (Ki ation rate : 68.5% Norm	means)	BIC6 26				
1	N(58272)	M(2 14 0 77)	3.1	N(0.2/0.13)	2 1.2	N(-0.26/-0.17)	12	N(0.59/0.4)	1.9.4	N(-0.07/-0.04)
2	N(0.14.0.02)	N(0.13, 0.02)	3.9	N(0.58/0.39)	4 9.2	N(0.28/0.18)	4,2	N(0.39/0.4)	3 1.4	N(0.05/0.03)
2	N(0.14, 0.02)	$\mathcal{N}(0.23, 0.02)$	1 2	N(0.81/0.6)	+,2,3	11(0.20/0.10)	4.3	N(0.86/0.66)	0,1,4	11(0.00/0.00)
4	N(0.19.0.05)	N (0.23, 0.2)	4,0	11(0.01/0.0)			4,0	14(0.00/0.00)		
-4	1 (0.19, 0.05)	74 (0.1, 0.05)								

	Marginal distrib	utions				Vine tree	struc	ture		
Variable	Cluster 1	Cluster 2		Clu	ster 1			Clus	ter 2	
				Tree 1		Tree 2		Tree 1		Tree 2
				BCW						
				General VCMM (Hi	ierarchic	cal)				
		C	lassifi	cation rate : 82.2% No	ormalise	d BIC : -7.15				
1	snorm(2.04, 0.74, 2.25)	lnorm(1.47, 0.45)	4,1	BB8(1.59,0.87/0.15)	2,1;4	R90J(-1.08/-0.04)	2,1	R90C(-0.38/-0.16)	3,1;2	SC(0.19/0.09)
2	$\Gamma(37.47, 293.68)$	$\Gamma(38.33, 266.39)$	4,2	T(0.46, 9.34/0.3)	3,2;4	F(-0.38/-0.04)	3,2	C(1.53/0.43)	4,2;3	SJ(1.1/0.05)
3	sstd(0.19, 0.16, 7.49, 84.9)	sstd(0.46, 0.19, 9.25, 1.26)	4,3	SBB8(5.71,0.9/0.65)			4,3	BB1(1.87,1.22/0.58)		
4	snorm(0.08, 0.04, 1.22)	logis(0.19, 0.03)		04/0 1 11 1 1						
		Genera	l VCM lassifi	1M (Copula with single cation rate : 74.8% No	e param ormalise	d BIC : -7.10				
1	snorm(2.04, 0.72, 2)	sstd(5.49, 2.92, 3.31, 125, 19)	4.1	F(1.62/0.18)	2.1:4	R90J(-1.06/-0.03)	3.2	C(1.35/0.4)	4.2:3	SJ(1.21/0.11)
2	lnorm(-2.06, 0.17)	$\Gamma(41.71, 296.14)$	4.2	T(0.54.9.03/0.36)	3.2:4	T(-0.05.21.79/-0.03)	4.1	J(1.29/0.14)	3.1:4	F(-0.8/-0.09)
3	sstd(0.21, 0.18, 7.98, 308.6)	snorm(0.45, 0.18, 1.21)	4.3	F(11.19/0.69)	- , ,	( , , ,	4.3	SG(2.42/0.59)	- / /	( / /
4	snorm(0.09, 0.05, 1.46)	logis(0.19, 0.03)						· · · ·		
				Gaussian VCMM (H	lierarchi	cal)				
		C	lassifi	cation rate : 77.9% No	ormalise	d BIC : -6.61				
1	$\mathcal{N}(2.03, 0.71)$	$\mathcal{N}(4.77, 2.64)$	4,1	N(0.19/0.12)	2,1;4	N(-0.08/-0.05)	3,1	N(-0.15/-0.09)	4,1;3	N(0.31/0.2)
2	$\mathcal{N}(0.13, 0.02)$	$\mathcal{N}(0.14, 0.02)$	4,2	N(0.47/0.31)	3,2;4	N(0.01/0)	3,2	N(0.44/0.29)	4,2;3	N(0.27/0.17)
3	$\mathcal{N}(0.17, 0.11)$	$\mathcal{N}(0.5, 0.19)$	4,3	N(0.84/0.64)			4,3	N(0.52/0.35)		
4	N(0.08, 0.04)	$\mathcal{N}(0.19, 0.05)$		a 1100.01 /11						
			n .c	General VCMM (Kn	neans sc	ale)				
1	+	+	assin	E(1.95/0.9)	ormanse	a BIC : -7.21	4.1	50/0.2/0.00)	9.1.4	D270C/ 0 10 / 0 00)
1	ssta(4.24, 2.32, 4.32, 3.07)	E(42.24.242.25)	3,2	$\Gamma(1.85/0.2)$ N(0.20/0.26)	4,2;5	F(1.27/0.15)	4,1	C(0.2/0.09)	2,1;4	T(0.06.10.8/0.04)
2	lnorm(-0.8, 0.35)	1(42.24, 342.33)	4,1	F(4.27/0.41)	3,1,4	r(-1.37/-0.13)	4,2	SBB8(4.22.0.05/0.50)	3,2,4	1(-0.00,19.8/-0.04)
4	snorm(0.19, 0.04, 1.87)	snorm(0.14, 0.11, 351.0) snorm(0.07, 0.03, 0.85)	4,5	F(4.27/0.41)			4,5	55556(4.22,0.95/0.59)		
	3807 8 (0.13, 0.04, 1.01)	General	VCM	M (Copula with single	parame	ter) (Kmeans scale)				
		C	lassifi	cation rate : 83.9% No	ormalise	d BIC : -7.14				
1	lnorm(1.29, 0.52)	sstd(2.04, 0.86, 6.36, 2.48)	3,2	F(2.2/0.23)	4,2;3	R90J(-1.05/-0.03)	4,2	C(0.43/0.18)	1,2;4	R90C(-0.19/-0.09)
2	llogis(12.78, 0.15)	$\Gamma(43.58, 354.61)$	4,1	N(0.43/0.28)	3,1;4	F(-1.17/-0.13)	4,1	G(1.09/0.08)	3,1;4	N(-0.1/-0.07)
3	lnorm(-0.84, 0.39)	snorm(0.15, 0.11, 178.44)	4,3	F(5.19/0.47)			4,3	T(0.77,30/0.56)		
4	snorm(0.18, 0.04, 1.55)	snorm(0.07, 0.03, 0.81)								
			n .e	Gaussian VCMM (Ki	means s	cale)				
1	N( ( ( ( A ( D ( C ( )	Af(0.05.0.76)	lassifi	cation rate : 79.7% No	ormalise	d BIC : -6.00	4.1	NI(0.16 (0.1)	014	N(0.10/0.00)
1	$\mathcal{N}(4.64, 2.66)$	$\mathcal{N}(2.05, 0.76)$	2,1	N(-0.31/-0.2)	3,1;2	N(-0.02/-0.01)	4,1	N(0.16/0.1)	2,1;4	N(-0.12/-0.08)
2	$\mathcal{N}(0.15, 0.02)$	$\mathcal{N}(0.13, 0.02)$ $\mathcal{N}(0.17, 0.11)$	3,2	N(0.32/0.21) N(0.47/0.21)	4,2;3	N(0.17/0.11)	4,2	N(0.4/0.26) N(0.82/0.62)	3,2;4	N(-0.02/-0.01)
4	N (0.5, 0.19)	$\mathcal{N}(0.17, 0.11)$ $\mathcal{N}(0.08, 0.04)$	4,5	N(0.47/0.31)			4,5	14(0.03/0.02)		
	34 (0.13, 0.04)	34 (0.00, 0.04)		General VCMM (Hiera	archical	scale)				
		C	lassifi	cation rate : 86.9% No	ormalise	d BIC : -7.24				
1	snorm(1.99, 0.73, 2.37)	lnorm(1.36, 0.48)	4,1	F(0.51/0.06)	2,1;4	R270C(-0.15/-0.07)	2,1	N(-0.28/-0.18)	3,1;2	SC(0.1/0.05)
2	$\Gamma(40.23, 322.64)$	llogis(12.03, 0.14)	4,2	SG(1.27/0.22)	3,2;4	F(-0.39/-0.04)	3,2	SG(1.42/0.29)	4,2;3	SJ(1.16/0.08)
3	snorm(0.15, 0.12, 346.35)	sstd(0.48, 0.19, 7.12, 1.9)	4,3	${\rm SBB8}(4.22,\!0.96/0.6)$			4,3	SG(1.77/0.44)		
4	$\mathcal{N}(0.07, 0.03)$	sstd(0.19, 0.04, 9.19, 1.46)								
		General V	CMM	(Copula with single p	aramete	r) (Hierarchical scale)				
1	enorm(2.01.0.74.2.4)	eetd(4.45.2.5.4.36.2.70)	//assin	E(0.50/0.07)	2 1.4	B270C(010/000)	9.1	N(03/010)	3 1.9	SC(0.13/0.06)
2	Γ(39.12, 213.40)	llogis(12.5.0.14)	4.1	SG(1.27/0.22)	3 9.4	B270G(-0.19/-0.08)	3.2	SG(1.44/0.31)	4 2.3	SJ(1 11/0.06)
23	snorm(0.16, 0.12, 190 82)	sstd(0.48, 0.19, 7.18, 1.94)	4.3	T(0.77.30/0.56)	0,2,4	100(-1.00/-0.04)	4.3	SG(1.82/0.45)	1,2,0	55(1.11/0.00)
4	$\mathcal{N}(0.07, 0.03)$	sstd(0.19, 0.04, 9.8, 1.41)	1,0	= (0111,007 0100)			1,0	20(1102/0110)		
	, ,/	x		Gaussian VCMM (Hier	archical	scale)	L			
		C	lassifi	cation rate : 78.4% No	ormalise	d BIC : -6.61				
1	$\mathcal{N}(2.03, 0.73)$	$\mathcal{N}(4.73, 2.64)$	4,2	N(0.44/0.29)	1,2;4	N(-0.11/-0.07)	2,1	N(-0.27/-0.17)	3,1;2	N(-0.04/-0.02)
2	$\mathcal{N}(0.13, 0.02)$	$\mathcal{N}(0.15, 0.02)$	4,1	N(0.17/0.11)	3,1;4	N(-0.03/-0.02)	3,2	N(0.41/0.27)	4,2;3	N(0.25/0.16)
3	$\mathcal{N}(0.17, 0.11)$	$\mathcal{N}(0.5, 0.19)$	4,3	N(0.83/0.63)			4,3	N(0.51/0.34)		
4	$\mathcal{N}(0.08, 0.04)$	$\mathcal{N}(0.19, 0.05)$								

LBM

Wt BMI WBC

#### **C.3** Pair plots of AIS data with initialization and final clustering



Initializing clustering Random (50.2%)



Copula data for cluster 1 after initializing clustering

UM UM	Ø	Ø	0	<b>%</b>
0.81		Ø		
0.51	0,62			
0.059	0.092	6.11		٢
-0.33	4.13	ier)	¢012	

Copula data for cluster 2 after initializing clustering



Final clustering (General VCMM) Random (60.2%)



Copula data for cluster 1 after final clustering (General VCMM)



Copula data for cluster 2 after final clustering (General VCMM)



Final clustering Final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter)) Random (65.8%) Random (61.6%)





Copula data for cluster 1 after

1

final clustering

LIDAA

Copula data for cluster 1 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))

<b>F</b>		Ø	0	4
0.93		Ø	0	
0.55	0.57	ü		
0.19	0.14	0.19		Ø
-0.27	-0.19	4988	0.16	~

		Ø	$\bigcirc$	
0.95	tilili	Ø	0	
0.67	0.68	ddilla		
0.064	aati	2.098	WC	0
-0;16	410		0.20	147

Copula data for cluster 2 after final clustering

Copula data for cluster 2 after final clustering

(General VCMM (Copula with single pa- (Gaussian VCMM) rameter))

Figure C.1: The pairwise scatter plot of the subset of AIS data and its u data for each cluster, after initializing clustering with Random (First column), final clustering by general VCMM (Second column), General VCMM (Copula with single parameter) (Third column) and Gaussian VCMM (Fourth column) with Random and the EM algorithm reaching the highest classification rate. Different colours refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster.



Initializing clustering Kmeans (71.9%)



Copula data for cluster 1 after initializing clustering

UM The second second	Ø	Ø		٥
0,69		Ø		0
0.30	0.51			Ø
	-	9.094	alida	Ø
	0,30	0,45	<b>0:24</b>	, in the second se

Copula data for cluster 2 after initializing clustering



LBM Wt BMI WBC PBF

Final clustering (General VCMM) Kmeans (77.5%)



Copula data for cluster 1 after final clustering (General VCMM)

	Ø	Ø		٢
0.72	nim	Ø		0
0.38	0.55			0
	azib	4,090	wic	Ø
800	0.33	0.42	0.26	

Copula data for cluster 2 after final clustering (General VCMM)





Final clustering

Kmeans (91.4%)

Final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter)) Kmeans (76.7%)





Copula data for cluster 1 after Copula data for cluster 1 after final clustering final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))

	Ø	Ø		۲
0.75		Ø		Ø
0.41	0.56	T		Ø
Xan.	8	6.12	wic di ny ny litera	0
•#	0,33	0.41	0.28	



Copula data for cluster 2 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))

Copula data for cluster 2 after final clustering

Figure C.2: The pairwise scatter plot of the subset of AIS data and its u data for each cluster, after initializing clustering with Kmeans (First column), final clustering by general VCMM (Second column), General VCMM (Copula with single parameter) (Third column) and Gaussian VCMM (Fourth column) with Kmeans and the EM algorithm reaching the highest classification rate. Different colours refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster.

## [!htbp]



Initializing clustering Kmeans scale (68.1%)



Copula data for cluster 1 after initializing clustering

	Ø			٩
0.68		Ø		0
0.22	0.44			Ø
(0.646	408	-		Ø
-0.12	0,22	0.42	.0.18	~

Copula data for cluster 2 after initializing clustering



Final clustering (General VCMM) Kmeans scale (91.4%)



Copula data for cluster 1 after final clustering (General VCMM)

	ß	Ø	0	Ø
0.87		I	0	Ø
0.52	0.59		0	Ø
0.07%	a.18	0.18		0
0.26	0.40	0.45	0.33	747 

Copula data for cluster 2 after final clustering (General VCMM)





Final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter)) Kmeans scale (91.4%)





0.78 0.56 0.69 0.0 0.12 0.25 0.48 0.47

Copula data for cluster 1 after

final clustering

Copula data for cluster 1 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))

	P	Ø		Ø
0.87		T		ø
0:54	0.59	idili		Ø
c.ms	0,14	0.17		0
0.29	0.42	0.46	0.31	P47

		Ø	0	Ø
0.93	In the second	Ø	0	<b>A</b>
0.58	0.61			Ø
0.11	.0.19	0.16		0
0.34	0.42	0.47	0.29	~

Copula data for cluster 2 after final clustering

Copula data for cluster 2 after final clustering

(General VCMM (Copula with single pa- (Gaussian VCMM) rameter))

Figure C.3: The pairwise scatter plot of the subset of AIS data and its u data for each cluster, after initializing clustering with Kmeans scale (First column), final clustering by general VCMM (Second column), General VCMM (Copula with single parameter) (Third column) and Gaussian VCMM (Fourth column) with Kmeans scale and the EM algorithm reaching the highest classification rate. Different colours refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster.



Initializing clustering Hierarchical (85.3%)



Copula data for cluster 1 after initializing clustering

UM COMPANY	ſ	Ø		٥
0.89	iddan	Ø	0	
0.51	0.56	minim		Ø
0.11	à.16	0.18	Witc	0
0.20	0.32	0.39	0.28	~

Copula data for cluster 2 after initializing clustering



Final clustering (General VCMM) Hierarchical (91.4%)



Copula data for cluster 1 after final clustering (General VCMM)

1. The		Ø		Ø
0.92		Ø	0	R
0.58	0.61	idilidi		Ø
0.096	3.12	<b>6.15</b>	we	0
0.32	0.40	0.46	9.30	14

Copula data for cluster 2 after final clustering (General VCMM)



Final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter)) Hierarchical (92.4%)



Copula data for cluster 1 after

final clustering

rameter))

(

Hierarchical (90.5%) 0.78 0.56 0.69 ( D)

0.47

Copula data for cluster 1 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM)

0.48

0.25

Final clustering

		Ø		0	
9.92		Ø	0	<b>B</b>	
0.58	0.61			Ø	
0.10	.9.19	0.18		0	
0.34	0.41	0.47	9.30		

0.93 0.58 0.61 10.13 0.34 0.42 0.47



Figure C.4: The pairwise scatter plot of the subset of AIS data and its u data for each cluster, after initializing clustering with Hierarchical (First column), final clustering by general VCMM (Second column), General VCMM (Copula with single parameter) (Third column) and Gaussian VCMM (Fourth column) with Hierarchical and the EM algorithm reaching the highest classification rate. Different colours refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster. LBM

₹V

BMI

WBC

LBM Wt BMI WBC PBF

LBM

X

0.77

0.56

0.094

0.24

final clustering

0.69

0.13

0.47

Copula data for cluster 1 after

0.46

LBM Wt BMI WBC PBF

LBM

1M

BMI

WBC



Initializing clustering Hierarchical scale (81.3%)



Copula data for cluster 1 after initializing clustering

		Ø		0
0.87	-	đ		٨
0.51	0.59	Ň		0
0.000	419	0.54	wic	Ø
0.19	0.32	0:40	0.32	72

Copula data for cluster 2 after initializing clustering



Final clustering (General VCMM) Hierarchical scale (92.4%)



Copula data for cluster 1 after final clustering (General VCMM)

		Ø		Ø
0.93		Ø	0	<b>1</b>
0.60	0.62		0	Ø
0.12	.0.14	0.161	wic	0
0.37	0.45	0.48	0.29	~

Copula data for cluster 2 after final clustering (General VCMM)



Final clustering Final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) Hierarchical scale (91.4%) rameter)) Hierarchical scale (92.4%)



Copula data for cluster 1 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))

		Ø	0	Ø
0.93		Ø	0	<b>B</b>
0.60	0.62		0	Ø
0.12	:0.14.	0.16	wic	0
0.37	0.45	0.48	0.29	~

Copula data for cluster 2 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM)

0.93 0.59 0.62 0.12 0.47 0.30 0.44 0.29

Copula data for cluster 2 after final clustering

Figure C.5: The pairwise scatter plot of the subset of AIS data and its u data for each cluster, after initializing clustering with Hierarchical scale (First column), final clustering by general VCMM (Second column), General VCMM (Copula with single parameter) (Third column) and Gaussian VCMM (Fourth column) with Hierarchical scale and the EM algorithm reaching the highest classification rate. Different colours refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster.

rameter))

PSE

#### C.4 Pair plots of BCW data with initialization and final clustering



Initializing clustering Random (51.1%)



Copula data for cluster 1 after initializing clustering



Copula data for cluster 2 after initializing clustering





Copula data for cluster 1 after final clustering (General VCMM)



Copula data for cluster 2 after final clustering (General VCMM)



0.34 0.31 0.38



Copula data for cluster 2 after final clustering

Figure C.6: The pairwise scatter plot of the subset of Breast Cancer Wisconsin Diagnostic data and its u data for each cluster, after initializing clustering with Random (First column), final clustering by general VCMM (Second column), General VCMM (Copula with single parameter) (Third column) and Gaussian VCMM (Fourth column) with Random and the EM algorithm reaching the highest classification rate. Different colours refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster.



Final clustering Final clustering (General VCMM (Copula with single pa-(Gaussian VCMM) Random (78.4%) rameter)) Random (51.1%)



Copula data for cluster 1 after

final clustering

rameter))



167

Copula data for cluster 1 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM)



PSE ES EC ECP

ES

E



Initializing clustering Kmeans (62.8%)



Copula data for cluster 1 after initializing clustering



Copula data for cluster 2 after initializing clustering



Final clustering (General VCMM) Kmeans (63.6%)



Copula data for cluster 1 after final clustering (General VCMM)



Copula data for cluster 2 after final clustering (General VCMM)



Final clustering Final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter)) Kmeans (68.5%) Kmeans (63.0%)



0.22 0.34 0.24 0.38 0.71

Copula data for cluster 1 after Copula data for cluster 1 after final clustering final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))

Pic	Ø	<b>R</b>	Ø
0.j8-	65 - 11 - 111111 11 -	Ø	S
0.33	0.41		Ø
0.38	0,40	0.61	w

Copula data for cluster 2 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))



Copula data for cluster 2 after final clustering

Figure C.7: The pairwise scatter plot of the subset of Breast Cancer Wisconsin Diagnostic data and its u data for each cluster, after initializing clustering with Kmeans (First column), final clustering by general VCMM (Second column), General VCMM (Copula with single parameter) (Third column) and Gaussian VCMM (Fourth column) with Kmeans and the EM algorithm reaching the highest classification rate. Different colours refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster.



Initializing clustering Kmeans scale (82.2%)



Copula data for cluster 1 after initializing clustering



Copula data for cluster 2 after initializing clustering



Final clustering (General VCMM) Kmeans scale (85.1%)



Copula data for cluster 1 after final clustering (General VCMM)



Copula data for cluster 2 after final clustering (General VCMM)



#### Final clustering

Final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter)) Kmeans scale (79.7%) Kmeans scale (83.9%)



Copula data for cluster 1 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))



Copula data for cluster 2 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))

Copula data for cluster 2 after final clustering

0.28

Figure C.8: The pairwise scatter plot of the subset of Breast Cancer Wisconsin Diagnostic data and its u data for each cluster, after initializing clustering with Kmeans scale (First column), final clustering by general VCMM (Second column), General VCMM (Copula with single parameter) (Third column) and Gaussian VCMM (Fourth column) with Kmeans scale and the EM algorithm reaching the highest classification rate. Different colours refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster.

0.65 9.12 0.27

Copula data for cluster 1 after final clustering





PSE ES EC ECP



Initializing clustering Hierarchical (68.5%)



Copula data for cluster 1 after initializing clustering



Copula data for cluster 2 after initializing clustering



Final clustering (General VCMM) Hierarchical (82.2%)



Copula data for cluster 1 after final clustering (General VCMM)



Copula data for cluster 2 after final clustering (General VCMM)



#### Final clustering

(General VCMM (Copula with single pa- (Gaussian VCMM) rameter)) Hierarchical (74.8%)



Copula data for cluster 1 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))



Copula data for cluster 2 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))



Copula data for cluster 2 after final clustering

Figure C.9: The pairwise scatter plot of the subset of Breast Cancer Wisconsin Diagnostic data and its u data for each cluster, after initializing clustering with Hierarchical (First column), final clustering by general VCMM (Second column), General VCMM (Copula with single parameter) (Third column) and Gaussian VCMM (Fourth column) with Hierarchical and the EM algorithm reaching the highest classification rate. Different colours refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster.





Final clustering Hierarchical (77.9%)



Copula data for cluster 1 after final clustering



Initializing clustering Hierarchical scale (84.5%)



Copula data for cluster 1 after initializing clustering



Copula data for cluster 2 after initializing clustering



ES

EC

Final clustering (General VCMM) Hierarchical scale (86.9%)



Copula data for cluster 1 after final clustering (General VCMM)



Copula data for cluster 2 after final clustering (General VCMM)



#### Final clustering

(General VCMM (Copula with single pa- (Gaussian VCMM) rameter)) Hierarchical scale (87.8%)



Copula data for cluster 1 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))



Copula data for cluster 2 after final clustering (General VCMM (Copula with single pa- (Gaussian VCMM) rameter))

Copula data for cluster 2 after final clustering

Figure C.10: The pairwise scatter plot of the subset of Breast Cancer Wisconsin Diagnostic data and its u data for each cluster, after initializing clustering with Hierarchical scale (First column), final clustering by general VCMM (Second column), General VCMM (Copula with single parameter) (Third column) and Gaussian VCMM (Fourth column) with Hierarchical scale and the EM algorithm reaching the highest classification rate. Different colours refer to the points from different clusters. The number inside the round bracket is the classification rate, compared to the true cluster.









ES

EC

# References

- J. D. Banfield and A. E. Raftery. Model-based gaussian and non-gaussian clustering. *Biometrics*, 49(3):803–821, 1993. doi: https://doi.org/10.2307/2532201.
- T. Bedford and R. M. Cooke. Vines : a new graphical model for dependent random variables. Annals of Statistics, 30(4):1031–1068, 2002. doi: https://doi.org/10.1214/aos/1031689016.
- C. M. Bishop. *Pattern Recognition and Machine Learning*. Springer-Verlag New York, 2006. URL https: //www.springer.com/gp/book/9780387310732#aboutAuthors.
- C. Czado. Analyzing Dependent Data with Vine Copulas A Practical Guide With R. Springer International Publishing, 2019. doi: https://doi.org/10.1007/978-3-030-13785-4.
- U. J. Dang, A. Punzo, S. I. Paul D. McNicholas, and R. P. Browne. Multivariate response and parsimony for gaussian cluster-weighted models. *Journal of Classification volume*, 34:4–34, 2017. URL https://link.springer.com/article/10.1007/s00357-017-9221-2.
- B. A. Desmarais and J. J. Harden. Testing for zero inflation in count models: Bias correction for the vuong test. Stata Journal, 13(4):810–835, 2013. doi: https://doi.org/10.1177/1536867X1301300408.
- K. Ellis, S. Godbole, S. Marshall, G. Lanckriet, J. Staudenmayer, and J. Kerr. Identifying active travel behaviors in challenging environments using gps, accelerometers, and machine learning algorithms. *Frontiers in Public Health*, 2(36), 2014. doi: https://doi.org/10.3389/fpubh.2014.00036.
- C. Fraley. Algorithms for model-based gaussian hierarchical clustering. SIAM Journal on Scientific Computing, 57(1):270–281, 1998. doi: https://doi.org/10.1137/S1064827596311451.
- L. A. García-Escudero, A. Gordaliza, C. Matrán, and A. Mayo-Iscar. Avoiding spurious local maximizers in mixture modeling. *Statistics and Computing*, 25:619–633, 2015. doi: https://doi.org/10.1007/s11222-014-9455-3.
- J. A. Hartigan and M. A. Wong. Algorithm as 136: A k-means clustering algorithm. Journal of the Royal Statistical Society. Series C (Applied Statistics), 28(1):100–108, 1979. doi: https://doi.org/10.2307/2346830.
- N. Heal. Agile, hackathons, and the danger of local maxima, 2020. URL https://medium.com/swlh/agile-hackathons-and-the-danger-of-local-maxima-4c2255e8f63b.
- C. Jin, Y. Zhang, S. Balakrishnan, M. J. Wainwright, and M. Jordan. Local maxima in the likelihood of gaussian mixture models: Structural results and algorithmic consequences. *Neural Information Processing Systems (NIPS)*, 2016. URL https://arxiv.org/abs/1609.00978.
- C. Liu and D. B. Rubin. The ecme algorithm: A simple extension of em and ecm with faster monotone convergence. *Biometrika*, 81(4):633–648, 1994. doi: https://doi.org/10.2307/2337067.
- G. J. McLachlan and T. Krishnan. The EM Algorithm and Extensions, Second Edition. Wiley, 2007. doi: https://doi.org/10.1002/9780470191613.
- X.-L. Meng and D. B. Rubin. Maximum likelihood estimation via the ecm algorithm: A general framework. *Biometrika*, 80(2):267–278, 1993. doi: https://doi.org/10.2307/2337198.
- K. P. Murphy. Machine Learning : A Probabilistic Perspective. The MIT Press, 2012.
- A. Punzo and P. D. McNicholas. Parsimonious mixtures of multivariate contaminated normal distributions. Biometrical Journal, 58(6):1506–1537, 2016. doi: https://doi.org/10.1002/binj.201500144.
- W. M. Rand. Objective criteria for the evaluation of clustering methods. Journal of the American Statistical Association, 66(336):846–850, 1971. doi: https://doi.org/10.2307/2284239.

- Ö. Sahin and C. Czado. Vine copula mixture models and clustering for non-gaussian data. *Econometrics and Statistics*, 2021. ISSN 2452-3062. doi: https://doi.org/10.1016/j.ecosta.2021.08.011. URL https://www.sciencedirect.com/science/article/pii/S2452306221001052.
- L. Scrucca, M. Fop, T. B. Murphy, and A. E. Raftery. mclust 5: Clustering, classification and density estimation using gaussian finite mixture models. *The R Journal*, 8(1):205–233, 2016. doi: https://doi.org/10.32614/RJ-2016-021.
- E. Shireman, D. Steinley, and M. J. Brusco. Examining the effect of initialization strategies on the performance of gaussian mixture modeling. *Behavior Research Methods volume*, 49(1):282–293, 2017. doi: https://doi.org/ 10.3758/s13428-015-0697-6.
- A. Sklar. Fonctions de répartition à n dimensions et leurs marges. Publications de l'Institut Statistique de l'Universit e de Paris, 8:229–231, 1959.
- D. A. van Dyk and X.-L. Meng. On the orderings and groupings of conditional maximizations within ecm-type algorithms. *Journal of Computational and Graphical Statistics*, 6(2):202–223, 1997. doi: https://doi.org/10.2307/1390931.
- Q. H. Vuong. Likelihood ratio tests for model selection and non-nested hypotheses. *Econometrica*, 57(2): 307–333, 1989. doi: https://doi.org/10.2307/1912557.